The development paths and relationships of value creation and capture: an investigation of digital ventures using System Dynamics

Martin Alexander Lycko

A thesis submitted in partial fulfilment of the requirements of Nottingham Trent University for the degree of Doctor of Philosophy

October 2022

Copyright statement

The copyright in this work is held by the author. You may copy up to 5% of this work for private study, or personal, non-commercial research. Any re-use of the information contained within this document should be fully referenced, quoting the author, title, university, degree level and pagination. Queries or requests for any other use, or if a more substantial copy is required, should be directed to the author.

"The overwhelming importance of future profits is counterintuitive even in Silicon Valley. For a company to be valuable it must grow *and endure*, but many entrepreneurs focus only on short-term growth. They have an excuse: growth is easy to measure, but durability isn't. Those who succumb to measurement mania obsess about weekly active user statistics, monthly revenue targets, and quarterly earnings reports. However, you can hit those numbers and still overlook deeper, harder-to-measure problems that threaten the durability of your business.

[...]

If you focus on near-term growth above all else, you miss the most important question you should be asking: *will this business still be around a decade from now?* Numbers alone won't tell you the answer; instead you must think critically about the qualitative characteristics of your business."

(Thiel and Masters, 2014, p. 47)

Abstract

Purpose: While many digital ventures strive for high growth, they may neglect developing an ability to capture value, causing their failure or dependence on investors' funding. Therefore, this thesis aims to develop an understanding of the requirements and priorities allowing digital ventures to develop an ability to capture value while growing their value creation. It investigates how digital ventures' growth in value creation and ability to capture value develop over time and the relationships between these two performance outcomes.

Methods: This thesis develops a System Dynamics model to represent the complex system and processes that affect value creation and capture. The basis for this model is the synthesis of two contemporary growth process theories – growth paths theory and dynamic states theory – and their adaptation for the context of digital ventures. The model is validated using secondary data from four case studies. It simulates scenarios to identify development patterns and relationships of value creation and capture.

Findings: By synthesising growth paths and dynamic states theory, this thesis reveals the dynamic relationships between performance outcomes, resources, and capabilities. These elements change continuously through the feedback loops connecting them. It also identifies the influence of discontinuous changes in contextual variables regarding a venture's environment, business model design, capability development, and dominant logic on those feedback loops. The scenario simulations reveal two development patterns that depend on a venture's dominant logic. Ventures unwilling to increase their employee numbers exhibit goal-seeking development towards natural levels of value creation and capture. The trajectories and levels towards which ventures develop depend on the contextual factors. Ventures willing to grow by hiring more employees experience exponential growth in value creation. They develop towards a lower ability to capture value in a goal-seeking manner, pointing to adverse effects of growth on the ability to capture value during growth periods. When ventures begin to exploit their established size rather than growing aggressively, economies of scale improve their ability to capture value, highlighting a

iv

beneficial relationship between growth in value creation and value capture after periods of growth.

Implications: Conflicting with the common belief among digital entrepreneurs, this thesis indicates that growth has only minor positive effects on capturing value. Instead of trying to grow themselves profitable, digital ventures need to ensure that their conceptual factors allow them to develop an ability to capture value. They need to select an appropriate environment and business model before growing. They should also pursue continuous improvements to raise their natural performance levels. These two elements are required to create the conditions to grow value creation and develop an ability to capture value. Digital ventures may need to change their priorities. They may pursue lower growth rates to maintain their ability to capture value or obtain funding to sustain high-growth periods. Moreover, digital ventures also need to consider exploiting their established size rather than pursuing further growth.

Contribution: This thesis provides a comprehensive formal growth process theory for digital ventures focussing on two performance outcomes. It provides an overview of the complex system and processes driving value creation and capture using a System Dynamics model. It illustrates possible development paths of value creation and capture and their relationships for digital ventures, contributing to unsettled questions in the performance management literature and challenging digital entrepreneurs' beliefs. Those practitioners can use the model to approximate the development of their ventures, and researchers can adapt it to investigate growth and performance in non-digital companies.

Acknowledgements

I am grateful to the many people without whom this work would not have been possible. First and foremost, I would like to thank my two supervisors, Dr Kostas Galanakis and Dr Néstor Valero-Silva, who supported my work on this thesis by providing feedback, suggestions, and opportunities for further development. I thank them for the opportunity to pursue a PhD, the introduction to and continuous guidance in System Dynamics modelling, and their help to sharpen my ideas and their communication. This thesis has also benefitted from many invaluable discussions with countless academics and practitioners at conferences, workshops, and private conversations, who I would like to thank. Finally, I would like to thank my family and friends for their moral support through this journey.

List of selected publications

Journal articles related to this thesis

 Lycko, M.A. and Mahlendorf, M.D., 2017. Management-Control-Systeme in Start-ups. *Controlling & Management Review*, 61(6), pp.24-33.

Conference proceedings extracted from this thesis

- ¹Lycko, M.A., Galanakis, K. and Valeo-Silva, N., 2021. The Performance Development of Software-as-a-Service Ventures. In: *Proceedings of the 23rd UK System Dynamics Chapter Annual Conference 'Thinking Big'*, 25-26 May 2021.
- Lycko, M.A., Galanakis, K. and Valero-Silva, N., 2019. Strategic entrepreneurship for digital start-ups: Factors affecting exploration and exploitation along the growth process. In: 11th *International Process Symposium (PROS2019)*, Crete, 19-22 June 2019.
- Lycko, M.A., Galanakis, K. and Valero-Silva, N., 2018. Achieving growth by forming stakeholder relationships: an exploration of strategies for digital start-ups. In: *British Academy of Management (BAM2018) Conference*, Bristol Business School, University of the West of England, Bristol, 4-6 September 2018.
- ²Lycko, M.A., 2018. Achieving growth by forming stakeholder relationships: an exploration of strategies for digital start-ups. In: *British BAM2018 Doctoral Symposium*, Bristol Business School, University of the West of England, Bristol, 4-6 September 2018.

Awards

- Runner up for the Student Prize at the 23rd UK System Dynamics Chapter Annual Conference
- 2. Awarded best poster presentation at the BAM2018 Doctoral Symposium

Table of content

Acknowl	edgementsvi
List of selected publicationsvii	
Table of	content viii
List of fig	uresxiv
List of ta	olesxvii
List of ab	breviations xviii
CHAPT	ER 1: INTRODUCTION1
1.1. Di	gital ventures2
1.2. Pe	rformance outcomes and identification of research questions3
1.2.1.	Development of value creation and capture
1.2.2.	Relationship(s) of value creation and capture7
1.3. G	owth process theories as a conceptual basis10
1.3.1.	Organisational life-cycle theory
1.3.1. 1.3.2.	Growth paths theory
1.3.1. 1.3.2. 1.3.3.	Growth paths theory
1.3.1. 1.3.2. 1.3.3. 1.3.4.	Organisational life-cycle theory 11 Growth paths theory 13 Dynamic states theory 14 The complementary nature of growth paths and dynamic states
1.3.1. 1.3.2. 1.3.3. 1.3.4. theory	Organisational life-cycle theory 11 Growth paths theory 13 Dynamic states theory 14 The complementary nature of growth paths and dynamic states 16
1.3.1. 1.3.2. 1.3.3. 1.3.4. theory 1.4. TI	Organisational life-cycle theory 11 Growth paths theory 13 Dynamic states theory 14 The complementary nature of growth paths and dynamic states 16 nesis outline 18
1.3.1. 1.3.2. 1.3.3. 1.3.4. theory 1.4. TI	Organisational life-cycle theory 11 Growth paths theory 13 Dynamic states theory 14 The complementary nature of growth paths and dynamic states 16 nesis outline 18
1.3.1. 1.3.2. 1.3.3. 1.3.4. theory 1.4. TI	Organisational life-cycle theory 11 Growth paths theory 13 Dynamic states theory 14 The complementary nature of growth paths and dynamic states 16 16 18 ER 2: SYSTEM DYNAMICS MODELLING
1.3.1. 1.3.2. 1.3.3. 1.3.4. theory 1.4. TH CHAPT METHO	Organisational life-cycle theory 11 Growth paths theory 13 Dynamic states theory 14 The complementary nature of growth paths and dynamic states 16 16 18 ER 2: SYSTEM DYNAMICS MODELLING 20
1.3.1. 1.3.2. 1.3.3. 1.3.4. theory 1.4. TH CHAPT METHO 2.1. PH	Organisational life-cycle theory 11 Growth paths theory 13 Dynamic states theory 14 The complementary nature of growth paths and dynamic states 16 16 18 ER 2: SYSTEM DYNAMICS MODELLING 20 ilosophical underpinnings and methodological requirements 20
1.3.1. 1.3.2. 1.3.3. 1.3.4. theory 1.4. TH CHAPT METHO 2.1. PH 2.1.1.	Organisational life-cycle theory 11 Growth paths theory 13 Dynamic states theory 14 The complementary nature of growth paths and dynamic states 16 16 18 ER 2: SYSTEM DYNAMICS MODELLING 20 ilosophical underpinnings and methodological requirements 20 Process philosophy and research 20
1.3.1. 1.3.2. 1.3.3. 1.3.4. theory 1.4. TH CHAPT METHO 2.1. PH 2.1.1. 2.1.2.	Organisational Ine-cycle theory

2.2.1. Selection of System Dynamics as a modelling method25

2.2.2	2. System Dynamics modelling	
2.3.	Model development	30
2.3.	1. Model conceptualisation	
2.	.3.1.1. Literature review	
2.	.3.1.2. Causal loop diagrams	
2.3.	2. Model formalisation	
2.		
2.	.3.2.2. Variable formulation	
2.3.	3. Model verification	41
2.4.	Model simulation	42
2.4.	1. Case simulation and model validation	43
2.	.4.1.1. Case study identification and selection	
2.	.4.1.2. Model parameterisation and calibration	
2.	.4.1.3. Model validation	49
2.4.2	2. Scenario simulation	
2.4.	3. Analysis of results	

CHAPTER 3: THEORY INTEGRATION AND MODEL

CON	CEPTUALISATION	53
3.1.	Performance outcomes	54
3.1.	1. Value creation	56
3.1.	2. Value capture	58
3.2.	The business model and its elements	61
3.2.	1. The impact of resources on performance outcomes	62
3.2.	2. Resource transformation through activities	66
3.2.	3. Capability development	69
3.3.	Influences on the performance outcomes and business model	71
3.3.	1. Environmental influences	71
3.3.	2. Dominant logic of managers and entrepreneurs	72
3.4.	Implications for growth process and performance development	77
3.4.	1. Implications for growth states	77
3.4.	2. Continuous change within a growth state	78
3.4.	3. Performance development in different growth states	79

СНАРТЕ	CR 4: THEORY CONTEXTUALISATION AND MODE	L
FORMA	LISATION	82
4.1. Va	lue creation subsystem	83
4.1.1.	Separating value creation for different users and customers	84
4.1.2.	Value creation for each customer and user type	86
4.1.3.	Value drivers determining the product strength	87
4.1.3.1	. Technological quality	
4.1.3.2	. Network effects for customers	
4.1.4.	Customer usage intensity	93
4.1.5.	Calculation of reported figures and accounting performance	95
4.1.6.	Overview of subsystem inputs	98
4.2. Va	lue capture subsystem	99
4.2.1.	Share of value captured from customers	100
4.2.1.1	. Bargaining power to users and customers	102
4.2.1.2	. Customer switching costs	105
4.2.1.3	. Value provided by competitors and substitutes	107
4.2.2.	Share of value lost to input providers and erosion of resources.	108
4.2.2.1	. Operating of resource inputs	110
4.2.2.2	. Providers of capital	111
4.2.3.	Calculation of reported figures and accounting measures	112
4.2.3.1	. Calculation of reported figures	112
4.2.3.2	. Comparison to accounting performance measures	114
4.2.4.	Overview of subsystem inputs	117
4.3. Tee	chnology development subsystem	118
4.3.1.	Technological resources	119
4.3.2.	Technology development activities	120
4.3.2.1	. Technology development	121
4.3.2.2	. Technology maintenance	
4.3.3.	Technology development capabilities	123
4.3.4.	Overview of subsystem inputs	126
4.4. Cu	stomer and user subsystem	127
4.4.1.	Customer and user bases	128
4.4.2.	Marketing and selling	131
4.4.3.	Word-of-mouth marketing	133

4.4.4.	Customer lifetime	134
4.4.5.	Value delivering activities	
4.4.6.	Overview of subsystem inputs	
4.5. Firr	n managing subsystem	139
4.5.1.	Management of human resources	139
4.5.1.1.	Marketing and selling employees	
4.5.1.2.	Target marketing and selling employees	
4.5.1.3.	Hiring delay	
4.5.1.4.	Priorities of marketing and selling employees	
4.5.2.	Management of complementary assets	147
4.5.2.1.	Complementary assets and their adequacy	
4.5.2.2.	Target investment in complementary assets	
4.5.3.	Management of financial resources	150
4.5.3.1.	Financial resources	
4.5.3.2.	Equity and debt employed	
4.5.3.3.	Liquidity adequacy and target equity and debt raising	
4.5.4.	Overview of subsystem	156

CHAPTER 5: CASE SIMULATION AND MODEL VALIDATION 159

Firm managing subsystem	159
1. Alpha	
2. Beta	
1. Gamma	
2. Delta	
Technology developing subsystem	
1. Alpha	
2. Beta	
3. Gamma	
4. Delta	177
Customer and user subsystem	
1. Alpha	
2. Beta	
3. Gamma	
	Firm managing subsystem

5.3	3.4. Delta	
5.4.	Value creation subsystem.	
5.4	4.1. Alpha	
5.4	4.2. Beta	
5.4	4.3. Gamma	
5.4	4.4. Delta	
5.5.	Value capture subsystem	
5.5	5.1. Alpha	
5.5	5.2. Beta	
5.5	5.3. Gamma	
5.5	5.4. Delta	
СНА	APTER 6: SCENARIO SIN	IULATIONS
6.1.	Dominant logics with a caj	o on employee numbers200
6.1	1.1. Employee-capped base	case
(6.1.1.1. Model inputs for the emp	bloyee-capped base case
6 1	1.2 Different targets express	the of the employee-capped base case
0.1	1.2. Different targets expres	
6.1	1.3. Internal improvement i	or capability development210
6.2.	Dominant logics without a	cap on employee numbers212
6.2	2.1. Employee-uncapped ba	
6.2	2.2. Different targets expres	sed in the dominant logic216
6.2	2.3. Internal improvement f	or capability development219
6.3.	Transitioning between dor	ninant logics221
6.3	3.1. Transitioning base case	
6.3	3.2. Different targets expres	sed in the dominant logic
(6.3.2.1. Different targets through	out the simulation225
(6.3.2.2. Different restructuring m	agnitudes227
(6.3.2.3. Different number of grow	vth cycles
6.3	3.3. Internal improvement f	for capability development232
6.4.	Contextual influences and	robustness234
6.4	4.1. Different environmenta	l conditions235

6.4.1.1.

6.4.1.2. Dynamics environments	3
6.4.2. Different business model designs	2
6.4.2.1. Business models with network effects	3
6.4.2.2. Business models with different efficiency levels	5
6.5. Implications to the case firms248	3
CHAPTER 7: DISCUSSION251	l
7.1. Influences on value creation and capture251	l
7.2. Implications for the ability of digital ventures to growth value	
creation and capture value253	5
7.3. How do value creation and capture develop for digital ventures? 255	5
7.3.1. Dominant logics determine development patterns	5
7.3.2. Contextual factors determine trajectories and levels within patterns	
258	
7.4 What relationship(s) exist(s) between value creation and canture? 26()
7.4.1 Growth in value creation adversely affects the share of value	,
contured during periods of growth 261	1
7.4.2 Growth in value groation has minor positive effects on the share of	L
value contured after periods of growth	,
value captured after periods of growth	,
CHAPTER 8: CONCLUSION265	5
8.1. Review of the research process	5
8.2. Contribution to theory and practice	7
8.2.1. Contribution to growth process theories and management	7
8.2.2. Contribution to performance management literature and practice269)
8.2.3. Contribution to digital ventures	l
8.3. Limitations and future research272	2
REFERENCES	5
APPENDICESxix	

List of figures

Figure 1-1: Profit-growth transitions (own graphic based on data collected by Davidsson, Steffens and Fitzsimmons (2009))
Figure 1-2: Organisational life-cycle and examples of stage models developed by Miller and Friesen (1984) and Kazanjian and Drazin (1990)12
Figure 1-3: Growth paths identified by Garnsey and Heffernan (2005)13
Figure 1-4: Elements of a dynamic state (Levie and Lichtenstein, 2010, p. 332) 15
Figure 2-1: System Dynamics modelling process
Figure 2-2: Cone of causal loop diagrams (adapted based on Coyle, 1996, p. 44)
Figure 2-3: Example of a causal loop diagram
Figure 2-4: Example of stocks-and-flow-structure
Figure 3-1: Overview of model elements
Figure 3-2: Resource generation
Figure 3-3: Value creation
Figure 3-4: Share of value captured
Figure 3-5: Resource-performance relationships
Figure 3-6: Resource development through activities
Figure 3-7: Capability development
Figure 3-8: Impact of the dominant logic74
Figure 4-1: Model subsystems
Figure 4-2: Value creation subsystem
Figure 4-3: Value creation for users and customers
Figure 4-4: Rate of value creation for customers
Figure 4-5: Strength of product or service for customers
Figure 4-6: Technological quality
Figure 4-7: Network effects for customers
Figure 4-8: Customer usage intensity
Figure 4-9: Rolling annual revenue
Figure 4-10: Value capture subsystem
Figure 4-11: Share of value captured
Figure 4-12: Share of value captured from customers and users101
Figure 4-13: Bargaining power to customers
Figure 4-14: Customer switching costs
Figure 4-15: Value provided by competitors and substitutes107
Figure 4-16: Share of value lost to input providers and resource erosion

Figure 4-17: Cost of resource inputs	110
Figure 4-18: Cost of capital	111
Figure 4-19: Rolling annual expenses	113
Figure 4-20: Return on assets and net profit margin	115
Figure 4-21: Technology development subsystem	119
Figure 4-22: Technological resources	119
Figure 4-23: Technology development	121
Figure 4-24: Technology maintenance and time for development	122
Figure 4-25: Technology developing capabilities	124
Figure 4-26: Customer and user subsystem	128
Figure 4-27: Customer and user bases	129
Figure 4-28: Marketing and selling activities	132
Figure 4-29: Customer and user acquisition through word-of-mouth	133
Figure 4-30: Customer lifetime	135
Figure 4-31: Value delivering activities	137
Figure 4-32: Firm managing subsystem	139
Figure 4-33: Marketing and selling employees	140
Figure 4-34: Target marketing and selling employees	142
Figure 4-35: Managerial slack and hiring delay	145
Figure 4-36: Priorities of marketing and selling employees	146
Figure 4-37: Complementary assets and their adequacy	147
Figure 4-38: Target complementary assets	149
Figure 4-39: Financial resources	150
Figure 4-40: Equity and debt employed	152
Figure 4-41: Liquidity adequacy, shortfall, and capital targets	154
Figure 5-1: Firm managing outputs for Alpha	163
Figure 5-2: Firm managing outputs for Beta	166
Figure 5-3: Firm managing outputs for Gamma	169
Figure 5-4: Firm managing outputs for Delta	172
Figure 5-5: Technological resources of Alpha	174
Figure 5-6: Technological resources of Beta	175
Figure 5-7: Technological resources of Gamma	176
Figure 5-8: Technological resources of Delta	177
Figure 5-9: Customer subsystem outputs for Alpha	180
Figure 5-10: Customer subsystem outputs for Beta	

Figure 5-11: Customer subsystem outputs for Gamma1	183
Figure 5-12: Customer subsystem outputs for Delta1	184
Figure 5-13: Value creation outputs for Alpha	186
Figure 5-14: Value creation outputs for Beta	188
Figure 5-15: Value creation outputs for Gamma1	190
Figure 5-16: Value creation outputs for Delta	192
Figure 5-17: Value capture outputs for Alpha	193
Figure 5-18: Value capture outputs for Beta	195
Figure 5-19: Value capture outputs for Gamma1	196
Figure 5-20: Value capture outputs for Delta 1	197
Figure 6-1: Overview of simulations and findings1	199
Figure 6-2: Performance development of employee-capped base case	205
Figure 6-3: Effect of changes to employee-capped base case's dominant logic 2	209
Figure 6-4: Effect of changes to employee-capped base case's capability development	211
Figure 6-5: Performance development of employee-uncapped base case	214
Figure 6-6: Effect of changes to employee-uncapped base case's dominant logi	c 217
Figure 6-7: Effect of changes to employee-uncapped base case's capability development	220
Figure 6-8: Performance development of transitioning base case transitioning.2	223
Figure 6-9: Effect of changes to transitioning base case's dominant logic (targe	ets) 226
Figure 6-10: Effect of changes to transitioning base case's dominant logic (reductions)	228
Figure 6-11: Effect of changes to transitioning base case's dominant logic (cycles)	230
Figure 6-12: Effect of changes to transitioning base case's capability developm	ent 233
Figure 6-13: Effect of stable environment conditions	237
Figure 6-14: Effect of environmental dynamics	240
Figure 6-15: Effect of business models with network effects	245
Figure 6-16: Effect of business model's efficiency levels	247
Figure 7-1: Growth process framework	254

List of tables

Table 1-1: Empirical literature on the growth-profitability relationship9
Table 1-2: Comparison of growth paths theory and dynamic states theory16
Table 2-1:Overview of proposed System Dynamics modelling approaches29
Table 4-1: Outputs and inputs of the value creation subsystem98
Table 4-2: Outputs and inputs of the value capture subsystem117
Table 4-3: Outputs and inputs of the technology development subsystem127
Table 4-4: Outputs and inputs of the customer and user subsystem
Table 4-5: Outputs and inputs of the firm managing subsystem157
Table 5-1: Firm management inputs for Alpha
Table 5-2: Firm management inputs for Beta 165
Table 5-3: Firm management inputs for Gamma
Table 5-4: Firm management inputs for Delta
Table 5-5: Technology developing inputs for Alpha174
Table 5-6: Technology developing inputs for Beta 175
Table 5-7: Technology developing inputs for Gamma
Table 5-8: Technology developing inputs for Delta 177
Table 5-9: Customer and user subsystem inputs for Alpha179
Table 5-10: Customer and user subsystem inputs for Beta181
Table 5-11: Customer and user subsystem inputs for Gamma182
Table 5-12: Customer and user subsystem inputs for Delta
Table 5-13: Value creation inputs for Alpha 186
Table 5-14: Value creation inputs for Beta 188
Table 5-15: Value creation inputs for Gamma
Table 5-16: Value creation inputs for Delta
Table 5-17: Value capture inputs for Alpha
Table 5-18: Value capture inputs for Beta 194
Table 5-19: Value capture inputs for Gamma
Table 5-20: Value capture inputs for Delta
Table 6-1: Changes to employee-capped base case's dominant logic208
Table 6-2: Changes to employee-capped base case's capability development210
Table 6-3: Changes to reflect differently attractive stable environments236
Table 6-4: Changes to reflect different environmental dynamics239
Table 6-5: Changes to reflect business models with network effects244
Table 6-6: Changes to reflect different productivity levels 246

List of abbreviations

- CA Complementary assets
- Eq Equation
- FM Firm managing/Firm management
- FT Financial Times
- i.e. That is
- IPO Initial public offering
- MS Marketing and selling/marketing and sales
- OLC Organisational life-cycle theory
- SaaS Software-as-a-service
- TD Technology developing/technology development
- VD Value delivering
- WoM Word-of-mouth

Chapter 1: Introduction

The development of digital technologies and the rapid growth of the digital economy have led to the emergence and success of digital businesses as a new type of company (Hafezieh, Akhavan and Eshraghian, 2011). Digital ventures are young companies that use digital technologies as a central part of their business model (Kraus *et al.*, 2018; Nzembayie, Buckley and Cooney, 2019; Zaheer, Breyer and Dumay, 2019). They commercialise the internet and information and communication technology (Kraus *et al.*, 2018). Examples of such technologies include artefacts such as software or an app, platforms such as social networks, or infrastructure like cloud computing (Nambisan, 2017).

While some digital ventures have achieved incredible success, many others have failed (Zaheer *et al.*, 2019). In theory, digital ventures have a high growth potential due to their scalability, market demand, and access to resources (Kollmann *et al.*, 2016; Zaheer *et al.*, 2019). However, only 52 per cent of digital ventures reach the age of three years (Hathaway, 2013). Compared to the 59 per cent of all companies that survive their first three years (Stangler, 2010), digital ventures thus fail quicker and more often. Therefore, investigating and explaining the performance of digital ventures remains a critical research topic to improve their survival rates (Kraus *et al.*, 2018; Zaheer *et al.*, 2019; Zaheer, Breyer and Dumay, 2019).

This thesis investigates the performance developments and relationships of digital ventures. Thereby, it contributes to existing theory by exploring performance development patterns and the relationship between critical performance measures. It illustrates possible performance trajectories and the impact of growth for the benefit of digital managers and entrepreneurs, who may use this information to improve the survival and viability of their ventures. From a theoretical perspective, this thesis pursues these goals by integrating and formalising two complementary growth process theories: growth paths theory (Garnsey, 1998) and dynamic states theory (Levie and Lichtenstein, 2010). They describe the development and performance of companies over time (Davidsson, Achtenhagen

and Naldi, 2010). This thesis contextualises these two theories using the emerging literature on digital ventures to apply these general theories to the specific context.

The sections below outline the literature on digital business ventures, identify current research questions regarding their performance, present growth process theories as a conceptual basis to investigate these questions, and outline the approach taken in this thesis to answer them.

1.1. Digital ventures

Digital ventures face unique challenges and opportunities because of digital products and services' unique characteristics compared to physical ones (Hull et al., 2007; Kraus et al., 2018; Zaheer et al., 2019). These differences include the range of market opportunities, ease of market entry, pursuit of growth, competitive threats, and need to innovate continuously. The digital economy continues to develop, and new technologies emerge to address changing customer needs. Therefore, digital ventures can pursue a vast amount of growth opportunities (Kraus et al., 2018). Digital ventures can also establish their technologies at significantly lower costs than physical products. Thus, entering a market to pursue opportunities is much easier for digital companies (Kraus et al., 2018). Due to these opportunities and the ease to pursue them, digital ventures are noted for their growth potential (Zhang, Lichtenstein and Gander, 2015; Täuscher and Abdelkafi, 2018; Huang, Henfridsson and Liu, 2021). Such growth may also make digital ventures' products more attractive depending on their business model design due to, for example, network effects (Amit and Zott, 2001; Huang et al., 2017). However, these advantages also apply to competitors and new entrants. Therefore, digital ventures may struggle to sustain a competitive advantage and an ability to capture value (Steininger, Wunderlich and Pohl, 2013). Moreover, intellectual property rights for digital technologies are weak. While copyrights protect software, competitors can replicate software products with different software code. Digital ventures thus need to continuously improve their technology to stay ahead of competitors (Nambisan, 2017; Kraus et al., 2018; Zaheer, Breyer and Dumay, 2019). Digital technologies allow doing so by

2

remaining modular, reprogrammable, and editable even after selling them and scaling them up on the market (Nambisan, 2017).

These unique opportunities and challenges have distinct effects on a digital venture's success, which requires further investigation (Kraus et al., 2018; Zaheer et al., 2019; Zaheer, Breyer and Dumay, 2019). For example, different opportunities may be differently attractive and require careful selection and continuous adaptation by managers and entrepreneurs of digital ventures (Ries, 2011; Ojala, 2016). The ease and need to grow may entice them to scale up prematurely and pressure their resources (Marmer *et al.*, 2011). Continuous competitive threats may reduce growth and profitability (Teece, 2018). Thus, "a prominent research theme within digital entrepreneurship that needs to be investigated is digital start-up performance" (Zaheer et al., 2019, p. 263).

1.2. Performance outcomes and identification of research questions

Two critical performance measures for all companies, including digital ventures, are their growth in value creation and their ability to capture value (Hull et al., 2007; Huang et al., 2017; Hsieh and Wu, 2019). Value creation refers to a venture's generation of benefit for its customers. Customers assess the value of the venture's product subjectively based on its attributes, and their needs and circumstances. This value can be expressed in monetary terms through customers' willingness to pay for the product if there is a single source of supply (Brandenburger and Stuart, 1996; Bowman and Ambrosini, 2000; Bowman and Ambrosini, 2010). For example, an individual customer may determine the value of a product by specifying the amount he is willing to pay based on the product's performance, taste, or colour (Bowman and Ambrosini, 2000). On a company level, value creation thus refers to the sum of value created for all individual customers that bought a product or service in a period (Lepak, Smith and Taylor, 2007). The captured value represents the share of the created value received from customers and maintained by the venture as profit. The amount of value captured is equal or smaller than the amount of value created. Firstly, the revenue received from customers is usually lower than their willingness to pay due to competition. Secondly, companies incur the costs for suppliers and employees that help them

3

create customer value. Whereas customers benefit from value creation, the venture benefits from capturing value as it increases the company's resources (Brandenburger and Stuart, 1996; Lepak, Smith and Taylor, 2007; Bowman and Ambrosini, 2000; Bowman and Ambrosini, 2010). Whereas growth in value creation increases the maximum amount of value that could be captured by the venture, the ability to capture value specifies the portion of that value that the firm actually captures.

Performing well across both outcomes is critical for digital ventures. Firstly, value creation determines the potential for value capture. Therefore, growing the amount of created value is the essence of entrepreneurship, and ambitious entrepreneurs should strive to maximise the value they create for their customers (Hull *et al.*, 2007; Hermans *et al.*, 2012). The resources generated through the company's value creation and ability to capture value provide reserves and the means for reinvestments. Thus, companies should maximise the share of the value they capture (Sirmon and Hitt, 2003; Garnsey, Dee and Ford, 2006; Ketchen, Ireland and Snow, 2007). Together, the growth in value creation and the ability to capture value are holistic performance outcomes for companies. Therefore, they are considered the critical outcomes of business models, entrepreneurial activity, and strategic decision-making(Bowman and Ambrosini, 2000; Amit and Zott, 2001; Zott and Amit, 2007; Steffens, Davidsson and Fitzsimmons, 2009; Zott, Amit and Massa, 2011; Steininger, 2019).

However, some digital ventures may focus on growing their value creation without developing an ability to capture value. For example, digital entrepreneurs may focus on scaling up their customer and user base because they "typically face significant growth expectations" (Eisenmann, 2006; Marmer *et al.*, 2011; Huang *et al.*, 2017; Zaheer *et al.*, 2019; Huang, Henfridsson and Liu, 2021, p. 3). Thereby, they grow their value creation by creating value for more customers. However, their ability to grow value creation does not imply an ability to capture value. For example, they may enter competitive markets that pressure prices, incur short-term costs of growth, or grow sales of unprofitable products (Bowman and Ambrosini, 2000; Bowman and Ambrosini, 2007). To improve their ability to capture value, they need to strengthen their bargaining power to customers, possibly focusing on a smaller, more profitable market, and control costs – either option reduces their growth in value creation (Hirakubo and Friedman, 2008). Focusing on growth in value creation without an ability to capture value, digital ventures lose resources during the growth process. They may try to compensate for this resource loss by raising capital to avoid bankruptcy (Marmer *et al.*, 2011; Monteiro, 2019). However, capturing value is critical for their long-term survival and sustainability (Thiel and Masters, 2014). For example, during economic crises and after capital market bubbles, new capital may not be forthcoming. At that point, the venture needs to sustain itself with the resources it generates (Jain, Jayaraman and Kini, 2008; Ojala, 2016). Therefore, this thesis aims to develop an understanding of the requirements and priorities, allowing digital ventures to develop an ability to capture value while growing their value creation. It investigates two specific research questions that contribute to achieving this aim:

- 1. How do value creation and capture develop for digital ventures?
- 2. What relationship(s) exist(s) between value creation and capture?

To answer these questions and achieve its aim, this thesis uses System Dynamics modelling to synthesise growth paths theory and dynamic states theory. As the third section of this chapter describes in detail, growth paths theory focuses on value capture in a cumulative process that leads to changes in firm size. Dynamic state theory focuses on value creation in discontinuous growth states that describe firm characteristics. Synthesising these views allows for a comprehensive description of the growth process. The System Dynamics model, developed based on these theories, can illustrate the complex relationships illustrated in the theories and in performance development over time. Answering the research questions using System Dynamics can thus illustrate performance trajectories and the impact of growth on digital entrepreneurs and managers. It can also describe the mechanisms driving diverse performance developments and settle the discussion about growth-profitability trade-offs and benefits. The two subsections below review the literature regarding those two questions and outline the need to investigate them further.

1.2.1. Development of value creation and capture

A company's ability to grow value creation and capture value changes as it grows. Conceptual papers have previously argued that as companies grow, they become better at capturing value and worse at increasing the value they create (Ireland, Hitt and Sirmon, 2003; Ketchen, Ireland and Snow, 2007; Hitt et al., 2011). While this is true on average, companies develop diversely through different performance configurations (Steffens, Davidsson and Fitzsimmons, 2009; McKelvie and Wiklund, 2010). For example, Steffens, Davidsson and Fitzsimmons (2009) find that many, but not all, young firms can grow in sales. Moreover, their ability to turn higher sales into profitability differs diversely¹. In a similar study, the same authors ordered Australian and Swedish companies into five performance configurations based on their sales growth and return on assets: poor, growth, profit, middle, and star (Davidsson, Steffens and Fitzsimmons, 2009). They then analysed how companies transition from the growth and profit configurations, where companies excel at one outcome, to the star and poor configurations, where they excel at both or no outcome. Figure 1-1 visualises their Swedish data set with the transition of companies from growth and profit to star and poor configurations over two years.



Figure 1-1: Profit-growth transitions (own graphic based on data collected by Davidsson, Steffens and Fitzsimmons (2009))

¹ Empirical studies commonly use revenue and profitability measures as proxies for the more abstract concepts of value creation and the share of value captured (Porter, 1985; Garnsey, Dee and Ford, 2006; Steffens, Davidsson and Fitzsimmons, 2009).

Figure 1-1 shows that companies are more likely to excel at both performance outcomes (transition to star configuration) if they previously achieved profitability rather than sales growth. However, the data shows that firms' performance configurations develop diversely. For example, 32.8 per cent of companies in the profit configuration developed to the star configuration. Another 19 per cent developed to the poor configuration. Here, the data set becomes incomplete as the authors only required developments from profit and growth to poor and star configurations in their analysis. The study does not account for 48.2 per cent of profit companies' transitions. These companies either stayed within the profit configuration or developed to the growth or middle configurations (Davidsson, Steffens and Fitzsimmons, 2009). Other authors investigating the transition between performance configurations have also observed this diversity in development paths (Brännback *et al.*, 2009; Jang, 2011).

While these studies provide empirical insights into the diverse development of performance configurations, the two performance outcomes "evolve in complex, multidimensional ways that are not well understood" (Steffens, Davidsson and Fitzsimmons, 2009, p. 126). Because previous authors studied a process that unfolds over time using variance-based methodologies, the studies cannot explain why specific mechanisms would dominate the development of some firms but not that of others (Pentland, 1999; Langley et al., 2013). Future research should explore the impact of explanatory variables on these different paths (Brännback *et al.*, 2009). Moreover, these mentioned studies have not investigated the development of value creation and capture in digital ventures. Therefore, this thesis investigates how value creation and capture develop over time for digital ventures.

1.2.2. Relationship(s) of value creation and capture

The above illustration of the development of value creation and capture also points to the relationship between the two performance outcomes. Davidsson, Steffens and Fitzsimmons (2009) illustrate that companies with high growth and low profitability rates are more likely to subsequently perform poorly regarding both outcomes than companies with high profitability but low growth (Figure 11). Moreover, high growth companies with low profitability are also less likely than their counterpart to subsequently excel at both outcomes. Therefore, the authors argue that growth does not lead to profitability, but profitability may drive future growth.

However, empirical studies regarding the impact of profitability (i.e. ability to capture value) on growth (in sales and value creation) are inconclusive (Table 1-1). Some researchers argue profitability should drive growth as it illustrates that a company has developed a competitive advantage and its profit can fund its growth (Davidsson, Steffens and Fitzsimmons, 2009; Delmar, McKelvie and Wennberg, 2013; Nkwor and Ikpor, 2019). Other researchers find that profitability deteriorates growth because managers may forgo growth opportunities to maintain profitability or because profitable technology firms invest in product improvements rather than growth (Lee, 2018; Tong and Serrasqueiro, 2020). Yet others find no effect of profitability on growth, for example, because ventures can finance growth with external funding rather than profits (Markman and Gartner, 2002; Coad, Rao and Tamagni, 2011).

Relationship	Literature	
Profitability drives growth	Chandler and Jansen, 1992; Russo and Fouts, 1997; Roper, 1999; Mendelson, 2000; Brush, Philip and Hendrickx, 2000; Goddard, Molyneux and Wilson, 2004; Peng, 2004; Cowling, 2004;	
	Fitzsimmons, Steffens and Douglas, 2005; Cho and Pucik, 2005; Brännback <i>et al.</i> , 2009; Steffens, Davidsson and Fitzsimmons, 2009; Davidsson, Steffens and Fitzsimmons, 2009; Jang, 2011; Jang and	
	Park, 2011; Delmar, McKelvie and Wennberg, 2013; Adams <i>et al.</i> , 2014; Giotopoulos, 2014; Tien and Yang, 2014; Yazdanfar and	
	Öhman, 2015; Kachlami and Yazdanfar, 2016; Maichel-Guggemoos and Wagner, 2019; Nkwor and Ikpor, 2019; Simbaña-Taipe <i>et al.</i> ,	
	2019; Fernández-López <i>et al.</i> , 2019; Yadav, Pahi and Gangakhedkar, 2021; Eling, Jia and Schaper, 2021	
Profitability	Reid, 1995; Lee, 2014, 2018; Yoo and Kim, 2015; Tong and	
deteriorates growth	Serrasqueiro, 2020	
Profitability does not	Markman and Gartner, 2002; Coad, 2007; Bottazzi et al., 2010;	
affect growth	Coad, Rao and Tamagni, 2011; Arrighetti and Lasagni, 2013;	
	Federico and Capelleras, 2015; Ertan, Lewellen and Thomas, 2020; Léon, 2020	
Growth drives	Capon, Farley and Hoenig, 1990; Chandler and Jansen, 1992;	
profitability	Geroski, Machin and Walters, 1997; Russo and Fouts, 1997; Roper,	
	1999; Mendelson, 2000; Brush, Philip and Hendrickx, 2000; Peng, 2004; Cowling, 2004; Cho and Pucik, 2005; Coad, 2007;	
	Asimakopoulos, Samitas and Papadogonas, 2009; Bottazzi <i>et al.</i> , 2011; Delmar, McKelvie and Wennberg, 2013; Lee, 2014; Yazdanfar	
	and Öhman, 2015; Yoo and Kim, 2015; Federico and Capelleras,	
	2015; Senderovitz, Klyver and Steffens, 2016; Fuertes-Callén and	
	Cuellar-Fernández, 2019; Maichel-Guggemoos and Wagner, 2019; Nkwor and Ikpor. 2019: Yaday. Pahi and Gangakhedkar. 2021	
Growth deteriorates	Reid, 1995; Fitzsimmons, Steffens and Douglas, 2005; Brännback et	
profitability	bility <i>al.</i> , 2009; Davidsson, Steffens and Fitzsimmons, 2009; Steffens, Davidsson and Fitzsimmons, 2009; Jang, 2011; Jang and Park, 2011;	
	Nakano and Kim, 2011; Tong and Serrasqueiro, 2020	
Growth does not	Markman and Gartner, 2002; Bottazzi <i>et al.</i> , 2010; Lee, 2018; Ertan,	
affect profitability	Lewellen and Thomas, 2020	

Table 1-1: Empirical literature on the growth-profitability relationship

The research regarding the impact of growth on profitability is similarly inconclusive. Some researchers find a positive effect of growth on profitability because it, for example, creates learning effects and economies of scale (Katz and Shapiro, 1985; Delmar, McKelvie and Wennberg, 2013; Fuertes-Callén and Cuellar-Fernández, 2019). Others find the opposite, a negative effect of growth on profitability (Davidsson, Steffens and Fitzsimmons, 2009; Steffens, Davidsson and Fitzsimmons, 2009; Nakano and Kim, 2011). Theoretical arguments for such a relationship include, for example, the organisational turmoil created by growth and adjustment costs (Nicholls-Nixon, 2005; Greve, 2008; Penrose, 2009). Lastly, some authors find no effect of growth on profitability because the positive and negative effects of growth balance out one another (Markman and Gartner, 2002; Ertan, Lewellen and Thomas, 2020).

While most studies find a statistically significant positive relationship between growth and profitability, the strength of this relationship differs widely (Davidsson, Steffens and Fitzsimmons, 2009; Senderovitz, Klyver and Steffens, 2016). Some find only a very weak correlation between the two (e.g. Roper, 1999; Cho and Pucik, 2005). Others find a strong correlation between the two (e.g. Chandler and Jansen, 1992; Mendelson, 2000). However, despite the strong correlation, Chandler and Jansen (1992, p. 232) conclude that growth and profitability are "separate dimensions, and variables related to one are not necessarily related to the other, indicating the importance of considering both." Moreover, differences regarding the relationship can be observed for different industry contexts (Brush, Philip and Hendrickx, 2000; Delmar, McKelvie and Wennberg, 2013), growth and profitability measures (Coad, Rao and Tamagni, 2011; Yoo and Kim, 2015; Fuertes-Callén and Cuellar-Fernández, 2019), and firm characteristics like company size (Yadav, Pahi and Gangakhedkar, 2021), information technology systems (Senderovitz, Klyver and Steffens, 2016), and strategy (Schlichter, Klyver and Haug, 2021).

The literature has also not investigated this relationship for digital ventures, although firm and industry characteristics affect it. Managers and entrepreneurs of digital ventures may argue that scaling-up, i.e. growing quickly, improves their ability to capture value (Zhang, Lichtenstein and Gander, 2015; Huang *et al.*, 2017; Zaheer *et al.*, 2019). While some of the above literature supports this view, it is also contradicted by empirical evidence in general samples of small and medium enterprises and other types of technology companies (Fitzsimmons, Steffens and Douglas, 2005; Brännback *et al.*, 2009; Davidsson, Steffens and Fitzsimmons, 2009; Steffens, Davidsson and Fitzsimmons, 2009). Therefore, this thesis focuses on a specific investigation of the relationship between growth in value creation and the ability to capture value for digital ventures.

1.3. Growth process theories as a conceptual basis

This thesis builds on the conceptual basis of growth process theories to answer the two research questions. Growth process theories conceptualise growth as "a process of development [...] in which an interacting series of internal changes leads to increases in size" (Penrose, 2009, p. 1). These theories describe a company's internal changes and developments over time (Davidsson, Achtenhagen and Naldi, 2010). They can also be used to explain the development of value creation and capture (Pitelis, 2009).

1.3.1. Organisational life-cycle theory

Traditionally, organisational life-cycle (OLC) theory in the form of stage models has been the most influential growth process theory. Based on the analogy of living organisms, OLC theory assumes that companies are born, grow, and decline or renew themselves (Davidsson, Achtenhagen and Naldi, 2010). The theory postulates that all companies follow the same pattern of size changes (Levie and Lichtenstein, 2010). Stages of development have been added to the life-cycle to illustrate different firm characteristics as companies change in size. The theory defines each stage as a "unique configuration of variables related to organizational context and structure" (Hanks *et al.*, 1993, p. 7; Hoy, 2006). Thereby, different OLC models have identified different stages (Figure 1-2). For example, in their seminal article on OLC theory, Miller and Friesen (1984) identified different firm characteristics along the stages of birth, growth, maturity, decline, and revival. A model specifically developed for technology ventures has been proposed by Kazanjian and Drazin (1990). It includes the stages of conceptualisation, commercialisation, growth, and stability.



Figure 1-2: Organisational life-cycle and examples of stage models developed by Miller and Friesen (1984) and Kazanjian and Drazin (1990)

OLC theory provides some insight into the development of performance outcomes. For example, because companies develop new products and establish them on the market, the stages up to maturity/stability are more closely associated with growth in value creation (Miller and Friesen, 1984; Hanks *et al.*, 1993). During the maturity/stability and decline stage, companies are good at capturing value. While companies do not seek new opportunities during the maturity/stability stage, their established size should lead to economies of scale. During the revival stage, the company needs to enter new markets and launch new products. Thus, it increases the value it creates. Otherwise, adverse environmental factors such as increasing competition might cause the firm's decline (Miller and Friesen, 1984).

However, empirical studies have falsified OLC stage models due to their deterministic description of the growth process (Davidsson, Achtenhagen and Naldi, 2010). While changing firm characteristics could be identified as firms grow, research shows that one cannot order them in a single path that describes the development of all companies (Miller and Friesen, 1984; Stubbart and

Smalley, 1999; Stam, Garnsey and Heffernan, 2006). Moreover, stage models may be particularly unsuitable for digital ventures (Garnsey and Heffernan, 2005; Stam and Garnsey, 2006; Levie and Lichtenstein, 2010). Therefore, they "should no longer be used by scholars of entrepreneurship, for they act as a barrier to [the] advancement of research on the growth of entrepreneurial organizations" (Levie and Lichtenstein, 2010, p. 333).

Growth paths theory (Garnsey, 1998) and dynamic states theory (Levie and Lichtenstein, 2010) have been proposed as alternative theories explaining the growth process. They have been identified as the contemporary contributions that strive to advance growth process theories after OLC theory. Alternative approaches exist in the literature. However, these two theories have been highlighted as being "one of few recent efforts" and an "important vantage point". They also have been repeatedly mentioned in reviews of firm growth research (Davidsson, Achtenhagen and Naldi, 2010, p. 126; Leitch, Hill and Neergaard, 2010; McKelvie and Wiklund, 2010).

1.3.2. Growth paths theory

Growth paths theory aims to explain the diverse changes in size that companies experience. While OLC theory proposes a single growth pattern, growth paths theory identifies a range of development patterns. For example, Garnsey and Heffernan (2005) identified patterns in high tech companies, including continuous growth and growth setbacks, plateaus, or growth delays (Figure 1-3).



Figure 1-3: Growth paths identified by Garnsey and Heffernan (2005)

Based on Penrose' (2009) 'The Theory of Growth of the Firm', growth paths theory proposes the processes that may create these different growth patterns. It argues that digital ventures "match opportunities and resources in such a way as to initiate new activity that can create value" (Garnsey, 1998, p. 527). These opportunities emerge in the external environment, and a venture aims to pursue them utilising its resources and capabilities. If revenues from customers exceed the costs of inputs, the venture captures part of the value it creates for customers. These captured resources become part of the firm's financial slack. Thereby, they increase the venture's size, which the theory defines as the venture's total resources (Garnsey, Dee and Ford, 2006; Penrose, 2009). The financial slack may then be reinvested to pursue additional opportunities, creating an iterative process leading to different growth patterns (Stam, Garnsey and Heffernan, 2006). However, growth requires the venture's management to plan and execute the reinvestment activities. Therefore, the managerial slack slows down the iterative growth process in which resources and capabilities develop (Garnsey, 1998; Hugo and Garnsey, 2005; Penrose, 2009; Garnsey, Lubik and Heffernan, 2015; Miozzo and DiVito, 2016).

Growth paths theory advances growth process theories by highlighting the existence of different growth patterns. The theory also starts explaining these paths utilising Penrose's (2009) seminal work on company growth. However, the theory often remains abstract, reducing its implications for practitioners (Davidsson, Achtenhagen and Naldi, 2010). It requires further research to investigate and explain these different paths (Garnsey, Stam and Heffernan, 2006).

1.3.3. Dynamic states theory

Due to insufficient empirical evidence of stage models, researchers have proposed state models as an alternative theory for explaining the growth process. They relax the assumption that all firms grow through the same sequence of stages. Developing this conceptualisation further from a complexity theoretical perspective, Levie and Lichtenstein (2010) developed their theory of dynamic states. Their theory accepts that companies grow through different organisational configurations, i.e. states (Levie and Lichtenstein, 2010). However, dynamic

14

states theory rejects stage models' assumptions of a specific number and predetermined sequence of stages. Instead, each state represents the entrepreneur's attempts to match the venture's business model efficiently and effectively to the external environment. Thereby, the activities, resources, and capabilities; the collaborations and supply chain; and the market position of the business model (Figure 1-4) are continuously adapted to create customer value. Entrepreneurs and managers will adapt their business models based on their dominant logic. This logic expresses their ambition, understanding of the environment, and their belief of what it takes to be successful in that environment (Levie and Lichtenstein, 2010).



Figure 1-4: Elements of a dynamic state (Levie and Lichtenstein, 2010, p. 332)

Utilising Levie and Lichtenstein's (2010) original framework, Brown and Mawson (2013) identify a diverse range of trigger points that cause punctuated change and the transition between states. These may be endogenous (e.g. new product introduction, changing management), exogenous (e.g. macroeconomic changes), and co-determined (e.g. acquisitions, fundraising) trigger points. These trigger points cause punctuated changes between states (Harbermann and Schuilte, 2017).

Dynamic states theory presents a conceptualisation of states theory that avoids the criticism of stage models (Davidsson, Achtenhagen and Naldi, 2010). However, further research is required to explain the development experienced by companies in specific contexts and growth states (Levie and Lichtenstein, 2010).

1.3.4. The complementary nature of growth paths and dynamic states theory

Research suggests moving beyond OLC stage models to advance growth process theories (Davidsson, Achtenhagen and Naldi, 2010; Levie and Lichtenstein, 2010). Growth paths theory and dynamic states theory have been suggested as alternative conceptualisations of the growth process. They are complementary and compatible regarding their investigation of the growth process, underpinnings, description of the growth process, and performance outcomes (Table 1-2).

	Growth paths theory	Dynamic states theory
Main underpinning	Process philosophy and research	Complexity science and complex
		systems
Description of	Growth as a continuous and cumulative	Periods of stability for the firm with
growth process	process	punctuated change
	 Resources and capabilities 	Business model (including
	accumulate over time	activities, resources, and
	 Ventures pursue opportunities in 	capabilities; the collaborations
	the external environment and are	and supply chain; and strategy) is
	under its influence	adapted
	Planned and implemented by managers	Managers and entrepreneurs adapt the
	and entrepreneurs	business model based on their
		dominant logic
Element of the	Starts with <i>change in size</i> and proposes	Investigates the <i>change in firm</i>
growth process	mechanisms causing the change	characteristics (i.e. business model)
		without considering size
Focus on	Focuses on value capture as it increases	Focuses on value creation in its
performance outcome	the firm's size	conceptual framework

Table 1-2: Comparison of growth paths theory and dynamic states theory

Both theories make use of similar philosophical underpinnings. Growth paths theory uses process research and philosophy, which investigates the temporal sequence of events (Van de Ven, 1992; Langley, 1999; Tsoukas and Chia, 2002). Dynamic states theory draws from complexity science and complex systems, and considers a large number of interrelated elements producing surprising, non-linear, and emergent effects (Anderson *et al.*, 1999; McKelvey, 2004). These two perspectives are related as tools from complexity science can be used to investigate processes (Anderson and Meyer, 2016). These philosophical underpinnings are adopted in this thesis and further discussed in the methodology section. Due to their similar philosophical underpinnings, the theories' different conceptualisations can be integrated to develop a comprehensive growth process theory.

Due to their different underpinnings, each growth process theory presents a different account of the growth process. In growth paths theory, the growth process is a continuous process. The theory conceptualises it as an accumulation of resources and capabilities over time. This accumulation depends on the management's ability to plan and reinvest for further growth (Garnsey, 1998). In dynamic states theory, the growing firm is presented as experiencing periods of stability with punctuated changes. During periods of stability, it maintains its business model. During transitions between states, the venture adapts its business model based on its managers' dominant logic (Levie and Lichtenstein, 2010). Both development patterns can be observed in companies and are thus necessary to explain the complete growth process (Hamilton, 2012).

The growth process is composed of "internal changes [which] lead to increases in size" (Penrose, 2009, p. 1). Therefore, growth process theories need to account for two aspects: internal changes and increases in size. OLC theory has previously done so. However, growth paths and dynamic states theory each provide a complex account of a company's growth process that focuses on, respectively, the increase in size or the internal changes. Growth paths theory explains the change in size on an abstract level (Davidsson, Achtenhagen and Naldi, 2010). Dynamic states theory focuses on the firm's elements, conceptualised as the business model, that change over time while not accounting for size in its framework (Levie and Lichtenstein, 2010). A comprehensive theory of growth requires the integration of dynamic state theory's changes in firm characteristics and growth paths theory's size changes.

Lastly, the two theories are complementary regarding their focus on performance outcomes. Both theories mention value creation and capture throughout their conceptualisation. Growth paths theory focuses on value capture. It explains that the captured value becomes part of the venture's resources. Because it defines size as the total of all resources, value capture contributes directly to the venture's change in size (Garnsey, 1998). However, the theory acknowledges that capturing value requires value creation (Stam and Garnsey, 2006). Dynamic states theory focuses on value creation. It includes value creation in its framework (Figure 1-4). However, the theory acknowledges that ventures do not just need to create value

17

for customers. They also need to generate resources to sustain their growth state (Levie and Lichtenstein, 2010).

1.4. Thesis outline

This thesis thus utilises the theoretical foundations of growth paths theory and dynamic states theory to answer the above research questions about the development and relationship of value creation and capture for digital ventures. It strives to combine the two theories and overcome their individual limitations. It does so by building a System Dynamics model, which has been selected because it can answer the research questions while fitting to growth process theories' philosophical underpinnings and conceptualisations. This selection is further illustrated in Chapter 2 of this thesis. This methodology chapter also illustrates the foundations of the System Dynamics method and the process of developing, testing, and applying the model. This thesis thus follows the structure common in simulation research projects by placing the methodology chapter before the literature review.

Chapters 3 and 4 present the developed model and the theoretical basis used to develop it. The literature on growth paths theory, dynamic states theory and their foundational literature are combined to conceptualise the model (Chapter 3). Thereby, the thesis reveals the complex mechanisms affecting digital ventures' performance outcomes. Chapter 4 presents the formalisation and specification of the conceptual model for digital ventures as a mathematical model applying the principles of System Dynamics.

Chapters 5 and 6 simulate the model for case studies of real companies and hypothetical scenarios. Chapter 5 validates the model for four case studies of digital ventures. The chapter illustrates the model's ability to approximate the development of two software-as-a-service and two digital marketplaces over three to six years. Chapter 6 presents the simulation of the model for theory-driven but realistic hypothetical cases. Simulating different managerial decisions in different contexts allows identifying their impact on performance developments and relationships.
Chapter 7 discusses the implications of the model, the case studies, and the simulations for the two research questions and the aim of this thesis. Chapter 8 concludes this thesis with a discussion of its contribution, the strengths and limitations of its approach, and suggestions for future research.

Chapter 2: System Dynamics modelling methodology

This chapter presents the methodology and methods used to investigate the development and relationship of value creation and capture for digital ventures. Section one outlines the philosophical underpinnings of this research project. The subsequent section then proposes System Dynamics as a suitable and appropriate modelling method that suits these philosophical underpinnings, the research questions, and utilised theories. While the third section describes the development of the model, the fourth section focuses on its simulation for validation and insight generation.

2.1. Philosophical underpinnings and methodological requirements

Business and management scholars make use of different research philosophies that guide their research. These reflect a researcher's view on ontology and epistemology. Ontology refers to whether reality exists independently and objectively. Epistemology refers to what constitutes acceptable knowledge of that reality (Saunders, Lewis and Thornhill, 2012; Bryman and Bell, 2015). As illustrated in the introduction, this thesis builds on growth paths theory and dynamic states theory. These theories utilise process philosophy and complexity science, which adopt realism (ontology) and subjectivism (epistemology). This thesis also adopts these philosophy and complexity science below².

2.1.1. Process philosophy and research

Growth paths theory employs process philosophy and research (Garnsey, Stam and Heffernan, 2006). Central to process philosophy is the ontological view that

² The philosophical paradigm guides the entire research project. Therefore, it could also be argued, in reverse, that a researcher holds philosophical beliefs before selecting research questions and theories. The philosophy then implicitly triggers specific research questions and conceptualisations. These questions and conceptualisations would not be selected when holding another worldview (Bryman and Bell, 2015). For example, a researcher holding a process philosophical viewpoint may be more inclined to select a research question that investigates and explains change over time (Langley, 1999; Tsoukas and Chia, 2002; Chia, 2005).

reality consists of processes rather than entities. Processes have primacy over entities because entities only exist as manifestations that result from underlying processes at a point in time. Over time, processes change entities (Langley et al., 2013; Helin et al., 2014), such as digital ventures and their characteristics and performance. Process philosophy adopts an approach that is both realist (ontology) and subjectivist (epistemology). Thus, processes and manifestations of entities are assumed to exist objectively. However, our knowledge and understanding of these entities and processes are subjective and may differ (Chia, 1999). Process philosophy focuses on the emergence, change, and transformation that creates entities' manifestations and attributes (Chia, 1999; Tsoukas and Chia, 2002; Nayak and Chia, 2011). It prompts research that "is concerned with understanding how things evolve over time and why they evolve in this way" (Van De Ven and Huber, 1990, in Langley, 1999, p. 692). The introduction of this thesis is framed within such research questions. Process research then aims to identify the mechanisms and contingencies behind developments over time (Tsoukas and Hatch, 2001; Van De Ven and Poole, 2005).

In this context, the five main aspects that resonate with process philosophers are (Helin *et al.*, 2014):

- *Temporality*. While many organisational theories implicitly assume time, time is often not explicitly considered. Process research takes the passage of time into account to understand and explain evolving phenomena (Langley et al., 2013; Helin et al., 2014; Tsoukas, 2017).
- *Wholeness*. Rather than taking a reductionist approach that permits studying individual parts, process research postulates that change has a holistic and systemic effect. Therefore, process research needs to cover all parts of the whole, allowing connections between the parts, which increases complexity (Langley et al., 2013; Helin et al., 2014).
- Openness. Due to these connections, process research needs to emphasise the context of processes (Helin et al., 2014). This context gives rise to heterogeneity (Chia, 1999; Tsoukas and Chia, 2002; Hernes, 2008), which the introduction has illustrated for digital ventures' performance outcomes.
- *Force*. What drives and hinders processes are different forces, such as power, resistance, or pressure to achieve goals (Helin et al., 2014). These may be

endogenous or exogenous. Over time, they work as contingencies through the mechanisms underlying the processes (Pentland, 1999; Van De Ven and Poole, 2005).

• *Potentiality*. Potentiality reflects what might become a reality in the future through the processes (Helin et al., 2014). Thereby, what currently exists shapes what can be actualised as time passes. For example, current capabilities may determine a venture's growth potential. At the same time, current capabilities result from previous potentialities, such as learning and training (Hernes, 2008).

2.1.2. Complexity science and complex systems

Complexity science is concerned with systems composed of many elements and the behaviour created by the interrelationship between these elements (Dooley, 1997; McKelvey, 2004). Particularly dynamic states theory employs it explicitly to conceptualise new ventures' development process (Levie and Lichtenstein, 2010). Therefore, this thesis will need to employ research methods that can cope with complex systems to answer its research questions. Complexity science studies systems' behaviours and their diverse outcomes using formal models (McKelvey, 2004). Therefore, complexity science is well suited to investigate digital ventures' diversely developing performance outcomes (Davidsson, Steffens and Fitzsimmons, 2009; McKelvie and Wiklund, 2010).

The critical tool of complexity science is the notion of complex systems. Complex systems are composed of a large number of elements that interact dynamically through many connections. These relationships may be circular through positive and negative feedback (McKelvey, 2004; Lichtenstein and Plowman, 2009). Over time, this feedback causes non-linear behaviour where the system's outputs are disproportional to the system's inputs (McKelvey, 2004; Meyer, Gaba and Colwell, 2005). This non-linearity makes the systems particularly sensitive to their initial condition. Moreover, complex systems are open and affected by exogenous influences. Thus, different initial conditions and exogenous influences cause a system's behaviour over time (Dooley, 1997; McKelvey, 2004).

To investigate complex systems, researchers need to uncover a system's elements and the connections between elements, external influences, and time delays to investigate complex systems. Such research requires a combination of thick and thin descriptions (McKelvey, 2004):

- Thin descriptions use very few variables to explain outcomes using, for example, statistical analyses. Thin methods include traditional quantitative approaches, for example, simple and multi-variable regression analyses. These methods can generate generalisable results by utilising large samples. However, they often restrict the number of data points collected for each case in the sample (Saunders, Lewis and Thornhill, 2012; Bryman and Bell, 2015). These methods are thus unsuitable for investigating the development of complex systems because many relevant elements and their change over time are excluded (McKelvey, 2004). Some traditional quantitative methods allow for more complex relationships or longitudinal analyses. For example, structural equation modelling allows for more complex relationships between concepts and their measurements, and econometric methods allow for an investigation over time. However, both methods are still restricted to compared to the real structure of complex systems - very few variables and rely on statistical inferences rather than causality and feedback that is required to understand them (Langley, 1999; McKelvey, 2004; Davis, Eisenhardt and Bingham, 2007; Miller and Tsang, 2010).
- Thick descriptions provide detailed accounts of developments, for example, in a single firm. They use traditional qualitative methods such as case studies or narrative inquiries facilitated using interviews with executives and thematic analysis. These methods have the advantage of uncovering the complex systems and processes that cause company development over time. However, due to the magnitude of data collected for each case, these methods often restrict themselves to few cases (Saunders, Lewis and Thornhill, 2012; Bryman and Bell, 2015). Therefore, they are unable to adequately test and generalise their findings (McKelvey, 2004; Schwaninger and Grösser, 2008; Miller and Tsang, 2010).

Instead of relying on traditional methods, McKelvey (2004) suggests combining thick and thin research approaches using simulation modelling. Computer models

can provide comprehensive accounts of the mechanisms causing the development of individual cases. Such longitudinal case studies allow the uncovering of detailed, complex descriptions of their developments (Yin, 1981; Eisenhardt, 1989). They are thus able to provide thick descriptions. Researchers may also test simulation models using case study data. In addition, simulations allow the production of insight beyond cases and historical developments (McKelvey, 2004; Anderson and Meyer, 2016). Thereby, computer simulation can be generalised and provide thin descriptions. This thesis employs this combination of simulation modelling and case study testing to take advantage of their ability to generate rich and broad insight about the research questions that traditional quantitative or qualitative methods cannot generate.

2.2. Selection of System Dynamics modelling

Simulation modelling is the process of developing a formal model and experimenting with the model under varying conditions (Berends and Romme, 1999; Harrison *et al.*, 2007; Schwaninger and Grösser, 2008). A formal model is a computational representation of reality based on the underlying theory of the processes and system at investigation. Formal models express this theory mathematically (Davis, Eisenhardt and Bingham, 2007; Harrison *et al.*, 2007). Thereby, simulation modelling is situated at a *"sweet spot"* of theory development and testing (Davis, Eisenhardt and Bingham, 2007). Formal models expressing a theory can be tested and compared to actual, historical data. Moreover, researchers can use them for experimenting in a risk-free environment and without being constrained by the accurate measurement of constructs. Simulation modelling is thus a critical method to investigate complex systems (Langley, 1999; McKelvey, 2004; Anderson and Meyer, 2016).

2.2.1. Selection of System Dynamics as a modelling method

There are several modelling approaches, including agent-based modelling or fitness landscapes³. This project has selected System Dynamics. It is concerned with understanding complex, dynamic, time-dependent systems, and their behaviour. It is a rigorous modelling method that investigates the structure of systems with its circular causalities and feedback. These give rise to the system's non-linear behaviour (Coyle, 1996; Keating, 1999; Sterman, 2000; Morecroft, 2015). Davis, Eisenhardt and Bingham (2007) argue that simulation methods should be selected based on their fit to the underlying assumptions, research questions, and utilised theory. These aspects, which illustrate the suitability of SD for this research project, will be explored in the following paragraphs. They also compare System Dynamics to alternative simulation modelling techniques.

Underlying assumptions. System Dynamics is the simulation method that best fits this research project's philosophical underpinnings of process research and complexity science.

- Firstly, as its name suggests, System Dynamics focuses on the change of complex systems. It can provide thick descriptions of individual business cases by representing their continuous change across all relevant elements rather than focussing on, for example, individual agents or entities like agent-based modelling or discrete event simulations (Harrison *et al.*, 2007; Dangerfield, 2014). Like other simulation modelling approaches, the model's descriptions can be tested before generating data beyond observed cases (Sterman, 2000; Davis, Eisenhardt and Bingham, 2007).
- Secondly, System Dynamics addresses all five elements of process philosophy. It explicitly considers the temporality of processes by simulating delays and accumulation over time (van Oorschot *et al.*, 2013). Being designed to investigate systems, it can incorporate a high number of interacting elements, processes, and endogenous variables. It can also consider context explicitly through exogenous variables (Sterman, 2001; Azoulay, Repenning and Zuckerman, 2010; Rotaru, Churilov and Flitman, 2014). A system's development is entirely dependent on initial conditions,

³ For a comprehensive review, see Davis, Eisenhardt and Bigham (2007) and Lichtenstein (2011).

exogenous influences, and the defined relationships. Therefore, System Dynamics models account for potentiality through path-dependence (Größler, Thun and Milling, 2008).

Research Questions. System Dynamics suits the research questions by unveiling mechanisms, generating practical recommendations for individual companies, and simulating variables even where exact measurement is impossible.

Firstly, System Dynamics uncovers and illustrates complex causal relationships (Davis, Eisenhardt and Bingham, 2007). Thereby, System Dynamics models combine the strength of qualitative and quantitative methods. While qualitative methods such as Soft Systems Methodology can uncover complex relationships and visualise them, they are unable to test, generalise, and simulate identified relationships (Forrester, 1994; Wolstenholme, 1999; Coyle, 2000). Traditional quantitative methods such as structural equation modelling also include simple visualisations besides mathematical relationships. However, these are often restricted to few variables and linear, one-directional relationships. They are thus unsuitable for modelling the feedback loops that drive value creation and capture (Davis, Eisenhardt and Bingham, 2007; Harrison et al., 2007). Other simulation modelling techniques, such as agent-based modelling, represent models without graphical representations (Coyle, 1996; Größler, Thun and Milling, 2008; Lichtenstein, 2011). System Dynamics models illustrate complex relationships graphically and mathematically. The graphical representation improves understanding of the model and allows a qualitative analysis of relationships and feedback loops compared to other simulation modelling approaches. System Dynamics models scale this connection of graphical and mathematical relationships to very complex models compared to the simple relationships expressed by traditional quantitative methods (Wolstenholme, 1999; Coyle, 2000, Sterman, 2000). Moreover, to improve comprehension of large models, System Dynamics includes additional tools to communicate overall insights of the model (Morecroft, 1982; Sterman, 2000). Lastly, the link between graphical and mathematical relationships ensures that even large, complex models can be audited and tested more thoroughly before simulating the mathematical relationships (Barlas, 1996).

- Secondly, System Dynamics can generate practical recommendations for individual ventures by building scenarios and determining optimal policies through simulation (Sterman, 2000; Täuscher, 2018). System Dynamics does so for a focal case company, rather than staying on the industry or population level like other simulation modelling methods such as fitness landscapes or genetic algorithms (Davis, Eisenhardt and Bingham, 2007). Moreover, unlike quantitative methods such as structural equation modelling, it can consider the exact conditions of a focal company rather than just making general statistical inferences based on few variables. Therefore, System Dynamics models like the one developed in this thesis can be used by practitioners to monitor, forecast, and manage their businesses, ensuring the practical relevance of academic work (Vhzquez, Liz and Aracil, 1996; Winch and Arthur, 2002).
- Lastly, simulation models are not constrained by the exact measurement and data availability of all variables (Langley, 1999; Harrison et al., 2007). Value creation and the share of value captured, which are central to this thesis, cannot be measured accurately. Therefore, previous empirical studies used proxies to measure those performance outcomes. Such proxies commonly included revenue, revenue growth, and return on assets (Garnsey, Dee and Ford, 2006; Davidsson, Steffens and Fitzsimmons, 2009; Steffens, Davidsson and Fitzsimmons, 2009). System Dynamics allows calculating performance measures consistent with their theoretical conceptualisation and setting model inputs without numerical data for all variables. Instead, many System Dynamics models are parameterised based on analyst judgments and automatic model calibration when numerical values are unavailable (Sterman, 2000; Oliva, 2003, 2004; Walker and Wakeland, 2011). Therefore, System Dynamics is well suited to investigate the development of value creation and capture and can approximate those performance outcomes consistent with theory.

Utilised theories. Davis, Eisenhardt and Bingham (2007) argued that simulation modelling works best for simple theories that the selected modelling method can represent. Simple theories are underdeveloped. They may include only limited empirical or analytical grounding, few propositions, or an unformalised underlying logic (Davis, Eisenhardt and Bingham, 2007). Dynamic states and

growth paths theory are such theories. Researchers have proposed them only in the last decade (dynamic states theory) or the last two decades (growth paths theory). They require further exploration because this short time frame limits the original and direct contributions to the theories (Davidsson, Achtenhagen and Naldi, 2010). Simple theories such as growth process theories can guide the modelling process, and formal modelling can advance them by identifying additional factors, relationships, and logical flaws (Malerba et al., 1999; Davis, Eisenhardt and Bingham, 2007). Contemporary growth process theories represent growth as a complex, dynamic, and cumulative process (Garnsey, 1998; Levie and Lichtenstein, 2010). System Dynamics can capture this representation by incorporating constant change among variables governed by feedback, nonlinearity, path-dependent behaviour, and stock accumulation (Sterman, 2001; Levie and Lichtenstein, 2010). Thereby, System Dynamics fits better to the theories utilised in this thesis than other modelling approaches that are less cumulative and more stochastic or adaptive like genetic algorithms or fitness landscapes. Moreover, System Dynamics models are uniquely suited to account for accumulation over time and time delays that the utilised theories suggest, which many quantitative methods such as regression analysis or structural equation modelling cannot consider adequately (Barlas and Carpenter, 1990; Vhzquez, Liz and Aracil, 1996; Davis, Eisenhardt and Bingham, 2007; Sterman, 2018).

2.2.2. System Dynamics modelling

Seminal authors have proposed slightly different processes to develop System Dynamics models (Table 2-1). The process starts with articulating the problem that requires a dynamic solution. Modellers then conceptualise the problem or system using causal loop diagrams. Through these diagrams, modellers formulate and express their dynamic hypothesis, their theory that accounts for the behaviour of the system. Modellers then formulate stock-and-flow diagrams and a formal simulation model based on these conceptualisations. Modellers and this thesis use the term formalisation to refer to developing stock-and-flow diagrams and mathematical relationships (Sterman, 2000; Harrison *et al.*, 2007; Ford, 2019).

Modellers then test the formal model by simulating it and comparing model approximations to actual data. After successful testing, the model can be simulated for hypothetical scenarios to evaluate the impact of varying organisational policies (Coyle, 1996; Sterman, 2000).

Forrester, 1994	Coyle, 1996	Sterman, 2000; Morecroft, 2015	Martinez-Moyano and Richardson, 2013	This thesis
	Problem recognition	Problem articulation	Problem identification and definition	Introduction Research need
System description	Causal loop diagrams Qualitative analysis	Formulation of dynamic hypothesis	System conceptualisation	Model development: • Theory integration and model conceptualisation • Theory contextualisation and model formalisation
Conversion to equations	Simulation modelling	Formulation of simulation model	Model formulation	
Model simulation	Model testing	Testing	Model testing and evaluation	Model simulation:Case simulation
Design alternative policies	Policy design	Policy design and evaluation	Model use, implementation, and dissemination	and validation Scenario simulation
Educate and debate Implement changes in policies	Exploratory modelling and policy optimisation		Design of learning strategy/infrastructure	

Table 2-1: Overview of proposed System Dynamics modelling approaches

This thesis breaks the modelling process activities into three steps. As stated in the introduction, this thesis aims to develop an understanding of the requirements and priorities allowing digital ventures to develop an ability to capture value while growing their value creation. The remainder of this chapter describes the steps to develop and simulate the model. First, this thesis develops a model in two steps that correspond to the chapters presenting the model. One chapter conceptualises the model based on the literature on growth process theories and identifies the main variables and feedback loops affecting performance outcomes. The other chapter specifies the model based on the literature on digital ventures and converts details of stock-and-flow formulations into mathematical equations. Second, the model simulated. This thesis simulates the model for case companies to test the model's ability to approximate companies' developments over time. It then simulates hypothetical scenarios to generate data and insight regarding the development and relationships of value creation and capture. Through these steps, this thesis reveals the mechanisms affecting these performance outcomes and explores the causes of their diverse development. They form a continuous circle of model development and simulation. Each step builds on the others, with continuous learning and iterations, to develop, simulate, and apply the model (Figure 2-1). Each iteration of this continuous development cycle adds to understanding the influences on performance outcomes and their development.



Figure 2-1: System Dynamics modelling process

The two sections below illustrate the steps to develop and simulate the model. Each of their subsections considers an individual step within development and simulation.

2.3. Model development

This thesis develops a System Dynamics model based on established theory to understand the development and relationships of value creation and capture. By building the model based on established theory, this thesis ensures that the model is generally applicable and theoretically informed (Coyle, 1996; Langley, 1999; Sterman, 2000). However, the model goes beyond visualising and formalising the established theories. This thesis aims to develop the theories further. It integrates growth paths theory and dynamics states theory, focuses the theories on performance outcomes, and contextualises them for digital ventures.

Due to the complexity of the phenomena under investigation, this thesis presents the developed System Dynamics model on different levels of aggregation. These levels of aggregation reflect different levels of detail provided about variables. On the one hand, too many details may hinder the communication of the central feedback loops that drive behaviour. On the other hand, too few details may create doubts and confusion about the relationships between variables (Sterman, 2000). Therefore, Coyle (1996) argues that modellers should overcome this tension by presenting causal loop diagrams with different levels of detail. In his illustration (Figure 2-2), Coyle suggests four such levels of detail. However, these levels are not strict guidelines. Instead, the levels of detail illustrated in a modelling project need to be selected based on the model's purpose and complexity. The size of the ellipses reflects the level of detail of each causal loop diagram. Modellers may use different variable names across these ellipses to aggregate variables but must maintain the overall insight and relationships between the elements across different detail levels. Causal loop diagrams should reflect the most fundamental insight and significant results as the most aggregated version. Modellers present the formal model's computer code with its fine details about variables as the most detailed version.



Figure 2-2: Cone of causal loop diagrams (adapted based on Coyle, 1996, p. 44)

This thesis also uses these different levels of detail and aggregation. Two separate chapters present the model at different levels of detail. Thereby, this thesis utilises the advantages of different System Dynamics conceptualisation tools. The subsubsections below review these tools. The structure also aims to indicate how the different types of literature – growth process theories and digital ventures – have been used in this thesis.

- Chapter 3 of this thesis focuses on model conceptualisation based on growth process theories. It builds and illustrates the model with its most important relationships driving the changes in performance outcomes. It identifies these relationships by integrating and synthesising growth process theories and their background literature. This description includes only the most essential details of digital ventures. The final causal loop diagram provides a strategic overview and summary of feedback loops affecting value creation and the share of value captured.
- Chapter 4 of this thesis focuses on contextualising and formalising the model for digital ventures. Additional details are added to the model utilising the literature on digital ventures. The chapter presents these details along subsystems using stock-and-flow diagrams which provide additional information required for model formalisation. It then illustrates the equations of the formal model. The end of each subsystem provides a model boundary chart to illustrate its links to other subsystems and exogenous model inputs.

The two model development chapters work as a tandem and are complementary. While the first chapter provides the blueprint and outline for the second chapter, the second chapter provides detailed evidence for the aggregated links in the first chapter. While the first chapter focuses on growth process theories, the second chapter focuses on the literature on digital ventures. While the first chapter utilises a more discursive and strategic language and tone, the second chapter focuses on the precise nature of variables and equations. Thereby, both are the result of the iterative modelling process outlined above. This process also includes the verification of conceptualisations and formalisations. The three subsections below illustrate the activities executed to develop these two versions of the model and their verification.

2.3.1. Model conceptualisation

A System Dynamics model has been conceptualised based on the literature on growth paths theory and dynamics states theory, and illustrated using causal loop diagrams.

2.3.1.1. Literature review

This thesis has synthesised the literature on the two theories to identify their relevant concepts and relationships. Existing literature reviews on growth process theories have identified the seminal articles on growth paths and dynamic states theory (see Davidsson, Achtenhagen and Naldi, 2010; Leitch, Hill and Neergaard, 2010; McKelvie and Wiklund, 2010). The author has identified additional contributions to the theories through forward and backward citation analyses. He has pursued this citation analysis iteratively and as a snowballing process until no additional contributions to the theories were identified. Thereby, this thesis has developed a comprehensive literature database on the two growth process theories (Tranfield, Denyer and Smart, 2003; Denyer and Tranfield, 2009; Wohlin, 2014; Bandara *et al.*, 2015). This literature was treated as data and analysed thematically to develop a conceptual framework and specify its elements and relationships (Bandara *et al.*, 2015). Previous reviews on digital ventures have also used this approach (see Zaheer, Breyer and Dumay, 2019). For example, this

allowed the identifying of different resources, capabilities, and activities; different influences on performance outcomes; and different environmental factors.

In a second step, the author has identified secondary literature on which the growth process theories build. Growth is a holistic phenomenon that requires drawing from multiple business and management research fields (Leitch, Hill and Neergaard, 2010; McKelvie and Wiklund, 2010). Therefore, the author has evaluated the reference lists of seminal contributions to identify relevant background literature. This background literature provided additional insight into the development of the identified elements. For example, growth paths theory acknowledges that companies require complementary assets (Garnsey, 1998). Secondary literature further conceptualises complementary assets and their influence (e.g. Teece, 1986, 1998). This process adds details and relationships to the previously established framework (Bandara *et al.*, 2015).

2.3.1.2. Causal loop diagrams

The model conceptualisation chapter uses causal loop diagrams to communicate the variables and relationships identified in the synthesis of growth process theories and their foundational literature. Causal loop diagrams reveal the processes and mechanisms affecting variables of interest, including delays and exogenous variables (Figure 2-3). They capture System Dynamics' proposition that a system's behaviour emerges from feedback mechanisms among the variables (Sterman, 2000; Größler, Thun and Milling, 2008; Morecroft, 2015). Such causal loop diagrams illustrate the modeller's hypothesis about the causes of the system's and variables' dynamic changes. Therefore, they are commonly used to summarise a model and provide an overview of its most important insights. The causal loop diagrams used in the subsequent chapter of this thesis thus illustrate the most relevant feedback loops. As in other System Dynamics modelling processes, the causal loop diagrams developed in this thesis have also been analysed qualitatively. The combination of feedback loops is evaluated regarding their potential impact on performance outcomes over time (Forrester, 1994; Albin and Forrester, 1997; Wolstenholme, 1999; Sterman, 2000; Homer and Oliva, 2001).



Figure 2-3: Example of a causal loop diagram

Each causal loop presented in a causal loop diagram consists of a set of variables that influence another circularly. The polarity illustrates the relationship between two variables (indicated with + or - on the arrow). The balance of polarities determines whether a loop is balancing or reinforcing (indicated with a B or R). While balancing loops represent dampening and goal-seeking feedback mechanisms that counteract change, reinforcing loops cause their own growth (Sterman, 2000; Morecroft, 2015). For example, such a goal-seeking mechanism may entice a company to borrow more money to raise its cash balance to its liquidity goal. New borrowing increases a company's interest expenses. It then reduces the cash balance, creating a reinforcing loop due to even more borrowing. The causal effects of feedback loops may be delayed (double crossed lines), which explicitly account for temporality (Sterman, 2000; Morecroft, 2015). Moreover, a feedback loop may be affected by exogenous variables or parameters. These are not part of the loop itself but affect it (Coyle, 1996), such as the interest rate in the illustration above. These exogenous influences capture the openness of processes considered in process research (Azoulay, Repenning and Zuckerman, 2010).

2.3.2. Model formalisation

This thesis then builds on these causal loop diagrams to specify and formalise the model for the context of digital ventures. Because the literature on growth process theories applies to a diverse range of companies, the processes they describe remain abstract (Davidsson, Achtenhagen and Naldi, 2010). Adding literature on digital ventures allows specifying, for example, how resources or activities materialise for digital ventures. In addition, some elements included in growth

process theories may also not apply to digital ventures (e.g. inventories), while they require additional elements (e.g. different business model designs). The author has identified such adaptations by reviewing contradictory evidence between the growth process literature and digital venture literature. The identified variables and relationships in growth process theories have thus been challenged critically using the literature on digital ventures. This thesis uses stock-and-flow diagrams, subsystem diagrams, and model boundary charts to communicate this detailed model version. Equations are then formulated based on the stock-andflow diagrams.

2.3.2.1. Stocks-and-flow diagrams, subsystems, and model boundaries This thesis utilises stock-and-flow diagrams to illustrate the detailed version of the model. They capture additional information required for formal models, such as the nature of variables. Stocks, also called levels and represented by boxes, preserve their value. They accumulate through their inflows and outflows, illustrated by arrows and valves. Some System Dynamics modellers also refer to flows as rates to highlight that they represent instantaneous rates of changes, i.e. at any point in time, a rate variable illustrates the amount the stock changes per period. Consistent with process philosophy, stocks and flows thus capture that stocks as entities are not existent by themselves. Instead, they are a temporal manifestation of the continuous in- and outflow processes changing them. When modelling businesses using System Dynamics, balance sheet items should be modelled as stocks. Items from the income or cash flow statement should be flows or auxiliary variables. Therefore, most performance variables used by managers reflect outcomes achieved over a period of time. Examples include revenue or profit, which are usually modelled as flows or auxiliary variables (Coyle, 1996; Albin and Forrester, 1997; Sterman, 2000; Warren, 2002; Morecroft, 2015). The two balance sheet items – *cash* and *amount borrowed* – in the example above (Figure 2-3) are converted into stocks (Figure 2-4). They are not altered directly but only through flows such as revenues and expenses increasing or decreasing cash (Sterman, 2001).



Figure 2-4: Example of stocks-and-flow-structure

Stocks require initialisation to simulate models. The processes and causalities represented by causal loops will then change the initial value of stocks via the inflows and outflows over time (Größler, Thun and Milling, 2008). This process is path-dependent, meaning the initial value of a stock directs the behaviour of the system. In theory, modellers could reduce the complexity of all System Dynamics models to stocks and flows. However, they usually include auxiliary variables in the model to detail causal chains (Morecroft, 1982; Coyle, 1996; Albin and Forrester, 1997).

The detailed version of the model cannot be captured in a single stock-and-flow diagram because it contains too many variables and relationships. Each stocksand-flow diagram presented in Chapter 4 of this thesis thus captures a small part of the model. The chapter uses two additional tools to communicate how these individual parts of the model interact with another:

- *Subsystem diagrams* illustrate the major parts or sectors of the model, which group variables together. Such groups may be established based on, for example, organisational units or conceptual similarities between variables (Morecroft, 1982; Sterman, 2000). Such subsystem diagrams communicate the model's boundaries. They also help structure the model's presentation by signposting which model elements are described in which part of the documentation (Morecroft, 1982; Sterman, 2000).
- *Model boundary charts* are a critical tool to analyse System Dynamics models. These tables explicate and assess the model's endogenous (calculated in the model), exogenous (external influence affecting calculations), and

excluded variables (not considered in a model). By highlighting the model's scope, model boundary charts allow challenging the assumptions underlying the model's overall structure. Together with initial conditions, exogenous influences determine the balance of feedback loops within the model. Explicating these variables allows challenging whether exogenous influences should be part of the model's feedback structure (Albin and Forrester, 1997; Sterman, 2000). Chapter 4 of this thesis presents a model boundary chart at the end of each subsystem. These illustrate the connections to other subsystems and the exogenous parameters influencing this part of the model.

2.3.2.2. Variable formulation

The relationships between variables illustrated in stock-and-flow diagrams have been converted to mathematical equations to develop a formal model. This process went back and forth between the detailed stock-and-flow diagrams and equations. The author used Coyle's (1996) common module method to facilitate this iterative process. It models a set of connected variables utilising reappearing structures in System Dynamics models (e.g. s-curves, goal-seeking behaviour). Sterman's (2000) extensive System Dynamics modelling guide and previous papers have been used to identify appropriate structures. They have been used to rework stock-and-flow diagrams and formulate their equations. Moreover, the author assigned each variable a unit based on the concept it represents in reality.

The nature of variables identified in stock-and-flow diagrams determines which equations are allowed in System Dynamics models. This thesis has utilised different approaches to set equations for stocks, hard variables, and soft variables.

- System Dynamics formulates the accumulation process of stocks through its associated flows using integration. Therefore, the value of a stock is calculated by integrating its net flow and adding it to its initial value. The net flow represents the rate of change per unit of time, which the software calculates by subtracting the outflows from the inflows (Coyle, 1996; Sterman, 2000; Warren, 2002; Harrison *et al.*, 2007; Dangerfield, 2014; Morecroft, 2015).
- Flow and auxiliary variables are not constrained to integration. Their equations are derived by connecting their affecting variables based on the

underlying theory and conceptualisation. In the case of variables representing hard concepts, such as amounts of money or employees, equations can be derived by reviewing the concept each variable represents (Coyle, 1996; Sterman, 2000; Warren, 2002; Dangerfield, 2014; Morecroft, 2015). For example, the *liquidity gap* in Figure 2-4 might represent the absolute difference between the company's *liquidity goal* and its current *cash position*. Therefore, one may calculate it by subtracting the cash at hand from the liquidity goal. Modellers can calculate the *interest expenses* by multiplying the *amount borrowed* with the *interest rate*.

System Dynamics models can also account for soft or qualitative variables. These variables may include, for example, product quality or a company's capabilities. While analysts cannot assess these variables accurately and measure them numerically, some qualitative information might exist. System Dynamics modes should include soft variables because their exclusion would assign them the value and impact of zero – the only value and impact known to be false (Sterman, 1991, 2000, 2001, 2002; Forrester, 1994; Homer and Oliva, 2001; Größler, 2007; Kunc, 2017, 2018). Two approaches have been used in this thesis to formulate soft variables and connect them to the hard variables: Firstly, some authors have suggested using *indexed variables*. Thereby, the modeller assigns a soft variable the value of one at the beginning of the simulation. The modeller needs to normalise the impact of influencing hard variables to achieve this indexing, usually by dividing the current by the influence's initial value. Different influences can then be added up, multiplied, or weighted using parameters representing the influencing variables' effect sizes. Modellers may then use a standard rate to connect the soft variable to the hard variables of the system. This standard rate represents the effect of the indexed variable on a hard variable. The standard rate may be based on expert judgements or calibrated automatically by the modelling software (Hennessy, 1997; Sterman, 2000; Walker and Wakeland, 2011). Secondly, modellers suggest using soft variables as filters with ranges between zero and one. The variable has its most significant impact at a value of one and no impact at zero (Coyle, 2000; Warren, 2002; Dangerfield, 2014). Warren (2002) uses such filter variables to model capabilities compared to best practice or a theoretical maximum. A capability may, for example, reflect the average employees' ability to acquire new customers. Together with the best practice or maximum, this capability affects the outcome of the capability on a hard variable (Coyle, 1996; Sterman, 2000; Dangerfield, 2014).

Variables in System Dynamics models are assigned units, which modellers need to set according to the concepts they represent. Variables' units fulfil two critical purposes. Firstly, units aid modelling equations because an outcome variable's unit needs to correspond to the unit that emerges from its equations. Therefore, the units of the variables provide insight into how modellers need to combine variables in equations. When verifying the model (see Subsection 2.3.3), a dimensional consistency test ensures that all equations fulfil this requirement and the model's equations can easily be checked. Secondly, these units aid in simulating and validating the model because the units hint at the data points corresponding to the variables. This thesis has adhered to four rules when assigning units (Coyle, 1996; Sterman, 2000; Warren, 2002; Morecroft, 2015):

- The units of a flow have the corresponding stock's units per unit of time. Modelling business problems, balance sheet items are assigned units of dollars (in this thesis, irrespective of the actual currency). Their flows such as cash flow, depreciation, revenue are thus assigned units of dollars per period. Human resources are commonly measured in the number of employees, and their flows like hiring have the units of employees per period. Customers are measured in the number of customers. They are acquired and lost through variables with the units of customers per year (Coyle, 1996; Sterman, 2000; Warren, 2002; Dangerfield, 2014; Morecroft, 2015).
- Fractions are always dimensionless, shorted DMNL in models. Such fractions express, for example, the fraction of sales spend on research and development.
- Fractional flow rates have units of one per period. For example, these reflect the interest rate, which expresses how many units of money in interest expenses a business incurs for each unit of money per period. This term can be shortened to one per unit of time.
- The unit of time has been set to one year to align with data availability in annual reports.

2.3.3. Model verification

All models represent a simplified version of reality developed from a specific perspective. Model testing is thus concerned with building confidence in the model and ensuring the model is appropriate for its intended purpose (Coyle, 1996; Sterman, 2000; Morecroft, 2015). In the iterative model development process (Figure 2-1), modellers may then reconceptualise the model or alter its equation (Forrester and Senge, 1979; Sterman, 2000). Model testing can be split into verification and validation (Kleijnen, 1995; Sargent, 2013). The next section further discusses validation as part of the model's simulation (Subsection 2.4.1). Verification ensures that the model implements the underlying theories accurately by ensuring the model's equations are free of inconsistencies and errors (Sargent, 2013). Forrester and Senge (1979), Barlas (1996), Sterman (2000), and Morecroft (2015) define the verification tests for System Dynamics models:

- A first test to be executed is a *structure verification test*. It compares the modelled feedback loops with information about the system, such as academic literature (Barlas, 1996). This test also ensures that the model is free of inconsistencies, including inappropriate assumptions, such as negative stocks (Barlas, 1996; Sterman, 2000). For example, the number of employees of a venture cannot logically fall below zero. Minimum and maximum values have been set for variables to apply this test to the developed model. The Vensim software warns the modeller when a simulation breaches these values.
- A *parameter verification test* checks that exogenous parameters match the knowledge of the system conceptually and numerically. Conceptually, all parameters must represent identifiable elements of the real system. Numerically, one must then be able to approximate all parameters with sufficient confidence (Forrester and Senge, 1979; Barlas, 1996; Sterman, 2000; Morecroft, 2015). This thesis thus defines all exogenous variables to illustrate the elements they represent in reality. It proposes measures to estimate inputs utilising the information presented in annual reports (see Appendix C).

- A *boundary adequacy test* ensures that all relevant feedback loops are included in the model to satisfy the model's purpose (Forrester and Senge, 1979; Morecroft, 2015). The two tests previously introduced provide information regarding this inclusion. For example, negative stocks often indicate that a modeller has not considered a balancing feedback structure that reduces outflows to zero. Additionally, another look at the model's parameters is required. Modellers should include the feedback structure that causes inputs to change if these inputs are not constant over the period of interest (Coyle, 1996; Sterman, 2000). This test has been executed when validating the model with case data (Subsection 2.4.1) by assessing if input parameters are reasonably constant within a growth state.
- A *dimensional consistency test* analyses the model's equations. It ensures that the equations used to connect variables compute an output with the same units as assigned during the formalisation step (Forrester and Senge, 1979; Barlas, 1996). The simulation software can operate this test automatically (Coyle, 1996; Oliva, Sterman and Giese, 2003).
- Equations also need to compute reasonable outputs for all possible inputs. An *extreme conditions test* evaluates the equation's output under extreme input values (Forrester and Senge, 1979; Barlas, 1996; Sterman, 2000; Morecroft, 2015). These extreme conditions may be the parameter's minimum or maximum values assigned to them during formalisation or extraordinarily low and high numbers.

The thesis has used these five interrelated tests to verify the model thoroughly. They ensure that the formal model implements the interpretation and conceptualisation of growth process theories and the literature on digital ventures. If the model fails any of these tests, further model development takes place to refine the model. Thereby, the iterative process illustrated above (Figure 2-1) continued until a verified model was developed.

2.4. Model simulation

Causal loop diagrams and the other conceptualisation tools already provide insightful descriptions of the system and processes at work. Simulating the

corresponding formal model provides the means to test the model, confirm its structure, and generate further insight. Thereby, further confidence in the model's ability to approximate actual developments is established. Moreover, System Dynamics models often include feedback loops that work against another. In such cases, analysts cannot predict the system's behaviour because it depends on the balance of feedback loops. Simulating the model allows determining the balance of feedback loops and generating data through the model (Wolstenholme, 1999; Sterman, 2000; Homer and Oliva, 2001; Kunc, 2017). This thesis simulates the model for case studies of digital ventures to validate the model (Chapter 5). It then simulates hypothetical scenarios to generate further insight about value creation and capture (Chapter 6). Both types of simulation are discussed in the two subsections below. Simulations are then analysed to answer the research questions outlined in the introduction and achieve the aim of this thesis.

2.4.1. Case simulation and model validation

The model is first simulated for case companies to validate it. This validation tests the model's ability to approximate the development of companies over time. Similar to verification, validation allows establishing the level of confidence in the model's approximations (Kleijnen, 1995; Sargent, 2013). The selection of case companies, the model's set up for each company, and the comparisons of model and historical data is illustrated below.

2.4.1.1. Case study identification and selection

Case companies have been identified through an extensive and rigorous search and filtering process of two different sources. The first source is the Financial Times's (FT) 1000 fastest-growing companies lists. The lists have been published since 2017 and include companies that have submitted entries for inclusion in the ranking and companies identified through the FT's additional research. The lists cover the 1000 European-based companies with the highest organically achieved compound annual revenue growth rates over the three years before the submission period⁴. The second source used to identify companies was the NASDAQ's initial public offerings (IPO) calendar. It includes companies that went public on major US stock exchanges and NASDAQ-owned international stock exchanges. These have been included because the NASDAQ is the leading technology stock exchange (Geron, 2013).

The 2017, 2018, and 2019 editions of the FT's list (total of 3000 companies) and all companies on the IPO calendar that went public between January 2010 and December 2019 (total of 1868 companies) have been considered. These companies have been filtered through numerous steps. The first two steps have only been applied to the FT lists.

- From the FT lists, only companies based in the United Kingdom have been included to ensure that their annual reports can be accessed through the university's database subscriptions and to avoid language barriers.
- Because some companies may be included in multiple editions of the FT list, duplicates among the lists have been eliminated.
- Company descriptions and websites have then been scanned to identify digital ventures. Companies providing physical products or services have been eliminated from the sample as they do not fit the characteristics required for this investigation. Many companies eliminated through this step are natural resource, energy, manufacturing, electronics, or biotechnology companies.
- Financial technology companies have been excluded because their growth behaviour deviates significantly from producing companies. They are generally excluded from growth process theories (Penrose, 2009).
- Digital ventures with all required data points (see Appendix C) for model inputs and outputs over at least four years have been selected. Many companies included on the FT lists are exempt from publishing full annual reports due to their small size and private ownership. Thus, a significant number of companies had to be excluded through this filter.

⁴ For example, companies in the 2019 list are ranked based on their revenue growth rate between 2014 and 2017. To be considered in the ranking, companies must have achieved at least 100,000 EUR in revenues (or currency value equivalent) in the first year and 1.5 million EUR in revenues in the final year of the three-year period. Companies must also be independent, i.e. not branches or subsidiaries of other companies.

• Because this project aims to understand how digital ventures can develop an ability to capture value while growing their value creation, only companies with initially negative profitability have been selected. Companies growing to develop an ability to capture value often do so with their first product and organically. Therefore, companies that grow through acquisitions, diversification, or new product introduction have been excluded.

Through this filtering, four companies have been identified. These allow testing the model in various settings and across different industries, business models, locations, sizes, growth rates, and profitability measures. This diversity is required to ensure the model can approximate companies in various situations and identify the boundaries for which the model is suitable (Groesser and Schwaninger, 2012). For example, while two companies operate as software-as-aservice (SaaS) providers, two companies operate online marketplaces. This diversity allows testing the model for a variety of digital business models. While the FT companies are smaller with higher growth rates, the NASDAQ list includes more mature companies with lower growth rates. Therefore, the model can be tested against different development paths and sizes. As a requirement of this project's ethical approval, companies remain anonymous. The terms Alpha, Beta, Gamma, and Delta will be used throughout this thesis to refer to the four companies.

2.4.1.2. Model parameterisation and calibration

Initial values and exogenous parameters must be set to simulate the model for each case company (Sterman, 2000, 2001). System Dynamic modellers should use all available data, not just numerical time-series data (Forrester, 1987). Similarly, case studies can use qualitative and quantitative information (Yin, 1981; Eisenhardt, 1989). Therefore, annual reports have been used by System Dynamics modellers (Oliva, Sterman and Giese, 2003; Sterman *et al.*, 2007) and growth process researchers (Kor and Mahoney, 2000; Greve, 2008; Gabrielsson and Gabrielsson, 2013; Miozzo and DiVito, 2016). Annual reports include reliable quantitative financial and operational information as well as important qualitative management commentary (Cowton, 1998; Rowbottom and Lymer, 2010). For System Dynamics modellers, this balance allows setting model inputs based on the quantitative data without subjective assessments while also providing them with contextual information. For growth process researchers, annual reports have the advantage of being a reliable source of information because they are legally bound to include accurate information. They are also permanent documents that avoid hindsight bias (Nikula *et al.*, 2010; Saunders, Lewis and Thornhill, 2012; Bryman and Bell, 2015).

Annual reports have first been analysed qualitatively to identify growth states. As suggested by dynamic states theory, the management may alter the business model periodically. Thus, the firm exhibits periods of stability punctuated by abrupt change (Levie and Lichtenstein, 2010; Brown and Mawson, 2013; Harbermann and Schuilte, 2017). These periods of stability have been identified by analysing the qualitative information in annual reports using a time-ordered matrix (see Miles, Huberman and Saldana, 2014) and temporal bracketing (see Langley, 1999).

- A time-order matrix organises qualitative data along two axes: time and categories (Miles, Huberman and Saldana, 2014). The qualitative information in annual reports has been reviewed for important information about input variables and output variables. Relevant information about variables has been coded regarding the year they refer to (time) and the variables (categories). A coding scheme with years and variables has been set up in NVivo, and the matrix has been created using the software's matrix search function. This computer-aided qualitative analysis has the advantage of streamlining the process of qualitative data analysis. It allows researchers to maintain an overview of collected and coded information over hundreds of pages of annual reports and multiple reports per company. Computer-aided analysis has also improved data analysis using automatically generated matrixes (Yearworth, 2010; Bandara *et al.*, 2015; Gaur and Kumar, 2018).
- Temporal brackets in case companies' developments have been identified using these time-ordered matrixes. In Langley's (1999) temporal bracketing strategy, phases of development are identified. Elements of interest remain constant during each phase, while these elements change when transitioning between phases. As Langley (1999) argues, these phases are not a predictable, theoretical sequence but instead describe and order case data. Therefore,

temporal bracketing fits well the assumptions of growth states in dynamic states theory. These brackets have been identified by reviewing the statements made each year about variables. Years are summarised and divided into growth states based on constant input variables.

Parameter values must be set for all input variables based on the company's growth states. Dynamic states theory's conceptualisation of growth states indicates that some variables are constant during a growth state (Levie and Lichtenstein, 2010; Brown and Mawson, 2013; Harbermann and Schuilte, 2017). Therefore, most inputs variables were held constant during a growth state.⁵ Based on the qualitative analysis of annual reports, the analyst has assessed if the change in a growth state justifies a change in an input variable. If a change could not be justified, the variable has been held constant over multiple growth states. For example, a company no longer planning to scale up its customer base justifies lowering a target growth rate. However, it does not justify changing the market prices of resource inputs. System Dynamic modellers use a combination of techniques to estimate input variables. These fall into three categories (Sterman, 2000, 2001; Morecroft, 2015; Kunc, 2018):

- Hard, quantitative variables can usually be set based on case data with no or only minimal transformation. Such hard variable might, for example, refer to the initial number of employees or revenue in different years. Such hard variables are set based on annual reports' financial and operational data (Morecroft, 2015). Numerical data has been indexed to obscure the identity of case companies as required by the ethical approval for this project. Due to this indexing, many initial values have been set as one (further details on indexing in Chapter 4 with the relevant variables). Where hard input variables are constants over multiple years, the average over these years in which the variable was held constant has been used.
- Soft, qualitative variables such as product quality or switches implemented in the model to reflect different policies can be set based on the judgement of the analyst. Guidelines to set variables based on a qualitative analysis of annual

⁵ Only a few selected input variables have been allowed to change during growth states and from year to year. These variables relate to accounting figures such as taxation, which are subject to complex rules (i.e. tax law) that are not the subject of this thesis.

reports have been developed for such switches and soft variables. These guidelines are based on previous research and the model's equations. For example, initial capabilities, which are qualitative variables between zero and one, are assessed using the capability maturity model with five levels (Garud, Kumaraswamy and Sambamurthy, 2006; Belt *et al.*, 2009; van Oorschot *et al.*, 2013).

Some model parameters may be set using automatic model calibration. Such automatic model calibration is used in System Dynamics when no data is available to estimate an input variable. Examples may include standard rates that are used in some common System Dynamics structures such as s-curves. System Dynamics software includes features to estimate such unknown variables. The software identifies input values that achieve the closest fit between other observable variables and their historical data (Kleijnen, 1995; Berends and Romme, 1999; Sterman, 2000; Walker and Wakeland, 2011; Rahmandad and Sterman, 2012). This calibration is an important part of model validation. It tests if the modelled structure can replicate the behaviour observed in variables of interest (Lyneis and Pugh, 1996; Oliva, 2003, 2004; Homer, 2012). Oliva (2003) further highlights that modellers should evaluate calibrated inputs for their feasibility and consistency. Regarding feasibility, modellers should evaluate if set values are theoretically possible. Regarding consistency, modellers should evaluate if values align with other information available. For example, negative values for employee turnover would be a breach of feasibility. Employee turnover of over a hundred per cent may be inconsistent. Modellers should question it because industry averages are usually much lower (see Baron, Hannan and Burton, 2001; Gjerløv-Juel and Guenther, 2019).

Appendix C lists all hard, soft, and automatically calibrated variables. For variables set based on quantitative or qualitative information, the appendix also includes calculations, transformation/indexing, and other guidelines to set each variable. For automatically calibrated variables, the appendix includes details on the feasibility and consistency checks. The author has collected all input values for a company in a standardised Excel spreadsheet. The Vensim software reads the input values from the spreadsheet to run the model.

2.4.1.3. Model validation

Validation tests the model's ability to approximate the system's behaviour (Sargent, 2013). Modellers usually execute it through a behavioural reproduction test. It tests the model's ability to replicate the system's symptoms, fluctuations, and modes of behaviour (Forrester and Senge, 1979; Barlas, 1996; Sterman, 2000; Morecroft, 2015). This thesis validates the developed model through a subsystem strategy. It compares the development of hard, quantitative variables to case companies' historical data graphically (Chapter 5).

Researchers need to decide which variables will be compared to historical case data to validate the model. By simulating the model for each company with the company's unique model inputs, the model computes the development of each endogenous variable for each company. This thesis compares all subsystems' hard, quantitative outputs to historical data to holistically validate the model. Validating each subsystem ensures that individual parts of the model are functioning as intended. It allows the spotting of additional errors that are cancelled out when reviewing only value creation and capture as the focal performance outcomes of this thesis. Moreover, designing subsystems with hard, quantitative outputs ensures that these outputs are objectively measurable. It avoids subjective assessment and the researcher's bias when comparing outputs to case data (Homer, 1996, 2012; Oliva, 2003; Martinez-Moyano and Richardson, 2013).

The outcomes of each subsystem are compared to their historical counterparts graphically. Through a graphical comparison, the model's ability to follow the same development path over time can be established (Barlas, 1996; Sterman, 2000). This graphical comparison allows the researcher to review the whole development of variables. Compared to aggregated, statistical measures of fit, this graphical approach provides additional information. For example, it may highlight the timing of errors and mismatches between model, theory, and case data. The modeller can trace and fix the causes of errors along causal links to problematic structures and equations (Homer, 2012; Martinez-Moyano and Richardson, 2013). These were then improved in the iterative modelling process.

2.4.2. Scenario simulation

After the developed model has been validated using case data, it can be used for simulations to test and evaluate different policies. Thereby, the model is used to simulate scenarios investigating the impact of alterations on the system's behaviour. Compared to experiments in the real world, these virtual simulations have the advantage of being cost-effective. They are also replicable because the model results depend entirely on the model inputs without interference from cofounding or other influences. Simulation modelling also reduces safety concerns and adverse effects for participants as experiments do not affect them in reality (Coyle, 2000; Sterman, 2000; Warren, 2002; Davis, Eisenhardt and Bingham, 2007; Harrison *et al.*, 2007). In this project, the development of value creation and capture under varying scenarios will be simulated. Setting up System Dynamics simulations requires constructing a base case and identifying the policy changes that are compared to the base case.

Firstly, simulation researchers must create a base case of initial conditions and parameters that serves as the foundation for later comparisons. Practitioners of System Dynamics modelling in companies might, for example, use the company's current conditions and policies as a base case. This base case approximates a company's future development if everything stays as it is now (Sterman, 2000; Warren, 2002). Simulation researchers might also create artificial, hypothetical base cases (March, 1991). While the former approach ensures that the base case is realistic, the latter approach can isolate and create optimal conditions for theoretical relevance. Davis *et al.* (2007) argue that simulation research should strive for realism and theoretical relevance. Therefore, this thesis combines both approaches by creating a hypothetical, idealised company as a base case. The case is inspired by details of the case companies and their data. It is constructed with theoretical implications for a typical company in mind. In addition, parameter values are evaluated for their realism compared to those found among the case companies during model validation.

Secondly, researchers must decide which policies to test in the model. In System Dynamics, two important types of policy changes can be observed. Firstly, policy changes might concern the numerical values of variables. For example,

companies might employ new strategies. The model can reflect these through values for managerial targets, goals, priorities, and commitment levels. Secondly, policy changes might concern the structure of the model. In business contexts, such structure-affecting policy changes usually relate to management decision rules regarding, for example, which factors they consider when setting targets and implementing them. Modellers usually implement policy changes through Coyle's (1996, p. 106) 'policy switches'. These switches are variables set to either zero or one to activate or suppress certain parts of an equation (Coyle, 1996; Sterman, 2000; Warren, 2002; Harrison *et al.*, 2007). Both types of experiments are simulated using the model. Simulations will determine the effect of policy changes that change the model structure and the impact of different variable values within the same policy structure. These principles have been implemented to determine the structure of Chapter 6. Its sections are set based on different policy switches that cause different types of performance development. Within each section, policies related to numerical values are simulated.

2.4.3. Analysis of results

The analysis of simulation results focuses on value creation, the share of value captured, and financial resources as a cumulative performance measure. Each scenario has been analysed individually before comparing scenarios to synthesise answers to this thesis' research questions and aim.

Individual scenarios have been analysed by identifying behavioural patterns, the structures that cause the patterns, and the variables that affect structures. Firstly, the time-series graph for each performance outcome in a scenario has been analysed to identify development patterns. These patterns may align to patterns commonly observed in System Dynamics projects (Taylor, 1980). They include, for example, exponential growth, goal-seeking behaviour, s-curves, oscillation, and overshoots (Kim and Anderson, 1998; Sterman, 2000). Secondly, the model has been reviewed for the dominant structures that cause a scenario's performance development. For example, goal-seeking development requires a balancing feedback loop that becomes stronger over time (Kim and Anderson, 1998; Sterman, 2000). The identified structures are described to explain the

observed patterns. It is best practice in System Dynamics modelling that the dominant structures are also included in the causal loops that conceptualise and provide an overview of the model (Martinez-Moyano and Richardson, 2013). In the iterative modelling process, it has thus been ensured that they are considered and represented when the model is conceptualised in Chapter 3. Lastly, the exogenous variables influencing these structures have been considered. For example, a goal-seeking loop requires a goal towards which a variable develops.

Scenario results, particularly the exogenous leverage points, have also been compared across scenarios to answer this thesis' research questions and achieve its aim. The development patterns, structures, and influencing variables have been compared across scenarios to identify leverage points that cause different patterns and the diversity within those patterns (Meadows, 1997, 2009). This allowed this thesis to identify answers to its research questions by categorising different development patterns and revealing the positive and negative impact of growth on value creation and the ability to capture value. The presentation of scenario results (Chapter 6) is structured to illustrate those similarities and differences. Each section of the chapter describes one development pattern with diverse developments within each pattern. While the first section illustrates the development of a company focused on exploiting its current size, the second section illustrates the development of a company striving to scale up. This presentation eases the identification of answers to the research questions presented in the introduction. The discussion derives answers to these research questions and derives a framework helping digital ventures develop an ability to capture value while growing their value creation. This framework and the answers to the research questions also draw from the other chapters and modelling steps. This information is used to describe the conceptualisations (Chapter 3) and formal structures (Chapter 4) causing development patterns, and illustrating them in historical case study development (Chapter 5).

Chapter 3: Theory integration and model conceptualisation

This chapter conceptualises the model used to investigate the development and relationships of value creation and capture for digital ventures. The chapter provides a strategic overview of the model utilising causal loop diagrams illustrating the mechanisms affecting value creation and the share of value captured. These causal loop diagrams synthesise and integrate growth paths theory, dynamic states theory, and their background literature. Their relevant elements are: value creation and value capture as two critical performance outcomes; the business model including its resources, activities, and capabilities; influences from the external environment and the dominant logic of managers and entrepreneurs (Figure 3-1). The diagrams in this chapter illustrate the systemic relationship between these elements.



Figure 3-1: Overview of model elements

This first section conceptualises value creation and capture as the critical performance outcomes of this thesis. It then works backwards to identify the business model elements affecting performance outcomes, including resources, activities, and capabilities. This chapter then discusses influences on the identified relationships in the companies' environment and different dominant logics. As explained in the methodology chapter, the causal loop diagrams

communicate how the growth process theories have been synthesised on a general and aggregated level. Thus, this chapter illustrates the dynamic processes affecting performance outcomes over time. The final section of this chapter analyses the conceptual model qualitatively to obtain insight about the development of value creation and the share of value captured. The diagrams conceptualised here serve as a guideline for the next chapter, where these general feedback processes are contextualised, specified, and formalised for digital ventures (Wolstenholme, 1999; Sterman, 2000; Morecroft, 2015; Kunc, 2017).

3.1. Performance outcomes

Companies need to create customer value and capture part of that value to generate resources during the growth process (Figure 3-2). Whereas *value creation* refers to the total amount of customer value created over a period of time, the *share of value captured* refers to the fraction of the created value that is received and maintained by a company as profit (Teece, 1986; Garnsey, Dee and Ford, 2006; Hull *et al.*, 2007; Lepak, Smith and Taylor, 2007; Davidsson, Steffens and Fitzsimmons, 2009; Levie and Lichtenstein, 2010; Zott and Amit, 2010; Huang *et al.*, 2017; Hsieh and Wu, 2019). Growth process theories build on the existing literature on value creation and capture to conceptualise these performance outcomes. Teece (1986) and Zott and Amit (2010) illustrate value created by a company over a period, such as a financial year. However, a company only captures a part of that pie as profit. This part captured by the company represents the resources generated during the growth process.


Figure 3-2: Resource generation

Both performance measures are critical to the continuity of the firm during the growth process. While growth paths theory focuses on capturing value and generating resources (Garnsey, 1998; Garnsey, Lubik and Heffernan, 2015), dynamic states theory stresses value creation (Levie and Lichtenstein, 2010). Both theories acknowledge that both performance outcomes - creating value and capturing part of that value - are required for resource generation. These generated resources accumulate as *financial resources*. In growth paths theory, these generated resources increase the company's size and provide the means for further investments (Garnsey, 1998; Stam and Garnsey, 2006; Garnsey, Lubik and Heffernan, 2015). In dynamics states theory, these resources ensure the viability of a growth state (Levie and Lichtenstein, 2010). Therefore, companies should maximise the resources they generate by growing the amount of value they create or increasing the share of value they capture. However, as pointed out in the introduction of this thesis, existing theory cannot explain why these two performance measures develop diversely and what relationships exist among these two outcomes (Sapienza et al., 2006; Davidsson, Steffens and Fitzsimmons, 2009; Steffens, Davidsson and Fitzsimmons, 2009). The subsections below conceptualise the influences on value creation and capture. These influences are

part of the venture's resources and external environment. Managers need to change them to maximise their value creation and share of value capture.

3.1.1. Value creation

Value creation is the generation of customer benefit by providing solutions to customer needs (Amit and Zott, 2001; Levie and Lichtenstein, 2010). It reflects the monetary amount of value that the company has created for all its customers during a period. The theories separate value creation into the volume of products or services rendered over a period and the use value of each unit of product or service. For example, companies selling physical products create value by providing a certain number of products per period, with each product unit having a specific value. This value per unit of product is also called use value. It refers to the amount of money at which a customer is indifferent between having one unit of the product or having the money. It is the maximum amount of money a customer is willing to pay for a product (Porter, 1985, 1991; Peteraf and Barney, 2003; Lepak, Smith and Taylor, 2007). The product's use value reflects the value created per unit of product. Therefore, the total monetary amount of value created by a company also depends on the number of units sold over the period of interest (Zott and Amit, 2010). One can decompose the value created by a service company similarly. It depends on the value created every time a venture renders a service and the number of times the company renders it over a period.

The resources in the venture's business model and factors in its external environment affect *value creation* (Figure 3-3). The volume of products or services rendered in a digital context depends on the venture's *customer and user base*. The formalisation chapter further distinguishes between these different types of customers. The *use value* depends on the venture's *product quality* and the *usage intensity* at which customers use the venture's product or service. The customer and user base and product quality are reviewed as resources in the business model. The usage intensity is an environmental factor reflecting the behaviour of customers and users regarding the product. Thereby, the model implements arguments of the resource-based view that companies need to provide valuable products to customers. However, the customer in the venture's

environment assesses the value (Barney, 1991, 2001; Grant, 1991; Peteraf, 1993; Peteraf and Barney, 2003). In a digital context, this usage intensity changes through *feature adoption*. Customers learn more about a software product, use additional features and functionalities, and generate greater value from using the product over time. However, this adoption is limited to a maximum of available features and time (Afuah and Tucci, 2000; Amit and Zott, 2001; Harmon, Raffo and Faulk, 2004; DaSilva *et al.*, 2013; Haile and Altmann, 2016b).



Figure 3-3: Value creation⁶

To maximise value creation, managers and entrepreneurs can thus alter their resource base or select more attractive environments. Companies can scale up their customer and user base to create value for more customers. They can also improve their product quality to generate more value for each user or customer (Lepak, Smith and Taylor, 2007; Garnsey, Lubik and Heffernan, 2015; Huang *et al.*, 2017). These two approaches form the basis for the managerial targets discussed as part of dominant logics discussed below. At a constant customer/user

⁶ Elements considered in each section are highlighted in blue to allow the reader to follow the descriptions of the text in the diagrams. Parts introduced in previous sections are displayed in grey to build the model step by step.

base and product quality, value creation and resource generation will automatically develop towards a maximum through the balancing feedback loop of customers adopting features over time (B1). Companies can target users and customers that use the product more intensively and thus derive greater value from using it (Garnsey, 1998). Thereby, they reach a higher maximum of value creation or develop towards it more quickly.

3.1.2. Value capture

The created value presents "the overall size of the value pie" and the maximum value a company can capture. The share of value captured represents the fraction of the pie that a company receives and maintains as profit (Zott and Amit, 2010, p. 218). Within this metaphor, companies capture only part of the value they create for customers as revenue. Ventures also pass on some of the value to input providers as costs and expenses and they lose some value due to the erosion of resources (Nelson and Winter, 1978; Teece, 1986; Dierickx and Cool, 1989; Porter, 1991; Knott, Bryce and Posen, 2003; Lepak, Smith and Taylor, 2007; Zott and Amit, 2010). These three factors, and thus the *share of value captured*, depend on the resources of the venture and its external environment (Figure 3-4).



Figure 3-4: Share of value captured

Companies capture only part of the value they create for customers as revenue. While the total value created reflects customers' maximum willingness to pay, the actual price exchanged for a product or service is usually below this maximum willingness to pay. Customers capture the difference between their willingness to pay and the price as consumer surplus. The share of value captured from *customers* as revenue depends on the *product quality* as a resource relative to the strength of competitors and substitutes in the environment. A monopoly provider may be able to capture the entire use value created for customers. The product's price corresponds to customers' maximum willingness to pay in such cases. However, the availability of suitable alternatives from competitors and substitutes lowers the bargaining power to customers, forcing companies to lower prices. Thereby, companies capture a smaller fraction of value, while the consumer surplus increases (Porter, 1991; Peteraf and Barney, 2003; Garnsey, Dee and Ford, 2006; Lepak, Smith and Taylor, 2007). Moreover, as customers learn and adopt features of digital products and services, they develop *switching costs* that increase the bargaining power of the venture and allow it to capture more value (DaSilva et al., 2013). However, due to contract terms that are common for

digital products and services, effects on the share of value captured may be delayed (Harmon, Raffo and Faulk, 2004; Saaksjarvi and Lassila, 2005; Tyrväinen and Selin, 2011; Voigt and Hinz, 2016; Kohtamäki *et al.*, 2019).

Companies also lose value to input providers such as employees, suppliers, and investors. Through expenses, companies compensate input providers for their contribution to the production process. The *share of value lost to input providers* expresses these costs as a fraction of the value created. It depends on two factors: Firstly, the *efficiency of operations,* which reflects the venture's *value creation* per resource input. Secondly, the *cost per unit of input resource* in the external environment (Porter, 1985; Peteraf and Barney, 2003; Stam and Garnsey, 2006; Zott and Amit, 2007).

The venture may also lose resources due to the erosion of resources. During the process of value creation, the venture's resources erode due to their limited useful lifetime. For example, fixed assets depreciate and amortise. The *share of value lost to the erosion of resources* expresses these lost resources as a fraction of the *value creation* (Nelson and Winter, 1978; Dierickx and Cool, 1989; Porter, 1991; Knott, Bryce and Posen, 2003).

The causal loop diagram above separates the share of value captured into three fractions: the share of value captured from customers, the share of value lost to input providers, and the share of value lost to resource erosion. Variables in the external environment and the resource base affect these three fractions. Managers can alter them to maximise value capture. In the resource base, ventures can improve their product quality to improve their bargaining power to customers (Porter, 1985, 1991; Lepak, Smith and Taylor, 2007). They may also use resources more efficiently or over extended periods (Barney, 1991; Grant, 1991). In the environment, managers and entrepreneurs can select industries and business models with less competition and lower input costs (Porter, 1985, 1991; Lepak, Smith and Taylor, 2007). Moreover, they can target customers that develop greater switching costs (Porter, 1985). Improving the product quality, switching costs, and industry selection may affect value creation and capture simultaneously. Efficiency improvements affect only value capture. However, changes in the resource base and environment may not cause immediate

improvements in the share of value captured due to contract times. Moreover, dynamic processes can be observed in this diagram. As customers adopt features (B1), switching costs increase, which improve the value captured from customers over time. Thereby, the share of value captured increases because the venture creates more value at equal input usage and resource erosion.

3.2. The business model and its elements

The business model defines the architecture used by a company that "creates and delivers value to customers, and then converts payments received to profits" (Teece, 2010, p. 173). Thus, a company's business model determines value creation and the share of value captured as the two central performance measures (Teece, 2010; Zott and Amit, 2010; Standing and Mattsson, 2016; Guo et al., 2020). This relationship between the business model and the performance outcomes is acknowledged in growth paths theory (Hugo and Garnsey, 2002; Stam, Garnsey and Heffernan, 2006; Garnsey, Lubik and Heffernan, 2015) and dynamic states theory (Levie and Lichtenstein, 2010; Brown and Mawson, 2013; Gabrielsson and Gabrielsson, 2013; Ingley, Khlif and Karoui, 2017). Three different views exist in the literature on business models cited by growth process theories. One view conceptualises business models as activity systems or value chains, in which activities are the drivers of performance (Porter, 1991, 1996; McKelvey, 1999; Zott and Amit, 2010). The resource-based view argues that companies' performance depends on their resources and capabilities (Grant, 1996; Amit and Zott, 2001). The third view on business models sees them as the connecting mechanisms between the firm and its environment, with the place and position in that environment determining performance (Zott and Amit, 2007; Garnsey, Lubik and Heffernan, 2015). The concepts of resources, activities, and capabilities are central to growth paths theory, although the theory does not explicitly link them to the business model (Garnsey, 1998; Stam and Garnsey, 2006). Reflecting the elements identified in growth paths theory, this thesis considers the first two views - the activity systems and resource-based view - as providing the business model's elements. The environment as the third view of business models influences these elements and their relationships. The

subsections below describe resources, activities, and capabilities as elements of the business model and note influences from the external environment where they are present. Such influences have already been identified when conceptualising the performance outcomes. For example, they include the strength of competitors and substitutes or the costs of resource inputs.

3.2.1. The impact of resources on performance outcomes

This thesis proposes that, at any point in time, value creation and capture as central performance outcomes depend on the company's resources and its external environment. Resources are tangible and intangible assets used or controlled by the firm for its operations (Barney, 1991; Grant, 1991, 1996; Teece and Pisano, 1994; Amit and Zott, 2001; Penrose, 2009). This thesis identifies five types of resources in growth process theories that directly affect value creation and capture (Figure 3-5). Moreover, some resources depend on secondary resources in a digital context.



Figure 3-5: Resource-performance relationships

The above conceptualisation of value creation has already identified two resources affecting performance outcomes:

- The *customer and user base* reflects the number of customers and users serviced by a venture. A larger customer base provides additional demand and an outlet for the venture's product or service. Therefore, the larger this customer base, the more value the venture creates (Garnsey, 1998; Stam and Garnsey, 2006; Garnsey, Lubik and Heffernan, 2015). As further discussed in the next chapter, digital ventures may distinguish between customers, users, and premium users (Afuah and Tucci, 2000; Kollmann, 2006; Zott, Amit and Massa, 2011).
- The *product quality* also affects value creation through the use value. This quality usually depends on the quality of the product's technology and its features (Miozzo and DiVito, 2016; Hesse and Sternberg, 2017). However, digital ventures may also employ business models with additional value drivers. For example, digital ventures may create value using content. Their product quality thus depends on the quality of that content. The causal loop diagram above aggregates the *technology and content quality* because they develop through similar mechanisms. Other digital ventures employ business models, the customer and user base is a driver of the product quality too (Zhu and Iansiti, 2007; Kim, Oh and Shin, 2010; Hamari and Keronen, 2017). The next chapter further discusses these business model design choices and value drivers.

A venture also requires resource inputs to create value and must compensate providers of resources. Economists generally distinguish between labour and capital as inputs for companies (Nelson and Winter, 1982). The model considers labour as human resources. It distinguishes between three different types of capital based on the views of innovation and strategic management scholars. These are the venture's technology and content assets, complementary assets such as equipment, and financial resources held as liquid funds (Teece, 1986; Grant, 1991; Garnsey, 1998; Penrose, 2009).

• *Technological and content assets* reflect the quantity of technological resources and content assets maintained by the venture. Technological resources may include, for example, intangible assets capitalised on its

balance sheet. Content assets reflect the venture's content library, such as music, video, or written content available for users. Like their quality, these two types of assets are aggregated here. Once developed, they may amortise over time and increase the share of value lost to resource erosion (Nelson and Winter, 1978; Dierickx and Cool, 1989; Porter, 1991; Knott, Bryce and Posen, 2003).

- *Complementary assets* are resources required to commercialise a technology. Teece (1986, 1998) regards all resources and capabilities required to commercialise a technology as complementary assets. Growth paths theory focusses on, for example, equipment for production and customer service (Garnsey, 1998; Garnsey, Stam and Heffernan, 2006; Stam, Garnsey and Heffernan, 2006; Miozzo and DiVito, 2016). In a digital context, these might, for example, include hosting equipment and server capacity (Oliva, Sterman and Giese, 2003; Sterman *et al.*, 2007). Like the technological and content assets, complementary assets depreciate over time. Thus, they affect the share of value lost to the erosion of resources (Nelson and Winter, 1978; Dierickx and Cool, 1989; Knott, Bryce and Posen, 2003).
- *Human resources* reflect the firm's number of employees. These are critical for the growth process because they execute activities using other resources (Nelson and Winter, 1982; Garnsey, 1998; Penrose, 2009; Garnsey, Lubik and Heffernan, 2015). The more employees are required to execute the venture's activities, the lower the venture's efficiency, the higher the compensation paid to the input provider, and thus the higher share of value lost to input providers (Porter, 1985; Peteraf and Barney, 2003; Stam and Garnsey, 2006; Zott and Amit, 2007). Over time, the venture's human resources erode due to *employee turnover* (Hambrick and Crozier, 1985; Hamermesh and Pfann, 1996; Baron and Hannan, 2002; Garnsey and Heffernan, 2005).
- *Financial resources* refer to the company's cash position (Teece, Pisano and Shuen, 1997; Miozzo and DiVito, 2016). These resources provide the company with liquidity to invest in its activities and to compensate input providers. Financial resources also provide a buffer for shocks and miscalculations, such as drops in demand due to increasing competition or higher than expected costs (Garnsey, 1998; Garnsey and Heffernan, 2005;

Garnsey, Dee and Ford, 2006; Garnsey, Stam and Heffernan, 2006). Financial resources are generated through resource generation and by raising them from investors through the *capital employed*. This capital reflects the amount of money raised. Investors and lenders are compensated for their capital contribution. The more capital is employed to create a given amount of value, the lower the venture's efficiency and the higher the share value lost to input providers (Porter, 1985; Peteraf and Barney, 2003; Stam and Garnsey, 2006; Zott and Amit, 2007).

Resources are critical for the growth process besides their direct effect on value creation and capture. Penrose (2009) defines the growth process as a series of internal changes leading to a change in firm size. Growth paths theory builds on Penrose's conceptualisation of the firm (Garnsey, 1998; Stam, Garnsey and Heffernan, 2006). She defines the firm as "a pool of productive resources" (Penrose, 2009, p. 2). Because the firm is a collection of resources, its size is defined as the present value of its total resource (Garnsey, Dee and Ford, 2006; Stam, Garnsey and Heffernan, 2006; Penrose, 2009). However, one cannot calculate this present value accurately. Therefore, growth paths theory utilises other measures of resources to approximate a firm's size. These measures include, for example, total assets, fixed assets, capital raised, or the number of employees (Garnsey, 1998; Garnsey, Dee and Ford, 2006; Garnsey, Stam and Heffernan, 2006; Stam and Garnsey, 2006; Stam, Garnsey and Heffernan, 2006). Moreover, the internal changes occurring during the growth process refer to changes in the company's resource base. After starting with an initial resource base, the company's resources change continuously through exchange with the external environment and internal transformation (Garnsey, 1998; Stam and Garnsey, 2006; Miozzo and DiVito, 2016). Some of these dynamic changes in the company's resources as part of the growth process are already present in the above conceptualisation. For example, the complementary assets and human resources erode over time due to depreciation and turnover. Growth paths theory adds further that resources generated through creating value and capturing a part of that value increase the firm's financial resources (Garnsey, 1998; Stam and Garnsey, 2006; Garnsey, Lubik and Heffernan, 2015). Activities may then transform these financial resources (Garnsey, 1998; Stam and Garnsey, 2006).

3.2.2. Resource transformation through activities

Resources develop dynamically through depreciation, amortisation, and resource generation. Additionally, activities may affect them. Growth paths theory defines activities as tasks executed by the company's human resources utilising other resources to alter the firm's resource base (Garnsey, 1998; Hugo and Garnsey, 2001, 2002, 2005; Stam and Garnsey, 2006; Penrose, 2009; Garnsey, Lubik and Heffernan, 2015). Activities may alter the venture's resource base by transforming one resource into another, mobilising resources from the external environment, or creating resources from scratch. Different types of activities can be identified through their impact on the venture's resources (Warren, 2002). This thesis considers four different types of activities, which link to the above resources and have been identified in growth process theories (Figure 3-6).



Figure 3-6: Resource development through activities

A first activity links to the venture's technology and content. Ventures may create additional technological and content assets through *technology and content creation*. For many physical products, this takes the form of research and development, through which the venture's employees add or improve product features. Thereby, the venture transforms financial resources into technological resources and content assets (Porter, 1991; Penrose, 2009; Garnsey, Lubik and

Heffernan, 2015; Miozzo and DiVito, 2016). In a digital context, employees and customers may also create content (Amit and Zott, 2001; Sahut, Iandoli and Teulon, 2019). This creation increases the venture's technological and content assets. Moreover, the creation may lead to *technology and content improvements* that affect the product quality. However, due to technology s-curves and limits to content quality, the higher the venture's existing technology and content quality, the lower the effect of creation efforts on the change in quality (Tidd, Bessant and Pavitt, 1994; Dosi, Nelson and Winter, 2000; Smith, 2015). This relationship creates a balancing feedback loop (B2) that limits growth in value creation and an increase in the ability to capture value over time through product quality improvements.

A second activity regards the development of the venture's customer and user base. Ventures acquire new customers and users through *marketing and selling*. The venture's human resources carry out marketing and selling depending on their capability level. Their effectiveness also depends on the product quality relative to competitors' products and substitutes (Garnsey, 1998; Bhide, 2000; Garnsey, Lubik and Heffernan, 2015; Hesse and Sternberg, 2017). A reinforcing feedback loop emerges in which the existing customers and users improve the product quality directly through network effects and indirectly though content creation, which in turn increases the acquisition of additional customers and users, who improve the product quality further (R1). Moreover, as discussed in the next chapter, digital ventures may rely more on word-of-mouth marketing than companies providing physical products. Through word-of-mouth marketing, the venture's existing users and customers acquire new users and customers, which in turn acquire more user and customers, creating another reinforcing loop (R2). Thus, the venture's marketing activities and customer/user recommendations drive the increase in the customer and user base (Mikalef, Giannakos and Pateli, 2013; Huang et al., 2017).

Once acquired, digital ventures must deliver value to their customers and users. *Value delivering* in the context of physical products includes manufacturing. Digital ventures need to ensure the hosting of their website and service customers. Thereby, the venture's human resources provide the venture with a capacity to

deliver value to customers and users. However, if the company cannot deliver value to its customers, customers might decide to drop the venture's product or service (Garnsey, Dee and Ford, 2006; Stam, Garnsey and Heffernan, 2006; Penrose, 2009; Gabrielsson and Gabrielsson, 2013; Miozzo and DiVito, 2016). This creates a balancing feedback loop in which insufficient capacity leads to higher churn, reducing the customer and user base towards the venture's capacity (B3). Moreover, customers and users may *churn* by switching to competitors' products. This churn depends on the switching costs of customers and users and the product quality relative to competitors and substitutes (Zhao, Song and Storm, 2013; Voigt and Hinz, 2016). This relationship may be delayed due to contract times employed by digital ventures and creates another reinforcing feedback loop. Like the acquisition of new customer and users through marketing and selling, the venture's larger customer and user base makes the product more attractive, reducing churn and helping to sustain the customer and user base (R3).

The last activity affects the venture's human resources and capital assets that the other activities utilise. Firm managing refers to the activities executed to oversee and expand the venture's operations. Its current managerial and administrative employees provide the venture with a *firm management capacity* utilised to manage and oversee the venture's existing employees. The firm's management is also responsible for planning and executing the venture's expansion through fundraising, hiring, and investing into complementary assets. The management executes these activities based on their priorities conceptualised as part of the dominant logic. However, the venture's management can only perform these activities when its capacity is not fully occupied with managing its existing human resources. In such cases, the venture has what growth process theories call managerial slack. This slack represents time available to plan and execute growth and expansion activities. The management's capability and its managerial slack constrain the venture's capacity to expand through hiring, fundraising, and investing. This limited capacity to grow the venture is critical to growth paths theory and is referred to as the Penrose effect (Garnsey, 1998; Hugo and Garnsey, 2001, 2002, 2005; Garnsey, Dee and Ford, 2006; Stam and Garnsey, 2006; Penrose, 2009; Garnsey, Lubik and Heffernan, 2015). The firm's ability to grow depends on the strengths of a reinforcing and balancing feedback loop

determining managerial slack. On the one hand, the venture's existing human resources that require supervising reduce its managerial slack, inhibiting it to hire further (B4) and expand the firm. On the other hand, some of the firm's human resources are managers themselves who increase the venture's managerial capacity and allow it to grow the firm further including through additional hiring (R4).

3.2.3. Capability development

Besides human resources, the capabilities of employees also affect the outcomes of marketing and selling, value delivering, technology and content creation, and firm managing activities. This thesis defines capabilities as levels of productivity that determine the strength of the relationship between the venture's human resources and the effect of activities on resources. In growth paths theory, these productivity levels are achieved through the content of capabilities on an individual and organisational level. On an individual level, capabilities reflect the firm's employees' knowledge, skills, experience, and competence to execute activities. On an organisational level, these capabilities manifest in routines and processes which improve over time through learning, erode through hiring, and depend on the availability of sufficient complementary assets (Figure 3-7).



Figure 3-7: Capability development

In both growth process theories, capabilities improve over time through learning and routinisation (Garnsey, 1998; Hugo and Garnsey, 2005; Stam, Garnsey and Heffernan, 2006; Gabrielsson and Gabrielsson, 2013). Growth paths theory argues that new companies have not yet established efficient routines and processes. Instead, routines are experimental, and individuals are uncoordinated. However, capabilities improve as individuals accumulate experience working together and improve their routines and processes (Hugo and Garnsey, 2005; Stam and Garnsey, 2007; Garnsey, Lubik and Heffernan, 2015). Dynamic states theory also acknowledges such improvements through learning-by-doing (Gabrielsson and Gabrielsson, 2013). The rate of improvements depends on the firm's *dynamic capabilities*⁷. These are routines changing other routines and reflect the firm's "ability to change or reconfigure existing substantive capabilities" (Nelson and Winter, 1982; Nelson, 1995; Eisenhardt and Martin, 2000; Zott, 2003; Kor and Mahoney, 2004; Zahra, Sapienza and Davidsson, 2006; Gabrielsson and Gabrielsson, 2013, p. 1359). However, capabilities will develop towards a maximum through a balancing feedback loop (B5) because the higher the current level of capabilities, the lower the incentive to search for further improvements in knowledge and routines, and the more rigid the venture becomes. Therefore, the higher the firm's existing capabilities, the lower the firm dynamic capabilities (Porter, 1996; Dosi, Nelson and Winter, 2000; Eisenhardt and Martin, 2000; Bessant, Phelps and Adams, 2005; Bingham, Eisenhardt and Furr, 2007; Phelps, Adams and Bessant, 2007).

In addition to capability improvements, growth paths theory also highlights that mobilising additional resources, particularly human resources, harms capabilities. Routines and processes are firm-specific. Therefore, new employees lack a complete understanding of their specific role in routines (Fransman, 1994; McKelvey, 1999; Dosi, Nelson and Winter, 2000; Kor and Mahoney, 2004; Chiles, Bluedorn and Gupta, 2007; Federico and Capelleras, 2015). Thus, new

⁷ A range of different definitions of dynamic capabilities exists in the literature. For example, some define dynamic capabilities as all capabilities that alter the resource base or as reflecting the firm's ability to adjust to the environment. Please see Zahra, Sapienza and Davidsson (2006) for a comprehensive review and list of definitions. The definition of dynamic capabilities as measures of change in substantive capabilities has been selected due to its consistency with the role of capabilities as productivity measures depending on individuals' knowledge and organisational routines.

employees are generally less productive than employees with experience in the firm. Hiring new employees reduces the average capability of the firm until new employees have gone through the same learning process that existing employees have gone through (Slater, 1980; Garnsey, 1998; Tan and Mahoney, 2005; Stam, Garnsey and Heffernan, 2006; Penrose, 2009; Miozzo and DiVito, 2016; Harbermann and Schuilte, 2017).

Lastly, dynamic states theory and the resource-based view highlight that capabilities also rely on the venture's resources (Grant, 1991; Peteraf and Barney, 2003; Gabrielsson and Gabrielsson, 2013). More specifically, growth paths theory points towards the importance of holding adequate complementary assets. A venture with more human resources requires more complementary assets to equip those employees (Garnsey, 1998; Stam, Garnsey and Heffernan, 2006). If the venture's complementary assets are insufficient for its human resource to carry out their tasks, capabilities as productivity measures fall (Garnsey, 1998; Stam, Garnsey and Heffernan, 2006; Gabrielsson and Gabrielsson, 2013).

3.3. Influences on the performance outcomes and business model

The dominant logic and external environment have been identified as influences affecting the performance outcomes and the business model. While some of these influences expand the model through additional feedback loops, other variables affect the causal loops without being affected by the feedback processes themselves. The subsections below describe these environmental factors and dominant logic.

3.3.1. Environmental influences

The environment affects ventures' performance outcomes by altering its ability to create value and capture a part of that value directly and indirectly (Garnsey, 1998; Hugo and Garnsey, 2005; Levie and Lichtenstein, 2010; Brown and Mawson, 2013; Ingley, Khlif and Karoui, 2017). The resource-based view and the activity-systems view on business models have been utilised to identify the business model's elements. The third view on business models conceptualises it

as linking the company to its environment. In this view, a company's performance is affected by its position in its environment (Zott and Amit, 2007; Garnsey, Lubik and Heffernan, 2015). Growth paths theory conceptualises this environment as different groups of stakeholders that affect and are affected by the company (Garnsey & Heffernan, 2005; Garnsey, 1998). Following this conceptualisation, one can already identify influences from three stakeholder groups in the environment (Figure 6-7). Firstly, users and customers who generate use value based on their usage intensity. They may also contribute to value creation through product improvements and network effects. Secondly, input providers that the venture compensates for their human resources and capital through costs and expenses. Thirdly, competitors and substitutes who affect the venture's bargaining power and ability to acquire and hold on to customers. The detailed descriptions of the model in the next chapter discuss providers of complementary products as a fourth stakeholder group. The detailed model also includes a range of exogenous variables for each stakeholder group that determine the strength of the group's influences. These being exogenous effects is consistent with dynamic states theory, which considers the environment as given and the firm as adapting to its environment (Levie and Lichtenstein, 2010; Brown and Mawson, 2013). Thereby, these exogenous elements are a contextual influence that determines a venture's growth state. They allow some parts of the environment to change continuously (e.g. the user and customer base in the above model), while others change only with a growth state (e.g. costs per unit of resource input).

3.3.2. Dominant logic of managers and entrepreneurs

The dominant logic of managers and entrepreneurs determines expansion activities and drives resource allocation decisions. It reflects managers' and entrepreneurs' aspirations, beliefs, and decision-making logic (Prahalad and Bettis, 1986; Von Krogh, Erat and Macus, 2000; Levie and Lichtenstein, 2010). This decision-making logic adds additional feedback mechanisms to the model that determine the extent of hiring, fundraising, and investments into complementary assets based on managerial targets (Figure 3-8). As in dynamic states theory, these managerial targets reflect the exogenous part of the dominant logic. They affect the growth process from the outside (Levie and Lichtenstein, 2010). The venture forms targets that direct its firm management activities depending on these exogenous influences and its current resources and capabilities. This process takes place in three steps described below.



Figure 3-8: Impact of the dominant logic

Firstly, the venture's management determines the magnitude of targeted change in its resource base. This magnitude depends on its current resource stocks and the exogenous managerial targets. As outlined in the value creation subsection, digital ventures may create more value by improving their product quality and by increasing their user and customer base (Lepak, Smith and Taylor, 2007; Garnsey, Lubik and Heffernan, 2015; Huang *et al.*, 2017). Thus, ventures need to increase either or both resources stocks to increase value creation. The *target customer and user acquisition* represents the number of customers and users the venture aims to acquire within one year. The management determines it based on the current *customer and user base* and the *target growth rate*. It must also make up for customers it expects to lose through churn. Similarly, the venture determines its *target product improvements* through the current *product quality* and the *target improvement rate*.

Secondly, the venture's management determines the human resources and liquidity required to finance these acquisition and improvement targets. The higher the acquisition and improvement targets, the higher the *required liquidity* for recruitment, customer acquisition, development costs, and content creation (Garnsey, 1998; Miozzo and DiVito, 2016). Similarly, the higher the acquisition and improvement targets, the higher the *required human resources* at the venture's current capability levels (Harbermann and Schuilte, 2017). Here, two different types of dominant logic can be identified (Hesse and Sternberg, 2017):

- Some ventures restrict their human resources to a *cap on employee numbers*. Such entrepreneurs may be unwilling to accept the risks that come with growth or are comfortable with the current size of their companies. For example, because it provides them with an adequate return and covers their living expenses. Through a cap on employee numbers, they may maintain a company they can manage and stay in control of (Davidsson, 1989; Garnsey, 1998; Hesse and Sternberg, 2017).
- Other managers and entrepreneurs aim to grow their companies continuously and regardless of employee numbers. They may grow their customer base because they believe in their product and services, the existence of an even larger market, and their companies' ability to benefit from it (Schumpeter, 1928; Kotter and Sathe, 1978; Eisenhardt and Schoonhoven, 1990; Stuart and

Abetti, 1990; Fransman, 1994; Nicholls-Nixon, 2005; Steffens, Davidsson and Fitzsimmons, 2009; Hansen and Hamilton, 2011; Huang *et al.*, 2017). In such cases, the employee numbers depend on the targets that the venture strived to achieve and its capabilities. They are not capped to a maximum.

Lastly, the management needs to determine the gap between its target and current human and financial resources. It then mobilises resources to eliminate the difference and acquires sufficient complementary assets. The venture's current *financial resources* provide it with the means for further investments. If its current resources are below its *required liquidity*, the venture has *a liquidity gap* to fill through fundraising. Similarly, if its current *human resources* are below the *required human resources*, the venture forms a *hiring target* and closes the difference between current and target employee numbers (Sterman, 2000; Oliva, Sterman and Giese, 2003). The venture will also *invest in complementary assets* to equip the new employees sufficiently (Garnsey, 1998; Stam, Garnsey and Heffernan, 2006). Two factors may delay these activities. Firstly, they require management may delay hiring until the liquidity gap has been closed (Garnsey, 1998; Garnsey and Heffernan, 2005).

This three-step process of determining growth and improvement targets, identifying human resources and liquidity requirements, and closing gaps creates three feedback loops. Firstly, the process of closing the gap between targets and actual resources form balancing feedback loops. For example, the hiring target is reduced as the human resources increase through hiring (B6). However, hiring also reduces capabilities, which managers consider when determining hiring targets. With lower productivity levels, they will hire more employees in the next iteration, forming a reinforcing feedback loop (R5). Moreover, the hired human resources will grow the venture's customer base and improve its product. Thereby, goals and targets continuously shift upwards, requiring more employees in the next iteration, creating a reinforcing feedback loop (R6).

Dynamic states theory proposes that companies will maintain a dominant logic throughout a growth state (Levie and Lichtenstein, 2010; Brown and Mawson, 2013; Ingley, Khlif and Karoui, 2017). Similarly, scholars related to growth paths

theory argue that goals, aspirations, and decision rules are maintained over periods of time (Nelson and Winter, 1982; Hambrick and Crozier, 1985; Nelson, 1995; Lockett and Thompson, 2004). Therefore, the target rates and caps on employee numbers are assumed to be constant during a growth state and exogenous to the system.

3.4. Implications for growth process and performance development

The above conceptualisation has implications for our understanding of growth states and the performance development within them. A qualitative analysis of the causal loop diagrams can derive these insights. Below, implications for the identification of growth states and two types of dominant logic are identified. The continuous change in value creation and capture is then outlined. This section then combines these different views from the respective theories. Thereby, this thesis argues that the performance development of companies depends on the type of growth state employed by a venture.

3.4.1. Implications for growth states

Dynamic states theory conceptualises the growth process as a process punctuated by transitions between growth states. Within a growth state, some elements of the company remain constant and unchanged. These elements change periodically when the firm changes its growth state (Levie and Lichtenstein, 2010; Brown and Mawson, 2013). The causal loop diagrams above outline exogenous variables that change with a growth state. They regard the venture's environment and its dominant logic. Therefore, a growth state can be defined by the variables that reflect the environment and dominant logic. The firm's external environment and the managers' dominant logic remain the same within a growth state. A change in these exogenous variables is discontinuous and indicates a change in a growth state.

These models' exogenous variables provide managers with two options to affect their firm's development and performance. First, managers and entrepreneurs select their environment (Garnsey, 1998; Levie and Lichtenstein, 2010; Brown

and Mawson, 2013). For example, companies can target customers that realise greater value from their products or services. They may also position themselves in a niche with less competition or select input providers with lower bargaining power. Second, managers can pursue different targets with different employee limits. These resource allocation decisions alter the quantity and quality of resources in its business model over time (Prahalad and Bettis, 1986; Von Krogh, Erat and Macus, 2000; Levie and Lichtenstein, 2010). Formal modelling in the next chapter also uncovered other influences regarding capability development and the business model design. These are assumed to remain constant during a growth state like the environment and dominant logic.

While infinitely many growth states can exist, all growth states fall into two types. Levie and Lichtenstein (2010, p. 335) argue in their seminal contribution to dynamic states theory that there are infinitely many growths states. They derive this conclusion because the business model, which the dominant logic determines, and the environment can each manifest in infinitely many ways. The model can reflect this. For example, the venture's costs for its input resources in the environment can take any positive value. Similarly, ventures can pursue any growth rate and adapt their business model elements accordingly. However, despite infinite possibilities, two essential differences regarding growth states can be observed: managers and entrepreneurs that want to grow with and without hiring more employees (Hesse and Sternberg, 2017). While the former operates at their cap on employee numbers, the latter is operating below or without a cap on employee numbers.

3.4.2. Continuous change within a growth state

Unlike dynamic states theory, growth paths theory conceptualises the growth process as a continuous resource and capability accumulation process (Garnsey, 1998; Garnsey, Stam and Heffernan, 2006). This chapter has reviewed growth paths theory and dynamic states theory to define and conceptualise value creation, value capture, and business model elements. These elements are endogenous to the model, meaning that they change continuously as part of the feedback processes considered in the model (Sterman, 2000). The feedback loops above

illustrate these endogenous feedback processes. Performance outcomes and business model elements in the model develop dynamically. For example, the performance outcomes at any point in time depend on the quantity of resources in the business model and their value in its environment (Teece and Pisano, 1994; Garnsey, 1998; Stam and Garnsey, 2006). Creating value and capturing part of that value generates resources that increase the firm's resources. Therefore, the relationship between resources and performance outcomes is bidirectional (Garnsey, 1998; Stam and Garnsey, 2006). Through activities, which utilise resources, the firm alters its resource base (Hugo and Garnsey, 2005; Stam and Garnsey, 2007). The effectiveness of activities in altering resources depends on employee capabilities which develop through learning and destruction when executing activities (Hugo and Garnsey, 2005; Stam and Garnsey, 2006; Garnsey, Lubik and Heffernan, 2015; Miozzo and DiVito, 2016). The company needs to maintain a balance of the different types of resources within the resource base to prevent the impairment of capabilities and achieve its growth and improvement targets (Grant, 1991; Garnsey, 1998; Peteraf and Barney, 2003; Stam, Garnsey and Heffernan, 2006; Gabrielsson and Gabrielsson, 2013). These processes cause continuous change among the business model elements and the performance outcomes. They reflect resource and capability accumulation processes that growth paths theory considers the essence of the growth process (Garnsey, 1998; Stam and Garnsey, 2006).

3.4.3. Performance development in different growth states

The views of dynamics states theory and growth paths theory are complementary. This section has argued that two types of growth states exist regarding companies operating at or below their cap on employee numbers. It has also outlined the continuous changes in performance development and business model elements. Together, these two insights have implications for the performance development of companies. The dynamics among the business model elements and performance outcomes are different depending on the venture's type of dominant logic.

This thesis proposes that companies operating at their cap on employee numbers can improve their performance outcomes up to a maximum. In the above causal loop diagrams, companies can grow the value they create. Their existing employees will acquire new customers and improve the product. With increasing value creation and capped inputs, the share of value captured by such ventures will also continuously improve. Companies might even increase their growth rates in value creation due to increasing capabilities (B5). However, companies that want to maintain their current human resources are constrained by a maximum amount of value creation. Due to its value delivering capacity, companies will lose new customers that they cannot service adequately (B3). This capacity cannot be expanded by utilising productivity improvements due to the limitations of capability development (B5). Moreover, the technology s-curve prevents the product from being improved endlessly (B2). Due to these limitations, companies operating at their employee cap will eventually reach a maximum amount of value creation. At constant inputs, the share of value captured will also reach a maximum level. How quickly ventures will reach their respective maximums depends on the activities and capabilities and the idle potential of the venture. Examples of the former include the rate of acquiring customers and developing technology. Examples of the latter include possible capability improvements, its idle value delivering capacity, or remaining product improvements.

Those maximum levels do not constrain companies operating below or without a cap on employee numbers. Instead, such companies continuously increase the rate at which they acquire and service customers and develop their technology. These factors allow such ventures to increase the value they create continuously. However, a qualitative analysis of the causal loop diagrams above does not allow for hypothesising about the development of the share of value captured. On the one hand, the venture's increasing value creation should also increase the share of value it captured. Digital ventures' value creation and ability to capture value increase as it hires more employees who grow its customer and user base and improve its product. The resource generated through additional value creation can then be reinvested and reduce the liquidity gap that is filled before further growth (R7). However, on the other hand, the venture continuously increases the resource

inputs for its activities through hiring, reducing its share of value captured, and the resources generated. Thus, fewer resources are available for reinvestment and the liquidity gap that requires closing before additional hiring increases (B7). However, the performance development of a case company depends on the relative strengths of the reinforcing and balancing feedback loops. Therefore, a qualitative analysis is insufficient to evaluate if the share of value captured remains fixed, falls, rises, or incurs more complex patterns.

These limitations of the qualitative analysis also apply to other important aspects. Firstly, the performance development taking place when companies change between the two growth states is unknown. While companies may grow, they will eventually need to reduce their growth rates and transition to an employee-capped dominant logic. Secondly, the impact of environmental factors and managerial targets requires investigation. The same applies to the additional exogenous, contextual variables uncovered in the next chapter regarding the venture's business model design and capability development. Simulation can determine the development of performance outcomes for individual companies (Chapter 5) and hypothetical scenarios (Chapter 6). Moreover, they can evaluate the transitions between growth states and the impact of other influences like the environment and business model design (Sterman, 2000; Warren, 2002; Morecroft, 2015). Therefore, the next chapter will utilise the literature on digital ventures to specify the mechanisms conceptualised in this chapter. It will formalise a simulation model that can be tested and simulated for case companies and hypothetical scenarios.

Chapter 4: Theory contextualisation and model formalisation

The previous chapter has integrated growth paths theory and dynamic states theory to identify the mechanisms affecting value creation and capture. It has also derived insights regarding the development of value creation and capture from the causal loop diagrams. For example, it hypothesised that companies that cap their employee numbers would approach maximum values for their value creation and share of value captured. A formal model and its simulation can test this qualitative analysis. They can also provide further insight regarding the balance of feedback loops for companies without employee limits. This chapter uses stock-and-flow diagrams to illustrate such a formal model and explain its equations. As argued in the methodology section, this chapter thereby provides a more detailed version of the model. The additional details provided here adapt and specify the model for the digital context. The chapter illustrates this detailed model in six subsystems (Figure 4-1).



Figure 4-1: Model subsystems

The first two sections of this chapter focus on the value creation and capture subsystems. They contextualise the performance outcomes for digital ventures. As conceptualised in the previous chapter, the influences on these performance outcomes are traced back to variables in the environment and the venture's resources. The subsequent three sections focus on subsystems regarding different parts of the business model. The technology development subsystem and content creation subsystem⁸ drive product improvements and determine the technological resources and content assets. The customer and user subsystem determines the venture's human resources and capital assets. The resource outcomes of those business model subsystems are inputs to the performance outcome subsystems. As conceptualised in the previous chapter, these subsystems are affected by the venture's environment and dominant logic. Moreover, this chapter identifies additional exogenous variables defining a growth regarding capability development and business model designs.

4.1. Value creation subsystem

This section traces the influences on a digital venture's value creation back to the outputs of the customer and user, technology development, and content creation subsystems (Figure 4-2). The subsystem highlights the external influences from customers and complementary providers that affect value creation. It also considers the business model design choices that affect value creation. Finally, the subsystem derives revenue as an accounting performance measure used to validate the model using a variable from the value capture subsystem.

⁸ This thesis developed the content creation subsystem based on the literature on digital ventures. None of the identified case companies employs a content-based business model and the subsystem operates similar to the technology development subsystem. Therefore, the details of the content creation subsystem are illustrated in Appendix A.



Figure 4-2: Value creation subsystem

This section adapts the conceptualisation of value creation for digital ventures. It distinguishes different customer groups, alters the attachment of the use value to units of the product, considers different value drivers, and recognises the changing usage intensity. The overall argument of model conceptualisation that performance outcomes are dependent on the venture's resources and environment is maintained.

4.1.1. Separating value creation for different users and customers

While customers of many physical products and services are homogenous, some digital ventures such as marketplace or social networks may distinguish between users and customers (Afuah and Tucci, 2000; Kollmann, 2006; Zott, Amit and Massa, 2011). Other digital ventures, such as some software-as-a-service (SaaS) companies, may further distinguish between users and premium users who generate different amounts of value from a product or service (Teece, 2010; Roma and Dominici, 2016; Voigt and Hinz, 2016; Möller *et al.*, 2020). Because digital ventures provide different types of services to these customer and user groups, the value created for them needs to be distinguished (Figure 4-3).



Figure 4-3: Value creation for users and customers

The model calculates a digital venture's *rate of value creation* by combining the *rate of value creation for all users* and the *rate of value creation for customers* (Equation 4-1). It separates the rate of value creation for all users further into the *rate of value creation for users* and the *rate of value creation for premium users* (Equation 4-2). Each value creation rate represents how much value a venture generates for the respective customer and user group in monetary amounts per year.

rate of value creation		Eq 4-1
	= rate of value creation for all users	
	+ rate of value creation for customers	

rate of value creation for all users

Eq 4-2

= rate of value creation for users

+ rate of value creation for premium users

The actual monetary amount that reflects customers' maximum willingness to pay is unknown and cannot be accurately measured (Forrester, 1984; Sterman, 2015; Haile and Altmann, 2016b, 2016a). Therefore, the model treats the rate of value creation as an indexed qualitative variable. At the beginning of the simulation, the rate of value creation has a value of one. Thereby, analysts can easily assess the growth in value creation without knowing the exact monetary amount. For example, a measure of 110 after one year of simulations indicates a 10% growth.

4.1.2. Value creation for each customer and user type

A second difference in the digital context concerns the attachment of the use value to units of products. Firstly, units of products cannot be easily identified and defined due to the openness of digital products and services (Lyytinen, Yoo and Boland, 2016; Nambisan, 2017; Nambisan, Lyytinen and Majchrzak, 2017). Secondly, the size of the venture's customer and user bases, not the rate at which products are sold and shipped, is a critical component of digital ventures' strategies and driver of value creation (Huang *et al.*, 2017; Kelestyn, Henfridsson and Nandhakumar, 2017; Zaheer *et al.*, 2019). Lastly, digital ventures may not generate value per unit of product but per period of time. For example, SaaS companies usually offer subscriptions to their software products. Users generate value from these over the period during which they use the software (Afuah and Tucci, 2000; Harmon, Raffo and Faulk, 2004; Haile and Altmann, 2016b; Gu, Kannan and Ma, 2018). Therefore, the value created for customers depends on the number of customers and the rate at which value is created for the average customer (Figure 4-4)⁹.



Figure 4-4: Rate of value creation for customers

The model calculates the rate of value creation for customers by multiplying together the *customer base* and the *rate of use value creation for customers* (Equation 4-3).

rate of value creation for customers

Eq 4-3

= Customer base

* rate of use value creation for customers

⁹ The figure and the remainder of this section describe mechanisms for customers. The mechanisms are also applied and adapted to determine the value created for users and premium users. Appendix A provides these additional model details.

The rate of customer use value creation is a qualitative variable that reflects how much monetary value is generated from the product by the average customer. The model formalises the conceptualisation of the previous chapter by multiplying together the *strength of product for customers* and *customer usage intensity* (Equation 4-4). Each of these variables has an initial value of one to index value creation.

rate of use value creation for customers

Eq 4-4

- = Customer usage intensity
- * strength of product for customers

The following two subsections describe the product's strength in a digital context and the customer usage intensity. The customer base is an outcome of the customer and user subsystem.

4.1.3. Value drivers determining the product strength

The strength of physical products depends on their features, such as their technological performance (Lepak, Smith and Taylor, 2007). However, digital ventures may rely on a broader range of factors contributing to value creation (Afuah and Tucci, 2000; Amit and Zott, 2001; Kollmann, 2006; Doganova and Eyquem-Renault, 2009; Huarng, 2013; Ojala, 2016; Kraus *et al.*, 2018). These value drivers "refer to any factor that enhances the total value created by an e-business" (Amit and Zott, 2001, p. 494). This thesis has identified five value drivers by synthesising the literature on digital business models. Each may contribute to the quality of a venture's offer for its users and customers:

• *Technological quality*: A venture's product or service is underpinned by its technology in the form of software code. Digital ventures can improve their technological quality through further development by adding features, improving reliability, or improving the ease-of-use of the technology (Wernerfelt, 1984; Bontis and Chung, 2000; Amit and Zott, 2001; Arora, Fosfuri and Gambardella, 2001; Teece, 2007; Clarysse, Bruneel and Wright, 2011; Haile and Altmann, 2016b; Hamari and Keronen, 2017; Nambisan, 2017).

- *Content quality*: Some digital ventures may sell digital artefacts to their customers and users or provide access to them. Examples include newspaper articles, music, or videos. The higher the quality of these content items, the more value is created for users consuming them (Kim, Oh and Shin, 2010; Bradley *et al.*, 2012; Naldi and Picard, 2012; Kallinikos, Aaltonen and Marton, 2013; Kim and Kim, 2017; Nambisan, 2017).
- Direct network effects: Some digital ventures may provide value to users by facilitating engagement with a network of other users. Examples of such business models are particularly present on social networks, on which users engage with other users such as their friends or colleagues. The larger the number of users that one can engage with, the larger the value provided by the venture (Amit and Zott, 2001; Porter, 2008; Stampfl, Prügl and Osterloh, 2013; Steininger, Wunderlich and Pohl, 2013; Zhang, Lichtenstein and Gander, 2015; Kelestyn, Henfridsson and Nandhakumar, 2017; Täuscher and Abdelkafi, 2018). However, some scholars also argue that negative direct network effects exist for some digital ventures. In such cases, a larger number of the same type of user decreases the value for each user. This may occur, for example, due to more competition within a marketplace (Asvanund *et al.*, 2004; Zhu and Bao, 2018).
- Indirect network effects: Indirect network effects refer to improvements in customer value creation because two different types of users interact. This thesis distinguishes these two groups by using the terms users and customers. These network effects are present, for example, on multi-sided markets, in which ventures facilitate transactions between buyers and sellers. However, they are also present on social networks where digital ventures may allow advertising customers to market their products to non-paying users. In such cases, the value of the venture's offer increases with the size of the other type of user on the platform (Amit and Zott, 2001; Pagani, 2013; Staykova and Damsgaard, 2015; Gandia and Parmentier, 2017; Täuscher and Abdelkafi, 2018).
- *Complementary quality*: Other companies may use the openness of digital technologies to provide complementary products to the venture's technology. For example, some social networks allow game developers to provide games

from within the social network. Similarly, many software companies provide interfaces that allow integrating external applications to the venture's software. These complementary products increase the value of the digital venture's product or service depending on their quality (Afuah and Tucci, 2000; Amit and Zott, 2001; Tang, 2006; Song, Parry and Kawakami, 2009; Stampfl, Prügl and Osterloh, 2013; Nambisan, 2017; Eckhardt, Ciuchta and Carpenter, 2018; Teece, 2018).

These five value drivers may affect the strength of the venture's product or service for its users and customers (Figure 4-5). However, digital ventures may make use of these value drivers selectively. For example, SaaS companies may only employ the technological value driver. On the other hand, social networks rely primarily on network effects, while the technology still underpins their product.



Figure 4-5: Strength of product or service for customers

The model formalises each value driver as an indexed qualitative variable. If a venture relies on the value driver, its initial value is one. Otherwise, the value driver has a value of zero. The model multiplies all value drivers to calculate the product's strength (Equation 4-5). The HasValue function¹⁰ checks each driver's value and returns its value if it is greater than zero. If the value driver has the value of zero, the function returns the value of one. Thereby, unemployed value drivers do not affect the calculation of the product strength.

¹⁰ The HasValue function is a custom macro that has been programmed for the Vensim software as part of this thesis. It ensures that the function still returns an appropriate value for the product quality under extreme input values (their lower bound of zero). It uses an if-structure to determine if the variable is zero. If the variable's value is zero, the function returns its second parameter value. Otherwise, it returns the variable. Please see Appendix B for the equations of the macro.

strength of product or service for customers

- = HasValue(Technological quality, 1)
- * HasValue(Content quality, 1)
- * HasValue(Complementary quality, 1)
- * HasValue(direct network effects for customers, 1)
- * HasValue(indirect network effects for customers, 1)

The technological quality and network effects for customers are reviewed below. The complementary quality is an exogenous input. Let's suppose complementary products improve the value of the venture's offer. In that case, the quality of complementary products is initially set as one. The analyst then sets it every year to reflect the average complementary product's percentage improvement. For example, a value of 1.1 at a later point in time would indicate that the average quality of complementary products has improved by 10%. Appendix A provides further details on the product strength for users, network effects for users, and content quality.

4.1.3.1. Technological quality

The technological quality reflects the quality of the firm's software offering. It is formalised as a qualitative variable with the *initial technological quality* set as one. The venture transforms financial resources on its balance sheet into technological resources when it invests in technology. This indicates that the venture improved its product technologically (Huang and Kunc, 2012). The same technology s-curve conceptualised in the previous chapter exists for a digital venture's technology. Thus, the initial investments into technologies only yield minor quality improvements. With further investments, the rate of quality improvements per investment accelerates before it starts to decline as the s-curve reaches its maximum value (Sosa, Browning and Mihm, 2007; Nikula *et al.*, 2010). The structure and equation illustrated in Sterman's (2000) System Dynamics manual have been used to model this s-curve relationship (Figure 4-6).


Figure 4-6: Technological quality

In Sterman's (2000) structure, the technological quality accumulates based on the rate of technology improvements (Equation 4-6). These improvements depend on the rate of investment in technological resources and the effect of investments on technology improvements (Equation 4-7).

Technological quality	Eq 4-6
= initial technological quality	
+ \int rate of technology improvements	
rate of technology improvements	Ea 4-7

ate of technology improvements

ч

* effect of investments on technology improvements

= rate of investment in technological resources

The model calculates the effect of investments in technology on technological improvements based on the technological quality already achieved, the maximum technological quality, and the standard effect of investments on technology *improvements* (Equation 4-8). These elements are combined using the equation provided by Sterman (2000), which adjusts the standard effect by two factors. Initially, the effect size increases with a greater stock of technological quality. However, as the quality reaches the maximum, the second factor affecting the standard effect becomes smaller until no further improvements can be realised.

offect of investments	an tachnalagu improvementa	
effect of investments	on technology improvements	

Eq 4-8

= standard effect of investments on technology improvements

* Technological quality * (maximum technological quality

Technological quality)/maximum technological quality

The analyst needs to set the maximum technological quality after reviewing the types of improvements that the firm makes over the period of investigation. The standard effect has been calibrated automatically using the Vensim software.

4.1.3.2. Network effects for customers

Users and customers may both benefit from direct and indirect network effects. Therefore, four different network effects are considered in the model. The two relevant for customers are presented below: the *direct network effects for customers* and *indirect network effects for customers* (Figure 4-7). As argued above, different business models may employ different types of network effects (Stampfl, Prügl and Osterloh, 2013; Staykova and Damsgaard, 2015; Gandia and Parmentier, 2017; Täuscher and Abdelkafi, 2018; Zhu and Bao, 2018).



Figure 4-7: Network effects for customers

Each network effect has an initial value of one if the venture employs it and zero if the venture does not. This ensures the indexing of value creation. Sterman (2000) suggests normalising such quality factors. Therefore, each network effect is expressed by comparing the current *user base* and *customer base* to a reference point. These reference points are, respectively, the *initial user base, initial premium user base*, and *initial customer base* (Equation 4-9, 4-10).



indirect network effects for customers

- = IF THEN ELSE(customer indirect network effect setting
- = 0, 0, ((Premium user base + User base)/(initial user base

+ initial premium user base))^customer indirect network effect setting)

The model accounts for different business model designs through the *network effect setting* variables. If a venture does not employ a network effect, the analyst needs to set the respective variable as zero. The if-structure in the above equation then sets the network effect to zero (setting=0). For positive and negative network effects, the variable needs to be set as one and minus one. The power function then ensures that a greater user base increases network effects proportionally for positive network effect (setting=1). It also creates a reduction in network effects (setting=-1). Thereby, as suggested by Coyle (1996), these variables operate as policy switches. They adjust the feedback structure of network effects depending on the business model employed by a case company.

4.1.4. Customer usage intensity

As argued above, digital products create value for customers over time. However, the use value created per customer also changes with the intensity at which the average customer uses a product (Kim, Oh and Shin, 2010). For example, customers learn to use a software's interface and features over time (Amit and Zott, 2001; Deng and Wang, 2016). Users and customers may also personalise or customise digital technologies by adopting profiles on social networks or customising software for business processes (DaSilva *et al.*, 2013; Tucker, 2018). Existing customers will improve the value they generate from using a digital product towards a maximum. This maximum may reflect, for example, the number of features or content items available to the user. However, new customers lower the average value created for each customer because they have yet to go through the same learning, customisation, and adoption processes (Figure 4-8).



Figure 4-8: Customer usage intensity

The model combines two established structures to represent the processes of increasing usage intensity and adjustments to new customers (Equation 4-11). Firstly, it utilises Sterman's (2000) s-curve to model the process of increasing usage intensity towards a the *maximum customer usage intensity*. This is reflected in the *rate of improvements in customer usage intensity* (Equation 4-12). Secondly, it uses Warren's (2002) structure to adjust the quality of resources such as customer bases. This adjustment through the *rate of reduction in customer usage intensity* depends on the *new customer usage intensity* and the *fraction of new customers* (Equation 4-13, 4-14)¹¹.

Customer usage intensity

Eq 4-11

- = initial customer usage itensity
- + (rate of improvements in customer usage intensity
- rate of reduction in customer usage intensity)

rate of improvements in customer usage intensity

Eq 4-12

- = standard rate of adoption by customers * Customer usage intensity
- * ZIDZ(maximum customer usage intensity
- Customer usage intensity, maximum customer usage intensity)

¹¹ The XIDZ and ZIDZ functions prevent divisions by zero. The ZIDZ function requires a dividend and a divisor separated by a comma. The XIDZ function requires a return value as a third argument. If the divisor is zero, the XIDZ functions return the third argument, while the ZIDZ function returns zero. They ensure that the equations work under extreme conditions.

rate of reduction in customer usage intensity

- = (Customer usage intensity
- new customer usage intensity)
- * fraction of new customers

fraction of new customers

= XIDZ(rate of customer acquisition, Customer base, 1)

The *initial customer usage intensity* has a value of one to index value creation. The maximum usage intensity, the speed of adoption, and the usage intensity of new customers are constant throughout a growth state. The analyst needs to set them based on the case company's data. Analysts can use different data points to approximate them. These may include, for example, time spent using software, the rate of content consumption, the rate of engagement in a network or transactions, the number of complementary products used by the average, or the number of employees in a company using a product (Bontis and Chung, 2000; Amit and Zott, 2001; Haile and Altmann, 2016; Lyytinen, Yoo and Boland, 2016; Nambisan, 2017; Guggenberger *et al.*, 2020).

4.1.5. Calculation of reported figures and accounting performance

Practitioners and empirical studies commonly use a company's financial statements to evaluate its performance. They may use revenues to approximate value creation in the corresponding periods. Revenue growth between two consecutive periods is then used to approximate the growth in value creation (Porter, 1985; Garnsey, Dee and Ford, 2006; Steffens, Davidsson and Fitzsimmons, 2009). The model allows comparing the theory-driven conceptualisation of value creation to accounting measures such as revenue, which are widely available due to financial reporting requirements. However, there are conceptual differences between the values on financial statements and the variables in System Dynamics models. Reported figures need to be derived before calculating accounting performance measures to overcome these differences.

Eq 4-14

Eq 4-13

There is a significant conceptual difference between the rate variables specified above and the accounting figures shown on the firm's financial statements. Both measure a venture's performance over a period of time (Warren, 1999, 2002). However, accounting measures illustrate absolute amounts over a period. For example, revenue as the top line of an income statement reflects the amount of money captured from customers over the last financial year. On the other hand, the variables in a System Dynamics model express instantaneous rates of change (Sterman, 2000). After simulating a System Dynamics model for one year, the revenue variable in the model does not show how much value a company has generated over the past year. The variable shows the annualised rate of revenue generation at that point in time. Thus, the accounting measures are backwardslooking, i.e. report the revenue generated over the last twelve months. The instantaneous rates of change expressed by System Dynamics variables have the advantage of measuring performance in the moment. This allows managers to assess the performance of their firm timelier. However, System Dynamics modellers also need to model the reporting system (Sterman, 2005). Because "level variables are the only directly observable variables in a system; instantaneous rates of flow are not observable", modellers need to average accounting figures over the reporting period. This *reporting period* is set as one year to correspond to annual reports. One can then use these averages to calibrate and test the model by comparing these values to financial statements (Forrester, 1984, p. 8; Sterman, 2000, 2015; Oliva, Sterman and Giese, 2003). This relationship has been used to derive the *rolling annual revenue* (Figure 4-9).



Figure 4-9: Rolling annual revenue

The rolling annual revenue is the average of the *current rate of revenue generation* over the last year. Following Oliva, Sterman, and Giese (2003), a custom macro has been programmed for the Vensim software to derive these averages. It uses a stock with the current rate as an inflow and the rates no longer part of the reporting period as outflows (Equation 4-15)¹².

rolling annual revenue Eq 4-15 = ReportedFinancials (current rate of revenue generation,

reporting period)

The share of value captured from customers and users reflects the fraction of the created value that is captured as revenue. Multiplying it with the *rate of value creation* yields the *current rate of revenue generation* (Equation 4-16).

current rate of revenue generation

Eq 4-16

= rate of value creation

* share of value captured from customer and users

This share of value captured from customers and users is an input from the value capture subsystem presented in the next section. This equation and the use of a variable related to capturing value illustrate the limitations of revenue as a proxy of value creation used in previous empirical studies. Due to competition, the price paid by customers for a product is usually lower than their maximum willingness to pay. Therefore, the amount of value creation is usually greater than revenue. Moreover, a fall in revenue does not necessarily imply that the venture creates less value. Revenue may have dropped due to increasing competition (Lepak, Smith and Taylor, 2007). The model uses the strength of simulation models, which are not constrained by exact numerical measurement of concepts (Langley, 1999). This allows it to separate the rate of value creation and the rate of revenue generation. The model reflects value creation as theories conceptualise it without the use of revenue as a proxy.

¹² The macro includes a stock that averages the accounting figures over the reporting period. When less than a full period has passed in the model, and there is not a full year of data in the stock, the macro annualises the current content of the stock. Appendix B illustrates the full details of this macro.

4.1.6. Overview of subsystem inputs

This section has contextualised and formalised value creation as the first performance outcome for digital ventures during the growth process. The outcomes of the subsystem are the rate of value creation as the theoretical performance measure and revenue as a proxy (Table 4-1). This proxy can be compared to financial statements to test the model. The model derives these outcomes by accounting for different customer and user groups, value creation over time, diverse value drivers, and changing usage intensities. Through these mechanisms, variables related to a venture's resources have been identified as affecting value creation. The share of value captured is also required to calculate revenue. These inputs are discussed in the other sections of this chapter that cover the respective subsystems of the formal model.

• Rate of value creation as the theoretical performance measure		
Rolling annual revenue as an accounting proxy for value creation		
Exogenous inputs		
Business model design:		
Premium product advantage		
 Maximum technological quality, standard 		
effect of investment in technology on		
technological improvement		
• Customer direct network effect settings,		
customer indirect network effect settings, user		
base, initial premium user base direct network effect settings, user indirect		
network effect settings		
Initial content quality		
Environment – Customer and user behaviour:		
• Initial customer usage intensity, initial user		
usage intensity		
• New customer usage intensity, new user usage		
intensity		
• Maximum customer usage intensity,		
maximum user usage intensity		
• Standard rate of adoption by customers,		
standard rate of adoption by users		
Environment – Complementary products:		
Complementary quality		

Table 4-1: Outputs and inputs of the value creation subsystem

In addition to the variables calculated in other subsystems, the value creation subsystem relies on inputs set by the analyst based on information in the company's annual reports. Some of these variables are required to set up the model for a venture's business model. These variables activate or deactivate specific mechanisms in the model to represent, for example, freemium business models or network effects. The analyst also needs to set the maximum technological quality by reviewing the nature of changes executed on the venture's technology. The standard effect of investments into technology on the technological quality is set using automatic model calibration. Lastly, other variables reflect environmental conditions such as customer and user behaviour and complementary product quality.

4.2. Value capture subsystem

This section traces the influences on a digital venture's share of value captured back to variables in other subsystems (Figure 4-10). The share of value captured from customers and users depends on influences from the value creation subsystem, the business model design, and competition in the external environment. The shares of value lost to input providers and the erosion of resources depend on influences from the business model subsystems and the input providers in the external environment. Like the value creation subsystem, the model formalises the share of value captured based on theory. In addition, it derives profitability ratios using the share of value captured, which can be compared to case companies' financial statements.



Figure 4-10: Value capture subsystem

The share of value captured reflects the fraction of created value received and maintained by the venture as profit (Teece, 1986; Zott and Amit, 2010). As

conceptualised in growth process theories, it depends on the share of value captured from customers and users and the share of value lost to input providers and the erosion of resources (Figure 4-11).



Figure 4-11: Share of value captured

The formal model subtracts the *share of value lost to input providers and erosion of resources* from the *share of value captured from customers and users* (Equation 4-17). The share of value captured from customers and users represents the fraction of created value captured as revenue. The share of value lost to input providers expresses the venture's costs as a fraction of the value creation. The calculated difference thus expresses the *share of value captured* as the fraction of value maintained by the venture.

share of value captured

Eq 4-17

- = share of value captured from customers and users
- share of value lost to input providers and erosion of resources

Below, the share of value captured from customers and different shares of value lost have been adapted for the digital context. The model further formalises them to identify the input variables in the venture's resource base and environment.

4.2.1. Share of value captured from customers

While physical-product companies usually capture value from all customers, digital ventures may capture value more selectively from their customer and user bases. Firstly, despite creating value for different types of users and customers, digital ventures may not capture value from all types. For example, many digital marketplaces connecting buyers and sellers capture value only from sellers by charging a transaction fee, whereas the service is free for buyers. Similarly, social networks do not usually capture value from their users but capture value from their advertising customers (Afuah and Tucci, 2000; Muzellec, Ronteau and Lambkin, 2015; Staykova and Damsgaard, 2015; Remane *et al.*, 2016; Gandia and Parmentier, 2017). Secondly, digital ventures employing a freemium business model may capture value only from premium users but not from ordinary users. SaaS companies and some social networks and content providers may employ freemium models to allow users to test the product and entice them to purchase the upgraded version (Teece, 2010; Roma and Dominici, 2016; Voigt and Hinz, 2016; Möller *et al.*, 2020).

The model distinguishes different types of users and customers for value creation. Similarly, it distinguishes the types for value capture. The share of value captured from customers and users considers that value is only captured from customers and premium users, while no value is captured from standard users. In growth process theories, the share of value captured depends on the bargaining power of the venture to its customers (Porter, 1991; Peteraf and Barney, 2003; Garnsey, Dee and Ford, 2006; Lepak, Smith and Taylor, 2007). This logic has been applied to the value captured from customers and premium users (Figure 4-12).



Figure 4-12: Share of value captured from customers and users

The *bargaining power to customers* and *bargaining power to premium users* reflects the fraction of value captured from customers and premium users. The model calculates the average of these two bargaining powers, weighted by their share of value creation (Equation 4-18).

share of value captured from customers and users

- = Bargaining power to customers
 - * (rate of value creation for customers
 - /rate of value creation)
 - + Bargaining power to premium users
 - * (rate of value creation for premium users
 - /rate of value creation)

The value creation subsystem models the rates of value creation. The bargaining power and its influences are discussed below¹³.

4.2.1.1. Bargaining power to users and customers

The bargaining power to customers reflects the fraction of value captured from the average customer. As in the conceptual model, this bargaining power depends on the strength of the venture's product or service relative to the strength of competitor and substitute products (Porter, 1991; Peteraf and Barney, 2003; Garnsey, Dee and Ford, 2006; Lepak, Smith and Taylor, 2007). While this general principle is maintained, the formal model adjusts the mechanisms for two particularly prominent specifications in the digital context (Figure 4-13). Firstly, existing customers develop switching costs over time. These costs make existing customers less likely to switch to competitors or substitutes and thus increase the venture's bargaining power to customers. Moreover, the development of switching costs implies that ventures have a lower bargaining power to new customers. These new customers have not yet developed switching costs (Amit and Zott, 2001; Marquez and Blanchar, 2006; Kooskora, 2020). Secondly, some digital ventures lock customers into contracts for months, years, and some even longer than three years. Such contract periods are particularly prominent among SaaS ventures in a business-to-business context. This contract period delays the impact of competition, product improvements, and switching costs on the bargaining power (Harmon, Raffo and Faulk, 2004; Saaksjarvi and Lassila, 2005; Tyrväinen and Selin, 2011; Voigt and Hinz, 2016; Kohtamäki et al., 2019).

¹³ Just as for the value creation mechanisms, this subsection focuses on customers. The same mechanisms also exist for the value captured from premium users. To avoid repetition, these are presented in Appendix A.



Figure 4-13: Bargaining power to customers

The structure used to model the usage intensity has been adjusted to formalise the development of the bargaining power to customers. Over time, the *initial bargaining power to customers* changes through improvements in the bargaining power to existing customers and adjustments for the bargaining power to new customers (Equation 4-19). The *change in bargaining to existing customers* adjusts the overall bargaining power to the *bargaining power to existing customer contract period* (Equation 4-20). The *adjustment of bargaining power to new customers* reduced the bargaining power towards the *bargaining power to new customers* depending on the *fraction of new customers* (Equation 4-21).

Bargaining power to customers

Eq 4-19

- = initial bargaining power to customer
- + (change in bargaining power to existing customers
- adjustment of bargaining power to new customers)

change in bargaining power to existing customers

Eq 4-20

- = ZIDZ(bargaining power to existing customers
- Bargaining power to customers, customer contract period)

adjustment of bargaining power to new customers	
= (Bargaining power to customers	4-2
 bargaining power to new customers) 	1
* fraction of new customers	

The bargaining power to new customers depends on the strength of the product for customers relative to the value provided by competitors and substitutes (Equation 4-22). These two variables have been combined through an equation commonly used when considering competition, for example, to determine market shares (Sterman, 2000). The model uses this equation because it fulfils the theoretical underpinning of the bargaining power in growth process theories. When there is no competition, the venture's bargaining power is one, and it captures the entire value created. With increasing competition, the bargaining power reduces but never turns negative. The equation fulfils these conditions for extreme values and ensures the bargaining power always remains between zero and one. The model also considers *customer switching costs* to calculate the *initial bargaining power* and *bargaining power to existing customers*. The model uses switching costs as a multiplier that increases the strength of the product for customers. If there are no switching costs, the product strength of existing customers stays at the level for new customers (Equation 4-23, 4-24).

bargaining power to new customers	Eq 4-22
= ZIDZ(strength of product for customers,	
strength of product for customers	
+ value provided by competitors and substitutes)	

initial bargaining power to customer

Eq 4-23

- = ZIDZ(strength of product for customers
- * (1 + customer switching costs),
- strength of product for customers * (1
- + customer switching costs)
- + value provided by competitors and substitutes)

bargaining power to existing customers

- = ZIDZ(strength of product for customers
- * (1 + customer switching costs),
- strength of product for customers * (1
- + customer switching costs)
- + value provided by competitors and substitutes)

The value creation subsystem has reviewed the product strength. The switching costs and value provided by competitors and substitutes are described below. The contract period is set based on statements made in the company's annual reports and on its website.

4.2.1.2. Customer switching costs

The development of customer switching costs increases the bargaining power of the venture to its customers over time. These switching costs develop as customers increase their usage intensity. For example, Amit and Zott (2001) argue that the familiarity gained from learning to use a user interface or feature is a form of switching costs. DaSilva *et al.* (2013) describe customising settings and inserting data to cloud platforms as switching costs. Similarly, Tucker (2018) describes the effort made to customise user profiles on social networks as switching costs. The model has considered these processes of personalising, customising, and learning about a digital product or service via the usage intensity increases towards a maximum over time. The model expresses switching costs as the increase in the usage intensity (Figure 4-14). The model also formalises a process to approximate the average usage intensity by new customers to account for companies changing their target customers to ones with different usage intensities.



Figure 4-14: Customer switching costs

The model calculates customer switching costs as the increase in the *customer usage intensity* expressed as a fraction of the *average initial customer usage intensity* (Equation 4-25). The average rate's initial value is the usage intensity of new customers at the beginning of the simulation. It is then adjusted to reflect changes in the venture's target customers (Equation 4-26). The *change in average initial customer usage intensity* moves the current average towards the *new customer usage intensity* depending on the *fraction of new customers* (Equation 4-27). If the venture does not change its target customer segment, no adjustment takes place. Similarly, no adjustment takes place when no new customers are acquired.

customer switching costs	
= ZIDZ(Customer usage intensity	
– Average initial customer usage,	
Average initial customers usage intensity)	
customers usage intensity = new customer usage intensity +∫(change in average initial customer usage intensity)	Eq 4-26
age initial customer usage intensity = (new customer usage intensity - Average initial customer usage intensity) * fraction of new customers	Eq 4-27
	hing costs = ZIDZ(Customer usage intensity - Average initial customer usage, Average initial customers usage intensity) customers usage intensity = new customer usage intensity + \int (change in average initial customer usage intensity) nge initial customer usage intensity = (new customer usage intensity - Average initial customer usage intensity) * fraction of new customers

All inputs to this part of the model are derived in the value creation subsystem.

4.2.1.3. Value provided by competitors and substitutes

The number of competitors and substitute providers and the strength of their products affect the *value provided by competitors and substitutes* (Figure 4-15). The number of competitors can be expressed by the market conditions as monopolies, oligopolies, and perfect competition. Competitors' and substitutes' strength refers to the quality of the average competitor product or substitute (Porter, 1991; Peteraf and Barney, 2003; Lepak, Smith and Taylor, 2007; Zhu and Iansiti, 2007; Naudé and Liebregts, 2009; Ma and Kauffman, 2014; Ruutu, Casey and Kotovirta, 2017; Teece, 2018).



Figure 4-15: Value provided by competitors and substitutes

The model multiplies together the *strength of competitors or substitutes* and the *number of competitors or substitutes* to calculate the value provided by competitors and substitutes (Equation 4-28).

value provided by competitors and substitutes Eq 4-28 = Strength of competitors or substitutes * Number of competitors and substitutes

Similar to the use value created by the digital venture for its users and customers, both variables are treated as qualitative variables. The analyst needs to set the number of competitors and substitutes based on whether the market is a monopoly (0), an oligopoly (0.25), or under perfect competition (1). The analyst sets the strength of competitors and substitutes to represent their relative product strength based on the qualitative information provided in annual reports. For example, the average competitors' product might be equally good (1) or only have 80% of the functionality of the venture's product (0.8).

4.2.2. Share of value lost to input providers and erosion of resources

The venture loses value due to the erosion of its resources and the compensation paid to input providers. The overall logic for non-digital businesses conceptualised in growth process theories can also be applied to digital ventures. Below, these shares of value lost are formalised. Thereby, this subsystem adds further details explaining the influences of resources and the external environment on the shares of value lost (Figure 4-16).



Figure 4-16: Share of value lost to input providers and resource erosion

The *share of value lost to input providers and the erosion of resources* is the fraction of created value lost during the value creation process. It is composed of two fractions that are added up (Equation 4-29).

share of value lost to input providers and erosion of resources Eq 4-29

= share of value lost to input providers

+ share of value lost to erosion of resources

The *share of value lost to input providers* expresses the costs to compensate input providers as a fraction of the value created. The share of value lost to input providers is a fraction equal to or larger than zero. It aggregates the share of value lost to each type of input provider identified in the model's conceptualisation. These are operating input providers in the form of human resources and providers of capital. Thus, it adds up the *share of value lost to operating input providers of capital* (Equation 4-30). The share of value lost to operating input providers is calculated based on the *efficiency of operations* and the *cost of resource inputs* (Equation 4-31), while the share of

value lost to providers of capital is calculated based on the *efficiency of capital employed* and the *cost of capital* (Equation 4-32). Each statement of efficiency reflects how much value is created per unit of resource input. Therefore, the inverse of each efficiency statement determines how many units of a resource are required to create one unit of value. Multiplying the cost per unit of resource with the inverse of the efficiency thus determines the share of value lost to each input provider. The model uses the venture's *total number of employees* as its operating resource use (Equation 4-33) and the *capital employed* as its capital resource use (Equation 4-34). The model considers four different types of human resources that make up the venture's total number of employees: *value delivering employees*, *marketing and selling employees*, *technology developing employees*, and *firm managing employees* (Equation 4-35). Similarly, the model considers the *equity employed* and the *debt employed* (Equation 4-36). These resources are inputs from the firm managing subsystem. The costs per unit of resource are reviewed below.

share of value lost to input providersEq 4-30= share of value lost to operating input providers+ share of value lost to providers of capitalshare of value lost to operating input providersEq 4-31
$$= \frac{1}{efficiency of operations} * cost of resource inputsEq 4-32share of value lost to providers of capitalEq 4-32 $= \frac{1}{efficiency of capital employed} * cost of capitalEq 4-33efficiency of operations = $\frac{rate of value creation}{total number of employees}$ Eq 4-34total number of employeesEq 4-35 $= Firm$ managing employeesEq 4-35 $+ Marketing and selling employees$ Eq 4-36$$$

The second source of value slippage is the erosion of the venture's resources in its value creation process (Nelson and Winter, 1978; Dierickx and Cool, 1989; Porter, 1991; Knott, Bryce and Posen, 2003). The *share of value lost to the erosion of resources* expresses the sum of all amortisation and depreciation expenses as a fraction of the value created (Equation 4-37).

share of value lost to erosion of resources

Eq 4-37

= (rate of amortisation of technological resources
+ rate of depreciation of complementary assets
+ rate of amortisation of content assets)
/rate of value creation

The rates of amortisation and depreciation are formalised in the technology development, content creation, and firm managing subsystems.

4.2.2.1. Operating of resource inputs

The cost of resource inputs reflects the annual costs per employee. Four different types of human resources and activities have been specified above. Each type of employee is associated with different costs, determined by labour market conditions (Garnsey, 1998). For example, programmers as part of the firm's technology developing employees are in high demand. Therefore, they can command a higher wage than, for example, marketing and sales employees. In addition to wages, the expenses per employee in the model also include other costs paid to suppliers that are part of employee activities. The model calculates the *cost of resource inputs* as the weighted average of the different human resource costs (Figure 4-17).



Figure 4-17: Cost of resource inputs

The model calculates the *cost of resource inputs* as a weighted average. It multiplies the number of employees by their respective costs and then divides the sum of those products by the total number of employees (Equation 4-38).

cost of resource inputs

Eq 4-38

= (Value delivering employees
* cost of value delivering inputs
+ Marketing and selling employees
* cost of marketing and selling inputs
+ Technology developing employees
* cost of technology developing inputs
+ Firm managing employees
* cost of firm managing inputs)
/total number of employees

Each input cost is exogenous to the model. They are calculated by dividing the firm's annual indexed expenses on the income statement by the firm's average number of indexed employees during the year. If the firm changes its growth state, the cost per input might change. For example, companies may internalise more activities, thereby hiring more employees while also altering the cost of that activity (Levie and Lichtenstein, 2010). The averages of values within growth states with the same input constellation have been used as model inputs.

4.2.2.2. Providers of capital

The *cost of capital* reflects the annual costs of each unit of capital employed. These costs are paid to the venture's providers of capital. While the *cost of equity* relates to the return provided to equity investors through dividends, the *cost of debt* reflects the interest rate provided to debt investors. The higher each cost of capital, the higher the overall cost of capital (Figure 4-18).



Figure 4-18: Cost of capital

The cost of capital is calculated as the weighted average of the cost of equity and the cost of debt (Equation 4-39).

```
cost of capital = (Equity employed * cost of equity + Debt employed
* cost of debt)/total capital employed
```

The cost of equity and debt are exogenous inputs to the model. They are calculated as inputs from the firm's financial statements by dividing the respective cost, i.e. the interest expense or dividends, by the average amount of debt or equity employed in the expense's year. They are primarily driven by capital market conditions (Garnsey, 1998), which the firm is too small to affect significantly. They are held constant over a growth state (Levie and Lichtenstein, 2010).

4.2.3. Calculation of reported figures and accounting measures

Empirical studies, managers, and investors may approximate the firm's ability to capture value using its return on assets. Thereby, the efficiency of using the resources recognised on the venture's balance sheet to generate net income is expressed (Ireland, Hitt and Sirmon, 2003; Davidsson, Steffens and Fitzsimmons, 2009; Delmar, McKelvie and Wennberg, 2013; Zhou, Park and Ungson, 2013). While most studies measure the ability to capture value with return on assets, growth paths theory has suggested profit margins as an alternative proxy (Garnsey, Dee and Ford, 2006). Similar to the value creation subsystem, the model allows comparing the theory-driven conceptualisation of value capture to accounting performance measures. This requires the conversion of model rates to annualised, reported figures.

4.2.3.1. Calculation of reported figures

As for revenue in the value creation subsystem, the instantaneous rates of change of the model need to be converted to annualised figures (Forrester, 1984; Warren, 1999, 2002; Sterman, 2000, 2005, 2015; Oliva, Sterman and Giese, 2003). The same logic applied to revenue has been applied to determine the *rolling annual operating expenses* and *rolling annual financing expenses* (Figure 4-19).

112



Figure 4-19: Rolling annual expenses

The rolling annual operating expenses and rolling annual financing expenses are the averages of the *current rate of operating expenses* and *the current rate of financing expenses* over the reporting period of one year. The model uses the macro utilised in the value creation subsystem to derive these averages (Equation 4-40, 4-41).

rolling annual operating expenses Eq 4-40 = Reported Financials(current rate of operating expenses, reporting period)

rolling annual financing expenses Eq 4-41 = ReportedFinancials(current rate of financing expenses , reporting period)

The *current rate of operating expenses* includes compensations made to input providers and depreciation and amortisation expenses. In the model, the *share of value lost to operating input providers* and the *share of value lost to erosion of resources* represents these costs as fractions of the created value. The current rate at which operating expenses are incurred can be obtained by multiplying the sum of the fractions with the rate of value creation (Equation 4-42).

current rate of operating expenses

Eq 4-42

= rate of value creation

- * (share of value lost to operating input providers
- + share of value lost to erosion of resources)

The current rate of financing expenses expresses the rate at which the venture incurs financing expenses to compensate capital providers. The share of value *lost to providers of capital* expresses these expenses as a fraction of the value created. Multiplying this fraction with the rate of value creation determines the rate of financing expenses. However, there is a conceptual difference regarding financing expenses between the financial statements and growth process theories. Accountants consider dividends irrelevant to profit. Growth process theories consider dividends as a necessary compensation made to investors for their contributed capital (Nelson and Winter, 1982; Lockett and Thompson, 2004; Penrose, 2009). Therefore, return on assets and profit margins as proxies inflate the resources generated in the growth process compared to the conceptualisations in growth process theories. To account for this conceptual difference, dividends paid must be disregarded in financing expenses when calculating accounting measures of company performance to compare them to financial statements. The rate of dividend payment can be determined by the venture's equity employed and the cost of equity (Equation 4-43). They are subtracted from the financing expenses (Equation 4-44).

current rate of financing expenses

Eq 4-43

- = rate of value creation
- * share of value lost to providers of capital
- rate of divident payment

rate of divident payment = Equity employed * cost of equity Eq 4-44

The equity employed is formalised as part of the firm managing subsystem below, while the cost of equity is formalised above.

4.2.3.2. Comparison to accounting performance measures

The reported revenue as a measure of value creation on the firm's income statement is already formalised in the value creation subsystem. The reported figures in this subsystem allow deriving the return on assets and net profit margin as accounting measures (Figure 4-20). They can be used as proxies for a company's ability to capture value and to test the model.



Figure 4-20: Return on assets and net profit margin

The *rolling annual operating profit* is the rolling annual revenue less the rolling annual operating expenses (Equation 4-45). The *rolling annual profit before tax* is the rolling annual operating profit less the rolling annual financing expenses (Equation 4-46). The *rolling annual net income* is calculated by subtracting the *rolling annual taxation* from the rolling annual profit before tax (Equation 4-47). The amount of taxes paid is an input from the venture's income statements. The *rolling annual return on assets* is calculated by dividing the net income by the *total assets* (Equation 4-48). Similarly, the venture's *rolling annual net profit margin* is calculated by dividing the venture's net income by its revenue (Equation 4-49).

rolling annual operating profit	
= rolling annual revenue	
 rolling annual operating expenses 	
olling annual profit before tax Ec	q 4-46
= rolling annual operating profit	
 rolling annual financing expenses 	
olling annual net income Ec	q 4-47
= rolling annual profit before tax	
 rolling annual taxation 	

rolling annual return on assets =
$$\frac{\text{rolling annual net income}}{\text{total assets}}$$
Eq 4-48rolling annual net profit margin = $\frac{\text{rolling annual net income}}{\text{rolling annual revenue}}$ Eq 4-49

The venture's total assets recognised on its balance sheet need to be determined to calculate return on assets. However, the conceptualisation of resources in the subsequent sections and those used to determine assets as balance sheet positions differ. A wide range of resources are considered in the model. However, a balance sheet considers only those resources that can be expressed in financial terms, i.e. those measured in amounts of money. Resources considered in the model in monetary amounts are *financial resources*, *technological resources*, complementary assets, and content assets (Equation 4-50). Other resources are considered in the model because they are relevant to the growth process and affect the performance outcomes such as human resources or the venture's product quality. However, these resources are not considered on a balance sheet and therefore not included when calculating the return on assets. On the other hand, the balance sheet may also include items not considered in the model, such as goodwill. These accounting measures are not considered relevant in growth process theories and do not affect value creation and capture (Grant, 1991; Penrose, 2009). These *other assets* are added to the balance sheet resources considered in the model to determine a return on assets figure comparable to financial statements (Equation 4-51).

total assets = assets in model + other assets
$$Eq 4-51$$

The value of other assets is an input to the model imported from the venture's financial statements. The assets stocks included in the model are derived in the firm managing, technology developing, and content creation subsystems.

4.2.4. Overview of subsystem inputs

This section has contextualised, formalised, and specified value capture for digital ventures. In addition, the return on assets and the net profit margin have been calculated using the share of value captured. These accounting measures calculated in the model can be compared to ventures' financial statements to test the model (Table 4-2). In addition, this section has worked backwards from the share of value captured to identify the inputs required from other subsystems. These include the venture's resources considered in the technology development, content creation, and firm managing subsystems. Moreover, variables from the value creation subsystem weigh different shares of value captured, approximate switching costs, and derive accounting measures.

Ou	tputs		
•	Share of value captured as the theoretical performance measure		
•	• Rolling annual return on assets and rolling annual net profit margin as accounting		
Ing	Inputs from other subsystems Exogenous inputs		
Te	chnology developing and content creation	Business model designs:	
sut	osystems:	• Customer contract period,	
•	Technological resources, rate of amortisation of technological resources	premium user contract period	
•	Content assets, rate of amortisation of content	Environment – Competition:	
	assets	• Strength of competitors and	
Fir	m managing subsystem:	substitutes	
•	Marketing and selling employees, value delivering employees, technology developing employees, firm	• Number of competitors and substitutes	
	managing employees	Environment – Input	
•	Equity employed, debt employed	providers:	
•	Complementary asset, rate of depreciation of	Cost of value delivering inputs cost of marketing	
_	Complementary asset	and selling inputs cost of	
• Financial resources		technology developing	
•	Rate of value creation, rate of value creation for customers, rate of value creation for premium users Strength of product of customers, strength of product of users Customer usage intensity, user usage intensity New customer usage intensity, new user usage	 inputs, cost of firm managing inputs Cost of equity, cost of debt Accounting measures: Other assets, rolling annual taxation 	
•	Fraction of new customers, fraction of new premium users, premium user base, rate of premium user acquisition Premium product advantage		

Table 4-2: Outputs and inputs of th	he value capture subsystem
-------------------------------------	----------------------------

In addition, the share of value captured depends on exogenous inputs regarding the business model design, environmental growth state inputs, and accounting measures. Firstly, the analyst needs to set contract terms for premium users and customers. Secondly, the analyst needs to evaluate the strength and number of competitors and substitute providers and set the costs of different resource inputs. Lastly, the model requires accounting measures that are not theoretically relevant to value creation and capture but ensure consistency of variables for testing purposes.

4.3. Technology development subsystem

Technological resources reflect the venture's software code that underpins its product or service. They determine the technological quality, one of the value drivers in the value creation subsystem. As conceptualised in the previous chapter, resources develop through activities, which rely on human resources and capabilities (Figure 4-21). Thus, the development activities rely on the venture's technology development capabilities and the employees made available by the firm's management. However, in a digital context, the established technology requires maintenance, which reduces the availability of employees for new development. These development and maintenance activities depend on the technological quality calculated in the value creation subsystem, the technology development capabilities, and the human resources in the firm managing subsystem.



Figure 4-21: Technology development subsystem

The following subsections formalise the development of technological resources, the impact of technological development and technological maintenance activities, and the development of technology developing capabilities.

4.3.1. Technological resources

The model's stock of technological resources reflects the technological assets capitalised on the venture's balance sheet. These assets are an accounting measure of the accumulated development efforts regarding the software that underpins a venture's product or service (Arora, Fosfuri and Gambardella, 2001; Nambisan, 2017). They increase through investments into technology and decrease through amortisation (Figure 4-22).



Figure 4-22: Technological resources

From their initial value, technological resources accumulate through investments and amortisation (Equation 4-52). The *rate of amortisation of technological*

resources depends on the *lifetime of technological resources* (Knott, Bryce and Posen, 2003; Teece, 2007). For example, if a resource has an estimated lifetime of five years, one-fifth of the value is amortised every year. Therefore, the model calculates the amortisation rate by dividing the stock of technological resources by their lifetime (Equation 4-53).

Technological resources	Eq 4-52
= initial technological resources	
+ \int (rate of investments in technolog	39
 rate of amortisation of technologic 	al resources)
rate of amortisation of technological resources	Eq 4-53
technological resources	
= lifetime of technological resources	

The initial technological resources are derived from the venture's financial statements. The resources in monetary amounts are indexed with the initial rate of value creation being one. Similarly, the lifetime in years is an exogenous input derived from the firm's financial statement. This approach is consistent with previous System Dynamics models such as the one developed by Huang and Kunc (2012), who developed a generic model for start-ups, and Oliva, Sterman, and Giese (2003), who developed a model replicating the growth of e-commerce companies. The rate of investments into technology is considered in the next subsection.

4.3.2. Technology development activities

Digital ventures' technology developing activities create their technological resources. Technology developing activities in a digital context relate to product improvements generated through software development, for example, by programming additional features (Bontis and Chung, 2000; Harmon, Raffo and Faulk, 2004; Baliyan and Kumar, 2014; Teece, 2018; Zaheer *et al.*, 2019). Moreover, in a digital context, the created technology requires continuous maintenance.

4.3.2.1. Technology development

The model defines technological resources as those resources capitalised on the venture's balance sheet. Therefore, the rate of investment in technology reflects the rate at which capital is invested into technological resources. The venture records those investments as capitalised software development expenses on its cash flow statement (Afuah and Tucci, 2000; Ruutu, Casey and Kotovirta, 2017; Teece, 2018). The model implements Sterman's (2000) resource-productivity-flow relationship to formalise activities, including technology development. The flow of technological investments depends on the executing resources and their productivity. The resources are the available development time of the venture's technology developing employees. Their productivity is represented by their ability to develop technology (Figure 4-23).



Figure 4-23: Technology development

The model calculates the *rate of investment in technology* by multiplying together the *technology developing employees*, their *fraction of time available for development*, and the *ability to develop technology* (Equation 4-54).

rate of investment in technology

Eq 4-54

- = Technology developing employees
- * fraction of time available for development
- * ability to develop technology

The number of technology developing employees is determined in the firm managing subsystem. The ability of each employee to develop technology is formalised as a capability below. The fraction of time available for development is determined by the firm's technology maintenance requirements.

4.3.2.2. Technology maintenance

Physical products are not usually altered by the development team after the product has been manufactured and shipped. Rather, the development team will develop a new version or different product. However, digital products and services are never finalised. Instead, software is altered continuously even after it has been sold and while the customer is using it due to two reasons. Firstly, digital products and services often include programming errors or security problems fixed over time (Sosa, Browning and Mihm, 2007; Rahmandad and Weiss, 2009; Woodard et al., 2013; Zhang and Niu, 2013; Cristofaro, 2020). Secondly, because digital products and services are open and interact with a range of other software such as operating systems, a change in these related products or services may require updating the venture's product (Nambisan, 2017; von Briel, Recker and Davidsson, 2018; Zaheer et al., 2019; Cristofaro, 2020). The venture's technology developing employees thus spend time debugging and updating existing features. This requirement reduces the time available for new developments. The more sophisticated the venture's existing technology, the more time will be spent on debugging and updating. The venture's technology developing employees execute this maintenance based on their capacity to maintain the technology (Figure 4-24).



Figure 4-24: Technology maintenance and time for development

This *capacity to maintain technology* depends on the venture's number of *technology developing employees* and their *ability to maintain technology* (Equation 4-55). It reflects the technological quality the employees can maintain. The excess over capacity is available for new developments. The model calculates the available time fraction by expressing the difference between the capacity and quality as a fraction of the capacity (Equation 4-56). The equation is stable under extreme conditions: When there is no technology yet, employees can

spend their entire time developing new technology. The max function ensures that the fraction of time does not turn negative when capacity is insufficient. In such cases, no time is available for new development.

capacity to maintain technology

Eq 4-55

= Technology developing employees

ability to maintain technology

fraction of time available for development Eq 4-56

= MAX(0, ZIDZ(capacity to maintain technology

- Technological quality, capacity to maintain technology))

The firm's management determines the technology developing employees. The existing technological quality is calculated in the value creation subsystem, and the ability to maintain technology is formalised below.

4.3.3. Technology development capabilities

In the model conceptualisation chapter, capabilities as measures of productivity depend on employee competence to execute activities and the adequacy of complementary assets (Subsection 3.2.3). While the average employee's competence improves through learning, it erodes when hiring new employees (Slater, 1980; Garnsey, 1998; Tan and Mahoney, 2005; Hugo and Garnsey, 2005; Stam, Garnsey and Heffernan, 2006; Stam and Garnsey, 2007; Penrose, 2009; Gabrielsson and Gabrielsson, 2013; Miozzo and DiVito, 2016; Harbermann and Schuilte, 2017; Hesse and Sternberg, 2017). This general logic has been applied to employees' technology developing competence with their ability to develop new and maintain existing technology (Figure 4-25)¹⁴.

¹⁴ The same structure has also been applied to model the capabilities of marketing and selling, value delivering, and firm managing employees. Appendix A illustrates these additional structures to avoid repetition.



Figure 4-25: Technology developing capabilities

The model uses the structure proposed by Warren (2002) for employee skills and their impact on resources to formalise capabilities. The structure has been selected because it fits well to the theoretical underpinnings of growth paths theory, including capability improvement through learning and destruction through hiring. Warren proposes that the skill of the average employee in the firm can be represented as a qualitative variable on a scale from zero to one. The average employee's competence develops through *technology developing competence adjustment* for new employees (Equation 4-57).

Technology developing competence

Eq 4-57

- = initial technology developing competence
- +) (technology developing competence improvements
- technology developing competence adjustment)

The competence improvements slow down as the employees move towards the maximum possible competence of one. This has been implemented by Sterman's (2000) s-curve (Equation 4-58). As in previous implementations of the s-curve, the model calculates the improvement rates using a standard rate adjusted for two factors. The standard rate is the inverse of the technology developing *(TD) years of improvement* required to achieve full competence levels. While the

improvement speeds up initially as the competence levels improve, it slows down as employees move towards the maximum possible. The competence is also adjusted when the firm hires new employees. The magnitude of the reduction depends on the *fraction of new development employees* and *new TD employee competence* (Equation 4-59, 4-60). If the venture hires no new employees, no adjustment takes place. If the venture replaces all employees, the capability is adjusted to reflect the capability of new employees.

technology developing competence improvements	
= (1/TD years of improvement)	
* Technology developing competence * (1	
 Technology developing competence) 	
	4
technology developing competence adjustment	q 4-59
= (Technology developing competence	
 new TD employee competence) 	
fraction of new development employees	
fraction of new development employees Ec	q 4-60
= XIDZ(rate of hiring tech dev employees,	

Technology developing employees, 1)

The employees' abilities are calculated based on their competence. This requires scaling the competence from a dimensionless scale between zero and one to a quantitative, hard ability. Warren (2002) uses effect variables to scale the competence level to hard variables. This approach has also been used in the model. The *effect of competence on ability to develop technology* and the *effect of competence on ability to develop technology* and the *effect of competence on ability to maintain technology* convert the employees' average competence as a soft variable to their ability to develop and maintain the technology. Thereby, these effect sizes reflect the best possibly efficiency of the venture's business model design in the activity-based view on business models (Warren, 2000, 2002; Zott and Amit, 2010). They represent the best practice throughput of the venture's activities when fully developed capabilities are achieved. The hard abilities express, for example, the technology capitalised per employee per period. In addition, the model considers the adequacy of

complementary assets when calculating abilities (Equation 4-61, 4-62). This adequacy expresses the fraction of adequately equipped employees.

ability to develop technology

Eq 4-61

- = Technology developing competence
- * effect of competence on ability to develop technology
- * adequacy of complementary assets

ability to maintain technology

Eq 4-62

- = Technology developing competence
- * effect of competence on ability to maintain technology
- * adequacy of complementary assets

The initial competence and new employee competence is assessed based on the capability maturity model (see Paulk *et al.*, 1993) and descriptions of the venture's processes in annual reports. Some companies provide a time in years required to train new employees, which is used for the rate of improvement. Otherwise, the analyst sets the time required based on the training time of equally mature companies. The effects that transform the capability from soft to hard variables are set using Vensim's automatic model calibration.

4.3.4. Overview of subsystem inputs

This section has contextualised, formalised, and specified the subsystem through which digital ventures' technological resources develop. The outcome of this subsystem is the venture's stock of technological resources (Table 4-3). They can be compared to a case company's capitalised development expenses on its balance sheet. They develop through investments and amortisation, which are also used in other subsystems. The venture's technology developing activities and capabilities depend on its employees, complementary assets, and its existing technological quality. These inputs are determined by the venture's management and in the value creation subsystem.
Outputs	
Technological resources	
Inputs from other subsystems	Exogenous inputs
Firm managing subsystem:	Capability development:
 Technology developing 	Initial technology developing competence
employees, rate of hiring value	TD years of improvement
delivering employees	• New TD employee competence
 Adequacy of complementary 	Business model design:
assets	• Effect of competence on ability to develop
Value creation subsystem:	technology, effect of competence on ability to
 Technological quality 	maintain technology
	Accounting measures:
	Initial technological resources
	Lifetime of technological resources

Table 4-3: Outputs and inputs of the technology development subsystem

In addition, the technology development subsystem relies on a range of inputs set by the analyst based on accounting measures like the initial technological resources and their lifetime. Moreover, inputs regarding the development of capabilities and business model design are required. These include initial capability levels, improvement times, and effect sizes.

4.4. Customer and user subsystem

Customer and user bases are another critical resource for digital ventures. These customers and users may use the venture's software product or engage with another on its platform (Kollmann, 2006; Muzellec, Ronteau and Lambkin, 2015; Standing and Mattsson, 2016; Huang *et al.*, 2017). The outcomes of this subsystem are the indexed numbers of customers, users, and premium users (Figure 4-26). They are inputs to the value creation subsystem as the target of value creation and as a driver of network effects. The customer and user bases increase through the venture's marketing and selling activities and world-of-mouth marketing. They decrease depending on the customer lifetime, which depends on the venture's ability to deliver value. In addition, the development of the customer and user bases depends on variables related to customer and user behaviour and competition in the external environment. The abilities to acquire and deliver value to customers and users depends on respective employee capabilities.



Figure 4-26: Customer and user subsystem

In the following sections, the customer and user bases are formalised. Their development through marketing and selling activities, word-of-mouth marketing, the customer lifetime, and value delivering activities are illustrated.

4.4.1. Customer and user bases

As the performance outcomes sections describe, some digital ventures such as multi-sided platforms distinguish between users and customers. Other digital ventures may distinguish between users and premium users (Afuah and Tucci, 2000; Kollmann, 2006; Teece, 2010; Zott, Amit and Massa, 2011; Roma and Dominici, 2016; Voigt and Hinz, 2016; Möller *et al.*, 2020). The model presented in this thesis distinguishes between customers, users, and premium users to account for these different business model designs (Figure 4-27). These different bases accumulate through marketing, word-of-mouth, and churn. Additionally, users and premium users may convert into another category (Gary *et al.*, 2008; Wagner, Benlian and Hess, 2014; Holm and Günzel-Jensen, 2017; Gu, Kannan and Ma, 2018).



Figure 4-27: Customer and user bases

The venture's *user base, premium user base*, and *customer base* reflect the number of users of each type using the venture's product or service at a point in time. The sum of all bases is indexed at one at the beginning of the simulation to ensure the indexing of the rate of value creation. If the type is not present for a venture, the respective base has a value of zero. The relevant bases accumulate through their respective rates of acquisition and churn. Moreover, users and premium users are affected by conversion rates into the other category (Equation 4-63, 4-64, 4-65).

User base = initial user base Eq 4-63 + \int (rate of user acquisition + rate of conversion to users - rate of converion to premium – rate of user churn) Premium user base Eq 4-64 = initial premium user base + \int (rate of premium user acquisition + rate of conversion to premium - rate of conversion to users - rate of premium user churn) Customer base = initial user base

+ \int (rate of customer acquisition - rate of customer churn)

The *rate of conversion to premium* reflects the annualised number of users that switch to the upgraded version of the product. However, premium users may also drop the upgraded version. The annualised number of users that exchange the premium for the standard version of the product is reflected in the *rate of conversion to users*. Each rate decreases its originating stock while it increases its target stock. Both rates of conversion depend on the stock of users from which they convert and the *user to premium conversion rate* and *premium to user conversion rate*. These rates express, respectively, the fraction of users or premium who switch per period (Equation 4-66, 4-67).

rate of conversion to premium	Eq 4-66
= User base * user to premium conversion rate	

rate of conversion to users

Eq 4-67

= Premium user base * premium to user conversion rate

Users, premium users, and customers may stop using the venture's product or service. The *rate of user churn*, *rate of premium user churn*, and *rate of customer churn* express the annualised number at which the respective type stops using the product. These rates decrease their respective customer and user bases. They depend on the *customer lifetime, user lifetime, and premium user lifetime.* These lifetimes measure the average number of years a user stays with the venture (Oliva, Sterman and Giese, 2003; Hadiji *et al.*, 2014; Runge *et al.*, 2014). This loss of customers and users is formalised like the amortisation in the technology development subsystem. The model calculates the churn of users and customers by dividing the respective user or customer base by the respective lifetime (Equation 4-68, 4-69, 4-70).

rate of user churn = ZIDZ(User base, user lifetime)	Eq 4-68
rate of premium user churn	Eq 4-69
= ZIDZ(Premium user base, premium user lifetime)	

The *rate of user acquisition, rate of premium user acquisition,* and *rate of customer acquisition* are outcomes of two mechanisms. Firstly, the marketing and selling (MS) activities executed by the venture's employees (Oliva, Sterman and Giese, 2003; Huang *et al.*, 2017). Secondly, word-of-mouth (WoM) marketing, in which the venture's existing customers and users acquire further ones (Mikalef, Giannakos and Pateli, 2013; Huang *et al.*, 2017). The respective acquisition rates are the sum of the customers and users acquired through these two mechanisms (Equation 4-71, 4-72, 4-73).

rate of user acc	quisition	Eq 4-71
	= rate of user acquisition through MS	
	+ rate of user acquisition through WoM	
rate of premiur	m user acquisition	Eq 4-72
	= rate of premium user acquisition through MS	
	+ rate of premium user acquisition through WoM	
rate of custome	er acquisition	Eq 4-73
	= rate of customer acquisition through MS	
	+ rate of customer acquisition through WoM	

The subsections below discuss the acquisition of customers and users through marketing and selling, through word-of-mouth marketing, and their respective lifetime.

4.4.2. Marketing and selling

Like providers of physical product, digital ventures may rely on a broad range of marketing and sales techniques. Consumer businesses may be able to acquire many customers with a limited number of employees through digital marketing. Business-to-business ventures may rely on non-digital sales techniques like trade shows or direct marketing (Reuber and Fischer, 2011; Tyrväinen and Selin, 2011). Because digital ventures may distinguish between customers, users, and premium users (Rangaswamy *et al.*, 2020), marketing and sales are directed at acquiring these three groups. Therefore, the adapted and detailed marketing and

selling activities consider that the venture's employees may acquire these different groups. The model also considered managerial priorities to acquire the different groups (Figure 4-28).



Figure 4-28: Marketing and selling activities

As for technology developing activities, the model implements Sterman's (2000) resource-productivity-flow relationship to formalise marketing and selling. The *rate of customer acquisition through marketing and selling*, and the other rates, reflect the number of indexed customers the venture acquires per period. Each rate depends on the *marketing and selling employees*, the *focus to acquire* the respective type, and *ability to acquire* the type (Equation 4-74, 4-75, 4-76). The abilities reflect the rate at which the average indexed employee acquires the different customer types. The priorities reflect the fraction of time spent on acquiring each type.

rate of user acquisition through MS	Eq 4-74
= Marketing and selling employees	
* ability to acquire users * focus on acquiring users	
rate of premium user acquisition through MS	Eq 4-75
= Marketing and selling employees	
* ability to acquire premium users	
* focus on acquiring premium users	
rate of customer acquisition through MS	Eq 4-76
= Marketing and selling employees	
 ability to acquire customers 	
* focus on acquiring customers	

The abilities to acquire customers, users, and premium users depend on the competence of the venture's marketing and selling employees.¹⁵ The priorities are formalised in the firm managing subsystem.

4.4.3. Word-of-mouth marketing

Companies offering physical products may already benefit from word-of-mouth marketing. Their customers may recommend the product and thereby acquire further customers. Some digital ventures offering their product and services to consumers may benefit particularly from it. On the one hand, the reach and ease of accessing digital products makes the adoption for new customers quicker. On the other hand, network effects incentivise existing users and customers to refer and recommend digital products (Song, Parry and Kawakami, 2009; Reuber and Fischer, 2011; Mikalef, Giannakos and Pateli, 2013; Roma and Dominici, 2016; Kuester, Konya-Baumbach and Schuhmacher, 2018; Zaheer, 2020). This word-of-mouth marketing depends on the existing customer and user bases and the number of new users and customers that each existing one acquires (Figure 4-29).



Figure 4-29: Customer and user acquisition through word-of-mouth

The model calculates the *rate of customer acquisition through WoM* by multiplying the number of customers and users with the rates at which each existing customer and user acquires new customers (Equation 4-77). It uses the

¹⁵ These are formalised similarly to the technology developing competence. This specific capability is not formalised again but presented in Appendix A to avoid repetition.

same principles to derive users' and premium users' WoM acquisition rates (Equation 4-78, 4-79).

rate of customer	r acquisition through WoM	Eq 4-77
	= Customer base	
	* rate of customer acquisition per customer	
	+ (User base + Premium user base)	
	* rate of customer acquisition per user	
rate of user acqu	uisition through WoM	Eq 4-78
	= Customer base	
	* rate of user acquisition per customer + (User base	
	+ Premium user base)	
	* rate of user acquisition per user	
rate of premium	user acquisition through WoM	Eq 4-79
	= Customer base	
	 rate of premium user acquisition per customer 	
	+ (User base + Premium user base)	

* rate of premium user acquisition per user

The customer and user bases are calculated above. The analyst sets the rates of acquisition per customer and user based on information provided in annual reports. They reflect the number of indexed users or customers acquired by each existing indexed user or customer per period.

4.4.4. Customer lifetime

The customer lifetime reflects the number of years that the average customer maintains his relationship with the venture and uses its product¹⁶. As discussed in the value capture subsystem, some digital ventures lock customers into long-term contracts. The length of those contracts provides a minimum customer lifetime (Ruutu, Casey and Kotovirta, 2017). Above this minimum length, the customer lifetime depends on the competition and service adequacy of the venture (Figure

¹⁶ The supplementary material regarding this subsystem in Appendix A uses the same structure to derive the user lifetime and premium user lifetime.

4-30). For example, more customers will drop the venture's product when competitors or providers of substitutes improve their products. Additionally, the venture needs to ensure that it has sufficient capacity to service customers and users adequately. For example, if a SaaS product is not available or customer service is insufficient, more customers will drop the product (Warren, 2002; Oliva, Sterman and Giese, 2003; Currie, Joyce and Winch, 2007; Tyrväinen and Selin, 2011; Zhao, Song and Storm, 2013; Voigt and Hinz, 2016).



Figure 4-30: Customer lifetime

The model uses Sterman's (2000) structure for multiplicative effects to formalise the customer lifetime by adjusting a *standard customer lifetime* for two effects. The model adjusts it for the *customer capacity adequacy* and the *relative strength to competition for customers*. In addition, it sets the *customer contract period* as a minimum lifetime (Equation 4-80).

customer lifetime

Eq 4-80

- = MAX(customer contract period, standard customer lifetime
- * customer capacity adequacy
- * relative strength to competition for customers)

The customer capacity adequacy reflects the fraction of its customers that the venture can adequately service. It is calculated by dividing the *capacity to service customers* by the *customer base*. In cases where the capacity exceeds the customer base, the adequacy has been capped at one (Equation 4-81). The relative strength to competition for customers expresses the strength of the venture's product, including switching costs, compared to competitors and substitutes

(Equation 4-82). The model formalises this relationship using the equations employed for the bargaining power in the value capture subsystem. It ensures that the equation works for extreme values. For example, the customer lifetime remains at its standard value if there is no competition and reduces to the contract period as competition increases.

customer capacity adequacy = MIN (1, ZIDZ(capacity to service customers, Eq 4-81 Customer base))

relative strength to competition for customers Eq 4-82

= ZIDZ(strength of product for customers

* (1 + customer switching costs), strength of product for customers

* (1 + customer switching costs)

+ value provided by competitors and substitutes)

The strength of the product for customers is calculated in the value creation subsystem. The switching costs and value provided by competitors and substitutes are calculated in the value capture subsystems. The customer base is discussed above, while the ability to service customers is discussed below as part of the venture's value delivering activities. The Vensim software sets the standard customer lifetime using automatic model calibration.

4.4.5. Value delivering activities

Because the fully digital products considered in this thesis are non-physical, digital ventures do not require inventories or manufacturing capacities considered part of the value delivering activities for physical products (Afuah and Tucci, 2000; Hull *et al.*, 2007; Hafezieh, Akhavan and Eshraghian, 2011; Bradley *et al.*, 2012). Instead, digital ventures' capacity constraint depends on their ability to host their product and service their customers and users appropriately (Afuah and Tucci, 2000; Oliva, Sterman and Giese, 2003; Liao *et al.*, 2015; Kohtamäki *et al.*, 2019). The capacities to service customers and users depend on the venture's value delivering employees, their abilities to service customers and users, and their focus on servicing each type (Figure 4-31).



Figure 4-31: Value delivering activities

The *capacity to service customers* reflects the number of indexed customers the venture can adequately service. The model calculates it by multiplying the venture's *value delivering employees*, their *ability to service customers*, and their *focus on servicing customers* (Equation 4-83). While the ability reflects how many customers each employee can service, the focus reflects the fraction of time spent servicing customers. The model applies the same principles to the *capacity to service users*, which reflects the indexed number of users and premium users the venture can service adequately (Equation 4-84).

capacity to service customers

Eq 4-83

- = Value delivering employees
- * ability to service customers
- * focus on servicing customers

capacity to service users

Eq 4-84

= Value delivering employees * ability to service users* focus on servicing users

The model determines the venture's value delivering employees and their focus in the firm managing subsystem. The abilities to service users and customers depend on the employee competences. It is formalised like the technology development competence in Appendix A.

4.4.6. Overview of subsystem inputs

This subsystem has formalised the processes through which the venture's customer base, user base, and premium user base develop (Table 4-4). The total of all customer and users is indexed at one. These bases then develop through the venture's marketing and selling activities, word-of-mouth marketing, and

customer churn. The model requires a range of inputs from other subsystems to calculate these flows. These include the employees engaged in the firm's marketing and value delivering activities, their hiring rates, and priorities. Additionally, variables from the value creation and capture subsystem are required.

Outputs		
Customer base, user base, premium user base		
Inputs from other subsystems	Exogenous inputs	
Firm managing subsystem:	Environment – Customer and user behaviour:	
 Marketing and selling employees, Value delivering employees Rate of hiring marketing and selling employees, rate of hiring value delivering employees Focus on acquiring customers, focus on acquiring users, focus on acquiring premium users Focus on servicing users, priority on servicing customers 	 Initial customer base, initial user base, initial premium user base User to premium conversion rate, premium to user conversion rate Rate of user acquisition per user, rate of user acquisition per customer, rate of premium user acquisition per user, rate of premium user acquisition per customer, rate of customer acquisition per user, rate of customer acquisition per user, rate of customer acquisition per customer Standard customer lifetime, standard user 	
Adequacy of complementary	lifetime Canability developments	
	Capability development:	
 Strength of product for customers, strength of product for users Premium product advantage Value capture subsystem: 	 Initial marketing and setting competence, initial value delivering competence MS years of improvement, VD years of improvement New MS employee competence, new VD employee competence 	
Value provided by competitors	Business model design:	
 and substitutes Customer contract period, premium user contract period Customer switching costs, user switching costs 	 Effect of competence on ability to acquire customers, effect of competence on ability to acquire users, effect of competence on ability to acquire premium users Effect of competence on ability to service customers, effect of ability on ability to service service users 	

Table 4-4: Outputs and inputs of the customer and user subsystem

The customer and user subsystem also depends on a range of input set by the analyst based on annual reports and variables set through automatic model calibration. The analyst needs to extract initial values, conversion rate, and wordof-mouth acquisition rates from annual reports. Standard lifetimes are then calibrated using the Vensim software. The analyst needs to set the variables related to capability development. These are the same types of variables already presented for the technology development capabilities. Lastly, the Vensim software is used to calibrate effect sizes that reflect the maximum throughput of a business model's activities.

4.5. Firm managing subsystem

The conceptualisation chapter illustrates that a digital venture's management is responsible for managing and expanding the firm's human resources, complementary assets, and financial resources. These outcomes are the inputs to the other subsystems described above. The three sections below illustrate these three managerial areas. They derive these outcomes by formalising how the management forms targets and implements them depending on their capabilities (Figure 4-32). The targets themselves depend on the capabilities in the other subsystems and the management's dominant logic.



Figure 4-32: Firm managing subsystem

The first subsection covers the venture's human resource management. The subsequent subsections then cover its complementary assets and financial resources management.

4.5.1. Management of human resources

The venture may need to hire additional employees to achieve its growth and technology improvement goals. The employees across the venture's four

activities are the first outcome of the firm managing subsystem. This section describes the formalisation of human resources using the example of marketing and selling employees¹⁷. It covers the stock of marketing and selling employees, influences on hiring employees, and the management's priorities for marketing and selling activities.

4.5.1.1. Marketing and selling employees

The model employs a structure that System Dynamics modellers often use to model employees (see Sterman, 2000; Morecroft, 2015). The structure adjusts the initial number of employees through outflows of employees leaving due to lay-offs and turnover and an inflow of new employees through hiring (Figure 4-33).



Figure 4-33: Marketing and selling employees

The model indexed the number of *marketing and selling employees* at one at the beginning of the simulation. This number then accumulates through the rates of hiring, lay-offs, and employees leaving (Equation 4-85). The model uses the same equations to determine these flows as Sterman (2000) and Morecroft (2015) use. The number of employees leaving the organisation per period depends on the number of employees and the *employee turnover rate* (Equation 4-86). This rate reflects the fraction of employees that leave the company every year. The hiring and layoffs rates adjust the number of employees to the *target marketing and*

¹⁷ Appendix A covers the other three types of human resource stocks to avoid repetition. These additional stocks regard the venture's value delivering, technology developing, and firm managing employees.

selling employees. The number of employees that need to be hired or laid off is expressed by the *target change in marketing and selling employees*. The model calculates it as the difference between the management's target and the current number of employees (Equation 4-87). If this difference is positive, the venture hires new employees over a *hiring delay* until the number of employees equals the target and the target change is zero. The previous chapter has also argued that the venture may delay its hiring until its fundraising efforts have secured sufficient financial resources for a planned expansion (Subsection 3.3.2). The *liquidity adequacy* adjusts the *rate of hiring marketing and selling employees* depending on the secured funding. Moreover, the venture needs to replace employees is negative, the venture lays employees off through the *rate of marketing and selling employees* lay offs over the delay to lay off (Equation 4-89).

Marketing and selling employees	Eq 4-85
= initial marketing and selling employees	
+ \int (rate of hiring marketing and selling employees	
 rate of marketing and selling employees lay offs 	
 rate of marketing and selling employees leaving) 	
rate of marketing and selling employees leaving	Eq 4-86
= Marketing and selling employees	
* employee turnover rate	
target change in marketing and selling employees	Eq 4-87
= target marketing and selling employees	
 Marketing and selling employees 	
rate of hiring marketing and selling employees	Eq 4-88
= MAX(0, target change in marketing and selling employees/hiring delay)	
* liquidity adequacy + rate of markeing and selling employees leaving	
rate of marketing and selling employees lay offs	Eq 4-89
= MAX(0, -target change in marketing and selling employees	
/delay to lay off)	

The delay to lay off is set to 0.1 for all companies. The turnover rate is set based on companies' annual reports and publicly available information. The liquidity adequacy is discussed as part of the financial management below. The two subsequent subsections formalise the target employees and hiring delay that drive the firm's increase in human resources.

4.5.1.2. Target marketing and selling employees

The target marketing and selling employees capture the number of employees the management wants to employ to achieve its growth goals for customers, users, and premium users. Previous System Dynamics models have derived such targets based on the management's goals and the expected productivity of each employee (Sterman, 2000; Morecroft, 2015). The model implements these principles to determine all employee targets. Its target number of marketing and selling employees thus depends on the customer, user, and premium user growth goal and employees' abilities to acquire them (Figure 4-34). Moreover, as argued during model conceptualisation, some digital ventures may employ dominant logics that cap their employee numbers and are unwilling to hire more employees (Subsection 3.3.2).



Figure 4-34: Target marketing and selling employees

The *target marketing and selling employees* are calculated by adjusting the *employees required to acquire customers and users* for the company's *MS employee cap* (Equation 4-90). If the company has not set a cap employee numbers (MS employee cap =-1), the company plans to hire all employees

required to achieve its goals. Otherwise, it hires the smaller number of required employees or the cap (Davidsson, 1989; Kirkwood, 2009; Hesse and Sternberg, 2017). The *employees required to acquire customers and users* expresses the number of employees required to achieve the company's customer and user growth goals. The model calculates this number through the sum of employees required in order to achieve each of the marketing and selling goals being pursued (Equation 4-91).

target marketing and selling employees

Eq 4-90

= IF THEN ELSE(MS employee cap = -1, employees required to acquire customers and users, MIN(MS employee cap, employees required to acquire customers and users))

employees required to acquire customers and users

Eq 4-91

- = employees required to acquire customers
- + employees required to acquire users
- + employees required to acquire premium users

The model bases its employee requirements on the principles of Sterman (2000) and Morecroft (2015). It calculates the number of employees required to achieve the venture's customer growth goal based on the goal's magnitude and the expected employee productivity¹⁸. The *target customer acquisition rate* and the *reported ability to acquire customers* reflect this goal and productivity (Equation 4-92). The reported ability to acquire customers captures the employees' *ability to acquire customers* but acknowledges that the managers do not know productivity levels accurately. Rather, its knowledge is affected by the *reporting delay*. As suggested by Sterman and Morecroft, the reported rate is derived using Vensim's smooth function (Equation 4-93). The rate at which the venture wants to acquire customers depends on the venture's current *customer base* and its annual *target customer growth*. However, its employees do not need to acquire customers generated through word-of-mouth by other users and customers (Equation 4-94). The venture must make up for customer loss through churn. The model

¹⁸ Appendix A illustrated these equations for users and premium users to avoid repetition.

incorporates this expected churn through the *reported customer lifetime* that acknowledges the reporting delay (Equation 4-95).

employees requ	ired to acquire customers	Eq 4-92
	= ZIDZ(target customer acquisition rate ,	
	reported ability to acquire customers)	
. 1 1.1.		E 4.02
reported ability	to acquire customers	Eq 4-93
	= SMOOTH(ability to acquire customers, reporting delay)	
target customer	r acquisition rate	Eq 4-94
	= Customer base * (target customer growth	
	+ ZIDZ(1, reported customer lifetime)	
	 rate of customer acquisition per customer 	
	 rate of customer acquisition per user) 	

reported customer lifetime = SMOOTH(customer lifetime, reporting delay) Eq 4-95

For all companies, the reporting delay has been set at 0.25 to reflect common quarterly reporting intervals. The MS employees cap and target customer growth are set by reviewing the strategic and operational statements in the venture's annual reports. They express the venture's dominant logic during a growth state. The model uses the same structure and equations to determine the employees required to acquire the target number of users and premium users (see Appendix A).

4.5.1.3. Hiring delay

The hiring delay expresses the time it takes the venture to increase its employee number to its target. Like other activities in the previous subsystems, this outcome of the firm management activities depends on employee inputs and their productivity (Sterman, 2000). Here, the inputs are the available management time expressed by the management employees and their managerial slack. Their productivity is expressed through their ability to hire (Figure 4-35).



Figure 4-35: Managerial slack and hiring delay

The more *firm management employees* are employed by the venture, the higher the *managerial slack*, and the higher their *ability to hire*, the lower the hiring delay (Equation 4-96)¹⁹. The *managerial slack* expresses the time fraction available to the management to determine and execute growth plans, including hiring. It is the fraction of *managerial capacity* not occupied with managing the venture's operational employees (Equation 4-97). This managerial capacity depends on the available firm managing employees and their *ability to manage the firm* (Equation 4-98).

hiring delay = XIDZ(1, Firm managing employees * managerial slack Eq 4-96 * ability to hire, 100)

managerial slack

- = MAX(0, (managerial capacity
- (Marketing and selling employees
- + Value delivering employees
- + Technology developing employees))
- /managerial capacity)

managerial capacity

Eq 4-98

Eq 4-97

= Firm managing employees * ability to manage the firm

¹⁹ The XIDZ function prevents division by zero when the product of firm managing employee, managerial slack, and the ability to hire is zero. In such cases, no hiring should take place. The function returns a hiring delay of 100 (years). In light of the timeframe of a few years for which the model is used, this leads to a neglectable amount of hiring.

The different employee stocks follow the same structure explained here for marketing and selling employees. The ability to hire and the ability to manage the firm depend on the management employees' capabilities²⁰.

4.5.1.4. Priorities of marketing and selling employees

Digital ventures' marketing and selling employees face three potential goals. Therefore, the available employees must prioritise whether to spend time on marketing to customers, users, or premium users (Figure 4-36).



Figure 4-36: Priorities of marketing and selling employees

These three priorities are calculated by expressing the number of employees required to achieve a goal as a fraction of employees required to achieve all goals (Equation 4-99, 4-100, 4-101).

focus on acquiring customers	Eq 4-99
= employees required to acquire customers	
/employees required to acquire customers and users	
focus on acquiring users	Eq 4-100
= employees required to acquire users	
/employees required to acquire customers and users	
focus on acquiring premium users	Eq 4-101
= employees required to acquire premium users	
/employees required to acquire customers and users	

The marketing and selling activities in the customer and user subsystem use these priorities. When target and actual employees match and the management's

²⁰ These capabilities are formalised using the same structure illustrated in the technology development subsystems. They are not repeated here but in Appendix A to avoid repetition.

estimates about productivity levels are correct, these equations ensure that all goals are achieved. Otherwise, each goal is achieved at the same proportion.

4.5.2. Management of complementary assets

Capabilities rely on the provision of adequate equipment, reflected in the firm's complementary assets (Garnsey, 1998; Stam, Garnsey and Heffernan, 2006). The following section describes the development of the firm's complementary assets, their adequacy, and the formation of investment targets into them.

4.5.2.1. Complementary assets and their adequacy

The venture's complementary assets reflect its property and equipment on its balance sheet. For digital ventures, complementary assets may include, for example, servers to host the firm's website, and the equipment and software used by its employees (Rosemann, Andersson and Lind, 2011). Their adequacy depends on the required complementary assets and the assets held by the venture. The management can increase complementary assets by acquiring them. They then depreciate over time (Figure 4-37).



Figure 4-37: Complementary assets and their adequacy

The *adequacy of complementary assets* expresses the venture's *complementary assets* as a fraction of the *required complementary assets* capped at a value of one (Equation 4-102). The required complementary assets depend on the *total number of employees* and the *complementary assets required per employee* (Equation 4-103).

adequacy of complementary assets

= MIN(1, Complementary assets

/required complementary assets)

required complementary assets

= total number of employees

* complementary assets required per employee

The venture's complementary assets accumulate from their initial level through the *rate of investment in complementary assets* and the *rate of depreciation of complementary assets* (Equation 4-104). The depreciation rate is calculated through the same equations used for the amortisation of technological resources in the technology development subsystem. The model divides the complementary asset stock by the *lifetime of complementary assets* (Equation 4-105). The *rate of investment in complementary assets* is calculated like the rate of hiring human resources. The investment rate achieves a *target investment in complementary assets* over a *delay to acquire complementary assets*. It also accounts for the *liquidity adequacy*, which prevents investments before fundraising has been completed (Equation 4-106).

Complementary assetsEq 4-104= initial complementary assets+
$$\int$$
 (rate of investment in complementary assets- rate of depreciation of complementary assets- rate of depreciation of complementary assetsrate of depreciation of complementary assetsEq 4-105= Complementary assets/lifetime of complementary assetsrate of investment in complementary assetsEq 4-106= (target investment in complementary assets/delay to acquire complementary assets)* liquidity adequacy* liquidity adequacy

The lifetime of complementary assets in years, the initial complementary assets, and the complementary assets per employee are exogenous inputs derived from the firm's financial statement. While the acquisition delay uses the same

Eq 4-102

Eq 4-103

4 104

r

equations as the hiring delay (see Appendix A), the target investment is formalised below.

4.5.2.2. Target investment in complementary assets

When expanding the firm by hiring new employees, the venture must also acquire the complementary assets required to equip its increasing number of employees. Thus, it acquires the difference between the available complementary assets and those required for its target number of employees (Figure 4-38).



Figure 4-38: Target complementary assets

The required complementary assets for the firm, including its expansion, depend on the *target operating employees* and *target firm managing employees* as well as the *complementary assets required per employee*. The difference between the required and held *complementary assets* is acquired by the venture (Equation 4-107). This targeted investment cannot fall below zero if the venture holds excess complementary assets.

target investment in complementary assets	Eq 4-107
= MAX(0, (target operating employees	
+ target firm managing employees)	
* complementary assets required per employee	

Complementary assets)

The complementary assets held and required per employee have been introduced above. The target employee numbers are explained above for marketing and selling employees and Appendix A for all other employee types.

4.5.3. Management of financial resources

Besides managing the firm's human resources and complementary assets, the venture also needs to manage its financial resources and liquidity.

4.5.3.1. Financial resources

As for other ventures, financial resources refer to a venture's cash position on its balance sheet (Mudambi and Treichel, 2005; Kollmann, 2006). They increase through generating resources and raising capital (Figure 4-39). They deplete through investments into technological resources, complementary assets, and content assets (Hugo and Garnsey, 2001).



Figure 4-39: Financial resources

The venture's financial resources accumulate through the *rate of financial resources generation*, *rate of capital raising*, and *rate of investments* (Equation 4-108).

Financial resourcesEq 4-108= initial financial resources
$$+ \int$$
 (rate of financial resource generation+ rate of capital raising – rate of investments)

The financial resources generated reflect those resources received and maintained by the venture in the form of cash. The *rate of resource generation* measures the absolute rate of change in balance sheet resources (Teece, 1986; Garnsey, 1998; Hugo and Garnsey, 2001, 2002; Garnsey, Stam and Heffernan, 2006; Zott and Amit, 2007, 2010; Garnsey, Lubik and Heffernan, 2015). The model calculates it by multiplying together the *rate of value* creation and the *share of value captured* (Equation 4-109). However, the share of value captured includes non-cash expenses such as depreciation and amortisation (Nelson and Winter, 1978; Dierickx and Cool, 1989; Porter, 1991; Knott, Bryce and Posen, 2003). These are added back to determine the *rate of financial resource generation* (Equation 4-110).

rate of resource generation	Eq 4-109
-----------------------------	----------

= rate of value creation * share of value captured

rate of financial resource generation

Eq 4-110

- = rate of resource generation
- + rate of amortisation of technological resources
- + rate of depreciation of complementary assets
- + rate of amortisation of content assets

Growth process theories recognise that digital ventures need to access resources from the external environment such as investors and lenders (Garnsey, 1998; Santos and Eisenhardt, 2009; Gabrielsson and Gabrielsson, 2013; Garnsey, Lubik and Heffernan, 2015; Miozzo and DiVito, 2016). Ventures mobilise these resources by raising capital as equity and debt (Penrose, 1960; Garnsey, 1998; Heirman and Clarysse, 2004; Garnsey and Heffernan, 2005; Clarysse, Bruneel and Wright, 2011; Hesse and Sternberg, 2017). The *rate of equity raising* and *rate of debt raising* increase the *rate of capital raising*. The *rate of equity repurchasing* and the *rate of debt repayments* reduce the venture's capital raising (Equation 4-111). All variables measure the rate at which capital is raised and repaid in monetary amounts per period. The next subsubsection further specifies these.

rate of capital raising

Eq 4-111

= rate of equity raising - rate of equity repurchasing+ rate of debt raising - rate of debt repayments

Over time, the venture invests its financial resources into technological resources, complementary assets, and content assets. The *rate of investments* is the sum of these individual investment rates (Equation 4-112).

rate of investments

Eq 4-112

- = rate of investment in technological resources
- + rate of investment in complementary assets
- + rate of investment in content assets

The technology development, firm management, and content creation subsystems specify these individual investment rates.

4.5.3.2. Equity and debt employed

The venture's balance sheet assets like its technology, complementary assets, and financial resources require funding and financing. The counterpart of the balance sheet assets is the debt and equity employed (Figure 4-40). Both types of capital are recognised in monetary amounts on the firm's balance sheet as liabilities and equity. They accumulate by raising new equity and debt. Repayments of debt and repurchases of equity lower the respective capital employed. These activities also affect the company's financial resources illustrated above (Garnsey, 1998; Garnsey, Dee and Ford, 2006; Stam and Garnsey, 2007; Gabrielsson and Gabrielsson, 2013; Garnsey, Lubik and Heffernan, 2015; Ingley, Khlif and Karoui, 2017). The capital employed affects the venture's ability to capture value because digital ventures may compensate their capital providers.



Figure 4-40: Equity and debt employed

The equity and debt employed by a digital venture result from the accumulation of raising and repaying each type of capital. The equity employed accumulates through the *rate of equity raising* and the *rate of equity repurchasing* (Equation 4-113). The debt employed accumulates through the *rate of debt raising* and the *rate of debt raising* and the *rate of debt repayments* (Equation 4-114).

Equity emplopyed

= initial equity employed
+
$$\int$$
 (rate of equity raising – rate of equity repurchasing)

Ea 4-113

Debt employed = initial debt employed Eq 4-114 + \int (rate of debt raising - rate of debt repayments)

The *rate of equity raising* and *rate of debt raising* have been calculated in the same manner as the rates of hiring and investments into complementary assets. The model calculates them by achieving the *target equity raising* and the *target debt raising* over the *delay to raise capital* (Equation 4-115, 4-116). The *rate of debt repayment* and *rate of equity repurchase* have been calculated using the respective stock of equity and debt and the *fraction of debt repaid* and *fraction of equity repurchased* per year (Equation 4-117, 4-118).

rate of equity raising = target equity raising/delay to raise capital	Eq 4-115
rate of debt raising = target debt raising/delay to raise capital	Eq 4-116
rate of equity repurchasing	Eq 4-117
= Equity employed * fraction of equity repurchased	

rate of debt repayments = Debt employed * fraction of debt repaid Eq 4-118

Most digital ventures do not repurchase equity, and debt repayment depends on the agreements with their lenders (Mudambi and Treichel, 2005). Therefore, the fractions of repurchasing and repayments are inputs based on the firm's financial statements. The rates of debt and equity raising depend on the management's target and delay to raise capital. The target is discussed below. The delay is formalised in the same manner as the hiring and investment delay (see Appendix A).

4.5.3.3. Liquidity adequacy and target equity and debt raising

Digital ventures may need to raise capital to finance their customer growth, user growth, technology improvement, and content creation plans. Companies use this capital to finance costs incurred before the expansions and improvements begin to pay off. The raising of this capital thereby has two crucial implications. Firstly, digital ventures typically raise capital before executing their expansion plans. Therefore, fundraising needs may delay hiring and investments into complementary assets (Garnsey, 1998; Hugo and Garnsey, 2005; Garnsey, Dee and Ford, 2006; Brown and Mawson, 2013; Miozzo and DiVito, 2016; Ingley, Khlif and Karoui, 2017). The model considers this delay through the liquidity adequacy. It reduces hiring and investment rates depending on the available financial resources and liquidity requirements. Secondly, depending on the venture's financial resources, a liquidity shortfall is created. The venture fills the shortfall by raising more equity and debt (Figure 4-41).



Figure 4-41: Liquidity adequacy, shortfall, and capital targets

The *liquidity adequacy* expresses the venture's *financial resources* as a fraction of the *liquidity requirements*. The model caps the adequacy at one (Equation 4-119). The venture's liquidity requirements have been split into the areas of expansion and improvement that might require financing. These are requirements for customer acquisition, user and premium user acquisition, technology improvements, and content creation (Equation 4-120). For example, the liquidity requirements for customer acquisition depend on the venture's *target customer acquisition rate*, the *reporting period* it raises funds for, and the *financial*

resources required per customer (Equation 4-121)²¹. The acquisition rate expresses the rate at which the venture's employees need to acquire customers to achieve its growth goals. The human resource management section has covered it in detail. The reporting period has been set as one and the financial resources per customer acquisition are an input set through automatic model calibration.

liquidity adequacy	Eq 4-119
= MIN(1, XIDZ(Financial resources, liquidity requirements, 1))	

liquidity requirements

Eq 4-120

- = liquidity requiements for customer acquisition
- + liquidity requirements for technology improvements
- + liquidity requirements for user acquisition
- + liquidity requirements for content creation

liquidity requirements for customer acquisition

Eq 4-121

- = target customer acquisition rate
- * financial resources required per customer
- * reporting period

The venture needs to raise its *liquidity shortfall*. It is the difference between the *liquidity requirements* and *financial resources* unless resources already exceed the requirements (Equation 4-122). It can then decide if this capital will be raised in the form of equity or debt. The *equity debt preference* is used to model this choice. It reflects the fraction of capital the venture aims to raise in equity. Thus, its *target equity raising* is determined by the liquidity shortfall and its equity preference (Equation 4-123). The difference between one and the equity debt preference is the fraction of capital to be raised in the form of debt. Therefore, multiplying this difference with the liquidity shortfall generates the *target debt raising* (Equation 4-124).

liquidity shortfall = MAX(0, liquidity requirements – Financial resources)	
target equity raising = liquidity shortfall * equity debt preference	Eq 4-123

²¹ The liquidity requirements for users, technology improvements, and content are shown in Appendix A using the same structure and principles to avoid repetition in this chapter.

The preference between equity and debt is a model input calculated by evaluating the fractions of capital raised by the venture each year.

4.5.4. Overview of subsystem

The outcomes of this subsystem are the venture's employees across its activities, its financial resources and types of capital employed, and its complementary assets (Table 4-5). This section has illustrated how these outcomes develop from their initial value based on the management's dominant logic and capabilities. Inputs from other subsystems are required to set goals in absolute values and determine the resources that need to be acquired to achieve them. For example, the targets for marketing and selling employees depend on a wide range of customer and user subsystem inputs. These determine acquisition targets and employees' ability to acquire customers and users. Similarly, the venture's targets for value delivering employees depend on inputs from the customer and user subsystem and the content creation subsystem (see Appendix A). The venture's target for technology developing employees depends on inputs from the technology developing employees depends on inputs from the accumulation of financial resources.

Outputs						
Marketing and selling employees, value delivering employees, technology developing						
employees, firm managing employees						
Complementary assets						
• Financial resources, equity employed, debt employed						
Inputs from other subsystems	Exogenous inputs					
Value creation subsystem:	Dominant logic – Targets and limits:					
Rate of value creation	• Target customer growth, target user					
Technological quality, maximum	growth, target premium user growth,					
technological quality, effect of	target technology improvement, target					
investments technology improvements	content library growth					
Value capture subsystem:	• MS employee cap, VD employee cap,					
• Share of value captured	TD employee cap, FM employee cap					
Customer and user subsystem:	Dominant logic – Human resource					
• Customer base, user base, premium user	management:					
base	• Initial marketing and selling employees,					
• Customer lifetime, user lifetime,	initial value delivering employees,					
premium user lifetime	initial technology development					
• user to premium conversion rate.	employees, initial firm managing					
premium to user conversion rate	employees					
• Ability to acquire customers, ability to	Employee turnover rate					
acquire users, ability to acquire	Dominant logic – Asset requirements:					
premium users	• Complementary assets required per					
• Rate of user acquisition per user, rate of	employee					
user acquisition per customer, rate of	• Financial resources required per					
premium user acquisition per user, rate	customer, financial resources required					
of premium user acquisition per	per user, financial resources required					
customer, rate of customer acquisition	per technology improvement, financial					
per user, rate of customer acquisition	resources required per content item					
per customer	• Equity debt preference					
• Ability to service users, ability to	• Fraction of equity repurchased, fraction					
service premium users, ability to create	of debt repaid					
content	Capability development:					
Content creation subsystem:	• Initial firm management competence					
• Rate of investment in content assets,	• FM years of improvement					
rate of amortisation of content assets	• New FM employee competence					
• Ability to create content, rate of	Business model design:					
external content creation	• Effect of competence on ability to					
Technology development subsystem:	manage employees					
Rate of investment in technological	Accounting measures:					
resources, rate of amortisation of	• Initial financial resources, initial					
technological resources	complementary assets, initial equity					
• Ability to develop technology ability to	employed, initial debt employed					
maintain technology	• Lifetime of complementary assets					

Table 4-5: Outputs and inputs of the firm managing subsystem

External inputs to the firm managing subsystem relate to managerial targets and limits that express the venture's dominant logic during a growth state. The subsystem also requires inputs for initial employee and capital stocks, turnover rates, and complementary asset requirements. The analyst sets these based on information contained in annual reports and financial statements. The analyst also needs to set capability development variables using the same procedures as for other capabilities (see Appendices A and C). The effect size of the capability and funding requirements are set through automatic model calibration. The lifetime of complementary assets is directly imported from financial statements.

Chapter 5: Case simulation and model validation

The previous two chapters conceptualised and formalised a System Dynamics model that reveals the processes affecting digital ventures' ability to create value and capture a share of that value. The two chapters have integrated growth paths theory and dynamic states theory, contextualised the theories for digital ventures, and proposed a formal simulation model. This simulation model requires validation to establish the level of confidence in the model's approximations of company developments. Thereby, this chapter also establishes confidence in the ability of the proposed mechanisms to account for the dynamic changes in performance outcomes. This chapter simulates the model for the four case companies identified through the rigorous and extensive sampling process described in the methodology chapter (Subsubsection 2.4.1.1). The methodology also describes the process of setting up the model for each of the four case companies (Subsubsection 2.4.1.2). The complete model is run for each company. However, the presentation of the model's inputs and outputs is split based on the subsystems presented in the model formalisation chapter (Chapter 4). The presentation works backwards from the firm management over the technology developing and user/customer subsystems to the performance outcome subsystems. This structure allows for illustrating how the outputs from the former parts of the model serve as inputs to the latter parts. Each figure illustrates the subsystem's outcomes for one case company. To establish confidence in the model, figures include the outcomes calculated by the model (blue lines) and the company's actual historical data (red lines).

5.1. Firm managing subsystem

This section assesses the model's ability to approximate the outcomes of the firm managing subsystem. The subsystem depends on exogenous inputs regarding the management's dominant logic, capability development, business model design, and accounting measures (Table 4-5). It uses these inputs and those from other subsystems to determine the venture's human resources and capital assets. These variables are inputs to the activities of ventures to market to and service

customers and users and develop its technology. The variables are also cost drivers in the value capture subsystem.

5.1.1. Alpha

Alpha is a UK-based company that provides communication software integrating into customer relationship management and human resource management platforms. The software is a cloud-based solution using a SaaS business model. Alpha's customers use the product to improve the productivity of their marketing, sales, customer service, and human resource departments. Data for Alpha is available for six years, during which time the company develops through three different growth states. For the first two years and a quarter (t=2.25), the venture aims to grow its customer base by 161% per year (Table 5-1). Afterwards, the venture changes its dominant logic. It reduces its customer growth goal to 20% and holds its marketing and selling employees constant at 7.25 indexed employees. All other employee numbers remain uncapped. During those first two growth states, Alpha's development employees integrate its software into additional platforms. While this increases Alpha's addressable market, it does not improve its product quality for existing customers. This changes when Alpha enters its third growth state after three and a half years of simulation (t=3.5). In this final growth state, the venture also targets technology improvements of 20% per year, reflecting major improvements.

The firm managing subsystem also requires inputs regarding the management's capability development and capital assets. The venture has an established management team and hires additional experienced managers. Therefore, the initial competence levels of the firm and new employees are set respectively at 0.5 and 0.3 on the capability maturity scale (see Appendix C). Employees take about two years to develop full proficiency. This training time has been set after reviewing the other case companies. A longer period has been set to reflect that the venture is less mature and has less developed training programmes than the other companies. The Vensim software estimated the impact of employee competence on hard variables. The software estimates that each fully equipped and capable indexed firm managing employee can manage about 26 indexed

160

operational employees. The venture needs to equip its employee with sufficient complementary assets and maintain sufficient liquidity. The venture requires about 0.7% of initial total assets per indexed employee in complementary assets. The software estimates that the venture raises about 24% of initial total assets for each indexed customer it plans to acquire. It also raises 67% of initial total assets for each indexed technology improvement it plans to achieve. Moreover, an employee turnover rate of 30% has been set to reflect high employee turnover in Alpha. Alpha does not employ a content-based business model or distinguish between users and customers. Therefore, targets regarding users and content (see Appendix D) are not relevant for the venture and are therefore set to zero.

Dominant logic – targets and HR		tiı	me: 0 - 2.25	t: 2.25 - 3.5	time:	3.5 - 6
Target customer growth			161%	20%		
Target user growth		0%				
Target technology improvement		0% 20%)%	
Marketing and selling (MS) employee ca	р	-1 7.25				
Value delivering (VD) employee cap		-1				
Tech. developing (TD) employee cap		-1				
Firm managing (FM) employee cap		-1				
Capabilities - firm managing Capital asset requirements						
initial firm managing competence	0.	.5	complementa	ary assets per emp).	0.7%
firm managing years of improvement		2 liquidity goal per customer		24%		
new FM employee competence	0.	.3	liquidity goal per user			0%
Employee turnover rate	30%	%	liquidity goa	l per tech improve	ement	67%
effect of competence on ability to manage the firm			26			

Table 5-1: Firm management inputs for Alpha²²

With a dominant logic focussed on customer growth in its first growth state, theory suggests that Alpha will hire more marketing and selling employees to achieve its growth targets. It will also hire value delivering employees to service the acquired customers. The venture's marketing and selling employees should remain at their capped value in the second and third growth state. However, these marketing and selling employees will still acquire additional customers that the venture needs to service. Therefore, value delivering employees may continue to increase. Moreover, the venture's technology developing employees should only

²² All simulations have a starting time of zero rather than a specific year in order to obscure company identities. If input values have been altered with growth states, these states are presented with this relative time point in the tables. For example, the column titled "time: 0 - 2.25" includes values set for the company between the start of the simulation and time point 2.25. If values span across growth states, the variable's row also spans across growth states in the table. Initial values and constants are presented without time indications. This applies, for example, to the capabilities and asset requirements in Table 5-1. Please also note that values are rounded.

rise significantly when Alpha decides to pursue new development goals in its third growth state. Due to these additional hires throughout its activities, more managing employees will be required to manage the firm's human resources. The venture also needs additional complementary assets to equip employees appropriately. The venture needs to raise additional capital to finance those assets and maintain sufficient liquidity because it loses resources throughout the years (see value capture subsystem).


Figure 5-1: Firm managing outputs for Alpha²³

The model and company data match theoretical expectations in the different growth states (Figure 5-1). The model and company history show that technology development employees increase substantially to achieve the target improvements in the venture's product in the third growth state. The venture's marketing and selling employees peak during the second year to achieve its customer growth goals. The venture's value delivering employees increase throughout the years

²³ The blue lines visualise the model's calculations and the red lines visualise the case companies' actual historical data. The same applies to all graphs and figures in this chapter.

due to its increasing customer base (see customer and user subsystem). The model also accounts for the rise in the venture's firm managing employees. While the model slightly overestimates firm managing employees, it underestimated value delivering employees. Besides this minor mismatch, the model's approximations of complementary assets increase with the historical data but remain stable when complementary assets fall for the case company. This might be due to the accounting treatment of complementary assets. Alpha may depreciate complementary assets even though it is still using the equipment. In the model, it replaces depreciated equipment through new investments that might not occur in reality. The equity and debt employed fit well to the historical data. However, the model cannot account for the volatility of historical financial resources. This mismatch might occur because the theory and thus the model do not account for all differences between profit as an amount of value captured and cash flow. Cash flow might also be affected by, for example, customers delaying payment or prepaying expenses. These additional processes lead to timing differences in capturing value and accumulating the captured value as financial resources, which the model and theory do not consider.

5.1.2. Beta

Beta is a US-based company that offers communication software for businesses and their teams. Beta claims that its SaaS product helps its customers to improve productivity and reduce costs. Three years of data are available for Beta, during which the company develops through two growth states. In both growth states, Beta pursues major improvements to its technology. Thus, a technology improvement goal of 20% has been set (Table 5-2). During the first growth state, lasting for two and a half years (t=2.5), the company aims to grow its customer base by 7% per year. During the second growth state, lasting until the end of the third year, the company restructures. Beta reduces its employees across all activities to between 0.8 and 1.25 indexed employees. The employee turnover rate is set to 20%. In addition, the venture requires 1.3% of initial total assets in complementary assets per indexed employee. The venture raises 154% of initial total assets to finance one indexed improvement in its technology and 7.1% of

164

initial total assets to finance its the acquisition of one indexed customer. Due to the higher maturity of Beta as a company compared to Alpha, initial capability levels of 0.7 have been selected to reflect the company's processes. New employees lack integration into processes. Therefore, a capability level of 0.3 has been set for new employees. The venture states that it takes about nine to twelve months to train new employees. Therefore, the training time is set to one year. Lastly, the Vensim software has set the variables converting employees' competence. It estimates that each indexed firm managing employees can manage six indexed employees at fully developed capabilities and sufficient equipment.

Dominant logic – targets and HR		time: 0 - 2.5	time: 2.5	time: 2.5 - 3	
Target customer growth		7%	0%		
Target user growth		0	%		
Target technology improvement		20)%		
Marketing and selling (MS) employee ca	р	-1	0.8		
Value delivering (VD) employee cap		-1	1		
Tech. developing (TD) employee cap	-1 0.8		0.8		
Firm managing (FM) employee cap	-1 1.25				
Capabilities - firm managing		Capital asset requi	rements		
initial firm managing competence	0.7	complementary asset	s per emp.	1.3%	
firm managing years of improvement	1	liquidity goal per cus	tomer	7.1%	
new FM employee competence	0.3	liquidity goal per use	r	0%	
Employee turnover rate	20%	liquidity goal per tecl	n. improvem.	154%	
effect of competence on ability to manage	e the fir	m		6	

Table 5-2: Firm management inputs for Beta

Theory suggests that the venture increases its marketing and value delivering employees to increase its customer base during the first growth state. In addition, the venture needs to hire additional managers to supervise the larger number of employees, acquire additional complementary assets, and raise more capital to maintain its liquidity targets. During the last growth state, theory suggests a decrease in all these variables to account for its restructuring.



Figure 5-2: Firm managing outputs for Beta

The model and historical data support the expectations from theory during the two growth states (Figure 5-2). As expected, its marketing, value delivering, and firm managing employees reach a high at the end of the second year, just before the end of its first growth state. The venture lays off technology developing employees because fewer employees can achieve the same output due to learning effects. The model can also approximate the venture's financial resources, equity, debt, and complementary assets well. As for Alpha, the model smooths out financial resources and does not capture their complete volatility. Moreover, the

venture overestimates complementary assets in the final year. This mismatch can be explained by a sale of complementary assets that the model does not consider.

5.1.1. Gamma

Gamma provides a platform connecting advertisers and online video publishers. While advertisers use Gamma's platform to distribute video adverts, publishers can use the platform to monetise their video content. While the company is based in the United States, it expanded internationally two years before the simulation period. Three years of data are available for Gamma, during which the company develops through two different growth states. Initially, Gamma plans to increase the number of advertisers (customers) on its platform by 33% per year (Table 5-3). After two years and three quarters (t=2.75), the company abandons its customer growth goal and restructures. It introduces caps on its employee numbers of between 0.6 and 1.6 indexed employees. This state lasts until the end of the third year. No additional years of data are available because the company has been taken over. The turnover rate has been set to 30% to reflect the high turnover that the company notices in its international subsidiaries. Moreover, the company pursues minor improvements to its technology throughout the years of simulation.

The company needs to maintain about 0.9% of initial total assets in complementary assets per indexed employee. It raises 32% of initial total assets to finance acquiring one indexed customer. Due to a similar maturity of Gamma, Beta's initial capability levels, improvement rates, and new employees capabilities are used. The Vensim software estimated the effect sizes that transform the management capabilities from a soft scale to a hard impact on the system. Each fully equipped and capable indexed employee can manage nearly eight indexed operating employees.

	Table 5-3: F	irm management	inputs	for Gamma
--	--------------	----------------	--------	-----------

Dominant logic – targets and HR		time: 0 - 2.75	time: 2.75 - 3	
Target customer growth		33%	0%	
Target user growth		0	%	
Target technology improvement		10	1%	
Marketing and selling (MS) employee ca	р	-1	0.95	
Value delivering (VD) employee cap		-1	1	
Tech. developing (TD) employee cap	-1 1.6		1.6	
Firm managing (FM) employee cap	-1 1.25		1.25	
Capabilities - firm managing		Capital asset require	ements	
initial firm managing competence	0.7	complementary assets	s per emp.	0.9%
firm managing years of improvement	1	liquidity goal per cust	tomer	32%
new FM employee competence	0.3	liquidity goal per user	[0%
Employee turnover rate	30%	liquidity goal per tech	. improvem.	0%
effect of competence on ability to manage	e the fir	m		7.7

Due to Gamma's customer growth goal in the first growth state, it should be expected that the venture increases its marketing and selling employees. In addition, its value delivering employees should increase with its customer base. Lastly, its technology developing employees should increase slightly to maintain the continuously developing technology stock. Theory suggests that the venture also needs to increase its firm managing employees, complementary assets, and capital employed. These are required to manage, equip, and finance the increase in employees.



Figure 5-3: Firm managing outputs for Gamma

The theory, model, and actual data coincide with most of these expectations (Figure 5-3). The firm's marketing, value delivering, and technology developing employees increase in the first two years and fall with the restructuring in the final year. Its firm managing employees and complementary assets follow the patterns of operating employees. Moreover, the model tracks well the equity and debt employed by the venture. However, there are differences regarding the timing of increases and decreases of the venture's employees. For example, in the model, marketing and selling employees increase until the growth state transition.

The historical data shows that marketing employees increase for only one year and start falling before the transition. These differences indicate a mismatch between quantitative and qualitative company data. The management has not officially changed strategy and restructured during the second year of simulation. However, its marketing and selling employees are already falling. Gamma also reduces its value delivering employees to the cap introduced as part of the restructuring in the following year. Thus, it seems that the management is restructuring earlier than it is communicating. This leads to tracking errors in other types of employees and complementary assets that build on the employee numbers. A final difference for Gamma's firm managing outcomes concerns its financial resources. While historical data shows an increase in financial resources with the restructuring, the model's financial resources continue to fall. These resources accumulate through capturing value and raising capital. Because the model approximates the capital well, it is an error carried forwards from capturing value (see value capture subsystem below).

5.1.2. Delta

Delta is a European company operating a marketplace for travel bookings. While the company provides an advertising platform for travel providers, users can find and compare travel options on the platform. Three years of data are available for the company, during which it develops through two growth states (Table 5-4). Throughout its development, Delta aims to increase its customers (travel providers) by 48% per year. Like Gamma, Delta develops through a period of growth until one year and three quarters (t=1.75). Delta aims to increase its user base (travellers) during this first growth state by 35% per year. It then restructures and reduces its employee numbers to between 0.75 and 1.8 indexed employees. Throughout these years, the company improves the user experience on its website. However, most of its development focuses on optimising the back end of its platform. Therefore, a target technology improvement goal of 10%, reflecting minor improvements, has been set. The company is proud of its entrepreneurial culture, including promoting junior staff and its ability to train them. However, Delta also outlines the risk of this culture, including promoting people without the necessary skills and experience. To reflect Delta's entrepreneurial culture, the capability level of new employees has been set to 0.1, while its training time has been set to half a year. While new employee competence is lower than for other case companies, the training time is quicker too. The lower turnover rate of 20% has been set to reflect that the company does not report adverse employee relationships. The Vensim software has set the venture's ability to manage employees to eight indexed operational employees. The venture requires about 0.2% of initial total asset in complementary assets per indexed employee. It raises capital to finance customer acquisition and user acquisition targets.

Dominant logic – targets and HR		time: 0 - 1.75	time: 1.7	5 - 3
Target customer growth		48	%	
Target user growth		35%	0%	
Target technology improvement		10	%	
Marketing and selling (MS) employee ca	р	-1	0.75	
Value delivering (VD) employee cap		-1	1.8	
Tech. developing (TD) employee cap		-1	1.25	
Firm managing (FM) employee cap	-1 1.		1.4	
Capabilities - firm managing		Capital asset requir	rements	
initial firm managing competence	0.7	complementary assets	per emp.	0.2%
firm managing years of improvement	0.5	liquidity goal per cust	tomer	30%
new FM employee competence	0.1	liquidity goal per user	•	5.4%
Employee turnover rate	20%	liquidity goal per tech	. improvem.	0%
effect of competence on ability to manage	e the fir	m		8

Table 5-4: Firm management inpu	<u>uts for Delta</u>
---------------------------------	----------------------

Like the theoretical expectations for Gamma, Delta should exhibit constant or increasing technology developing employees to achieve its minor improvements. Its marketing and selling employees should increase during the first growth state to achieve customer and user growth targets. With the development in marketing employees, the venture's firm managing employees should increase to manage the additional employees. These additional employees also require equipment through complementary assets, and capital should be raised to finance the growth and technology improvements.



Figure 5-4: Firm managing outputs for Delta

The model and historical data support these theoretical expectations (Figure 5-4). Technology developing employees increase during the first year, indicating that additional employees were required to achieve the technology improvement goals and maintain the existing technology. The venture's marketing and value delivering employees peak during the first growth state and fall afterwards. With the increasing employee numbers, firm managing employees increase during the first growth state and fall with the restructuring. The model can approximate some of these developments. It reflects well the increase in firm managing employees. Moreover, it can approximate well the fall of marketing and selling employees. However, the model's approximations of value delivering and management employees is off. The model does not reflect the peak in value delivering employees after one year because its user base does not capture a peak in users (see customer and user subsystem). This underestimation also causes firm managing employees to remain below the actual data. The venture also raises capital to finance its growth and improvement goals. These are well approximated in the model. However, the model's approximations of complementary assets and financial resources are off. These are errors carried forward from the mismatch for value delivering and firm managing employees. With too low employee stocks, the model assumes that the venture also requires fewer complementary assets. Because the venture invests less money into complementary assets, its financial resources are inflated. Moreover, the model overestimates revenue (see value capture subsystem), increasing financial resources further.

5.2. Technology developing subsystem

This section illustrates the model's ability to approximate the development of a venture's technological resources. Each case company's inputs are illustrated (see Table 4-3). The subsystem also depends on the technology developing employees. The section then compares the technological resources calculated in the model to the indexed assets reported on a venture's balance sheets.

5.2.1. Alpha

Parameters regarding capabilities need to be set to simulate the technology developing subsystem. The same capability inputs as for firm managing activities have been set for Alpha. Through automatic model calibration, the Vensim software set the effects of employee competence on developing and maintaining the venture's technology. At fully developed capabilities, each indexed employee can create 13% of initial total assets of new software per year and maintain 69% of the venture's initial technological quality.

Capability development			
initial technology developing competence	0.5	effect of competence on ability to develop technology	13%
TD years of improvement	2	effect of competence on ability to	600/
new TD employee competence	0.3	maintain technology	69%

Table 5-5: Technology developing inputs for Alpha

Alpha does not focus on developing new technology in its first two growth states. Instead, it maintains its existing technology and integrates it into additional platforms. Therefore, theory suggests that no new development should take place during the first two growth states. Due to amortisation, technological resources should decrease during this time. Technological resources should only increase with the third growth state when the venture hires additional employees to develop its technology and improve its quality.



Figure 5-5: Technological resources of Alpha

The developments suggested by theory in the growth states are reflected in the venture's technological resources over time and approximated well by the model (Figure 5-5). The model calculates that the venture's technological resources decrease until the fourth year of simulation. Thus, during the first four years, the amortisation of technological assets is larger than the development of new technology. Afterwards, the technological resources increase until the end of the simulation as Alpha improves its product. In reality, the venture improves the reliability of its software by reworking its backend and launching a series of new features for its existing customers, such as a mobile app.

5.2.2. Beta

The capability inputs described for firm managing capabilities have also been used for Beta's other activities. In addition, one indexed employee, fully trained and equipped, can maintain the venture's initial technology about twice and develop new technology worth 5% of initial total assets per year (Table 5-6).

Table 5-6: Technology developing inputs for Beta

Capability development					
initial technology developing competence	0.7	effect of competence on ability to develop technology	5%		
TD years of improvement	1	effect of competence on ability to	0100/		
new TD employee competence	0.3	maintain technology	212%		

Beta's technological resources should increase as the company plans major feature releases to improve its technology. However, Beta's technological resources also amortise over time. The overall development of resources should thus depend on the balance of new developments and amortisation. Particularly the layoff in its second growth state may lead to falls in technological resources. In that growth state, employee levels may have fallen below the threshold required to maintain the technology adequately and develop new features.



Figure 5-6: Technological resources of Beta

The model and historical case data support the theoretical arguments (Figure 5-6). While the venture was able to develop some new technology in its first year, technological resources decreased in the last two years. The model can approximate well the decrease in technological resources. It captures a small decrease in the first year, where the actual data increases. This development indicates that technological improvements take place particularly in the first year.

5.2.3. Gamma

Gamma's firm managing capability inputs have also been used for its other activities. In addition, one fully trained and equipped indexed employee can maintain 121% of the venture's initial technology and develop new technology worth 34% of initial total assets per year (Table 5-7).

Table 5-7: Technology developing inputs for Gamma

Capability development			
initial technology developing competence	0.7	effect of competence on ability to develop technology	34%
TD years of improvement	1	effect of competence on ability to	
new TD employee competence	0.3	maintain technology	121%

Because the venture pursues minor improvements in its technology and is very capable of maintaining it, technological resources may increase if new developments outweigh amortisation. With the restructuring and lower technology developing employees, the rise in technological resources should slow down or even fall during the final growth state and year.



Figure 5-7: Technological resources of Gamma

The model and data support the theory (Figure 5-7). The historical case data shows that the venture can increase its technological resources over the first two years of simulation and slightly decrease in the third year of simulations. These

changes align with its growth states, and the model approximates them well with only a minor mismatch in the first year.

5.2.4. Delta

The firm managing capability inputs have also been used for Delta's other activities. The Vensim software has set the impact of the capabilities on the hard variables of the system with one fully trained and equipped indexed employee being able to maintain 172% of the venture's initial technology and develop new technology worth 1.5% of initial total assets per year (Table 5-8).

Table 5-8: Technology developing inputs for Delta

Capability development					
initial technology developing competence	0.7	effect of competence on ability to develop technology	1.5%		
TD years of improvement	0.5	effect of competence on ability to	1.500/		
new TD employee competence	0.1	maintain technology	172%		

Because technology development was a minor priority for Delta, theory suggests that the venture's technology developing resources increase if they are in excess of amortisation.



Figure 5-8: Technological resources of Delta

The model and data support this theoretical claim (Figure 5-8). Technological resources increase during the first year where the venture hires additional employees and remain about constant afterwards with a slight increase in the final

year. The model timing is slightly off, reflecting the increase in technological resources in the second rather than first year.

5.3. Customer and user subsystem

This section assesses the model's ability to approximate the venture's customer base, user base, and premium user base. These developments depend on the exogenous inputs for the customer and user subsystem particularly regarding customer and user behaviour and employee capabilities. They also utilise inputs from the firm managing, value creation, and value capture subsystems (see Table 4-4).

5.3.1. Alpha

Because Alpha only provides value to customers, the customer base has an initial value of one (Table 5-9). Many variables related to users and premiums users, their lifetimes, acquisition through marketing and word-of-mouth, and servicing are zero. The same values as for Alpha's technology developing capabilities have been used. Through calibration of the model, a standard customer lifetime of eight years has been set. Each indexed employee can acquire 1.5 indexed customers per year and service seven indexed customers at fully developed capabilities.

User and customer behaviour					
initial customer base	1	rate of customer acquisition per customer	0		
initial user base	0	rate of customer acquisition per user	0		
initial premium user base	0	rate of user acquisition per customer	0		
user to premium conversion rate	0	rate of user acquisition per user	0		
premium to user conversion rate	0	rate of premium user acquisition per cust.	0		
standard customer lifetime	8	rate of premium user acquisition per user	0		
standard user lifetime	0				
Capability development					
initial marketing selling competence	0.5	initial value delivering competence	0.5		
MD years of improvement	2	VD years of improvement	2		
new MS employee competence	0.3	new VD employee competence	0.3		
effect of competence on ability to acq	uire c	ustomers	1.5		
effect of competence on ability to acq	uire u	sers	0		
effect of competence on ability to acquire premium users					
effect of competence on ability to serv	vice ci	ustomers	7		
effect of competence on ability to serv	vice u	sers	0		

Table 5-9: Customer and user subsystem inputs for Alpha

Alpha focussed on growing its customer base during the first growth state. Thus, the venture increases its value delivering and marketing and selling employees in the first two years. With higher inputs to the venture's activities, theory suggests that the venture's customer base increases during that time. Firstly, more marketing and selling employees increase the rate at which Alpha acquired customers. Secondly, additional value delivering employees increase the venture's service capacity, increasing the customer lifetime due to better service. However, theory also suggests that quick hiring should lead to a drop in the venture's productivity. Therefore, the customer base should not rise as quickly as the venture's employees. All else being equal, customer growth in the last two growth states should be lower than during the first due to fewer marketing employees. The acquisition of new customers should accelerate due to improving capabilities of those employees. However, an increase in competition (see value capture subsystem) negatively affects customer acquisition and lifetime.



Figure 5-9: Customer subsystem outputs for Alpha

Historical data from Alpha and the model reflect the theory (Figure 5-9). During the first growth state, Alpha's customer base increases nearly seven-fold from its initial index of one until the end of the second year. As expected, this is a lower increase than the firm's marketing and selling employees, which increase ten-fold during the same period (Figure 5-1). The model approximations reflect the exponential increase in the customer base. However, the model underestimates this increase. It then carries this error forward and continues to underestimate the customer base during the second and third growth states.

5.3.2. Beta

Like Alpha, Beta does not distinguish between customers, users, and premium users. Therefore, only the customer stock is used in the model. Variables related to users and premium users have been set as zero (Table 5-10). Moreover, variables related to word-of-mouth marketing have also been set as zero as the company claims to acquire its customers through its marketing and selling employees. The Vensim software estimates a standard customer lifetime of about seven years. Capability levels for Beta have been set in line with those for its other activities. Lastly, one indexed employee with fully developed capabilities can acquire 1 and service 1.5 indexed customers respectively.

User and customer behaviour					
initial customer base	1	rate of customer acquisition per customer	0		
initial user base	0	rate of customer acquisition per user	0		
initial premium user base	0	rate of user acquisition per customer	0		
user to premium conversion rate	0	rate of user acquisition per user	0		
premium to user conversion rate	0	rate of premium user acquisition per cust.	0		
standard customer lifetime	7.3	rate of premium user acquisition per user	0		
standard user lifetime	0				
Capability development					
initial marketing selling competence	0.7	initial value delivering competence	0.7		
MD years of improvement	1	VD years of improvement	1		
new MS employee competence	0.3	new VD employee competence	0.3		
effect of competence on ability to acq	uire c	ustomers	1		
effect of competence on ability to acq	uire u	sers	0		
effect of competence on ability to acquire premium users					
effect of competence on ability to service	vice ci	ustomers	1.5		
effect of competence on ability to service	vice u	sers	0		

Table 5-10: Customer and user subsystem inputs for Beta

Because the venture's marketing and selling employees have increased only slightly over the years, theory suggests that Beta's customer base also grows in minor amounts. This small growth would also fulfil the venture's small customer growth goal of 7% per year. Customers may decrease in the final growth state if the fewer marketing employees cannot acquire sufficient customers to make up for churn.



Figure 5-10: Customer subsystem outputs for Beta

Historical case data and the model support theory. They indicate a small increase of Beta's customer base in the first growth state (Figure 5-10). In the second growth state, the historical data shows a slight decrease in the customer base. The

model does not fully capture this decrease. However, the model approximates the customer base very well, and the mismatch in the final year is minor.

5.3.3. Gamma

Because Gamma services customers (advertisers) and users (media owners), both initial values have been set (Table 5-11). The customer and user bases are thus both initialised at 0.5. Because the venture acquires new customers and users through its marketing and selling activities rather than through word-of-mouth, all word-of-mouth variables have been set to zero. Moreover, the Vensim software estimates standard customer and user lifetimes of 3 and 5.4 years respectively. Each fully equipped and capable marketing and selling employee can acquire 1.9 indexed customers and 6.2 indexed users per year. Its value delivering employees can service 4.4 and 1 indexed customers and users respectively. The same capability settings as for the other activities of Gamma have been utilised for these types of employees.

User and customer behaviour				
initial customer base	0.5	rate of customer acquisition per customer	0	
initial user base	0.5	rate of customer acquisition per user	0	
initial premium user base	0	rate of user acquisition per customer	0	
user to premium conversion rate	0	rate of user acquisition per user	0	
premium to user conversion rate	0	rate of premium user acquisition per cust.	0	
standard customer lifetime	3	rate of premium user acquisition per user	0	
standard user lifetime	5.4			
Capability development				
initial marketing selling competence	0.7	initial value delivering competence	0.7	
MD years of improvement	1	VD years of improvement	1	
new MS employee competence	0.3	new VD employee competence	0.3	
effect of competence on ability to acqui	re cust	omers	1.9	
effect of competence on ability to acquire users				
effect of competence on ability to acquire premium users				
effect of competence on ability to servi	ce custo	omers	4.4	
effect of competence on ability to servi	ce users	5	1	

Table 5-11: Customer and user subsystem inputs for Gamma

Gamma aims to increase its customer base in its first growth state. It increases its marketing and selling employees during this period to achieve this goal. Therefore, theory suggests an increase in customers during the first growth state. After its restructuring, the increase in the customer base should slow down or even decrease depending on employee ability to make up for customer loss due to churn. Because growing the user base is not a priority for Gamma, the user base should remain constant.



Figure 5-11: Customer subsystem outputs for Gamma

The model and data coincide with the theory and align, with minor differences (Figure 5-11). The venture's customer base increases by nearly 50% in the historical data and the model's approximations during the first growth state. In the historical data, Gamma then loses customers. However, the customer base in the model remains constant. Gamma's user base remains roughly constant throughout the years of simulation in the historical and model-generated data.

5.3.4. Delta

Because Delta provides its services to customers and users, both stocks' initial values have been set to 0.5 (Table 5-12). The Vensim software set the standard customer lifetime to 7.6 years. The short standard user lifetime of 0.5 years reflects that most users only stay a short time to make a booking and then discontinue the use of Delta's platform. Delta's initial marketing and selling employees can acquire 12.5 indexed customers and 7 indexed users per year. Its employees can service 1.7 indexed customers and 1.6 indexed users. The variables for its capability levels and development are the same as for Gamma in the previous activities.

User and customer behaviour					
initial customer base	0.5	rate of customer acquisition per customer	0		
initial user base	0.5	rate of customer acquisition per user	0		
initial premium user base	0	rate of user acquisition per customer	0		
user to premium conversion rate	0	rate of user acquisition per user	0		
premium to user conversion rate	0	rate of premium user acquisition per cust.	0		
standard customer lifetime	7.6	rate of premium user acquisition per user	0		
standard user lifetime	0.5				
Capability development					
initial marketing selling competence	0.7	initial value delivering competence	0.7		
MD years of improvement	0.5	VD years of improvement	0.5		
new MS employee competence	0.1	new VD employee competence	0.1		
effect of competence on ability to acq	uire cust	omers	12.5		
effect of competence on ability to acq	uire user	S	7		
effect of competence on ability to acquire premium users					
effect of competence on ability to serv	vice custo	omers	1.7		
effect of competence on ability to serv	vice users	3	1.6		

Table 5-12: Customer and user subsystem inputs for Delta

With its initial focus on user and customer growth and additional hires in marketing and sales, Delta should acquire new users and customers in the first growth state. After the restructuring, its customer base is likely to increase further due to its long lifetime. On the other hand, Delta's user base may decrease. Due to their low lifetime, the lower number of employees may be unable to acquire sufficient customers to make up for churn.



Figure 5-12: Customer subsystem outputs for Delta

The theoretical expectations are supported by the historical data (Figure 5-12) which indicate a rise in the customer base throughout the years and a peak in the user base in the first year, with a fall after its restructuring. The increase in the customer base is replicated well by the model. Its approximation of the user base is good except for the first year, likely due to a too low number of marketing and selling employees.

5.4. Value creation subsystem

This section tests the value creation subsystem introduced in the first section of the model formalisation chapter. Each subsection presents the exogenous inputs for this subsystem (see Table 4-1). Additionally, the subsystem depends on the outputs of the technology development and customer/user subsystems. The section presents value creation as a theoretical performance outcome and compares it to the revenue calculated in the model and the revenue reported by the case companies.

5.4.1. Alpha

Customers generate value from Alpha's product only through its technology. Therefore, all policy switches related to network effects are set to zero (Table 5-13). The venture claims to have developed a feature-rich application at the beginning of the simulation. Reviewing improvements made throughout the simulation period, it has been estimated that the venture can still improve its product by 50%. Therefore, the maximum technological quality has been set as 1.5. The Vensim software determined that each indexed dollar of investment into technology has an effect of 0.05 on the technological quality. With the restructuring plan introduced with its second growth state, the venture also begins to target larger customers. The average indexed number of employees per customer using the software has been used to approximate the usage intensity. After an initial value of one, this number of employees quadruples in the first growth state. The larger customers make the product available to 50% more employees. The variables related to users, premium users, and complementary products are set to zero.

Business model designs					
Customer direct network effect settings	0	0 Premium product advantage			0
Customer indirect network effect settings	0	Maximum	technological qu	ality	1.5
User direct network effect settings	0	Standard e	ffect of investme	nt in	0.05
Customer indirect network effect settings	0	technology	technology on tech. improvement		
Initial customer usage intensity	1	Initial user usage intensity			0
User and customer behaviour	t:	t: 0 - 2.25 t: 2.25 - 3.5 t: 3.5		- 6	
Maximum customer usage intensity		4 6			
New customer usage intensity		1		3	
Standard rate of adoption by customers		1		1	
Maximum user usage intensity			0		
New usage intensity			0		
Standard rate of adoption by users	0				
Complementary product	time: 0 - 6				
Quality of complementary products			0		

Table 5-13: Value creation inputs for Alpha

With these inputs and the inputs for the other subsystems, three different drivers of value creation can be identified for Alpha. Firstly, its rising customer base throughout the years, Secondly, an acceleration due to the larger customers in the second growth state. And lastly, a further acceleration in the third growth state due to technology improvements. However, the latter two value drivers are limited due to the technology s-curve and the maximum customer usage intensity.



Figure 5-13: Value creation outputs for Alpha

Based on these inputs, the model approximates that Alpha increases the value it creates (Figure 5-13). Consistent with theory, an increase in the growth in value creation can be observed when the venture enters its second growth state. Interestingly, Alpha's growth in value creation accelerates when it reduces its customer growth target and pursues growth through a more attractive customer segment instead. At that point, the venture begins to target larger customers that generate greater value from the product. The development of value creation is

also reflected in the venture's revenue. The model calculates it through the rate of value creation and the share of value captured from customers and users. Alpha's revenue grows like the rate of value creation but remains at a lower amount. This relationship is expected and has been explained by conceptualising value creation as the upper limit for revenue in the model development chapters. The revenue calculated within the model approximates well the revenue reported by the venture. However, the model's revenue has a small negative bias in the first two growth states and a positive bias in the third state.

5.4.2. Beta

As for Alpha, user variables and network effect switches are set to zero (Table 5-4). Customers derive value through Beta's technology, which the venture believes is very mature and industry-leading. Therefore, the venture's maximum technological quality has been set to 1.2 to reflect that its technology will only improve 20% through further development. The Vensim software estimated that each indexed dollar invested into technology improves the technological quality by 50. Throughout both growth states, Beta does not change its customer segment and does not indicate that customers change their usage behaviour of the software over time. Therefore, the variables related to customer behaviour have been set as one, reflecting their initial values. The same applies to complementary products. While Beta's software is compatible with a range of other software products, the company does not indicate a change in the quality of those products. Therefore, their quality is held constant as one.

Table 5-14:	Value	creation	inputs	<u>for</u>	Beta

Business model designs				
Customer direct network effect settings	0 Premium product advantage			0
Customer indirect network effect settings	0	0 Maximum technological quality		
User direct network effect settings	0 Standard effect of investment in		50	
Customer indirect network effect settings	0	technology on tech	. improvement	50
Initial customer usage intensity	1	1 Initial user usage intensity		0
User and customer behaviour	time: 0 – 2.5 time: 2.5 – 3		3	
Maximum customer usage intensity]	l	
New customer usage intensity]	l	
Standard rate of adoption by customers]	l	
Maximum user usage intensity		()	
New usage intensity		()	
Standard rate of adoption by users	0			
Complementary product	time: 0 - 3			
Quality of complementary products]		

The value created by Beta should increase with the technology improvements and growth in the customer base during the first growth state. Because technological improvements occur mainly in the first year, value creation should increase primarily in year one. Beta's revenue should also increase because competition does not change over the three years of investigation. However, the contract times should delay the effect of increased value creation on revenues (for competition and contract times, see value capture below). In the second growth state, the customer base stops growing. Moreover, Beta's technology developing employees are inadequate to maintain and develop new value-creating features. Therefore, both drivers of value growth state.



Figure 5-14: Value creation outputs for Beta

The model and case data support the theoretical claims (Figure 5-14). Value creation and revenue both grow with the customer base and technology

improvements. The increase in revenue is delayed and takes place during the second year. The rate of value creation is constant and nearly stable in the second and third years. The revenue reflects this in year three due to contractual delays.

5.4.3. Gamma

Providing a marketplace, Gamma's customers and users benefit from interacting with the other group (Table 5-15). Therefore, indirect network effects settings have been set to reflect these positive network effects. However, due to competition within each base for advertising inventory and adverts, both bases also experience negative direct network effects. Gamma describes its technology as leading and sophisticated. Therefore, the maximum technological quality for mature products of 1.2 has been selected. The Vensim software set the impact of an indexed dollar invested into technology on the technological quality at 32. The venture does not indicate a change in the usage intensity of its user. However, the venture notices that new advertisers (customers) spend significantly less on the platform and increase their spending over time. This is particularly true for its international customers, who derive less value from the product. The company explains this with different privacy laws inhibiting its targeting algorithms and a lack of local media properties abroad. The analyst has set the new user usage intensity to 0.8, the rate of adoption to 2, and the maximum usage intensity to 1.2, to track the usage intensity of the average customer.

Business model designs				
Customer direct network effect settings	-1	-1 Premium product advantage		
Customer indirect network effect settings	1	Maximum technolo	gical quality	1.2
User direct network effect settings	-1 Standard effect of investment in		22	
Customer indirect network effect settings	1	technology on tech	. improvement	32
Initial customer usage intensity	1	1 Initial user usage intensity		1
User and customer behaviour	time: 0 - 2.75 time: 2.75 -		3	
Maximum customer usage intensity		1.	2	
New customer usage intensity		0.	8	
Standard rate of adoption by customers		2	2	
Maximum user usage intensity]	-	
New usage intensity]	-	
Standard rate of adoption by users	1			
Complementary product	time: 0 - 3			
Quality of complementary products		1		

Table 5-15: Value creation inputs for Gamma

Based on the theory represented by the model, different value drivers determine Gamma's value creation. The first driver concerns Gamma's increasing technological quality. The company also increases its customers. This increase means Gamma creates value for more customers and more value per user due to network effects. However, the increasing customer base and technological quality only apply in the first growth state. Therefore, it should be expected that the rate of value creation increases slower or even falls during the second growth state. The rise in value creation may also not affect revenue because Gamma does not capture value from users. Therefore, value creation and revenue may not develop as closely as they did for Alpha and Beta. Value creation and revenues may also be disconnected for Gamma because of an increase in competition reducing its bargaining power to its customers (see value capture subsystem).



Figure 5-15: Value creation outputs for Gamma

The model replicates the theoretical expectation that value creation develops with the customer base and technology (Figure 5-15). Gamma's rate of value creation increases during the first two years and then remains stable in the model until the end of the third year. This behaviour tracks the growth in the customer base closely. Revenue has shown to be an appropriate proxy for value creation in the SaaS companies Alpha and Beta. However, it does not seem appropriate for Gamma, which its platform business model can explain. Gamma does not capture value from publishers (users) even though it creates more value for them. Therefore, there is a disconnection between value creation and revenue. The model can approximate the historical revenue well. Minor differences can be observed, particularly in the initial year of simulation. The difference at the beginning of the simulation is due to the annualisation of revenue when no full year of data is available in the model. The model approximates historical revenue well when this is not an issue.

5.4.4. Delta

Operating a platform connecting customers (travel providers) and users (travellers), customers and users both benefit from indirect network effects (Table 5-16). However, because travel providers compete for bookings on the platform, a larger number of travel providers make it less valuable for each customer. The same applies to user who compete for providers' limited inventory. Therefore, negative direct network effects hinder value creation for customers and users. Delta's maximum technological quality has been set to 1.2 to reflect its high maturity. Moreover, the Vensim software has set the impact of each investment on the quality to 110.

<i>Table 5-16:</i>	Value	creation i	inputs	for Delta
10000 0 100	/	0.00000		0. 2000

Business model designs				
Customer direct network effect settings	-1	-1 Premium product advantage		
Customer indirect network effect settings	1	1 Maximum technological quality		
User direct network effect settings	-1	-1 Standard effect of investment in		
Customer indirect network effect settings	1	technology on tech. improvement	110	
Initial customer usage intensity	1	1 Initial user usage intensity		
User and customer behaviour	time: 0 - 1.75 time: 1.75 -		- 3	
Maximum customer usage intensity		1		
New customer usage intensity		1		
Standard rate of adoption by customers		1		
Maximum user usage intensity		1		
New usage intensity		1		
Standard rate of adoption by users		1		
Complementary product		time: 0 - 3		
Quality of complementary products		0		

Theory suggests three different value drivers for Delta: the change in its user base, the change in its customer base, and its improving technology. The growth in Delta's customer base outweighs the fall in its user base. In addition, its technology improves. Therefore, Delta's value creation should grow throughout the years.



Figure 5-16: Value creation outputs for Delta

The rate of value creation supports the theoretical expectations of increases throughout the years (Figure 5-16). As for Gamma, revenue is not a good measure of value creation for Delta. The mismatch between value creation and capture for multi-sided business models explains this mismatch (see Gamma). The model can approximate the development of revenue directionally but overestimates it in all years.

5.5. Value capture subsystem

This section tests the value capture subsystem introduced in the second section of the model formalisation chapter. It depends on the human resources and capital assets, which require compensation to their input providers, depreciation and amortisation, and the value creation subsystem. Each subsection sets the exogenous inputs for each company (see Table 4-2) and compares model outputs to historical profitability measures.

5.5.1. Alpha

Alpha locks its customer into one-year contracts (Table 5-17). The company does not change its interaction with input providers and activity system throughout its growth states. Therefore, its costs per indexed employee remain constant. Depending on the activity, Alpha's costs per indexed employee are between 4.8% and 54.9% of initial total assets per year. Moreover, because the venture does not issue dividends, its cost of equity is zero. It pays 8.17% interest on its debt. With the change in its customer base as part of its second growth state, the company also experiences more intense competition for those larger customers. While Alpha operates in an oligopoly throughout the simulated years, it faces stronger competitors for its new target customer group. It is estimated that competitors' products are equal to the venture's product during the second and third growth states.

Table 5-17: Value capture inputs for Alpha

Business model designs					
Customer contract period	1	Premium user contract period			
Input providers		t: 0 - 2.25	t: 2.25 - 3.5	t: 3.5 - 6	
Cost of value delivering inputs			7.2%		
Cost of marketing and selling inputs			4.9%		
Cost of technology developing inputs			4.8%		
Cost of firm managing inputs		54.9%			
Cost of equity		0.00%			
Cost of debt			8.17%		
Competition		t: 0 - 2.25	t: 2.25 - 3.5	t: 3.5 - 6	
Number of competitors			0.5		
Strength of competitors		0.6	1		

The model calculates the share of value captured, return on assets, and net profit margin for six years using these inputs and those from the other subsystems. In the first growth state, value creation and input compensation increase. Therefore, it is not possible to predict performance development based on theory. This development depends on the balance of the two increases. However, with some caps on employee numbers and increasing value creation, the value captured should increase towards a maximum during the last two growth states.



Figure 5-17: Value capture outputs for Alpha

Throughout the years of simulation, the model and net profit margin indicate improvements in performance outcomes while the venture loses resources every year (Figure 5-17). After one year of simulations, Alpha loses about five times its revenues. However, its share of value captured and net profit margin improve towards zero in a goal-seeking manner. Therefore, the model and historical data confirm the theory in the second and third growth states, where predictions were possible. However, the return on assets does not follow this behaviour. It fluctuates around -0.4 and -1, indicating that the venture loses about between half and all its assets per year. The model can approximate Alpha's trends in return on assets. However, the measure does not seem to be a suitable indicator for the share of value captured. Conceptual differences, such as the wrong basis to compare performance, have been noted in the model formalisation chapter that explain this difference (Subsubsection 4.2.3.2).

5.5.2. Beta

Beta locks its customers into contracts between one and three years. The middle value of two years has been set in the model (Table 5-18). Its costs per indexed employee remain constant and between 8.13% and 29.74% of total assets per year throughout its two growth states. While the venture does not pay any dividends, it pays about 4% interest on its debt. Many large software companies provide or develop similar applications, and many specialised companies compete with Beta. Therefore, the venture operates in a highlight competitive market, and the number of competitors has been set as one. It has been estimated that Beta's product is equal to the average competitors' product throughout the years.

Business model designs				
Customer contract period	2	Premium user contract period		
Input providers		time: 0 – 2.5	time: 2.5 – 3)
Cost of value delivering inputs		23.3	30%	
Cost of marketing and selling inputs		29.74%		
Cost of technology developing inputs		20.48%		
Cost of firm managing inputs		8.13%		
Cost of equity		0.0	0%	
Cost of debt		4.08%		
Competition		time: 0 – 2.5	time: 2.5 – 3	5
Number of competitors			1	
Strength of competitors		1		

Table 5-18: Value capture inputs for Beta

Due to constant competition, theory suggests that the bargaining power and share of value captured from customers should not change during the years of investigation. Therefore, changes in the share of value captured should only emerge from the shares of value lost. Due to its decreasing complementary assets and technological resources, the share of value lost to depreciation and amortisation should decrease over time and performance outcomes should improve. However, theory does not provide accurate predictions for the share of value lost to input providers in the first growth state. While some types of inputs decrease, other types increase, and the development of the share of value captured depends on the net effect of these changes. In the second growth state, all inputs and thus the compensation to input providers decrease while value creation and revenues remain nearly constant. Therefore, improvements in performance outcomes should be expected.



Figure 5-18: Value capture outputs for Beta

Overall, the model's and case history's performance measures support these theoretical expectations in the first growth state. Historical accounting measures improve throughout all years (Figure 5-18). The values calculated in the model can replicate this trend. However, its share of value captured diverges from this trend in the second year, where its rate of value creation does not grow.

5.5.3. Gamma

Gamma's customers can adjust their marketing spending for video campaigns dynamically. Therefore, a contract time of 0.25 years has been set to reflect that large enterprise customers with quarterly periods will review their spending in those intervals (Table 5-19). Throughout the growth states, the venture's indexed employees produce expenses of between about 4% and 60% of initial total assets per year. While it does not compensate provides of equity, Gamma pays 2.64% of interest on its debt. The company operates in a competitive market. Gamma claims that its product has features and advantages that competitors cannot match.

Therefore, the initial strength of competitors has been set to 0.8. However, the company notes an increase in competition after two years of simulation. It has been reflected by increasing the strength of competitors to 1.

Business model designs					
Customer contract period	0.25	Premium user contract period			
Input providers		time: 0 – 2.75 time: 2.7		3	
Cost of value delivering inputs		59.15%			
Cost of marketing and selling inputs		37.22%			
Cost of technology developing inputs		4.74%			
Cost of firm managing inputs		13.05%			
Cost of equity		0.0	0%		
Cost of debt		2.6	4%		
Competition		time: 0 – 2.75	time: 2.75 – 2	3	
Number of competitors		0.8 1			
Strength of competitors		1	1		

Table 5-19: Value capture inputs for Gamma

With increasing employee inputs during the first growth state, Gamma's development of the share of value captured depends on the balance of increasing employee costs and the change in value creation. With its restructuring in the final year, Gamma should improve its share of value captured.



Figure 5-19: Value capture outputs for Gamma

The model's approximation of the share of value captured and accounting measures reflect the theory (Figure 5-19). During the first two years, Gamma's historical accounting measures fall. They rise after the restructuring. The share of value capture also reflects this pattern. However, the model's approximation of accounting measures diverges from them in the final year. This may be due to the mismatch between the timing of the restructuring and the fall in employee numbers. Due to later restructuring, the model's performance outcomes are low for too long. This mismatch then causes the underestimation of financial resources (see firm managing subsystem).

5.5.4. Delta

Delta operates in a continuously competitive marketplace. Its operations cost between 0.53% and 121% of initial total assets per indexed employee per year. This high cost reflects Delta's need to continuously acquire users with low lifetimes through marketing. Moreover, the company pays about 11.5% in interest on its debt (Table 5-20).

Table 5-20: Value capture inputs for Delta

Business model designs				
Customer contract period	0.125	Premium user contract period		
Input providers		time: 0 - 1.75 time: 1.75 - 3		
Cost of value delivering inputs		0.53%		
Cost of marketing and selling inputs		121.25%		
Cost of technology developing inputs		6.29%		
Cost of firm managing inputs		5.30%		
Cost of equity		0.0	0%	
Cost of debt		11.5	52%	
Competition		time: 0 - 1.75	time: 1.75 - 3	
Number of competitors		1		
Strength of competitors		1		

Similar to Gamma, improvements in value capture performance outcomes should be expected when the venture restructures.



Figure 5-20: Value capture outputs for Delta

The historical case data and the model's approximations of accounting measures support the theoretical expectations. The share of value captured peaks just after Gamma restructured at the end of the second year (Figure 5-20). The net profit margin and return on assets improve during the final year. This delay is expected as accounting figures are annualised, and the restructuring only affects costs during the last quarter of Gamma's financial year.

Chapter 6: Scenario simulations

The previous chapter validated the model developed in this thesis. It established a sufficient degree of confidence in the model's ability to approximate the development of the case companies. The model can now facilitate simulations that reveal further insight regarding value creation and capture, their development, and the relationships between them. These simulations build on the model's conceptualisation, formalisation, and testing. The model conceptualisation using growth process theories and causal loops (Chapter 3) identified two types of dominant logic that affect the model's feedback loops. These two types refer to digital ventures that operate at their cap on employee numbers and those that strive to grow by hiring more employees. Model validation (Chapter 5) identified a third, combined case. It regards digital ventures transitioning from a dominant logic without employee caps to one with caps. These three dominant logics form the basis of the simulations in this chapter (Table 6-1). Each type of dominant logic shows a different performance development pattern, further investigated and confirmed using the exogenous variable groups identified during detailed, formal model development (Chapter 4). On the one hand, these groups include elements the management can influence while running their venture. These are the growth and improvement targets of the dominant logic and capability development variables expressing internal improvements. On the other hand, these variables relate to contextual influences in the venture's environment and business model design²⁴.

²⁴ These are the input variable groups outlined in the tables at the end of each subsystem in Chapter 4. In addition to the four groups utilised in this section, the formal modelling has also included a variable group related to accounting inputs. The model included these to ensure consistency for testing purposes between accounting figures calculated in the model and annual reports. Because they are not theoretically relevant (see the derivation of accounting measures in Chapter 4), they are not investigated further in this chapter.
	Pass save	Different targets (Dominant logic)			
	Base case	Internal improvements (Capability development)			
With a cap on employees	Goal-seeking development to natural performance levels	 No ability to diverge from performance development Small improvements possible but same pattern 			
Without a cap on employees	Growth in value creation at the expense of value capture	 More ambitious targets cause larger fall in value capture Need to balance customer growth and tech. improvement Small improvements possible but same pattern 			
Transitioning	Growth pays off after periods of growth	 Higher payoff for companies that pursued greater growth Trade-offs for companies with different growth cycles Need for continuous improvement 			
Contextual influences and robustness	Stable environments: Same Dynamic environments: Cor Network effects: Same patter Effect sizes: Same patterns of	patterns with different natural performance levels ntinuous shift in natural performance levels erns with different natural performance levels with different natural performance levels			

Figure 6-1: Overview of simulations and findings

The first section of this chapter develops a base case and investigates the performance development of a digital venture operating at its cap on employee numbers. It develops towards natural performance levels in a goal-seeking manner. These natural performance levels reflect limits to size, growth, and profitability. To raise these limits while maintaining their employee cap, companies can pursue different targets (reflected in the model by dominant logic variables) or improve their internal operations (reflected by capability development variables). While internal improvements enable a modest rise in performance outcomes, neither option allows the venture to alter its performance levels, ventures need to grow without capping their employee numbers and hiring more employees.

The second subsection investigates the same scenarios for growth-focussed companies without a cap on employee numbers. It shows the immediate and longterm relationships between growth in value creation and the share of value capture during the period of growth. Companies can improve their growth in value creation and limit adverse effects on the share of value captured by balancing customer growth and technology improvement targets. Operational improvements have the same effect. However, they cannot entirely escape the adverse effects of high growth, such as organisational turmoil and bankruptcy risks. Instead, they must eventually stop growing and introduce a cap on employee numbers. The third section of this chapter simulates this transition. It highlights that growth pays off after, not during, periods of growth. High growth has particularly adverse effects on value capture during the growth period. However, it also has more significant payoffs after periods of growth. Companies can further improve this payoff through continuous internal improvements.

The fourth section of this chapter confirms these development patterns along the contextual variables identified during detailed model development. It demonstrates these development patterns in different environmental conditions and across different business model designs. The final section of this chapter reflects on the simulations and their implications for the case companies in the previous chapter.

6.1. Dominant logics with a cap on employee numbers

This section illustrates and explains the performance development of digital ventures operating at their cap on employee numbers. Thereby, managers implement their desire to reduce delegation, remain in control of their ventures, or avoid growth risks (Davidsson, 1989; Hesse and Sternberg, 2017). However, not just owner-managers of digital ventures that want to maintain control limit their employee numbers. The case studies in the previous chapter show that larger companies also implement employee caps as part of restructuring activities to control costs. The performance development of companies with such employee caps is simulated under varying conditions in this section. Due to the model's indexing, simulations apply to smaller, owner-managed and larger, more mature firms. The first subsection, below, illustrates the model inputs and the development of a base case for comparison in further simulation. This base case exhibits a goal-seeking performance development towards natural levels for the rate of value creation and the share of value captured. However, managers and entrepreneurs may strive to improve the performance of their digital ventures beyond these natural performance levels. The detailed model developed (Chapter 4) outlines two options: pursuing more ambitious growth/improvement targets and internal improvements by affecting variables related to capability development. The second and third subsections alter input variables related to

these two options to determine the development of performance outcomes under these two conditions.

6.1.1. Employee-capped base case

A reference mode is required to generate insights from simulations. This reference mode captures the baseline behaviour of the system and serves as a comparison for the outcomes of other simulations (Sterman, 2000; Warren, 2002). As the methodology argues, simulations should strive to be theoretically relevant and realistic (Davis, Eisenhardt and Bingham, 2007). This thesis investigates the performance developments and relationships that allow companies to develop an ability to capture value while growing their value creation. Therefore, the base case has been created to reflect a company that achieves an ability to capture value as it increases its value creation. The following illustrates the input variables and the performance development of this base case.

6.1.1.1. Model inputs for the employee-capped base case

The model input variables for the base case have been obtained in an interactive process going back and forth between all simulations in this chapter. This process ensures they are theoretically relevant. The values for input variables have been compared to the case studies to ensure they are within reasonable, realistic ranges. Input variables have been selected to (1) ensure the venture is initially lossmaking, (2) allow feedback processes that improve the share of value captured, (3) employ a dominant logic that capitalises on these processes, and (4) ensure theoretical relevance. The summaries below outline the most important settings for each of these four criteria²⁵.

The level of competition and inputs costs have been used to create an initially loss-making base case:

• All operating input costs per indexed employee have been set at 25% of initial value creation to ensure that the company is initially loss-making. Thereby,

²⁵ Please see Appendix E for a complete and detailed list of input values.

all four costs together equal the initial rate of value creation of one, which is the maximum value the venture can capture.

• The initial strength of the average product of competitors and substitute providers is set as equal to the venture's product. The venture is operating in an oligopoly. These two settings reduce the share of value captured into the negative.

The following input values have been set to allow for feedback mechanisms that improve the ability to capture value over time:

- The maximum customer usage intensity is 1.5, the usage intensity of new customers is 0.5, and the rate of feature adoption is one year. These values allow the venture's customers to improve their usage intensity and develop switching costs over a period like those seen in case companies.
- The maximum technological quality has been set as 1.5. This potential allows the venture to improve its technological quality and value created per customer by up to 50%.
- The customer contract terms have been set at one year. This value is consistent with lengths among case companies in the previous chapter. It delays the effect of relative use value improvements on the ability to capture value.
- The venture's initial capability levels are assumed to have reached the level of "defined" on the capability maturity model (0.5). Capability improvement times have been set at one year, which is reasonable compared to values observed in case companies. The combination of these variables allows the venture to improve its employee productivity over time.

In the base case with a cap on employee numbers, the venture pursues the following objectives with the following constraints:

• The venture pursues a target customer growth rate of 25% per year and a target technology improvement rate of 10% per year. These values reflect that the venture is in the scaling-up process and focuses on growing its customer base while also improving its product.

• Employee numbers are capped at their initial level. Thus, the base case venture operates at its employee limits. All changes in performance outcomes are thus due to internal developments rather than changes in activity inputs.

The following settings have been placed to ensure that the model creates theoretically valuable insights rather than cofounding results:

- Accounting measures imported to the model, such as other assets or taxation, have been set as zero. They have been included in the model to achieve consistency between case companies' financial statements and model outputs for testing purposes. However, they are not theoretically relevant for the investigation below.
- All effect sizes that transform capabilities from soft to hard scales have been set to ensure that the venture runs at full capacity and achieves its goals with its initial employees and capability levels. Thus, the model is not affected by initial surges in hiring or layoffs to adjust capacities just after the beginning of the simulation. These settings ensure that the venture's performance development is due to theoretically relevant processes rather than eliminating operational imbalances.
- Initial financial resources have been set as one. The venture does not raise any capital to finance growth and technology improvements or return any debt or equity to investors. Thus, the venture's financial resources can be used as a cumulative measure of company performance.
- To create a simple base case, the business model design of a SaaS venture is used. The venture creates customer value based on its technological quality (see Alpha and Beta during model testing). The robustness of results for different business model designs is tested as a contextual influence in the final sections of this chapter.

6.1.1.2. Performance development of the employee-capped base case With the above inputs, the base case's performance outcomes develop in a goalseeking manner towards natural performance outcomes (Figure 6-2). As required for its set-up, the company does not initially capture value. Instead, its initial loss amounts to about 40% of the value it creates. Its share of value captured then improves and turns positive after about one year of simulations. While the

venture's share of value captured remains negative, its financial resources deteriorate. Once it has developed an ability to capture value, its financial resources accumulate. Because the rate of value creation and share of value captured increase throughout the five years of simulation, the rate at which financial resources accumulate accelerates. As discussed in the model development chapters, the financial resources represent cash and cash equivalents on the company's balance sheet. The venture could, for example, pay these out to investors. Two important theoretical insights from this employee-capped base case (BaseC) are reviewed below. Firstly, even digital ventures operating at their employee caps can grow and improve their share of value captured. However, their performance outcomes are limited, and the venture develops towards limits in a goal-seeking manner.



Figure 6-2: Performance development of employee-capped base case

A first theoretical insight from the BaseC scenario is that even companies unwilling to hire more employees can grow their rate of value creation and improve their share of value captured. The model explains the growth in value creation by the existing employees acquiring additional customers, improving the technology, and the improving customer usage intensity. The venture may even grow at accelerating rates due to capability improvements over time. This growth in value creation also improves the share of value captured. In a stable environment, the share of value captured from customers should increase as the SaaS venture improves its technology relative to its competitors. Thus, revenue as a fraction of value creation increases. Because the venture caps its employee headcount, it effectively caps the costs in absolute amounts too. At increasing rates of value creation, these costs expressed as a fraction of a growing rate of value creation decrease the share of value lost to input providers. Thus, the share of value captured increases due to an increasing share of value captured from customers and a declining share of value lost to input providers.

However, these improvements in performance outcomes are restricted for companies that operate at their employee cap. Instead, performance outcomes develop in a goal-seeking manner towards natural performance levels. Here, goalseeking behaviour does not imply that the management has targeted these rates of value creation and share of value captured. As the inputs for the base case illustrate, the management was seeking improvements in its technological quality and increases in its customer base of, respectively, 10% and 25% per year. Without interfering feedback in the model, such targets alone would have caused continuous, exponential growth in value creation. Instead, goal-seeking refers to the behaviour of variables in System Dynamics models that are pushed towards specific values by negative, balancing feedback (Coyle, 1996; Ford, 2019). In the model developed in this thesis, these balancing feedback loops include, for example, churn rates increasing as capacity limits are exceeded. Other balancing loops are the s-curves regarding, for example, the customer usage intensities or technological quality. While capacity limits and the rates of technology development and customer acquisition increase through improving capabilities over time, a balancing loop also restricts capability improvements. Therefore, managements that cap their employee numbers will eventually fall short of achieving their customer growth and technology improvement goals. While growth in value creation is thus possible, these balancing loops introduce a maximum rate of value creation towards which the system develops. These limits on the rate of value creation also restrict the processes which can improve the share of value captured discussed above.

The natural performance levels achieved by the venture may differ from the target that the management wants to achieve. For example, the base case's management targets a 25% growth in its customer base per year. The goal-seeking development towards a natural rate of value creation implies that growth in value creation reduces to zero over time. Therefore, the venture's management may strive to improve its performance levels and allow it to perform better than its natural levels. Without lifting the cap on employee numbers, the model allows the management to alter two groups of variables. Firstly, variables related to their targets, which are part of the dominant logic. Secondly, variables related to the venture's capability development, which the management may alter to improve internal operations. The two subsections below illustrate the impact of changes to these variables on performance outcomes and development.

6.1.2. Different targets expressed in the dominant logic

Companies may pursue different targets even while restricting their employee headcount to initial levels. The model conceptualisation chapter has outlined two decision points: the target customer growth rate and target technology improvement rate. These reflect the management's goals and aspirations for percentage growth in its customer base and technological quality per year. They affect management's hiring decisions and thus the inputs to all other activities. Three different scenarios have been designed to investigate the impact of changing growth and improvement goals on companies with employee caps (Table 6-1). The venture targets an increased technology improvement rate of 20% per year in a first technology-focused scenario (TechC). In a second scenario, the venture is scaling up its customer base quicker by targeting a customer growth rate of 50% per year (ScaleC). The ambitious company in the third scenario targets the higher technology improvement and customer growth rate (AmbitiousC).

Variables	BaseC	TechC	ScaleC	AmbitiousC
Target customer growth	25%	25%	50%	50%
Target technology improvement	10%	20%	10%	20%

Table 6-1: Changes to employee-capped base case's dominant logic²⁶

Regardless of their targets, companies that have hired the maximum number of desired employees still develop towards performance levels in a goal-seeking manner (Figure 6-3). The simulated companies with higher targets do not differ significantly from the base case operating at its employee limit. The TechC, ScaleC, and AmbitiousC scenarios develop towards similar rates of value creation, shares of value capture, and financial resources.

²⁶ This chapter abbreviates the scenario names when referring to them. The word standards for the scenario conditions. The letter indicates if employee numbers were capped (C), uncapped (U), or transitioning (T). For example, here the AmbitiousC scenario indicates more ambitious targets while capping employee numbers. The AmbitiousU scenario in the second section indicates the same targets but with uncapped employee numbers. The base case is shown in tables to compare changes to input values.



Figure 6-3: Effect of changes to employee-capped base case's dominant logic²⁷

The simulations show that ventures operating at their employee caps can only achieve growth and improvement targets up to maximum rates. Because the venture will hire no additional employees, inputs to marketing, value delivering,

²⁷ Please not that in this figure, the lines for scenarios TechC and ScaleC are the same as for BaseC. Therefore, only one line is visible.

and technology development are limited. Therefore, the venture cannot increase its technology improvement and customer acquisition rate by hiring additional employees because achieving these rates is impossible through the current productivity improvements. While companies can improve their internal operations (see next subsection), they may also have to hire additional human resources to achieve more ambitious targets (see next section).

6.1.3. Internal improvement for capability development

Pursuing more ambitious targets does not allow digital ventures with employee limits to significantly alter their natural performance levels. Instead, the management may seek alternative routes to improve its performance outcomes. The other option that employee-capped ventures have in the developed model is to improve their internal operations and raise the productivity of their employees. Because capabilities have been conceptualised as productivity levels, the management may thus strive to affect variables related to capability development. In the conceptual model, capabilities deteriorate through hiring and improve through learning based on dynamic capabilities, with which they form a balancing loop. The detailed description of the model introduces a range of exogenous variables affecting the development of capabilities. These include the time required to improve productivity levels and train new employees, the capability levels of new employees, and the employee turnover rate. Three scenarios investigate the impact of changing these variables on performance outcomes and their development (Table 6-2). The first scenario lowers capability improvements times from one year in the base case to half a year (TrainingC). The second scenario reduces the capability levels of the average new employee from 0.3 to 0.1 on the capability maturity model scale (NewEmpsC). The final scenario investigates the impact of reducing the employee turnover rate from 30% to 10% (TurnoverC).

Table 6-2: Changes to employee-capped base case's capability development

Variables	BaseC	TrainingC	NewEmpsC	TurnoverC
Capability improvement times	1	0.5	1	
New employee capability level		0.3	0.1	0.3
Employee turnover rate		10%		

210

Shorter improvement times and lower turnover rates improve both performance outcomes compared to the base case. Lower capability levels of new employees reduce performance outcomes (Figure 6-4). However, company performance outcomes still develop in a goal-seeking manner towards different natural performance levels.



Figure 6-4: Effect of changes to employee-capped base case's capability development

Changing variables related to capability development as part of internal improvements alters natural performance levels. Lowering turnover rates, lowering training time, and increasing the competence of new employees increases the capability level of the firm. For example, at lower turnover rates, trained employees are replaced less frequently by employees who lack experience. Thus, the average productivity level of employees increases. With these increased capability levels, the venture's employees can achieve a greater output in their activities. As shown when investigating different targets, the constraints for companies operating at their employee cap are not the management's target but what its current employees can achieve. Therefore, the venture's employees can acquire and service more customers and improve the technology faster with improved capability levels. These higher capacities increase the rate of value creation towards which the venture develops. Because costs are constant due to restricted inputs, the share of value captured improves with the rate of value creation.

However, changing capability development variables does not change the fundamental goal-seeking patterns of performance development. Because the venture still caps employee inputs to its activities, changes to the outputs of activities can only be expanded through capability improvements. Internal improvements, as explained above, increase the capabilities and employee productivity of the firm. However, improvements are governed by a balancing feedback loop and limited to best practice. Thus, capabilities settle at a higher level. These higher capability levels ultimately determine the rate of value creation and thus the share of value captured towards which the firm develops.

6.2. Dominant logics without a cap on employee numbers

The simulations above have focussed on companies that cap their employee headcount to maintain control over the companies. Ventures with such caps exhibit goal-seeking development towards natural performance levels. While internal improvements can raise these performance levels, ventures cannot diverge from goal-seeking behaviour while maintaining their employee cap. Thus, the ventures that want to improve performance levels further need to uncap their

employee headcount. In practice, such digital ventures focus on scaling up their customer base and improving their technology even if their ambitious goals require increasing their headcount.

The simulations in this section explore the performance development of digital ventures without a cap on employee numbers. This section first outlines the performance development of a base case without employee caps. It illustrates growth in value creation at the expense of the share of value captured. The venture can thus diverge from its goal-seeking development in value creation. However, it does so at the expense of capturing value and accumulating financial resources. To grow faster or reduce the adverse effects on value capture, managers may pursue different growth/improvement targets and internal improvements. Simulations show that an ambitious and customer-focussed dominant logic can increase the growth in value creation. However, such ambitious ventures also exhibit a more severe fall in the share of value captured and the venture's financial resources. Internal improvements and a balance of targets can accelerate growth and soften the adverse effects on the share of value captured. However, the share of value captured remains lower than in employeecapped scenarios. Therefore, companies may eventually stop growing in order to improve their share of value captured, which the next section of this chapter reviews.

6.2.1. Employee-uncapped base case

The simulations in this section use the inputs of the base case introduced in the previous section. They only eliminate the employee caps of that base case to simulate companies that grow their customer base and improve their product without capping their employee headcount. Simulating this base case for employee-uncapped companies (BaseU) illustrates that such companies can increase their value creation rate exponentially. BaseU diverges from the goal-seeking behaviour uncovered in the previous section (Figure 6-5). However, it does so at the expense of the share of value captured and financial resource accumulation. This growth in value creation, the trade-off with the share of value captured, and the impact on financial resources are reviewed below.



Figure 6-5: Performance development of employee-uncapped base case

A first insight is that companies with a growth-based dominant logic that do not cap their employment numbers can increase the value they create. They are not constrained by all the limits that affect companies with employee caps. Instead, they can grow their customer base and improve their technology, increasing their value creation. Companies achieve this growth by hiring additional employees to acquire more customers, service those new customers, improve their technology, and manage the larger firm.

The simulation also shows trade-offs between growth in value creation and the share of value captured. Compared to the BaseC, BaseU develops towards a lower share of value captured. The model explains this lower level through the lower capabilities of the average employee. Every time new employees are hired to achieve updated growth and improvement goals, capabilities fall due to new employees' lack of firm-specific knowledge. Therefore, more employees are required to acquire, service, and manage a growing firm. The share of value captured also develops towards a set level. On the one side, capability improvements and capacity filling up should improve the share of value captured. On the other hand, the continuous expansion lowers capabilities and creates new capacity. The share of value captured develops to the level at which these opposing forces balance. Due to a lower share of value capturing a share of that value.

Due to the lower share of value captured, these companies' financial resources lag behind the venture with employee caps. This deeper fall poses a risk of bankruptcy. BaseU's financial resources remain above zero. However, if the company had lacked an ability to capture value for longer, its financial resources would have fallen below zero. Therefore, growing ventures need to ensure they raise sufficient capital to finance their growth and avoid bankruptcy. Using the financial resources to measure cumulative performance, BaseC has been more successful than BaseU. After five years, BaseC has more than tripled its financial resources but BaseU has just over doubled them.

The simulation above shows that while companies without employee caps can achieve exponential growth in value creation, they do so at the expense of the share of value captured and financial resource accumulation. Managers of digital ventures may thus strive to improve the rate of growth in value creation and soften the adverse effect on the share of value captured. As outlined in the previous section, the model allows them to pursue different targets as part of their dominant logic or through internal improvements affecting capability

development. The two subsequent subsections investigate the impact of these changes on the performance outcomes of companies without employee caps.

6.2.2. Different targets expressed in the dominant logic

Managers and entrepreneurs of digital ventures may pursue more ambitious and different customer growth and technology improvement targets. They may do so to increase their growth in value creation or alter the development of their share of value captured. The same scenarios as for employee-capped companies (Table 6-1) are utilised to simulate different targets reflected in the dominant logic. The performance development patterns with more ambitious target rates are similar to BaseU. They include exponential growth in value creation and development towards a lower share of value captured. However, ScaleU and AmbitiousU also experience an immediate fall in the share of value captured (Figure 6-6). The different scenarios show different growth rates in value creation and different levels towards which the share of value captured develops. Moreover, ventures need to balance customer growth and technology improvement.



Figure 6-6: Effect of changes to employee-uncapped base case's dominant logic

A first important insight of the higher target growth and improvement rates is that they exaggerate the performance development observed in the BaseU scenario. Firstly, higher targets accelerate the growth in value creation, identifiable through the value creation's curves' steepness. Ventures achieve higher growth in value creation by hiring additional employees. Secondly, companies with more ambitious targets develop towards lower levels of value capture because they continuously need to hire more employees, resulting in lower average capability levels. An additional effect concerns the immediate fall in the share of value captured that affects ScaleU and AmbitiousU. Their shares of value captured immediately decreases due to timing differences between inputs that it must pay for and the growth in value creation. The venture must hire additional employees with lower productivity due to their lack of firm-specific knowledge to achieve more ambitious growth goals. While the venture pays these employees as soon as they are hired, these employees' outputs lag behind the average employee until they have been trained. Lastly, due to the lower share of value captured, financial resources become suppressed for longer. For ScaleU and AmbitiousU, they even turn negative. Real companies would have gone out of business at that point.

Companies that balance customer growth and technology improvements can soften the impact on the share of value captured. While focusing on scaling up the customer base allows exponential growth in value creation, improving the technology improves the share of value captured. Ventures that focus on customer growth can grow exponentially, whereas ventures that focus on their technology can only improve it along the technology s-curve. Thus, focusing resources on achieving more ambitious customer growth allows ScaleU to grow its rate of value creation faster and to higher levels. However, TechU's share of value captured reaches a value above BaseU's. The impact of technology development can explain these differences. With increasing technological quality, the venture's product becomes, relative to competitors and substitutes, more attractive to customers. Thereby, TechU increases its bargaining power to customers, resulting in more value being captured from them. Balancing both targets has the additional advantage of spreading the development costs over more customers. Thus, both targets are complementary and allow AmbitiousU to achieve exponential growth in value creation, while softening the negative impact on the share of value captured compared to ScaleU.

The simulations in this subsection have shown that managers can increase the growth in value creation by changing their dominant logic's targets. However, the more ambitious these goals, the steeper the immediate fall in value capture. Moreover, the level towards which the share of value captured develops may be

lower. This poses a significant risk of bankruptcy for companies that pursue ambitious customer growth goals. They must thus raise sufficient funding to remain solvent. Additionally, managers may improve performance through internal improvements.

6.2.3. Internal improvement for capability development

Managers and entrepreneurs of digital ventures may strive to improve their internal operations and employee productivity to increase their growth in value creation. These may also soften the impact of growth on the share of value captured and the financial resources. The impact of lower training times (TrainingU), employee turnover (TurnoverU), and new employees' productivity (NewEmpsU) are explored for companies without employee caps. The same scenario set-up for companies with employee caps (Table 6-2) investigates internal improvements. These companies can significantly soften the fall of the share of value captured (Figure 6-7).



Figure 6-7: Effect of changes to employee-uncapped base case's capability development

Digital ventures without employee caps can soften the fall in the share of value captured through internal improvements (affecting variables related to capability development). As explained for BaseC, shorter training times, lower employee turnover, and a higher level of new employees' competence increase the capability level of the firm. This level expresses the venture's average employee productivity. However, whereas capability improvements affected both

performance outcomes of employee-capped companies, their impact on employee-uncapped companies has a focus on the share of value captured. At higher productivity levels, the venture achieves its customer growth and technology improvement goals with fewer employees. Therefore, ventures that improve internally need to compensate fewer input providers for their contribution to activities. Compared to BaseU, internal improvements thus soften the fall in the share of value captured when they initialise the growth process. The venture also develops towards higher levels for the share of value captured. Minor differences in the rate of value creation can also be observed but these should entirely depend on the venture's targets. These differences occur because the management is never perfectly aware of the venture's actual productivity levels. Instead, their productivity estimates adjust over time to their actual values. However, by then, productivity has moved on to a new value. For example, with faster training times, the venture's management always underestimates productive levels and hires more employees than were required to achieve its targets. Due to more employees with improved capabilities, the management overachieves, leading to slight variations in the rate of value creation.

Thus, digital ventures could significantly soften the negative effect of growth on the share of value captured. However, the share of value captured and financial resources as a cumulative performance measure still suffer during the growth process.

6.3. Transitioning between dominant logics

A venture's share of value captured suffers during the growth in value creation. Ventures may thus decide to transition from a dominant logic focused on scaling up and improving the technology without employee caps to a dominant logic with employee caps. Research also argues that they may do so due to market constraints, fulfilled ambitions, or internal constraints (Levie and Lichtenstein, 2010; Brown and Mawson, 2013). Such transitions have also been observed and described in the case simulations during model validation, where all companies restructure and cap or reduce their employee numbers after periods of growth. A prime example of this development is Alpha. It grows exponentially during its

first growth state and then maintains a constant number of marketing employees for nearly four more years. This constant employee number is below its headcount in the growth period. This section investigates such transitions, which indicate that growth pays off after periods of growth. These benefits are particularly high if higher growth rates have previously been pursued. In addition, this section explores different growth cycles, different restructuring magnitudes, and the impact of internal improvements.

6.3.1. Transitioning base case

This subsection investigates the impact of transitions from dominant logics without a cap on employee numbers to dominant logics with a cap. The simulation follows BaseU for another five years. During the first five years, the transitioning base case (BaseT) exhibits the same conditions and development as BaseU. After five years, the venture's employee numbers are capped at the level they reached. The simulation is then continued for another five years with those caps on employee numbers. Thus, BaseT reflects a venture that grows for five years and then no longer increases the inputs to its operations. The simulation shows that the venture returns to the goal-seeking development of performance outcomes discovered for companies operating at their cap on employee numbers (Figure 6-8). The figure illustrates that growth pays off after periods of growth due to higher levels in both performance outcomes. These lead to a much-accelerated accumulation in financial resources.



Figure 6-8: Performance development of transitioning base case transitioning

BaseT illustrates a long-term trade-off between the growth in value creation and the share of value captured. Once companies introduce employee caps, their performance outcomes develop in a goal-seeking pattern again. However, their value creation rate after the growth period develops towards a higher level. Metaphorically, companies have created a much larger value pie from which they can now capture value.

The digital venture develops towards a higher share of value captured due to economies of scale. For example, BaseT can spread technology development costs and the amortisation of technological resources over more customers and value creation. Combined, the higher rate of value creation and share of value captured cause significantly faster growth in its financial resources once BaseT transitions. Thus, periods of growth pay off once the venture stops growing. Companies must decide whether the higher rate of value creation and share of value captured after growth outweigh the lower share of value captured during the growth period. They must also avoid the risk of bankruptcy during the growth period mentioned above. The initial period of growth pays off for BaseT because its financial resources accumulated over the ten years of simulation overtake the base cases with and without employee caps.

The above transition from an uncapped to a capped dominant logic illustrates that growth pays off after periods of growth. However, it raises questions about maximising this benefit of growth. As argued before, the management's levers are the target expressed in their dominant logic and pursuing internal improvements affecting capability development. These are reviewed in the two subsections below.

6.3.2. Different targets expressed in the dominant logic

The transitioning base case has illustrated that companies benefit from growth after they stopped growing employee inputs. The first two sections of this chapter have argued that companies may pursue different growth and improvement targets. Transitions introduce additional complexities to the development of value creation and capture during the growth process. Firstly, companies may not just maintain but also reduce their employee numbers when transitioning. Secondly, companies may also switch back and forth between dominant logics. Below, the impact of different targets, reductions, and growth cycles are explored.

6.3.2.1. Different targets throughout the simulation

Managers can pursue different targets. These targets do not affect companies with employee caps significantly because they remain constrained by their employee productivity. However, they significantly affect employee-uncapped companies. Simulations have identified that companies can pursue more ambitious targets and grow their rate of value creation faster. However, they do so at the expense of the share of value captured. Different target magnitudes expressed in the dominant logic should thus be tested for transitioning companies. The simulations for employee-uncapped companies (see subsection 6.2.2) have been continued for another five years. The simulation during the first five years is the same as illustrated for companies without employee caps. For the second five years of simulation, their employee numbers have been capped to their value after the first five years of simulation. These simulations show that more ambitious targets lead to greater payoffs after periods of growth. They exaggerate the development observed for BaseT (Figure 6-9).



Figure 6-9: Effect of changes to transitioning base case's dominant logic (targets)

Ambitious and customer-focussed targets exaggerate the impact on value creation and capture during the growth period. However, they also exaggerate the benefits to both performance outcomes after the period of growth. Like BaseT, all transition scenarios return to goal-seeking development towards new natural performance levels. However, their ambitious growth targets have increased company sizes to much higher levels during the first five years. Therefore, their natural rate of value creation is also much higher. In addition, these companies benefit from the economies of scale that increase their share of value captured. Because both performance outcomes are higher, AmbitiousT and ScaleT accumulate financial resources much quicker. Thus, while ambitious growth has more severe adverse effects during the period of growth, its payoff after periods of growth is also significantly higher.

6.3.2.2. Different restructuring magnitudes

The transitioning base case and different targets have assumed that ventures maintain the number of employees at the end of the growth period. This has allowed for theoretically relevant demonstrations of the impact of transitions and targets. However, as the simulations in the validation chapter have shown, employee caps are often lower. Ventures introduce lower caps as part of restructuring programs to reduce costs. Striving for realistic simulations, such reductions should be simulated. The scenario set-up below follows the same structure as the transitioning base case. However, they do not assume that the employee numbers at the end of the growth period are the cap. Instead, they simulate reductions in employee numbers at the point of transition. Two scenarios are designed to facilitate this process: a cap of 20% (Reductions20) and of 50% (Reductions 50) below the employee numbers at the end of the growth period. The simulations show the expected goal-seeking behaviour in the employee-capped period. However, companies develop towards different natural performance levels. While laying off employees increases the share of value captured quickly, it results in lower levels of value creation in the long term (Figure 6-10). Thus, companies introducing layoffs present another trade-off between value creation and capture.



Figure 6-10: Effect of changes to transitioning base case's dominant logic (reductions)

A benefit of reducing rather than maintaining employee numbers after growth periods is a quick increase in the share of value captured. The two scenarios with employee reductions show step-like changes in the share of value captured. The more significant the reduction, the higher the rapid improvement in the share of value captured. Timing differences between losing cost-generating and valuecreating resources explain these step-changes. The venture can quickly lay off its employees and reduce its costs. However, it can hold on to value creating resources like the customer base and technological quality that these employees have created before being laid off.

However, the simulations also show that layoffs lead to lower rates of value creation towards which companies develop. Like companies that have grown at higher rates before transitioning, companies that maintain more employees can establish a larger size. On the other hand, companies which lay off employees give up more of their size potential. In all previous simulations, companies have increased their rate of value creation to their natural performance levels. The Reduction50 scenario above shows that companies may also fall towards their natural performance level. For example, rather than filling up its capacity limit, the venture falls towards its capacity limit. After layoffs, its employees could not service all customers appropriately, increasing churn rates and reducing the customer base. The smaller size also affects the share of value captured due to economies of scale in the long term. The companies with lower reductions show a slightly higher share of value captured at the end of the simulation.

6.3.2.3. Different number of growth cycles

Another method to balance value creation and capture might be switching from employee-uncapped to -capped growth states more often. So far, the transition companies only develop through one growth cycle with unconstrained growth in the first five years and employee caps in the second five years. However, managers may increase the number of growth cycles they pursue. Scenarios with two and five growth cycles have been developed to investigate the impact of more regular switches. In 2Cycles, the company switches between dominant logics with and without a cap on employee numbers every two and a half years. In 5Cycles, the company switches dominant logics every year. Both scenarios begin the simulation with a dominant logic without employee caps. The employee caps are set to the number of employees at the end of the previous period without caps. Overall, these different dominant logics show that the positive and negative effects of the cycles have an early advantage in the rate of value creation.

However, more growth cycles lead to higher levels of value creation at the end of the simulation. These outcomes are reversed for the share of value captured, further indicating the trade-offs between value creation and capture. The financial resources as a cumulative performance measure are best for the base case with only one growth cycle. These three elements are discussed below.



Figure 6-11: Effect of changes to transitioning base case's dominant logic (cycles)

Pursuing fewer growth cycles leads to a higher level in value creation up to about seven and a half years of simulation. This initial outperformance in value creation for fewer growth cycles can easily be explained because the simulation assumes growth for the first five years. The simulations with higher growth cycles assume that the ventures grow without employee caps only for half of that time (2Cycles) or three years (5Cycles). Companies capping their employee numbers can also grow at decreasing rates and are limited to their natural performance levels. Therefore, the high number of growth cycles allows the ventures to scale up to ultimately higher rates of value creation.

While more growth cycles lead to an earlier improvement in the share of value captured, the base case with one growth cycle overtakes them once it stops growing. The theory outlined above can explain these initial improvements in the share of value captured. Growth leads to lower capability levels and thus lower shares of value captured. The company growing for the first five years without employee caps suffers from those lower levels for the entire first five years. The ventures stopping their growth more frequently can improve their share of value captured during the periods in which they cap their employee numbers. However, in the periods in which these companies grow again, their share of value captured falls again. This causes oscillation in performance outcomes with improvements when the ventures cap employee numbers and deterioration when they grow without caps. These adverse effects during uncapped years also explain the lower share of value captured for more growth cycles in the second half of the simulation. Here, the companies with more growth cycles suffer from the adverse effects of growth on the share of value captured. Even in years in which it caps employee numbers, the share of value captured remains below the base case with one growth cycle. In these years, the return to growth periods occurred before all improvements in the share of value captured took place.

The base case with one growth cycle leads to the highest financial resources as a cumulative performance measure. The above illustrations for different growth cycles demonstrate the trade-offs between growth in value creation and the share of value captured. While companies with more growth cycles can increase their share of value captured earlier, they ultimately fall behind. However, they can

increase their rate of value creation to higher levels. Due to these mixed results for the individual performance outcomes at different points in time, the financial resources as a cumulative measure of performance indicate the superior approach. Over the ten years of simulation, the company with one growth cycle accumulates the most resources. Taking a longer view might lead to different results, depending on the subsequent actions of companies with more growth cycles. As illustrated in the previous section, their higher rate of value creation might start paying off if they stopped growing. This leads them ultimately to a faster rate of resource accumulation that eventually overtakes the company with only one growth cycle.

6.3.3. Internal improvement for capability development

Another route to maximise the payoff after growth periods concerns internal improvements by affecting variables related to capability development. The same scenarios affecting training times, new employee capabilities, and employee turnover are utilised to investigate the impact of internal improvements on BaseT. These simulations show the need to pursue internal improvements continuously throughout both types of dominant logic. Improvements lead to minor variations in the rate of value creation, improved shares of value captured, and accelerated financial resources accumulation (Figure 6-12).



Figure 6-12: Effect of changes to transitioning base case's capability development

The transitioning base case is a combination of both dominant logics one after another. Therefore, the benefits of internal improvements can be explained by the benefits during the two individual dominant logics. During the first five years, the transitioning base case is growing without employee caps. It thus benefits from internal improvements by achieving its growth and improvement targets with fewer employees. These improvements increase the share of value captured at similar rates of value creation. During the second five years, the transitioning case caps its employee numbers. It now benefits from its capped employees achieving more output in their respective activities. Here, the venture benefits from an increased rate of value creation and lower costs when expressed as a fraction of value creation. Combined, those continuous improvements throughout the periods lead to lower falls in financial resources. They reduce the risk of bankruptcy during growth and accelerate the accumulation of financial resources in the long term.

6.4. Contextual influences and robustness

The above section has illustrated that digital ventures with employee caps experience goal-seeking performance development for their rate of value creation and share of value captured. The management can alter the level towards which their companies develop through the variables available in the model. These variables are targets expressed in the dominant logic and internal improvements affecting capability development. However, ventures with employee caps cannot diverge from a goal-seeking pattern. They can only grow their value creation exponentially by eliminating caps on employee numbers. The simulations have shown that they do so at the expense of the share of value captured. The higher the growth rates targeted by the venture, the lower the share of value captured. While companies can soften the negative impact on the share of value captured through internal improvements, their performance still lags behind companies that cap their employee numbers. Thus, companies may introduce employee caps after periods of growth. Ambitious growth paired with internal improvements pays off especially once the company stops increasing its headcount. To pursue this enquiry, the simulations used two types of influences from the model development chapters. Firstly, the different types of dominant logic, which the model conceptualisation chapter outlined, and secondly, the variables for managerial influences regarding targets and capability development which the model formalisation chapter uncovered. However, model formalisation has also identified contextual influences in the business model design and external environment. The management determines these when launching their ventures
and before growing its business. This section confirms the robustness of the above performance development patterns in different environments and for different business model designs. The section further investigates the influences of these contextual factors on performance development in employee-capped and employee-uncapped companies²⁸.

6.4.1. Different environmental conditions

The model development chapters have identified contextual factors in the external environment that affect performance outcomes directly or indirectly. The development of the model on its detailed, formal level has grouped these environmental influences into four stakeholder groups: customers and users, inputs providers, competitors and providers of substitute products, and providers of complementary products. So far, the model has assumed that exogenous variables related to these environmental groups remain constant at the base case level introduced at the beginning of this chapter. The following simulations investigate how changes in the environment affect the development of value creation and capture. They confirm and assess the performance development in different stable and dynamically changing environments.

6.4.1.1. Stable environments

In strategic management, the industry-structure approach has investigated the impact of environmental conditions on the performance of companies (Porter, 1985, 1991). In growth paths theory, entrepreneurs select their environment by deciding which product to launch in which market. Thereby, they select – although often unconsciously – their target customers. With these customers, entrepreneurs also select the constellation of suppliers, competitors, and complementary products that determine the attractiveness of an environment (Garnsey, 1998; Garnsey and Heffernan, 2005). The first contextual scenarios thus focus on simulating differently attractive environments (Table 6-3). Under *favourable* environmental conditions, the venture was able to identify and target

²⁸ This section does not test the performance development of transitioning companies because they combine the capped and uncapped scenario developments (see Section 6.3).

superior customer segments and enter a market in which it can link its product to complementary products. The model reflects these conditions through a higher maximum usage intensity of customers and a higher rate of feature adoption. Customers also have a longer lifetime, acquire new customers for the venture by recommending its product (i.e. WoM), and the quality of complementary products is higher. The venture faces suppliers with higher bargaining power, stronger competition, and more competitive market conditions in the *hostile* environment. The parameters reflect these conditions through higher input costs and more competitors with superior products.

Variables	Hostile	Base	Favourable
Maximum customer usage intensity	1.5		2
Standard rate of adoption	1		2
Standard customer lifetime	5		8
Rate of customer acquisition per customer	0%		10%
Quality of complementary products	1		2
Costs of each type of input	30%	2	5%
Strength of competitors and substitutes	1.5	1	
Number of competitors and substitutes	1	1 0.5	

Table 6-3: Changes to reflect differently attractive stable environments

While the performance outcomes improve in the favourable environment, the hostile environment erodes performance (Figure 6-13). These findings apply to ventures that cap (top row) and do not cap employee numbers (bottom row). While some variables have shifted the natural performance levels towards which a venture develops, other influences have also shifted the initial level. These simulations confirm the behaviours and patterns identified in the previous section and validate them across differently attractive environments.



Figure 6-13: Effect of stable environment conditions

The simulations of the differently attractive environments show that favourable environmental conditions improve performance for employee-capped and uncapped digital ventures. Hostile environmental conditions have the opposite effect. Thereby, performance development maintains the patterns identified in the previous sections. The scenarios thus validate the findings across differently attractive environments. The employee-capped company develops towards higher levels of value creation and capture under favourable environmental conditions. It develops towards lower levels in hostile environments. Similarly, the employeeuncapped company exhibits faster growth in value creation and higher shares of value captured under favourable conditions. Its growth in value creation and the share of value captured are lower in hostile environments. Due to superior performance in favourable environments, financial resources accumulate faster. They deplete quicker or rise slower in hostile environments. Especially in the hostile environment, the company's financial resources fall below zero. This fall indicates that these companies have gone bankrupt before the end of the 5-year simulation. Injecting more capital into the ventures under these hostile conditions would have allowed them to operate longer without going bankrupt. However, because the companies never develop an ability to capture value, they will ultimately fail and just burn through more capital. Digital ventures should avoid competitive markets and strong competitors to avoid such hostile conditions. They should also rely on low-cost inputs for their activities and link their product to established complementary offers. To further benefit from favourable market conditions that improve performance outcomes, digital ventures should strive to select favourable environments. They should target customers and users who make intensive use of their product, quickly adopt new features, and link their product to established complementary offers.

Tracing the impact of these different environmental conditions in the favourable and hostile scenario on performance outcomes reveals two different types of influences. Some variables that affect performance outcomes directly shift the curves by lifting their initial and maximum levels. Such variables include, for example, input costs. In addition to such direct environmental influences, performance outcomes depend on the resources of the venture. Environmental factors affecting resource stock accumulation cause a change in the maximum level and development speed towards it. However, they do not change the initial performance level. For example, a higher customer lifetime reduces the number of customers lost every year. Ventures targeting customers with higher lifetimes can thus increase their customer base quicker and to a higher level. This improved targeting increases their natural performance level and the speed of development towards it. Variables related to competition combine both types of influences. They affect performance outcomes directly and via resource stock accumulation. With higher levels of competition, the venture captured less value from customers while its customer base also grows slower.

6.4.1.2. Dynamics environments

A critical strength of simulation modelling as a research tool is eliminating confounding factors to isolate mechanisms of interest (Davis *et al.*, 2007; Harrison *et al.*, 2007). For example, the model isolates and investigates how

performance would develop based on internal changes while holding the environment constant and assuming it is perfectly stable. The previous sections' simulations and the above favourable and hostile scenarios assume such constant and stable environments. While these are theoretically relevant to identifying internal changes affecting performance outcomes, business environments are dynamically changing. Therefore, two scenarios with changing environments have also been developed to increase the realism of simulations (Table 6-4). In the *improving* environment, customer variables and complementary qualities improve from their base case to the favourable levels over time. This scenario represents environments in which the target customers become increasingly familiar with digital technologies, for example, by having seen similar technologies and applications before. The scenario reflects that providers of complementary products launch and improve their complements over time. In the worsening environment, competition and input costs increase over time to hostile levels. This scenario reflects environments in which a venture has discovered an initially attractive environment. However, over time, this attractiveness also entices competitors to enter the market and suppliers to raise their prices to capitalise on the venture's success. The model reflects these improving and worsening conditions by setting the initial and end value of the variables. The initial values are the base case values introduced at the beginning of this chapter. The end values are the favourable and hostile scenario values illustrated above. The model uses linear interpolation between these initial and end values.

Variables	Hostile	Base	Favourable	
Maximum customer usage intensity	1.5		1.5 -> 2	
Standard rate of adoption	1		1 -> 2	
Standard customer lifetime	5		5 -> 8	
Rate of customer acquisition per customer	0%		0 -> 10%	
Quality of complementary products	1		1 -> 2	
Costs of each type of input	25% -> 30% 2		5%	

Table 6-4: Changes to reflect different environmental dynamics²⁹

²⁹ For improving and worsening conditions, the left and right values reflect, respectively, the value at the beginning and end of the simulation. For example, the term "5 -> 8" for customer lifetime represents that the customer lifetime is five years at the beginning of the simulation. At the end of the simulation, it is eight years. During the five years of simulation, the model interpolates between these two values linearly. For example, after one year of simulation, the customer lifetime reflection was 5.6 years.

Strength of competitors and substitutes	1 -> 1.5	1
Number of competitors and substitutes	0.5 -> 1	0.5

In improving and worsening environmental conditions, the behavioural patterns for companies with and without caps on employee numbers change (Figure 6-14). Improving conditions lead to, for example, exponential growth in value creation even for companies with caps on employee numbers. Worsening conditions lead to a deterioration in value creation and capture over time. However, while these developments look different from the patterns discussed in the chapter so far, they can be explained through them.



Figure 6-14: Effect of environmental dynamics

In changing environments, such as the improving and worsening scenario, the patterns of performance development seem fundamentally altered. The improving environment leads to exponential growth in value creation in companies that cap and do not cap their employee numbers. Improving environmental conditions prevent the balancing feedback loops causing goal-seeking development from becoming dominant. For example, at a constant maximum usage intensity, each customer will eventually develop to this maximum. However, with this maximum level increasing in the improving scenario, customers can consistently increase their usage intensity further. In the worsening environment, the rate of value creation grows at a lower rate for the employee-uncapped company because the management continuously overestimates its employees' productivity. All other performance outcomes peak about halfway through the simulation before deteriorating again or about remaining stable. The venture's productivity improvements (and higher inputs in WorseningU) allow it to initially improve the rate at which it acquires customers and improves its technology. However, when no further productivity improvements are possible, the venture loses against everimproving competitors and increasing prices for input resources. Like the hostile scenario, companies in the worsening scenarios may not develop an ability to capture value and go bankrupt. Additional capital would have expanded the lifetime of these ventures. However, they will continue to burn through their capital and never turn profitable as their share of value captured continues to erode.

While these performance development patterns look different from those identified in this chapter already, the established patterns can explain and account for them. One can separate the dynamically changing environments into a series of different environments. The scenario set-up included beginning and end values for inputs with linear interpolation between these values. One can also interpret this interpolation as creating a unique environment for every step of the simulation. As the investigation of differently attractive but stable environments shows, each of these different environments has a different performance level. These levels follow the patterns outlined throughout this chapter. Thus, the improving and worsening scenario for employee-capped companies reflects goal-seeking development towards ever-improving and ever-decreasing natural performance levels. Similarly, the two scenarios for employee-uncapped companies reflect a series of developments towards ever-increasing or ever-decreasing shares of value captured.

It is also to note that performance in the improving and favourable scenarios and the hostile and worsening scenarios will converge once the environmental conditions are constant. Comparing the graphs in the two subsubsections, readers can identify that the value creation and the share of value captured graphs of the favourable and improving scenarios move towards one another. The same applies to the worsening and hostile scenarios. Thus, if the final environment is the same, the performance of companies with the same characteristics will eventually be the same. However, due to initial differences in value creation and capture, financial resources will maintain a difference. When performance outcomes have converged, the rate of cash generation of both companies is equal. However, because of superior value creation and capture in the favourable and worsening environments, financial resources get a head start compared to the improving or hostile environments. Thus, companies should seek to identify initially attractive environments rather than striving to make an industry attractive. They should also strive to slow down adverse environmental changes such as increasing competition or input costs.

6.4.2. Different business model designs

Besides the environment, the business model design has been identified as a contextual element determined before digital ventures scale up their product. A range of value drivers, mechanisms through which digital ventures create value for customers and users, have been identified during model formalisation. Similarly, the model includes variables that convert capabilities from their qualitative, soft scales between zero and one to impacts on the model's hard variables. As the model inputs for the base case at the beginning of this chapter illustrate, simulations assume the business model design of a SaaS company. Effect sizes have been set to ensure simulations are not affected by surges in hiring or layoffs just after the beginning of the simulation. However, companies may design different business models, which may impact performance development. Thus, the development patterns observed in this chapter require validation regarding active value drivers and transformational effect sizes. This subsection confirms the robustness of development patterns regarding these two

contextual elements. It provides further insight into the development of value creation and capture for different business model designs.

6.4.2.1. Business models with network effects

Model formalisation identified different value drivers for digital ventures (Chapter 4). The model validation chapter (Chapter 5) illustrated how companies might use these value drivers in practice. It distinguished and tested two different types of business models. Firstly, SaaS business models (Alpha and Beta) that rely on their technology as a driver of use value creation. Secondly, digital ventures which also rely on network effects to create customer value (Gamma and Delta). The detailed model development has further distinguished direct, indirect, positive, and negative effects. So far, the simulations in this chapter have assumed the business model of a SaaS venture that relies solely on its technological quality. Thus, the robustness of identified development patterns needs to be confirmed for network effects. Two scenarios have been developed and compared to the base case with a SaaS business model. In both new scenarios, the ventures distinguish between paying customers and non-paying users (Table 6-5). The first scenario is set up to investigate a social *network*'s performance development, in which value is created for users based on positive direct network effects. Customers who pay for the product, for example, advertisers, benefit through indirect network effects from the user base. However, there are negative direct network effects between customers as more advertisers increase the competition for users on the platform. The second scenario represents a *marketplace* that brings buyers and sellers together. Through positive indirect network effects, both groups benefit from one another. However, they are adversely affected by negative direct network effects.

Variables	Base	Network	Marketplace	
Initial customer base	1	.5	.5	
Initial user base	0	.5	.5	
Target user growth rate	0	25%	25%	
Customer direct network effect settings	0	-1	-1	
Customer indirect network effect settings	0	1	1	
User direct network effect settings	0	1	-1	
User indirect network effect settings	0	0	1	
effect of competence on ability to acquire	0	1.5	15	
users	Ū	1.5	1.0	
effect of competence on ability to service	0	2	2	
users	, v		_	

Table 6-5: Changes to reflect business models with network effects

The changing business models employing varying value drivers show the same development patterns illustrated throughout this chapter (Figure 6-15). The employee-capped scenarios (top row) show goal-seeking developments of performance outcomes. The employee-uncapped companies (bottom row) show exponential growth in value creation at the expense of the share of value captured. However, differences in the growth rates and shares towards which the performance outcomes develop can be observed, which are explained below.



Figure 6-15: Effect of business models with network effects

A first difference regarding the business model design is the increase in the rate of value creation for social networks that employ mostly positive network effects. The marketplaces and base cases do not exhibit this increase in the rate of value creation. The increased rate of value creation for the network is explained by employing more positive than negative network effects. Thereby increasing user and customer bases affect value creation in two ways. The venture creates value for more users and customers and it increases the value created per user and customer. This second effect is missing in SaaS and marketplace business models. While customers in a SaaS business model do not directly benefit from other customers using the platform, positive and negative network effects cancel out for marketplaces. A second difference regards the lower share of value captured for companies that serve customers and users. The lower share of value captured for networks and marketplaces can be explained by these business models creating value for users and customers. However, they only capture value from customers while users receive the product for free.

6.4.2.2. Business models with different efficiency levels

Business models also differ regarding the variables that convert capabilities from their soft, qualitative scale to a hard impact on the system. The effect sizes of the base case were set to balance the venture and avoid surges in hiring and layoffs at the beginning of the simulation. In reality, these effect sizes represent an employee's output with fully developed capabilities and thus the best-practice throughput of activities (Warren, 2002). However, digital ventures may design their activities differently and, therefore, have different best-practice throughputs. Three scenarios investigate the impact of changing these effect sizes (Table 6-6). Effect sizes have been lowered by 50% and 20% and increased by 20%.

Table 6-6: Changes to reflect different productivity levels

Variables	Lower50	Lower20	Base	Higher20
effect of competence on ability to develop technology	0.2	0.32	0.4	0.48
effect of competence on ability to maintain technology	2	3.2	4	4.80
effect of competence on ability to acquire customers	0.75	1.2	1.5	1.8
effect of competence on ability to service customers	1	1.6	2	2.4
effect of competence on ability to manage the firm	4.5	7.2	9	10.8

The venture with higher effect sizes maintains the development patterns of value creation and capture observed throughout this chapter (Figure 6-16). It develops in a goal-seeking manner towards higher natural performance levels when capping its employee numbers. It also exhibits similar growth in value creation but develops towards an improved share of value capture when it does not cap employee numbers. The venture with 20% lower effect sizes also follows these patterns but with development towards lower performance outcomes. However, the venture with a 50% reduction in effects sizes diverges from these development patterns. Its rate of value creation develops towards zero, while its share of value captured falls throughout the simulation and does not seem to develop towards a specific value in a goal-seeking manner. While this development pattern looks different on a superficial level, it can be explained through the mechanisms outlined throughout this chapter.



Figure 6-16: Effect of business model's efficiency levels

Ventures with higher effect sizes will develop towards higher natural performance levels for both outcomes when capping their employee numbers. They develop towards a higher level of the share of value captured when not capping their employee headcount. These patterns are like those identified when investigating capability levels in the previous sections. For employee-capped companies, the same number of employees can achieve more. They thus increase value creation and the share of value captured at constant absolute costs. For employee-uncapped companies, fewer employees can achieve the venture's targets. Thus, the venture increases its share of value captured at a similar rate of value creation. In either case, the improved rate of value creation and share of value captured leads to a quicker financial resource accumulation.

However, with lower capabilities, the rate of value creation and share of value captured may deteriorate over time for employee-capped and -uncapped

companies. Under conditions in which the management is firefighting and has no time to implement growth and improvement plans, the rate of value creation falls. Superficially, this looks like a performance development pattern different from those discussed throughout this chapter. However, like different patterns in dynamic environments, the deterioration can be fully explained by the patterns already discussed in this chapter. For example, goal-seeking patterns for both performance outcomes were expected for employee-capped companies. However, it has not been stipulated that companies improve towards their natural performance levels. Just as in worsening environments, company performance may also deteriorate to their natural level. Similarly, companies with layoffs may fall towards their natural rate of value creation. With smaller effect sizes like in the above scenario, the companies may have a natural rate of value creation of zero. This indicates that the venture has developed a business model that is not sustainable and will never turn profitable (even when competition is not increasing). In such cases, the development in the rate of value creation is still goal-seeking but deteriorates to its goal of zero rather than improving towards a higher rate. This development towards zero also explains the development of the share of value captured. As outlined in the first section of this chapter, employeecapped companies have constant absolute costs. Expressed as the fraction of an increasing rate of value creation, the share of value captured increases too. However, the constant amount of costs increases when expressed as a fraction of decreasing value creation. Thus, this fall is consistent with the theory outlined throughout this chapter and presents a special case of it.

6.5. Implications to the case firms

The simulations in this Chapter elaborate on development patterns observed in the case companies. They explain the development patterns of the case companies without interference from noise in data like the sale of assets. Moreover, they provide a basis to evaluate companies' different policies regarding their goals, contextual factors, and restructuring activities.

The simulations show that companies are right to pursue growth and technological improvements simultaneously. As the simulation of different

targets shows, the combination is superior for growing companies. On the one hand, technology improvements increase the share of value captured. On the other hand, ventures can spread development costs and amortisation over more customers. Therefore, for example, Alpha is correct to start pursuing such improvements in its final growth state. However, it should have pursued improvements throughout the years as Beta, Gamma, and Delta did.

The companies are also correct to transition away from dominant logics without caps on employee numbers. Instead, introducing caps on employee numbers allows the processes that improve value creation and capture to improve both performance outcomes simultaneously. The simulations of such transitions show that they accelerate financial resources accumulation. While growth will slow down, this transition may allow the companies to develop an ability to capture value. Alpha is a primary example of this development. Once it transitions, its marketing and selling employees remain nearly constant for more than four years. During this time, its share of value captured reaches its natural performance level and turns nearly positive, while it takes further action to improve its value creation (see below).

However, unless necessary to sustain themselves, companies should not restructure. Instead, they should have maintained their employee numbers. The simulation of restructuring shows that restructuring can quickly improve the share of value capture. However, companies do so at the expense of their long-term performance. Their lower inputs may not allow the companies to maintain their established size. Therefore, value creation may decrease over time, and they may give up established economies of scale. This outlook raises the question, why executives felt the need for restructuring. An explanation might be that there might not have been enough time for the dynamic processes to improve the share of value captured. Executives might have felt that the company would go bankrupt if it did not quickly improve profitability, i.e. the share of value captured. Executives thus require more foresight. Rather than growing and giving up a part of their venture's size potential, they should have foreseen financial difficulties and started transitioning earlier without restructuring. The model developed in this thesis might help future managers and entrepreneurs to do so. They may use

the model to approximate company developments and act earlier. The model may help them to measure performance in the moment without waiting for accounting reporting cycles (Subsubsection 4.1.5).

The case companies should also take additional actions to develop an ability to capture value. An example of such actions is Alpha. The company has taken additional steps to optimise its business model by switching market segments. They identified a superior customer segment during its growth process. The company has done well to optimise its contextual factors. It has grown its value creation and improved its share of value captured significantly, thereby nearly reaching an ability to capture value, despite lowering its target customer growth rate. However, simulations show that Alpha should have pursued these more attractive customers from the outset. Thereby, it would not have spent time and money on acquiring a less attractive customer segment. Nevertheless, Alpha excels in optimising contextual factors compared to the other companies. Beta's, Gamma's, and Delta's developments, even with caps on employee numbers, are worrying because restructuring has not allowed the company to develop an ability to capture value. Instead, they remain loss-making. These losses indicate that the companies have not optimised the contextual factors that determine the trajectory of their performance development. Therefore, their natural share of value captured toward which they develop is negative, and the companies can only survive through external funding. Managers and entrepreneurs of Beta and Gamma might have sold their respective companies due to this incapacity to develop an ability to capture value. Delta, which remains independent, needs to select a more attractive environment, develop a more efficient business model, and pursue internal improvements to develop superior capabilities. These are requirements for the development of an ability to capture value sustaining the venture.

Chapter 7: Discussion

This thesis aims to develop an understanding of the requirements and priorities allowing digital ventures to develop an ability to capture value while growing their value creation. This discussion first reviews the influences on value creation and capture. It then proposes a framework outlining the main propositions for digital ventures' ability to grow their value creation and capture value. The introduction of this thesis has proposed two research questions regarding the development and relationships of value creation and capture for digital ventures. The final two sections of this discussion illustrate the implications of the previous chapters for the two research questions, compare the findings to previous research, outline implications for theory and practice, and link them to the framework's elements and relationships.

7.1. Influences on value creation and capture

This thesis has synthesised growth paths theory and dynamic states theory, adapted the theories for the context of digital ventures, and focused them on value creation and capture. Thereby, this thesis captures the development of companies and their performance comprehensively.

The developed model highlights that performance outcomes, resources, and capabilities change continuously. At any point in time, value creation and capture depend on the venture's resources and external environment (Sections 3.2.1, 4.1, 4.2). These resources themselves continuously change through activities and capabilities (Section 3.2.2). The formal model considers this continuous resource accumulation in separate subsystems. In the technology development subsystems, the venture's technological quality improves along the technology s-curve (Section 4.3). In the customer and user subsystem, the customer and user bases develop based on customer and user acquisition and churn (Section 4.4). In the firm management subsystem, the venture's growth and improvement goals (Section 4.5). Technology development, marketing and servicing customers and users, and firm management require employees and depend on their

capabilities. These capabilities continuously improve through learning and deteriorate with new hiring (Section 3.2.3). Thereby, this thesis reflects relationships considered in the literature on value creation and capture (Teece, 1986; Lepak, Smith and Taylor, 2007), the resource-based view (Grant, 1996; Amit and Zott, 2001), the industry-structure approach (Porter, 1985; Zott and Amit, 2010), and capability development (Slater, 1980; Dosi, Nelson and Winter, 2000). Bringing these relationships together, this thesis can reflect the continuous growth process illustrated in growth paths theory, particularly its resource and capability accumulation (Garnsey, 1998; Stam and Garnsey, 2006). The continuous changes in value creation, capture, and resources can also be observed for case companies and during scenarios simulations. Within a growth state, all case companies show continuously changing performance outcomes and resources during model testing (Chapter 5). Performance outcomes and financial resources also change dynamically during scenario simulations (Chapter 6).

These continuous development processes are affected by contextual influences regarding the environment, business model design, capability development, and the management's dominant logic (Section 3.3). The detailed, formal version of the model considers exogenous variables regarding each of those elements in its subsystems (Chapter 4). They reflect the discontinuous nature of the growth process captured in dynamic states theory (Levie and Lichtenstein, 2010; Brown and Mawson, 2013). Thereby, the model can reflect that some variables remain constant during one or more growth states and change with transitions between states. The punctuated changes in these variables when transitioning between growth states have been observed and confirmed for case companies (Chapter 5). Simulating growth states and transitions has shown that these variables, through their impact on feedback loops, steer the venture's resource and capability accumulation and thereby alter its value creation and capture development (Chapter 6).

7.2. Implications for the ability of digital ventures to growth value creation and capture value

Figure 7-1 reflects the implications of the model's structure and behaviour for digital ventures' growth in value creation and ability to capture value. It distinguishes between two types of growth states: scaling up and exploiting size. These states differ regarding a contextual influence reflecting a venture's growth state and the behaviour of performance outcomes. On the one hand, companies in the scaling up state employ a dominant logic reflecting their willingness to hire more employees to achieve their growth and improvement goals. On the other hand, companies without that willingness exploit their current size. As outlined above, these two types of dominant logic and growth states affect the continuous change among business model elements and performance outcomes. While scaling up, ventures experience exponential growth in value creation. While exploiting size, they exhibit goal-seeking development in value creation. In both growth states, the share of value captured develops in a goal-seeking manner.

However, everything else being equal, companies scaling up develop towards lower shares of value captured. Moreover, the higher the targeted growth rate, the lower the share of value capture. Thus, digital ventures need to prioritise carefully between growth in value creation and their ability to capture value. Companies wanting to bootstrap need to carefully balance growth rates and their ability to capture value by reducing growth goals to maintain a positive share of value captured. If growth rates are so high that companies lose resources during the growth process, they need to raise venture capital. Theory points to a third type of scaling up, acquisitive growth (Lockett, Wiklund and Davidsson, 2007; Penrose, 2009). This thesis has not considered acquisitions, presenting an opportunity for future research. Digital ventures exploiting their current size may also develop towards different value creation and capture levels. For example, mature ventures that have previously scaled up developed economies of scale. These allow them to develop towards higher value creation and capture levels than lifestyle companies that have not previously scaled up. Lastly, restructuring ventures reduce their employee numbers after scaling up and so can quickly develop an ability to capture value. However, they do so at the expense of their long-term value creation and capture.



Figure 7-1: Growth process framework

Besides growth targets and their size, the other contextual factors affect a digital venture's rate of value creation and capture. These factors - regarding the venture's environment, business model design, capability development, and managerial targets – determine the trajectories of development patterns in either growth state. To optimise the development of value creation and capture, digital ventures may thus alter those factors. They present the requirements to grow value creation and develop an ability to capture value. When exploiting their size, digital ventures can create the conditions for ultimately higher rates of value creation and capture through these contextual factors. When scaling up, digital ventures can optimise them to lower the adverse effect of growth on value capture and do better at both outcomes simultaneously. However, environmental and business model design factors may present a challenge for reversal. Striving to grow in an unsuitable environment or with an unsuitable business model may drain resources and create lock-in and technological debt (Marmer et al., 2011; Marion et al., 2015; Ramadan et al., 2016; Quinones, 2017; Zaheer et al., 2019; Zaheer, Breyer and Dumay, 2019). Therefore, companies should get these two contextual factors right when establishing the firm and before progressing to one of the other two growth states. On the other hand, digital ventures can develop their internal operations, parts of their environment, and technology when operating. They should thus pursue these improvements continuously to further optimise performance outcomes.

The developed framework provides a growth process framework for managers and entrepreneurs. They may develop their ventures through its three stages iteratively while pursuing continuous improvements. Thereby, it avoids the criticism of OLC stage models and builds on and expands contemporary growth process theories. The framework does not assume that companies develop through these states in a particular order. Instead, companies may stay in a state or progress through them even circularly. For example, companies may return to the establishment state by diversifying and introducing new products, which have not been investigated in this thesis and present another area for future research. Moreover, the framework acknowledges that some activities should be pursued regardless of the growth state. Through these two characteristics, the framework addresses common criticism on organisational life-cycle theory (see Davidsson, Achtenhagen and Naldi, 2010). The framework also adds to growth process theories by aligning growth states with performance development patterns regarding value creation and capture. Thereby, the framework captures a comprehensive theory of growth that connects state characteristics with size changes (Penrose, 2009), while adding value capture as an additional performance outcome. This framework is valuable for practitioners because it illustrates the development patterns of their venture's performance in each state. It captures the answers to the two research questions of this thesis. These answers are reviewed below, linked to the framework and the previous chapters and compared to existing literature, with further implications for theory and practice identified.

7.3. How do value creation and capture develop for digital ventures?

This thesis has simulated the System Dynamics model reflecting the above influences on value creation and capture. Different scenarios have been created and evaluated to reveal the drivers of diverse performance development paths. Reviewing these different paths illustrates that the two different types of dominant logic determine development patterns, while contextual variables determine the trajectories within a pattern. The two subsections below review the findings leading to these two elements, answering how value creation and capture develop for digital ventures. The subsections compare the findings to previous research and outline implications for theory and practice.

7.3.1. Dominant logics determine development patterns

Reviewing model conceptualisation, formalisation, case simulations, and scenario simulations, two different types of performance development patterns can be identified. These reflect the part of managers' and entrepreneurs' dominant logic that captures if a venture is willing to hire more employees or is operating at its cap on employee numbers. These two types correspond to the scaling up and exploiting size growth states in the above framework (Figure 7-1). Model conceptualisation identified these caps on employee numbers as one element of the dominant logic (Subsection 3.3.2). The chapter hypothesised that these might also limit value creation and the share of value captured for companies operating at their cap on employee numbers. However, the chapter could not predict the development of value capture of companies without a cap on employee numbers (Subsection 3.4.3). Therefore, the formal model, which allows simulations and the observation of performance development, gives consideration to the dominant logic. It influences hiring and fundraising decisions in the firm managing subsystem (Section 4.5). Scenario simulations of the formal model show that companies operating at their cap on employee numbers can grow their value creation and improve their share of value captured up to a maximum in a goalseeking manner (Section 6.1). Companies below or without a cap on employee numbers can grow value creation exponentially. They still develop towards a share of value captured in a goal-seeking manner (Section 6.2). Simulations reveal these two types of behaviour across different contexts such as business model designs and environments (Section 6.4). Thus, a venture's type of dominant logic determines its development pattern. This development is also observable for case companies. A prime example is Alpha, which share of value captured develops in a goal-seeking manner over the observation period (Subsection 5.5.1).

These findings align with previous research on the importance of management's growth intentions and growth paths theory. Research on management's growth

intentions and outcomes has previously argued that "if an entrepreneur chooses to grow his or her business it is likely that growth will actually occur" (Davidsson, 1991; Delmar and Wiklund, 2008; Davidsson, Steffens and Fitzsimmons, 2009; Kirkwood, 2009, p. 485; Cassia and Minola, 2012; Hesse and Sternberg, 2017). This thesis aligns with this view as a manager's type of dominant logic reflects a willingness to increase the scale of operations needed to support growth in value creation. It adds that companies may grow, although at decreasing rates towards their natural performance levels, even when they do not strive to grow their rate of value creation. Thereby, this thesis replicates and explains two patterns observed in growth paths theory which has identified a range of growth paths. The growth of value creation in the framework's two growth states reflect growth paths theory's continuous and plateaued growth (Garnsey and Heffernan, 2005; Garnsey, Stam and Heffernan, 2006). Growth paths theory's other patterns can be reduced to these two patterns. For example, growth setbacks could be observed in worsening environments but have been explained by development towards everdecreasing performance levels (see Subsubsection 6.4.1.1). Similarly, validating and simulating for other types of companies may show delayed growth, which is particularly common, for example, among biotechnology companies with long development cycles (Garnsey, 1998). Samples of digital ventures before their market launch in the ideation stage may also show such behaviour. In addition to the development of value creation, this thesis has also revealed the goal-seeking development pattern of the share of value captured within the framework's two growth states.

These insights have implications for theory and practice. Firstly, theory needs to consider the dominant logic as the primary driver of company growth. Without the knowledge of an entrepreneur's growth ambitions, a theory cannot determine if growth was unfeasible or not desired. Secondly, researchers need to determine where on its development path a venture is to understand if performance changes are driven by to goal-seeking mechanisms or managerial decisions. Lastly, entrepreneurship theory, which considers growth the primary outcome of companies, should also consider value capture. As this thesis illustrates, growth is driven primarily by intentions (if capital is available to sustain the firm). Thus, the share of value captured can indicate if such growth is attractive for a venture.

This is also applicable to managers and entrepreneurs of digital ventures. The two growth states in the above framework that depend on their dominant logic (Figure 7-1) provide clear expectations of their performance development. Like researchers, they should not consider growth as the sole outcome of their venture. It is a managerial choice. Rather, they should also consider their venture's ability to capture value and expect it to develop in a goal-seeking manner regardless of the growth state. Therefore, they must ensure that the share of value capture towards which they develop is positive, i.e. create the requirements allowing the development of an ability to capture value This requires looking at the different trajectories within the two growth states (Subsection 7.3.2) and the penalty incurred for growth on value capture (Subsection 7.4.1).

7.3.2. Contextual factors determine trajectories and levels within patterns

While a company's dominant logics determines its performance development pattern, this thesis has illustrated the diversity of performance outcomes within the framework's two growth states (Chapter 6). These differences depend on the model's exogenous, contextual factors. Model conceptualisation (Section 3.3) and formalisation (Chapter 4) grouped these variables. They regarding the environment, business model design, capability development, and specific targets of the dominant logic. Simulations of case companies have shown that companies differ diversely regarding these elements and change contextual variables during state transitions (Chapter 5). Scenario simulations illustrate the influence of the dominant logics' targets and capability development, which the management can influence during the growth process (e.g. Subsections 6.1.2, 6.1.2). They also illustrate the impact of environmental factors and business model designs, which are hard to reverse once selected (Section 6.4). The simulations have shown that these contextual variables have different effects depending on the growth states of the framework. They affect both natural performance levels towards which companies develop when operating at their cap on employee numbers in the exploiting size state (Section 6.1). Thus, companies are required to create the conditions for ultimately superior value creation and capture through the contextual factors. For companies operating below or without employee caps in

the scaling up state, they determine the value creation growth rate and the level towards which the share of value captured develops (Section 6.2). Thereby, ventures can use the contextual factors to balance the two performance outcomes and create the requirements to grow at higher rates while maintaining an ability to capture value.

These contextual factors can account for the diversity in performance outcomes observed in empirical studies. The factors that determine these levels are in line with previous findings on company performance about the importance of the external environment (Porter, 1985, 2008; Romanelli, 1989; Zhang, Lichtenstein and Gander, 2015), business model designs (Amit and Zott, 2001; Zott and Amit, 2007, 2010; Teece and Linden, 2017; Täuscher and Abdelkafi, 2018), human resource management and capabilities (Nelson and Winter, 1982; Porter, 1996; Baron and Hannan, 2002), and the impact on managerial targets (Davidsson, 1991; Delmar and Wiklund, 2008; Kirkwood, 2009; Cassia and Minola, 2012; Hesse and Sternberg, 2017). This thesis was able to combine these separate streams of research into one model. Thus, it provides a holistic and systemic account of performance development consistent with previous research. The influence of these factors on performance levels implies that theory needs to consider all of them systematically too. Without considering all factors, firm development may not be fully accounted for. Managers must select and improve the contextual factors to create the requirements and conditions allowing their ventures to develop an ability to capture value while growing value creation. They do so when establishing their ventures and through continuous improvements, reflecting the two additional elements in the above framework.

When establishing their companies, managers and entrepreneurs must select appropriate environments and design effective business models. Due to their difficulty in reversing these factors, managers should select their environment, design their business model, and validate their selection before scaling up (Marmer *et al.*, 2011; Marion *et al.*, 2015; Ramadan *et al.*, 2016; Quinones, 2017; Zaheer *et al.*, 2019; Zaheer, Breyer and Dumay, 2019). Thus, the framework considers these factors in its establishment state before transitioning to scaling up or exploiting its size (Figure 7-1). Simulating the model has shown that customers using the product intensively, developing higher switching costs, and recommending the product led to superior performance outcomes. Moreover, digital ventures should strive to link their technology to complementary products. With its target customer segments, a venture also selects the competition for these customers, which it should strive to minimise. Common tools from strategic management may help entrepreneurs to do so, including, but not limited to, Porter's Five Forces, strategic group analyses, or industry life-cycles (Porter, 1985, 2008; Johnson, Scholes and Whittington, 2008; Grant, 2016; Aaker and McLoughlin, 2017). Selecting an environment, they should strive to identify blue oceans, i.e. uncontested market spaces with unmet customer needs (Kim and Mauborgne, 2005, 2017). Moreover, the venture should strive to design business models with low input costs and efficient activity systems. Digital entrepreneurs should validate their selection and design before transitioning to one of the other two growth states of the framework. They should explicate their assumptions and simulate their venture's development using the model developed in this thesis.

Digital ventures also need to pursue continuous improvements and alter their performance development through the magnitude of their targets while operating their business. Improving internal operations to accelerate capability development (e.g. improving employee training and turnover), external improvements (e.g. linking to complementary producers or incentivising customers to recommend the product), and technology development all positively affect the ability to capture value. For scaling up ventures, they soften the adverse effect of growth on value capture. For ventures exploiting their current size, they increase the levels of value creation and capture towards which the venture develops. Thus, such continuous improvements should be pursued in both of the framework's two growth states (Figure 7-1).

7.4. What relationship(s) exist(s) between value creation and capture?

This thesis has also modelled the complex relationships driving value creation and capture, and their connecting feedback mechanisms. The simulations in this thesis show that the impact of growth in value creation on the share of value captured changes over time. They show that growth in value creation adversely affects value capture during periods of growth but has minor positive effects after periods of growth. The two subsections below review the evidence of this thesis for these insights, compare them to previous literature, outline their implications for the theory and practice, and link them to the above framework.

7.4.1. Growth in value creation adversely affects the share of value captured during periods of growth

The simulations of the model developed in this thesis have revealed two tradeoffs between growth in value creation and the share of value captured during the framework's scaling up growth state. Firstly, ventures growing without or below their cap on employee numbers develop towards lower levels of value captured (Subsection 6.2.1). The higher the growth in value creation, the lower the share of value captured towards which the venture develops (Subsection 6.2.2). These lower levels have been explained by continuously mobilising resources from the external environment. These include, for example, new employees that lack firmspecific knowledge or new customers that are yet to develop switching costs. These influences are visible during model conceptualisation (Subsection 3.2.3) and each subsystem (e.g. Subsection 4.3.3), in which new hiring to reach goals reduces capabilities. Secondly, the increase in growth targets causes a temporary fall in the share of value captured (Subsection 6.2.2). This temporary fall can be explained by differences between inputs to the growth process and its outcomes. The venture requires more employees to achieve its more ambitious targets. As soon as it employs these, it incurs their associated cost, as conceptualised (Subsection 3.1.2) and formalised in the value capture subsystem (Section 4.2). However, it takes time until these new employees mobilise and establish the resources that create value.

These findings confirm some of the previous research on performance outcomes (e.g. Hambrick and Crozier, 1985; Nicholls-Nixon, 2005; Stam and Garnsey, 2006). The introduction has outlined literature finding positive, negative, and no impact of growth on profitability. During periods of growth, this thesis is thus aligned with the literature finding negative effects of growth. It has incorporated

the negative impact of growth on capabilities highlighted in growth paths theory into the model developed in this thesis. In addition, the model was able to show the impact of these lower capability levels on the share of value capture. It has also found similar mechanisms lowering customers' average usage intensity and switching costs. Superficially, these results conflict with the literature highlighting the benefit of growth, including economies of scale, network effects, and learning effects (Katz and Shapiro, 1985; Chandler and Jansen, 1992; Davidsson, Steffens and Fitzsimmons, 2009). As the next subsection illustrates, some of these benefits exist. However, they are outweighed by the negative effects that counteract them during the scaling up growth states. Moreover, not all benefits of growth outlined in the literature have been found. For example, learning effects could not be identified as an outcome of growth as new employees need to internalise this knowledge before becoming productive (Section 3.2.3). For theory, this implies that learning should be considered on an individual level rather than on an organisational level (Garnsey, 1998; Penrose, 2009).

The adverse effects of growth in value creation have implications for managers striving to grow their value creation and develop an ability to capture value. Firstly, they should strive to minimise the negative impact of growth when establishing their venture and through the continuous improvements in the framework (Figure 7-1). Thereby, they create the requirements to capture value while increasing their growth in value creation. Secondly, managers must carefully prioritise growth rates and the share of value captured. The higher the growth rates, the higher the temporary fall in value creation and the higher the funding required to sustain the venture. Entrepreneurs may thus need to raise sufficient capital to sustain periods of growth or reduce their target growth rates to sustainable levels, reflecting the venture capital-funded and bootstrapped companies in the scaling up state of the framework.

7.4.2. Growth in value creation has minor positive effects on the share of value captured after periods of growth

Rather than having a positive impact during the period of growth, growth seems to pay off after periods of growth. Simulations show this by exploring transitions between growth states with a dominant logic without caps on employee numbers to a dominant logic with a cap on employee numbers (Subsection 6.3.1). Thus, these companies are transitioning from scaling up to exploiting size states of the above framework (Figure 7-1). These simulations also show that the higher the growth targets during the growth period, the larger the benefit after growth (Subsection 6.3.2). These benefits take two forms. Firstly, growth scales up the venture and enables it to develop towards a higher natural rate of value creation when employees are maintained (Subsection 6.3.3). Secondly, growth creates economies of scale that slightly increase the share of value captured. Model conceptualisation shows that not all costs develop with the venture's user and customer base and the venture can spread, for example, development costs and amortisation over more customers (Subsection 3.2.2). Together these two mechanisms accelerate the accumulation of financial resources. Due to economies of scale, scaling up value creation through the customer and user base positively affects a venture's ability to capture value. However, these effects are minor compared to the overall development of the share of value captured in simulations, which companies achieve through, for example, technology development and improving usage intensities.

Researchers were right to propose that growth has a positive effect (e.g. Katz and Shapiro, 1985; Chandler and Jansen, 1992; Davidsson, Steffens and Fitzsimmons, 2009). The more nuanced approach of this thesis that acknowledges temporality and resource stock accumulation was required to understand when and how these positive effects occur. Once the managerial challenges of scaling up are no longer an issue, companies can reap its benefit through economies of scale. However, while this thesis could identify economies of scale, they only had small effects. This might be a special characteristic of the digital context. Firstly, the asset lightness of digital ventures and the ease of establishing them may explain the difference which was hardly significant. Because digital ventures do not hold many depreciating assets and technology can be created easily (Kraus *et al.*,

2018), there are few fixed costs to spread over more customers. Secondly, variable costs may be higher than practitioners expect. Digital entrepreneurs may claim that their products can be distributed at zero marginal costs, thus growing themselves profitable (Staykova and Damsgaard, 2015; Teece and Linden, 2017). However, digital ventures still require significant marketing, servicing, and managerial staff that are driven by the number of customers and users. During model validation (Chapter 5), the case companies show that digital ventures incur significant costs to acquire customers, service them, and manage these marketing and servicing employees. Such employee requirements have already been identified for e-commerce companies pursuing high growth strategies (Oliva, Sterman and Giese, 2003; Sterman *et al.*, 2007).

Together, the adverse effects of growth on the ability to capture value during growth periods and the positive effects after growth periods have implications for theory. Theory needs to consider size and growth conceptually different and acknowledge their temporal relationship. While size accumulates over time and positively affects the ability to capture value, growth as the change in size over a period has a negative impact. The minor positive impact of size implies that researchers need to determine not just if a statistically significant relationship exists but also the size of its effect. A positive relationship due to prior growth should be expected. However, other variables should be altered to develop an ability to capture value. The benefit of growth through a larger size with only minor improvements in the share of value captured has implications for managers striving to develop an ability to capture value. Contradicting the general views of practitioners that one can grow a digital venture into profitability, managers should scale up to exploit an already working business concept rather than make a business concept work through scaling up. After growth, they can reap its benefits by transitioning to the framework's exploiting size state.

Chapter 8: Conclusion

This thesis has investigated two research questions to understand the requirements and priorities allowing digital ventures to develop an ability to capture value while growing their value creation. This conclusion reviews if the methodology has achieved the overall aim, outlines the contributions of this thesis to theory and practice, and proposes areas for future research.

8.1. Review of the research process

This thesis has integrated growth paths theory (Garnsey, 1998) and dynamics states theory (Levie and Lichtenstein, 2010) and adapted the theories for digital ventures using System Dynamics modelling. The use of System Dynamics has allowed the tracing back of the influences on value creation and capture. Unlike traditional quantitative methods that only consider a few elements, the model developed in this way can provide researchers and practitioners with an overview of the complex system and processes driving performance development (Sterman, 2000; Winch and Arthur, 2002). It highlights the critical feedback loops that may improve or deteriorate the ability to create and capture value. It also highlights the contextual influences on the feedback loops and performance outcomes in the environment, business model design, capability development, and regarding the dominant logic. Managers can improve a venture's performance development and develop an ability to capture value while growing value creation by altering these variables. The established, rigorous processes of System Dynamics applied in this thesis allow a high degree of confidence in the model and its findings (Barlas, 1996). The theory-driven development based on growth paths theory, dynamic states theory, and their background literature ensures that relationships in the model are evidence-based and theoretically verified (Tranfield, Denyer and Smart, 2003). By employing a formal modelling methodology, the model also overcomes the limitations of empirical measures for value creation and capture (Langley, 1999; Harrison et al., 2007). It showed how the theoretical conceptualisation of value creation and capture differ from proxies used in empirical studies. Moreover, the model could account for hard/quantitative and soft/qualitative

variables simultaneously. The model thus investigates growth and company performance holistically. The developed model has then been validated by comparing its approximation to the historical data of four case companies. Utilising case companies has allowed this thesis to compare the model's approximation against detailed, longitudinal qualitative and quantitative company data.

By simulating hypothetical scenarios, this thesis has identified performance development patterns, relationships of performance outcomes, the dominant feedback loops that cause them, and the variables shifting loop dominance. Simulations have allowed investigating the leverage points that cause different performance development patterns and the trajectories within these patterns. They show that entrepreneurs must select contextual factors carefully and improve them to create the requirements in which their venture can develop an ability to capture value while growing value creation. They also need to be aware that growth reduces their ventures' ability to capture value temporarily. Thereby, the thesis could utilise the strength of computer simulations in which the modeller can investigate the impact of changes in selected variables while creating otherwise identical scenario conditions. This approach significantly contributed to the theoretical insights obtained in this thesis by eliminating confounding factors that other methodologies, such as case studies, need to consider (Coyle, 2000; Sterman, 2000; Warren, 2002; Davis, Eisenhardt and Bingham, 2007; Harrison et al., 2007). In practice, companies are likely to face situations that the model represents through multiple variables. In such cases, the impact on company performance depends on the balance of variable changes, which managers and entrepreneurs can approximate by simulating the model for their ventures.

Comparing ventures' performance development patterns, the structures causing patterns, and identifying the exogenous variables affecting structures has allowed this thesis to illustrate the importance of the management's dominant logic on performance development patterns. Moreover, using the above strength of simulations, this thesis could reveal the causes of diversity in performance development patterns and the impact of growth on the ability to capture value. These findings align and add to the literature discussed in the previous chapter.

They could also be utilised to develop a framework supporting digital ventures to develop an ability to capture value while growing value creation.

8.2. Contribution to theory and practice

This thesis has utilised growth process theories to investigate the performance development and relationships for digital ventures. Below, the contributions to growth process theories, the performance management literature, and the literature on digital ventures are outlined. Contributions to practice are outlined with these theoretical contributions.

8.2.1. Contribution to growth process theories and management

This thesis contributes to growth process theories by combining value creation and capture in one growth process theory, integrating two contemporary growth process theories to overcome the limitations of OLC theory, and by developing a complex, formal theory of the growth process.

Firstly, this thesis has focussed growth process theories on two performance outcomes: growth in value creation and the venture's ability to capture value. Growth process theories commonly focus merely on size changes as their focal performance outcome (Penrose, 2009; Davidsson, Achtenhagen and Naldi, 2010). However, growth and size alone are insufficient to sustain a company and growth may even reduce a company's ability to survive (Hirakubo and Friedman, 2008; Thiel and Masters, 2014). Therefore, this thesis has also considered a venture's ability to capture value, expressing if and what fraction of resources a venture is generating through its operating activities (Brandenburger and Stuart, 1996; Lepak, Smith and Taylor, 2007; Bowman and Ambrosini, 2000; Bowman and Ambrosini, 2010). Thus, unlike previous growth process theories, this thesis is not just measuring growth. It contributes to growth process theories by adding value capture as a second performance metric that determines growth's sustainability to growth process theories. Future researchers and managers/entrepreneurs of digital ventures should consider it an equally

important outcome during and of the growth process and not pursue growth at the expense of sustaining their company.

Secondly, this thesis has integrated two contemporary growth process theories to overcome the limitations of OLC stage models. OLC stage models were the dominant growth process theory that linked growth stage characteristics and companies' performance development (Hoy, 2006; Davidsson, Achtenhagen and Naldi, 2010). Because OLC stage models have been empirically challenged, contemporary growth process theories have been proposed. However, these are unable to link stage characteristics to development paths. While growth paths theory describes diverse development paths (Garnsey and Heffernan, 2005; Garnsey, Stam and Heffernan, 2006), dynamic states theory conceptualises growth states (Levie and Lichtenstein, 2010). This thesis specified the variables that define a growth state as dynamic states theory conceptualises them. Unlike dynamic states theory, which provides only an abstract definition of growth states (Davidsson, Achtenhagen and Naldi, 2010), this thesis contributes a set of variables that define and constitute growth states. They relate to the management's dominant logic, environmental factors, capability development, and business model design. These variables are exogenous inputs to the model that remain constant during a growth state. This thesis then conceptualised how these state variables affect the endogenous, continuous change among business model elements and performance outcomes that growth paths theory conceptualises. Thus, this model has contributed not only the underlying mechanisms that affect ventures during the growth process but also how they are affected by the elements that constitute growth states over which the management controls.

Thirdly, this thesis contributes a thoroughly tested System Dynamics model as an output that combines the strengths of quantitative, qualitative, and theoretical work. Similar to quantitative investigations of the growth process (e.g. Hamilton, 2012; Harbermann & Schuilte, 2017), this thesis has provided a clear and tested list of important variables and their relationships. Like qualitative and theoretical research (e.g. Garnsey, 1998; Levie and Lichtenstein, 2010; Hesse Sternberg, 2017), this thesis can account for the complex system that influences the growth

process. Future research, discussed below, may simulate and adapt this formal model for different scenarios and contexts as a starting point. Moreover, managers and entrepreneurs may use the model to approximate the development of their ventures, forming a practical contribution of the model developed in this thesis. They may form assumptions about a venture's model inputs and simulate the model to determine the venture's growth in value creation, ability to capture value, the penalty of growth on value capture, and the capital required to sustain growth periods (Vhzquez, Liz and Aracil, 1996; Winch and Arthur, 2002).

8.2.2. Contribution to performance management literature and practice

This thesis also contributes to the performance management literature by identifying two performance development patterns among diverse paths, clarifying the relationship between growth in value creation and the ability to capture value, and contributing to future research methodologically.

Firstly, this thesis identifies two development patterns and contributes to explaining the diverse performance developments of companies. Previous research could not clearly identify the factors that cause companies' diverse performance developments (Davidsson, Steffens and Fitzsimmons, 2009; Brännback et al., 2009; Jang, 2011). This thesis reduces the complexity of performance development paths by identifying and explaining two patterns within those diverse paths. It shows that a management's growth intentions determine the performance development patterns experienced by the company, causing either exponential growth at the expense of value capture or goal-seeking development in both performance outcomes. Contextual factors regarding the environment, business model design, capability development, and dominant logic determine the trajectories within these two patterns. This finding has managerial implications when striving to develop an ability to capture value. Managers need to select attractive environments, develop effective business models, and continuously improve them to ensure their natural performance levels are positive, thus creating the requirements to develop an ability to capture value.

Secondly, the thesis has illustrated the trade-offs and benefits of growth in value creation and the share of value captured, contributing to clarifying the inconclusive results of previous investigations. Previous research on the relationship between growth in value creation and the ability to capture value could not clearly identify if growth has a positive (Katz and Shapiro, 1985; Chandler and Jansen, 1992; Davidsson, Steffens and Fitzsimmons, 2009), negative (Fitzsimmons, Steffens and Douglas, 2005; Brännback et al., 2009; Davidsson, Steffens and Fitzsimmons, 2009; Steffens, Davidsson and Fitzsimmons, 2009), or no impact (Markman and Gartner, 2002; Bottazzi et al., 2010) on the ability to capture value. Similarly, it could not establish if an ability to capture value has a positive (Fitzsimmons, Steffens and Douglas, 2005; Brännback et al., 2009; Davidsson, Steffens and Fitzsimmons, 2009; Steffens, Davidsson and Fitzsimmons, 2009), negative (Reid, 1995; Lee, 2014, 2018), or no impact on growth (Markman and Gartner, 2002; Coad, 2002). This thesis has identified temporality as a possible explanation for the inconclusive and conflicting results. While growth has a negative impact on the ability to capture value during periods of growth, the increase in size that it causes has a positive impact. If an overall positive or negative effect is observed depends on firmspecific growth rates and delays in the system. This has managerial implications as growth goals are a managerial choice and part of the dominant logic. Therefore, executives can reduce their growth goals to improve the ability to capture value and manage their company's ability to achieve both performance outcomes.

In addition, this thesis also makes methodological contributions to the performance management literature. Firstly, this thesis' systemic investigation has allowed the incorporation of a wide variety of factors affecting value creation and capture into the developed model. These factors have previously been considered in different sets of literature, and the findings of this thesis align with them. This literature regards, for example, the importance of the external environment (Porter, 1985, 2008; Romanelli, 1989; Zhang, Lichtenstein and Gander, 2015), business model designs (Amit and Zott, 2001; Zott and Amit, 2007, 2010; Teece and Linden, 2017; Täuscher and Abdelkafi, 2018), human resource management and capabilities (Nelson and Winter, 1982; Porter, 1996; Baron and Hannan, 2002), and the impact on managerial targets (Davidsson,
1991; Delmar and Wiklund, 2008; Kirkwood, 2009; Cassia and Minola, 2012; Hesse and Sternberg, 2017). The contribution of this thesis is combining them into a single theory and model. Future research should consider all factors that determine performance patterns and trajectories. Secondly, this project has also given attention to the validity of proxies to measure value creation and capture. It has shown that revenue and profit margins are reasonable proxies for digital venture's performance outcomes, while researchers should avoid return on assets as a proxy for a digital venture's share of value captured.

8.2.3. Contribution to digital ventures

This thesis integrated growth process theories for and investigated company performance in the context of digital ventures. Therefore, it also contributes to current knowledge about this specific context and type of company.

The model as an output of this thesis represents a formal theory of growth for digital ventures. This thesis develops a growth process theory specific to digital ventures. Thus, it has fulfilled the call to contextualise general theories for digital ventures and their specific aspects (e.g. Kraus *et al.*, 2018; Nzembayie, Buckley and Cooney, 2019; Zaheer, Breyer and Dumay, 2019). Thus, thesis has, for example, included their asset-lightness, the importance and continuity of product development, and the interactions of different customer groups. While such issues were often mentioned in growth process theories, they were never considered essential parts of those theories. For example, as outlined in the introduction of this thesis, many OLC stage models (e.g. Kazanjian and Drazin, 1990) may consider research and development an initial step before other activities. As this thesis shows, it is a continuous activity increases growth in value creation and the share of value captured simultaneously. Future researchers investigating digital ventures may build on these identified issues and specifications.

Regarding performance outcomes, managers and entrepreneurs are alerted that their shared belief on growing a company into profitability is flawed. Instead, highlight is given to the importance of appropriate contextual factors to create the requirements to develop an ability to capture value. Moreover, digital entrepreneurs must be aware that their high growth intentions negatively affect their share of value captured and may increase capital requirements.

8.3. Limitations and future research

Future research may have occasion to further test the model developed in this thesis. Here the model has been tested against four companies after an extensive and rigorous identification and vetting process, thereby going beyond the validation executed for many System Dynamics models that are developed and validated for one company (e.g. Coyle and Exelby, 2000; Oliva, Sterman and Giese, 2003). However, in comparison with other quantitative methods, this sample size may be perceived as a limitation. Testing the model with further studies may reveal additional insight and areas for improvement. Researchers might, for example, use companies with alternative development paths or companies that are already profitable. Researchers should also focus on companies employing freemium and content-based business models. Lastly, future research may use primary, longitudinal data and work with companies that are less secretive about their operations. Thereby, the model can be improved based on direct feedback from practitioners (Luna-Reves and Andersen, 2003). As Sterman (2002, p. 501) argues, "all models are wrong". By means of future research, confidence in the model can be further enhanced.

Future research may also be able to build on the developed model and expand its scope of application to other types of companies. The research here has focussed on companies providing fully digital products and services. Further research could strive to generalise findings of this thesis and evaluate if they apply in other contexts. Four areas seem promising:

- A first area is to broaden the application of the model as regards diversification and acquisition strategies. Future research could consider the relationships between different product lines and the step changes in resources caused by acquiring other companies.
- A second step may be to expand the model to account for e-commerce and other companies combining digital and physical products. This would require, for example, the extension of the model through additional resource stocks

and activities that the digital ventures in this thesis do not require. Examples of such stocks and activities include different types of inventories, manufacturing, and logistics.

- Thirdly, future research could adapt the model for different contexts. Such research may use the development model and methodology to pursue such adaptations. Areas for such research were identified in the methodology section where of this thesis excluded, for example, electronics, natural resources, and biotechnology companies from the sample.
- Lastly, financial service companies are a promising context for future research. As noted in the methodology, growth process theories claim to be concerned with just producing companies and so exclude financial companies from their considerations (Penrose, 2009). The growth of investment companies may thus be unchartered territory and can therefore provide the opportunity to test existing growth process theories on firms which are usually excluded. The methodology and model proposed in this thesis may guide future researchers in doing so.

Rather than adding and adapting the model, future research would also be able to build on the findings of this research. This concerns improving the generality of the results of the findings and focussing on best practices to alter performance:

- A step for future research may confirm the findings of this thesis through alternative methods. Different research methods have different advantages and disadvantages. The strengths of the simulation methodology utilised here include, for example, the measurement of theoretical measures, accounting for hard and soft factors, and creating hypothetical scenarios. However, due to its complexity, it has been limited to small sample sizes. Additional research may further generalise and test the findings of this project, for example, through large datasets of companies.
- Future research may also focus on the methods that best improve company performance. The researcher here has attempted to outline what variables matter and what impact they have on performance developments and relationships. In the discussion, some guidelines have been provided to alter these variables. Future research could systematically evaluate published literature (Tranfield, Denyer and Smart, 2003) or identify best-practice

through qualitative and quantitative studies to identify how these variables could be altered by managers and entrepreneurs. Thereby, future research could develop a "playbook" that builds on the framework developed in the discussion chapter of this thesis.

When pursuing such lines of inquiry, it is recommended that future researchers adhere to the principles outlined in this thesis. They should evaluate the management's dominant logic as it is the primary driver of performance development patterns. Moreover, future researchers should control for all contextual factors outlined in this thesis. Researchers should also use revenue growth and profit margins as proxies of more theoretical performance measures.

References

- Aaker, D. A. and McLoughlin, D. (2017) *Strategic Market Management*. Edited by E. Edition. Chichester: John Wiley & Sons.
- Adams, M. et al. (2014) 'Firm size and growth in Sweden's life insurance market between 1855 and 1947: A test of Gibrat's law', *Business History*. Taylor & Francis, 56(6), pp. 956–974.
- Afuah, A. and Tucci, C. L. (2000) *Internet business models and strategies: Text and cases*. 2nd edn. New York: McGraw-Hill Higher Education.
- Albin, S. and Forrester, J. (1997) 'Building a System Dynamics Model', *MIT* System Dynamics in Education Project, p. 35.
- Amit, R. and Zott, C. (2001) 'Value creation in e-business', *Strategic Management Journal*, 22(6–7), pp. 493–520.
- Anderson, P. *et al.* (1999) 'Introduction to the Special Issue: Applications of Complexity Theory to Organization Science', *Organization Science*, 10(3), pp. 233–236.
- Anderson, P. and Meyer, A. D. (2016) 'Complexity Theory and Process
 Organization Studies', in Langley, A. and Tsoukas, H. (eds) *The SAGE Handbook of Process Organization Studies*. Beverley Hills: SAGE
 Publications, pp. 127–143.
- Arora, A., Fosfuri, A. and Gambardella, A. (2001) 'Markets for Technology and Their Implications for Corporate Strategy', *Industrial and Corporate Change*, 10(2), pp. 419–451.
- Arrighetti, A. and Lasagni, A. (2013) 'Assessing the Determinants of High-Growth Manufacturing Firms in Italy', *International Journal of the Economics* of Business, 20(2), pp. 245–267.
- Asimakopoulos, I., Samitas, A. and Papadogonas, T. (2009) 'Firm-specific and economy wide determinants of firm profitability: Greek evidence using panel data', *Managerial Finance*, 35(11), pp. 930–939.
- Asvanund, A. *et al.* (2004) 'An empirical analysis of network externalities in peer-to-peer music-sharing networks', *Information Systems Research*, 15(2), pp. 155–174.
- Azoulay, P., Repenning, N. P. and Zuckerman, E. W. (2010) 'Nasty, Brutish, and

Short: Embeddedness Failure in the Pharmaceutical Industry', *Administrative Science Quarterly2*, 55(3), pp. 472–507.

- Baliyan, N. and Kumar, S. (2014) 'Towards software engineering paradigm for software as a service', 7th International Conference on Contemporary Computing. IEEE, pp. 329–333.
- Bandara, W. *et al.* (2015) 'Achieving rigor in literature reviews: Insights from qualitative data analysis and tool-support', *Communications of the Association for Information Systems*, 37, pp. 154–204.
- Barlas, Y. (1996) 'Formal aspects of model validity and validation in system dynamics', *System Dynamics Review*, 12(3), pp. 183–210.
- Barlas, Y. and Carpenter, S. (1990) 'Philosophical roots of model validation: two paradigms', *System Dynamics Review*, 6(2), 148–166.
- Barney, J. B. (1991) 'Firm resources and sustained competitive advantage', *Journal of Management*, 17(1), pp. 99–120.
- Barney, J. B. (2001) 'Is the Resource-Based "View" a Useful Perspective for Strategic Management Research? Yes', *The Academy of Management Review*, 26(1), pp. 41–56.
- Baron, J. N. and Hannan, M. T. (2002) 'Organizational Blueprints for Success in High-Tech Start-Ups: Lessons from the Stanford Project on Emerging Companies', *California Management Review*, 44(3), pp. 8–36.
- Baron, J. N., Hannan, M. T. and Burton, M. D. (2001) 'Labor pains: Change in organizational models and employee turnover in young, high-tech firms', *American Journal of Sociology*, 10(4), pp. 960–1012.
- Bean, D. D. (2019) A small tech company tried it all to stop employee turnover.
 Only one thing worked. Available at: https://www.cnbc.com/2019/12/03/a-tech-firm-tried-it-all-to-stop-turnover-only-one-thing-worked.html (Accessed: 12 August 2021).
- Belt, P. *et al.* (2009) 'Organisational maturity and functional performance', *International Journal of Management and Enterprise Development*, 6(2), pp. 147–164.
- Berends, P. and Romme, G. (1999) 'Simulation as a research tool in management studies', *European Management Journal*, 17(6), pp. 576–583.
- Bessant, J., Phelps, R. and Adams, R. J. (2005) *A review of the literature addressing the role of knowledge and expertise at key stages of business*

growth and development.

- Bhide, A. (2000) The Origin and Evolution of New Businesses. Oxford: Oxford University Press.
- Bingham, C. B., Eisenhardt, K. M. and Furr, N. R. (2007) 'What makes a Process a Capability? Heuristics, Strategy, and Effective capture of Opportunities', *Strategic Entrepreneurship Journal*, 1, pp. 27–47.
- Bontis, N. and Chung, H. (2000) 'The evolution of software pricing: From box licenses to application service provider models', *Internet Research*, 10(3), pp. 246–255.
- Bottazzi, G. *et al.* (2010) 'Corporate performances and market selection: Some comparative evidence', *Industrial and Corporate Change*, 19(6), pp. 1953–1996.
- Bottazzi, G. *et al.* (2011) 'Corporate growth and industrial dynamics: Evidence from french manufacturing', *Applied Economics*, 43(1), pp. 103–116.
- Bowman, C. and Ambrosini, V. (2000) 'Value Creation Versus Value Capture: Towards a Coherent Definition of Value in Strategy', *British Journal of Management*, 11(1), pp. 1–15.
- Bowman, C. and Ambrosini, V. (2007) 'Identifying Valuable Resources', *European Management Journal*, 25(4), pp. 320-329.
- Bowman, C. and Ambrosini, V. (2010) 'How value is created, captured and destroyed', *European Business Review*, 22(5), pp. 479-495.
- Bradley, S. *et al.* (2012) 'Toward an evolution strategy for the digital goods business', *Management Decision*, 50(2), pp. 234–252.
- Brandenburger, A. and Start, H. (1996) 'Value-based business strategy', *Journal* of *Economics & Management Strategy*, 5(1), pp. 5-24.
- Brännback, M. *et al.* (2009) 'Growth and profitability in small privately held biotech firms: preliminary findings', *New Biotechnology*, 25(5), pp. 369–376.
- von Briel, F., Recker, J. and Davidsson, P. (2018) 'Not all digital venture ideas are created equal: Implications for venture creation processes', *Journal of Strategic Information Systems*. Elsevier, 27(4), pp. 278–295.
- Brown, R. and Mawson, S. (2013) 'Trigger points and high-growth firms: A conceptualisation and review of public policy implications', *Journal of Small Business and Enterprise Development*, 20(2), pp. 279–295.

Brush, T. H., Philip, B. and Hendrickx, M. (2000) 'The Free Cash Flow

Hypothesis for Sales Growth and Firm Performance', *Strategic Management Journal*, 21(4), pp. 455–472.

- Bryman, A. and Bell, E. (2015) *Business Research Methods*. 4th edn, *Business Research Methods*. 4th edn. Oxford: OUP Oxford.
- Capon, N., Farley, J. U. and Hoenig, S. (1990) 'Determinants of FinancialPerformance: A Meta-Analysis', *Management Science*, 36(10), pp. 1143–1159.
- Cassia, L. and Minola, T. (2012) 'Hyper-growth of SMEs: Toward a reconciliation of entrepreneurial orientation and strategic resources', *International Journal of Entrepreneurial Behavior & Research*, 18(2), pp. 179–197.
- Chandler, G. N. and Jansen, E. (1992) 'The founder's self-assessed competence and venture performance', *Journal of Business Venturing*, 7(3), pp. 223–236.
- Chia, R. (1999) 'A "Rhizomic" Model of Organizational Change and Transformation: Perspective from a Metaphysics of Change', *British Journal* of Management, 10, pp. 209–227.
- Chia, R. (2005) 'Organization Theory as a Postmodern Science', in Knudsen, C. and Tsoukas, H. (eds) *The Oxford Handbook of Organization Theory*. Oxford: Oxford University Press, pp. 113–142.
- Chiles, T. H., Bluedorn, A. C. and Gupta, V. K. (2007) 'Beyond creative destruction and entrepreneurial discovery: A radical Austrian approach to entrepreneurship', *Organization Studies*, 28(4), pp. 467–493.
- Cho, H.-J. and Pucik, V. (2005) 'Relationship between Innovativeness, Quality, Growth, Profitability, and Market Value', *Strategic Management Journal*, 26(6), pp. 555–575.
- Christensen, C. M. (1992) 'Exploring the Limits of the Technology S-Curve. Part I: Component Technologies', *Production and Operations Management*, 1(4), pp. 334–357.
- Clarysse, B., Bruneel, J. and Wright, M. (2011) 'Explaining growth paths of young technology-based firms: structuring resource portfolios in different competitive environments', *Strategic Entrepreneurship Journal*, 5(2), pp. 137– 157.
- Coad, A. (2007) 'Testing the principle of "growth of the fitter": The relationship between profits and firm growth', *Structural Change and Economic Dynamics*, 18(3), pp. 370–386.

- Coad, A., Rao, R. and Tamagni, F. (2011) 'Growth processes of Italian manufacturing firms', *Structural Change and Economic Dynamics*. Elsevier B.V., 22(1), pp. 54–70.
- Cowling, M. (2004) 'The growth Profit nexus', *Small Business Economics*, 22(1), pp. 1–9.
- Cowton, C. J. (1998) 'The use of secondary data in business ethics research', *Journal of Business Ethics*, 17(4), pp. 423–434.
- Coyle, G. (1996) *System Dynamics Modelling: A Practical Approach*. London: Chapman and Hall.
- Coyle, G. (2000) 'Qualitative and quantitative modelling in system dynamics: some research questions', *System Dynamics Review*, 16(3), pp. 225–244.
- Coyle, G. and Exelby, D. (2000) 'The validation of commercial system dynamics models', *System Dynamics Review*, 16(1), pp. 27–41.
- Cristofaro, M. (2020) 'E-business evolution: an analysis of mobile applications' business models', *Technology Analysis and Strategic Management*. Taylor & Francis, 32(1), pp. 88–103.
- Currie, W. L., Joyce, P. and Winch, G. (2007) 'Evaluating application service provisioning using system dynamics methodology', *British Journal of Management*, 18(2), pp. 172–191.
- Dangerfield, B. (2014) 'Systems thinking and system dynamics: A primer', in Brailsford, S., Churivol, L., and Dangersfield, B. (eds) *Discrete-Event Simulation and System Dynamics for Management Decision Making*. 1st edn. Wiley, pp. 29–51.
- DaSilva, C. M. *et al.* (2013) 'Disruptive technologies: A business model perspective on cloud computing', *Technology Analysis and Strategic Management*, 25(10), pp. 1161–1173.
- Davidsson, P. (1989) 'Entrepreneurship and After? A Study of GrowthWillingness in Small Firms', *Journal of Business Venturing*, 4(3), pp. 211–226.
- Davidsson, P. (1991) 'Continued entrepreneurship: Ability, need, and opportunity as determinants of small firm growth', *Journal of Business Venturing*, 6(6), pp. 405–429.
- Davidsson, P., Achtenhagen, L. and Naldi, L. (2010) 'Small Firm Growth', *Foundations and Trends in Entrepreneurship*, 6(2), pp. 69–166.
- Davidsson, P., Steffens, P. and Fitzsimmons, J. (2009) 'Growing profitable or

growing from profits: Putting the horse in front of the cart?', *Journal of Business Venturing*. Elsevier B.V., 24(4), pp. 388–406.

- Davis, J. P., Eisenhardt, K. M. and Bingham, C. B. (2007) 'Developing theory through simulation methods', *The Academy of Management Review*, 32(2), pp. 480–499.
- Delmar, F., McKelvie, A. and Wennberg, K. (2013) 'Untangling the relationships among growth, profitability and survival in new firms', *Technovation*, 33(8–9), pp. 276–291.
- Delmar, F. and Wiklund, J. (2008) 'The effect of small business managers' growth motivation on firm growth: A longitudinal study', *Entrepreneurship: Theory & Practice*, 32(3), pp. 437–457.
- Deng, Z. and Wang, Z. (2016) 'Early-mover advantages at cross-border businessto-business e-commerce portals', *Journal of Business Research*. Elsevier Inc., 69(12), pp. 6002–6011.
- Denyer, D. and Tranfield, D. (2009) 'Producing a Systematic Review', in *The SAGE Handbook of Organizational Research Methods*, pp. 671–689.
- Dierickx, I. and Cool, K. (1989) 'Asset Stock Accumulation and Sustainability of Competitive Advantage', *Management Science*, 35(12), pp. 1504–1511.
- Doganova, L. and Eyquem-Renault, M. (2009) 'What do business models do? Innovation devices in technology entrepreneurship', *Research Policy*, 38(10), pp. 1559–1570.
- Dooley, K. (1997) 'A Complex Adaptive Systems Model of Organization Change', *Nonlinear Dynamics, Psychology, and Life Sciences*, 1(1), pp. 69–97.
- Dosi, G., Nelson, R. R. and Winter, S. G. (2000) The Nature and Dynamics of Organizational Capabilities. Oxford: Oxford University Press.
- Eckhardt, J. T., Ciuchta, M. P. and Carpenter, M. (2018) 'Open innovation, information, and entrepreneurship within platform ecosystems', *Strategic Entrepreneurship Journal*, 12(3), pp. 369–391.
- Eisenhardt, K. M. (1989) 'Building Theories from Case Study Research', *The Academy of Management Review*, 14(4), pp. 532–550.
- Eisenhardt, K. M. and Martin, J. A. (2000) 'Dynamic capabilities: what are they?', *Strategic Management Journal*, 21(10–11), pp. 1105–1121.
- Eisenhardt, K. M. and Schoonhoven, C. B. (1990) 'Organizational Growth: Linking Founding Team, Strategy, Environment, and Growth Among U.S.

Semiconductor Ventures', *Administrative Science Quarterly*, 35(3), pp. 504–529.

- Eisenmann, T. R. (2006) 'Internet companies' growth strategies: determinants of investment intensity and long-term performance', *Strategic Management Journal*, 27, pp. 1183–1204.
- Eling, M., Jia, R. and Schaper, P. (2021) 'The magic triangle: growth, profitability and safety in the insurance industry', *Geneva Papers on Risk and Insurance: Issues and Practice*. Palgrave Macmillan UK, (0123456789).
- Ertan, A., Lewellen, S. and Thomas, J. K. (2020) 'Do profit margins expand for high growth firms?', *Journal of Management Accounting Research*, 32(3), pp. 117–135.
- Federico, J. S. and Capelleras, J.-L. L. (2015) 'The heterogeneous dynamics between growth and profits: the case of young firms', *Small Business Economics*, 44(2), pp. 231–253.
- Fernández-López, S. et al. (2019) 'Determinants of high-growth university spinoffs in Spain', Journal of Science and Technology Policy Management, 10(4), pp. 890–904.
- Fitzsimmons, J., Steffens, P. and Douglas, E. (2005) 'Growth and Profitability in Small and Medium Sized Australian Firms', in *AGSE Entrepreneurship Exchange*.
- Ford, D. N. (2019) 'A system dynamics glossary', *System Dynamics Review*, 35(4), pp. 369–379.
- Forrester, J. W. (1984) 'The system dynamics national model objectives, philosophy, and status', in *International System Dynamics Conference*.
- Forrester, J. W. (1987) 'Lessons from system dynamics modeling', *System Dynamics Review*, pp. 136–149.
- Forrester, J. W. (1994) 'System Dynamics, Systems Thinking, and Soft OR', *System Dynamics Review*, 10(2), pp. 1–14.
- Forrester, J. W. and Senge, P. M. (1979) *Tests For Building Confidence In System Dynamics Models*. Cambridge.

Founders Circle Capital (no date) In The Front Door, Out The Back: Attrition Challenges At High Growth Startups. Available at: https://www.founderscircle.com/high-startup-turnover-rate/.

Fransman, M. (1994) 'Information, Knowledge, Vision and Theories of the Firm',

Industrial and Corporate Change, 3(3), pp. 713–757.

- Fuertes-Callén, Y. and Cuellar-Fernández, B. (2019) 'Inter-relationship between firm growth and profitability in a context of economic crisis', *Journal of Business Economics and Management*, 20(1), pp. 86–106.
- Gabrielsson, P. and Gabrielsson, M. (2013) 'A dynamic model of growth phases and survival in international business-to-business new ventures: The moderating effect of decision-making logic', *Industrial Marketing Management*, 42(8), pp. 1357–1373.
- Gandia, R. and Parmentier, G. (2017) 'Optimizing value creation and value capture with a digital multi-sided business model', *Strategic Change*, 26(4), pp. 323–331.
- Garnsey, E. (1998) 'A theory of the early growth of the firm', *Industrial and Corporate Change*, 7(3), pp. 523–556.
- Garnsey, E., Dee, N. and Ford, S. (2006) *Clean Technology Ventures and Innovation, October.* 2006/01. Cambridge.
- Garnsey, E. and Heffernan, P. (2005) 'Growth setbacks in new firms', *Futures*, 37(7), pp. 675–697.
- Garnsey, E., Lubik, S. and Heffernan, P. (2015) 'Organizational emergence and new firm formation', in Wright, J. D. (ed.) *International Encyclopedia of the Social & Behavioral Sciences*. 2nd edn. Oxford: Elsevier.
- Garnsey, E., Stam, E. and Heffernan, P. (2006) 'New firm growth: Exploring processes and paths', *Industry and Innovation*, 13(1), pp. 1–20.
- Garud, R., Kumaraswamy, A. and Sambamurthy, V. (2006) 'Emergent by design: Performance and transformation at Infosys technologies', *Organization Science*, 17(2), pp. 277–286.
- Gary, M. S. et al. (2008) System Dynamics and Strategy: Towards Compelling Behavioural Theories of Strategic Dynamics.
- Gaur, A. and Kumar, M. (2018) 'A systematic approach to conducting review studies: An assessment of content analysis in 25 years of IB research', *Journal of World Business*. Elsevier, 53(2), pp. 280–289.
- Geron, T. (2013) The Top Tech IPOs Of 2012 And The Venture Capitalists Who Invested, forbes.com.
- Geroski, P. A., Machin, S. J. and Walters, C. F. (1997) 'Corporate growth and profitability', *Journal of Industrial Economics*, 45(2), pp. 171–189.

- Giotopoulos, I. (2014) 'Dynamics of Firm Profitability and Growth: Do Knowledge-Intensive (Business) Services Persistently Outperform?', *International Journal of the Economics of Business*. Routledge, 21(3), pp. 291–319.
- Gjerløv-Juel, P. and Guenther, C. (2019) 'Early employment expansion and longrun survival examining employee turnover as a context factor', *Journal of Business Venturing*. Elsevier, 34(1), pp. 80–102.
- Goddard, J. A., Molyneux, P. and Wilson, J. O. S. (2004) 'Dynamics of Growth and Profitability in Banking', *Journal of Money, Credit, and Banking*, 36(6), pp. 1069–1090.
- Grant, R. M. (1991) 'The resource-based theory of competitive advantage: implications for strategy formulation', *California Management Review*, 33(3), pp. 13–135.
- Grant, R. M. (1996) 'Toward a knowledge-based theory of the firm', *Strategic Management Journal*, 17(S2), pp. 109–122.
- Grant, R. M. (2016) *Contemporary strategy analysis: Text and cases edition.* John Wiley & Sons.
- Greve, H. R. (2008) 'A Behavioral Theory of Firm Growth: Sequential Attention to Size and Performance Goals', *The Academy of Management Journal*, 51(3), pp. 476–494.
- Groesser, S. N. and Schwaninger, M. (2012) 'Contributions to model validation: hierarchy, process, and cessation', *System Dynamics Review*, 28(2), pp. 157–181.
- Größler, A. (2007) 'A dynamic view on strategic resources and capabilities applied to an example from the manufacturing strategy literature', *Journal of Manufacturing Technology Management*, 18(3), pp. 250–266.
- Größler, A., Thun, J. H. and Milling, P. M. (2008) 'System dynamics as a structural theory in operations management', *Production and Operations Management*, 17(3), pp. 373–384.
- Gu, X., Kannan, P. K. and Ma, L. (2018) 'Selling the premium in freemium', *Journal of Marketing*, 82(6), pp. 10–27.
- Guggenberger, T. M. et al. (2020) 'Towards a unifying understanding of digital business models', Proceedings of the 24th Pacific Asia Conference on Information Systems: Information Systems (IS) for the Future, PACIS 2020, pp.

1–14.

- Guo, H. et al. (2020) 'Technology Push or Market Pull? Strategic Orientation in Business Model Design and Digital Start-up Performance', Journal of Product Innovation Management, 0(0), pp. 1–21.
- Hadiji, F. et al. (2014) 'Predicting player churn in the wild', *IEEE Conference on Computatonal Intelligence and Games, CIG.*
- Hafezieh, N., Akhavan, P. and Eshraghian, F. (2011) 'Exploration of process and competitive factors of entrepreneurship in digital space: A multiple case study in Iran', *Education, Business and Society: Contemporary Middle Eastern Issues*, 4(4), pp. 267–279.
- Haile, N. and Altmann, J. (2016a) 'Structural analysis of value creation in software service platforms', *Electronic Markets*, 26(2), pp. 129–142.
- Haile, N. and Altmann, J. (2016b) 'Value creation in software service platforms', *Future Generation Computer Systems*, 55, pp. 495–509.
- Hamari, J. and Keronen, L. (2017) 'Why do people buy virtual goods: A metaanalysis', *Computers in Human Behavior*. Elsevier Ltd, 71, pp. 59–69.
- Hambrick, D. C. and Crozier, L. M. (1985) 'Stumblers and stars in the management of rapid growth', *Journal of Business Venturing*, 1(1), pp. 31–45.
- Hamermesh, D. S. and Pfann, G. A. (1996) 'Adjustment Costs in Factor Demand', Journal of Economic Literature, 34(3), pp. 1264–1292.
- Hamilton, R. T. (2012) 'How firms grow and the influence of size and age', *International Small Business Journal*, 30(6), pp. 611–621.
- Hanks, S. H. *et al.* (1993) 'Tightening the Life-Cycle Construct: A Taxonomic Study of Growth Stage Configurations in Organizations', *Entrepreneurship: Theory & Practice*, 18(2), pp. 5–30.
- Hansen, B. and Hamilton, R. T. (2011) 'Factors distinguishing small firm growers and non-growers', *International Small Business Journal*, 29(3), pp. 278–294.
- Harbermann, H. and Schuilte, R. (2017) 'Analyzing non-linear dynamics of organic growth: Evidence from small german new ventures', *Journal of Small Business Strategy*, 27(2), pp. 36–64.
- Harmon, R., Raffo, D. and Faulk, S. (2004) 'Value-based pricing for new software products: strategy insights for developers', *Portland International Conference on Management of Engineering and Technology (PICMET)*

Proceedings, (August), pp. 1–24.

- Harrison, J. R. et al. (2007) 'Simulation Modeling in Organizational and Management Research', *The Academy of Management Review*, 32(4), pp. 1229–1245.
- Hathaway, I. (2013) Tech Starts: High-Technology Business Formation and Job Creation in the United States, Kauffman Foundation Research Series: Firm Formation and Economic Growth.
- Heirman, A. and Clarysse, B. (2004) 'How and Why do Research-Based Start-Ups Differ at Founding? A Resource-Based Configurational Perspective', *The Journal of Technology Transfer*, 29(3–4), pp. 247–268.
- Helin, J. et al. (2014) 'Process is how process does', in Helin, J. et al. (eds) The Oxford handbook of process philosophy and organization studies. Oxford: Oxford University Press.
- Hennessy, G. (1997) 'Modeling "Soft" Variables', *The Systems Thinker*, 8(7), pp. 6–7.
- Hermans, J. et al. (2012) Ambitious entrepreneurship: Antecedents and consequences.
- Hernes, T. (2008) Understanding Organization as Process: Theory for a Tangled World. Oxon: Routledge.
- Hesse, N. and Sternberg, R. (2017) 'Alternative growth patterns of university spin-offs: why so many remain small?', *International Entrepreneurship and Management Journal*. International Entrepreneurship and Management Journal, 13(3), pp. 953–984.
- Hirakubo, N. and Friedman, H. (2008) ' Dot-Bombs: Lessons from the Dot-Com Debacle', *Journal of Internet Commerce*, 1(2), pp. 89-102.
- Hitt, M. A. et al. (2011) 'Strategic Entrepreneurship: Creating Value for Individuals, Organizations, and Society', *The Academy of Management Perspectives*, 25(2), pp. 57–75.
- Holm, A. B. and Günzel-Jensen, F. (2017) 'Succeeding with freemium: strategies for implementation', *Journal of Business Strategy*, 38(2), pp. 16–24.
- Homer, J. B. (1996) 'Why we iterate: scientific modeling in theory and practice', *System Dynamics Review*, 12(1), pp. 1–19.
- Homer, J. B. (2012) 'Partial-model testing as a validation tool for system dynamics', *System Dynamics Review*, 28(3), pp. 281–294.

- Homer, J. and Oliva, R. (2001) 'Maps and models in system dynamics: A response to Coyle', *System Dynamics Review*, 17(4), pp. 347–355.
- Hoy, F. (2006) 'The complicating factor of life cycles in corporate venturing', *Entrepreneurship: Theory and Practice*, 30(6), pp. 831–836.
- Hsieh, Y. J. and Wu, Y. J. (2019) 'Entrepreneurship through the platform strategy in the digital era: Insights and research opportunities', *Computers in Human Behavior*. Elsevier Ltd, 95, pp. 315–323.
- Huang, J. et al. (2017) 'Growing on steroids: rapidly scaling the user base of digital ventures through digital innovation', MIS Quartely, 41(1), pp. 301–314.
- Huang, J., Henfridsson, O. and Liu, M. J. (2021) 'Extending Digital Ventures Through Templating', *Information Systems Research*.
- Huang, X. and Kunc, M. (2012) 'Modeling Start-ups using System Dynamics: Towards a generic model', in 30th International Conference of the System Dynamics Society.
- Huarng, K. H. (2013) 'A two-tier business model and its realization for entrepreneurship', *Journal of Business Research*. Elsevier Inc., 66(10), pp. 2102–2105.
- Hugo, O. and Garnsey, E. (2001) *Investigating the growth paths of young* technology-based firms: A process approach. 2001/01.
- Hugo, O. and Garnsey, E. (2002) 'The emergence of electronic messaging and the growth of four entrepreneurial entrants', in Oakley, R., During, W., and Kauser, S. (eds) *New Technology Based Firms in the New Millenium*.
 Amsterdam: Pergamon Press, pp. 97–123.
- Hugo, O. and Garnsey, E. (2005) 'Problem-solving and competence creation in the early development of new firms', *Managerial and Decision Economics*, 26(2), pp. 139–148.
- Hull, C. E. *et al.* (2007) 'Taking advantage of digital opportunities: a typology of digital entrepreneurship', *International Journal of Networking and Virtual Organisations*, 4(3), pp. 290–303.
- Ingley, C., Khlif, W. and Karoui, L. (2017) 'SME growth trajectories, transitions and board role portfolios: A critical review and integrative model', *International Small Business Journal: Researching Entrepreneurship*, 35(6), pp. 729–750.

Ireland, R. D., Hitt, M. A. and Sirmon, D. G. (2003) 'A Model of Strategic

Entrepreneurship: The Construct and its Dimensions', *Journal of Management*, 29(6), pp. 963–989.

- Jain, B. A., Jayaraman, N. and Kini, O. (2008) 'The path-to-profitability of Internet IPO firms', *Journal of Business Venturing*, 23(2), pp. 165–194.
- Jang, S. C. (2011) 'Growth-focused or profit-focused firms: Transitions toward profitable growth', *Tourism Management*, 32(3), pp. 667–674.
- Jang, S. C. (Shawn) and Park, K. (2011) 'Inter-relationship between firm growth and profitability', *International Journal of Hospitality Management*. Elsevier Ltd, 30(4), pp. 1027–1035.
- Johnson, G., Scholes, K. and Whittington, R. (2008) *Exploring Corporate Strategy*. 8th edn. Harlow: Pearson Education.
- Kachlami, H. and Yazdanfar, D. (2016) 'Determinants of SME growth: The influence of financing pattern. An empirical study based on Swedish data', *Management Research Review*, 39(9), pp. 966–986.
- Kallinikos, J., Aaltonen, A. and Marton, A. (2013) 'The Ambivalent Ontology of Digital Artifacts', *MIS Quarterly*, 37(2), pp. 357–370.
- Katz, B. M. L. and Shapiro, C. (1985) 'Network Externalities, Competition, and Compatibility', *The American Economic Review*, 75(3), pp. 424–440.
- Kazanjian, R. K. and Drazin, R. (1990) 'A stage-contingent model of design and growth for technology based new ventures', *Journal of Business Venturing*, 5(3), pp. 137–150.
- Keating, E. (1999) Issues to Consider While Developing a System Dynamics Model.
- Kelestyn, B., Henfridsson, O. and Nandhakumar, J. (2017) 'Scaling the User Base of Digital Ventures Through Generative Pattern Replication: The Case of Ridesharing', in 50th Hawaii International Conference on System Sciences, pp. 4776–4785.
- Ketchen, D. J., Ireland, R. D. and Snow, C. C. (2007) 'Strategic entrepreneurship, collaborative innovation, and wealth creation', *Strategic Entrepreneurship Journal*. Wiley, 1(3–4), pp. 371–385.
- Kim, C. and Kim, D. J. (2017) 'Uncovering the value stream of digital content business from users' viewpoint', *International Journal of Information Management*. Elsevier, 37(6), pp. 553–565.

Kim, C., Oh, E. and Shin, N. (2010) 'An empirical investigation of digital content

characteristics, value, and flow', *Journal of Computer Information Systems*, 50(4), pp. 79–87.

- Kim, D. H. and Anderson, V. (1998) Systems Archetype Basics: From Story to Structure, The Systems Thinker. Waltham: Pegasus Communications.
- Kim, W. C. and Mauborgne, R. (2005) 'Value innovation: A leap into the blue ocean', *Journal of Business Strategy*, 26(4), pp. 22–28.
- Kim, W. C. and Mauborgne, R. (2017) *Blue Ocean Strategy Reader*. Boston, MA.: Harvard Business Review Press.
- Kirkwood, J. (2009) 'To grow or not? Growing small service firms', Journal of Small Business and Enterprise Development, 16(3), pp. 485–503.
- Kleijnen, J. P. C. (1995) 'Verification and validation of simulation models', *European Journal of Operational Research*, 82, pp. 145–162.
- Knott, A. M., Bryce, D. J. and Posen, H. E. (2003) 'On the StrategicAccumulation of Intangible Assets', *Organization Science*, 14(2), pp. 192–207.
- Koch, T. and Windsperger, J. (2017) 'Seeing through the network: Competitive advantage in the digital economy', *Journal of Organization Design*. Journal of Organization Design, 6(1).
- Kohtamäki, M. *et al.* (2019) 'Digital servitization business models in ecosystems: A theory of the firm', *Journal of Business Research*. Elsevier, 104(June), pp. 380–392.
- Kollmann, T. (2006) 'What is e-entrepreneurship? fundamentals of company founding in the net economy', *International Journal of Technology Management*, 33(4), pp. 322–340.
- Kollmann, T. *et al.* (2016) *ESM European Startup Monitor 2016*. European Commission.
- Kooskora, M. (2020) 'The Role of an Entrepreneurial Mindset in Digital Transformation- Case Study of the Estonian Business School', in Soltanifar, M., Hughes, M., and Göcke, L. (eds) *Digital entrepreneurship: Impact of Business and Society*. Cham: Springer Nature, pp. 143–164.
- Kor, Y. Y. and Mahoney, J. T. (2000) 'Penrose's Resource-Based Approach: The Process and Product of Research Creativity', *Journal of Management Studies*, 37(1), pp. 109–139.
- Kor, Y. Y. and Mahoney, J. T. (2004) 'Edith Penrose's (1959) Contributions to the Resource-based View of Strategic Management', *Journal of Management*

Studies, 41(1), pp. 183–191.

- Kotter, J. and Sathe, V. (1978) 'Problems of Human Resource Management in Rapidly Growing Companies', *California Management Review*, 21(2), pp. 29– 36.
- Kraus, S. et al. (2018) 'Digital entrepreneurship: A research agenda on new business models for the twenty-first century', *International Journal of Entrepreneurial Behavior & Research*, 25(3), pp. 483–500.
- Von Krogh, G., Erat, P. and Macus, M. (2000) 'Exploring the link between dominant logic and company performance', *Creativity and Innovation Management*, 9(2), pp. 82–93.
- Kuester, S., Konya-Baumbach, E. and Schuhmacher, M. C. (2018) 'Get the show on the road: Go-to-market strategies for e-innovations of start-ups', *Journal of Business Research*. Elsevier, 83(October 2017), pp. 65–81.
- Kunc, M. (2017) 'System dynamics: A soft and hard approach to modelling', in 2017 Winter Simulation Conference (WSC), pp. 597–606.
- Kunc, M. (2018) System Dynamics: Soft and Hard Operational Research, OR Essentials.
- Labes, S., Erek, K. and Zarnekow, R. (2013) 'Common patterns of cloud business models', in *19th Americas Conference on Information Systems (AMCIS 2013)*.
- Langley, A. (1999) 'Strategies for Theorizing from Process Data', *The Academy* of Management Journal, 24(4), pp. 691–710.
- Langley, A. *et al.* (2013) 'Process studies of change in organizational and management: unveiling temporality, activity, and flow', *The Academy of Management Journal*, 56(1), pp. 1–13.
- Lee, S. (2014) 'The relationship between growth and profit: Evidence from firm-level panel data', *Structural Change and Economic Dynamics*. Elsevier B.V., 28, pp. 1–11. Available at: http://dx.doi.org/10.1016/j.strueco.2013.08.002.
- Lee, S. (2018) 'Growth, profits and R&D investment', *Economic Research-Ekonomska Istrazivanja*. Routledge, 31(1), pp. 607–625.
- Leitch, C., Hill, F. and Neergaard, H. (2010) 'Entrepreneurial and Business Growth and the Quest for a "Comprehensive Theory": Tilting at Windmills?', *Entrepreneurship: Theory & Practice*, 34(2), pp. 375–399.
- Léon, F. (2020) 'The elusive quest for high-growth firms in Africa: when other metrics of performance say nothing', *Small Business Economics*. Small

Business Economics.

- Lepak, D. P., Smith, K. G. and Taylor, M. S. (2007) 'Value creation and value capture: A multilevel perspective', *The Academy of Management Review*, 32(1), pp. 180–194.
- Levie, J. D. and Lichtenstein, B. B. (2010) 'A Terminal Assessment of Stages Theory: Introducing a Dynamic States Approach to Entrepreneurship', *Entrepreneurship: Theory & Practice*, 34(2), pp. 317–350.
- Liao, Y. W. *et al.* (2015) 'Understanding the dynamics between organizational IT investment strategy and market performance: A system dynamics approach', *Computers in Industry*. Elsevier B.V., 71, pp. 46–57.
- Lichtenstein, B. B. (2011) 'Complexity Science Contributions to the Field of Entrepreneurship', in *The Sage handbook of complexity and management*, pp. 473–495.
- Lichtenstein, B. B. and Plowman, D. A. (2009) 'The leadership of emergence: A complex systems leadership theory of emergence at successive organizational levels', *The Leadership Quarterly*, 20(4), pp. 617–630.
- Lipczynski, J. and Wilson, J. (2004) *The economics of business strategy*. Harlow: Financial Times Prentice Hall.
- Lockett, A. and Thompson, S. (2004) 'Edith Penrose's Contributions to the Resource-based View: An Alternative Perspective', *Journal of Management Studies*, 41(1), pp. 193–203.
- Lockett, A., Wiklund, J. and Davidsson, P. (2007) 'Re-Examining and Extending Penrose's Growth Theory: Updating Penrose for the 21st Century', *Frontiers of Entrepreneurship Research*, 27(13).
- Luna-Reyes, L. F. and Andersen, D. L. (2003) 'Collecting and analyzing qualitative data for system dynamics: Methods and models', *System Dynamics Review*, 19(4), pp. 271–296.
- Lyneis, J. M. and Pugh, A. I. (1996) 'Automated vs "Hand" Callibration of System Dynamics Models: An Experiment with a Simple Project Model', *Proceedings of the 1996 International System Dynamics Conference*, 1, pp. 321–324.
- Lyytinen, K., Yoo, Y. and Boland, R. J. (2016) 'Digital product innovation within four classes of innovation networks', *Information Systems Journal*, 26(1), pp. 47–75.

- Ma, D. and Kauffman, R. J. (2014) 'Competition Between Software-as-a-Service Vendors', *IEEE TRANSACTIONS ON ENGINEERING MANAGEMENT*. IEEE, 61(4), pp. 717–729.
- Maguire, S. *et al.* (2006) 'Complexity Science and Organizational Studies', in Clegg, S. R. et al. (eds) *The SAGE Handbook of Organization Studies*. 2nd edn. Thousand Oaks, pp. 165–214.
- Maichel-Guggemoos, L. and Wagner, J. (2019) 'Balancing growth, profitability and safety in the German insurance market', *Zeitschrift fur die gesamte Versicherungswissenschaft*, 108(5), pp. 421–442.
- Malerba, F. *et al.* (1999) "History-friendly" models of industry evolution: the computer industry', *Industrial and Corporate Change*, 8(1), pp. 3–40.
- March, J. G. (1991) 'Exploration and Exploitation in Organizational Learning', *Organization Science*, 2(1), pp. 71–87.
- Marion, T. J. *et al.* (2015) 'The evolution of interorganizational relationships in emerging ventures: An ethnographic study within the new product development process', *Journal of Business Venturing*. Elsevier Inc., 30(1), pp. 167–184.
- Markman, G. D. and Gartner, W. B. (2002) 'Is Extraordinary Growth Profitable? A Study of Inc. 500 High-Growth Companies', *Entrepreneurship: Theory & Practice*, 27(1), pp. 65–75.
- Marmer, M. *et al.* (2011) *Startup Genome Report Extra on Premature Scaling*. Available at: http://startupcompass.co.
- Marquez, A. C. and Blanchar, C. (2006) 'A Decision Support System for evaluating operations investments in high-technology business', *Decision Support Systems*, 41(2), pp. 472–487.
- Martinez-Moyano, I. J. and Richardson, G. P. (2013) 'Best practices in system dynamics modeling', *System Dynamics Review*, 29(2), pp. 102–123.
- McKelvey, B. (1999) 'Avoiding Complexity Catastrophe in Coevolutionary Pockets: Strategies for Rugged Landscapes', *Organization Science*, 10(3), pp. 294–321.
- McKelvey, B. (2004) 'Toward a complexity science of entrepreneurship', *Journal* of Business Venturing, 19(3), pp. 313–341.
- McKelvie, A. and Wiklund, J. (2010) 'Advancing Firm Growth Research: A Focus on Growth Mode Instead of Growth Rate', *Entrepreneurship: Theory*

and Practice, 34(2), pp. 261–288.

- Meadows, D. H. (1997) 'Leverage Points: Places to Intervene in a System', *Whole Earth*, 91(1), pp. 78–84.
- Meadows, D. H. (2009) *Thinking in Systems: A Primer*. Edited by D. Wright. London,: Earthscan.
- Mendelson, H. (2000) 'Organizational Architecture and Success in the Information Technology Industry', *Management Science*, 46(4), pp. 513–529.
- Meyer, A. D., Gaba, V. and Colwell, K. A. (2005) 'Organizing far from equilibrium: Nonlinear change in organizational fields', *Organization Science*, 16(5), pp. 456–473.
- Mikalef, P., Giannakos, M. and Pateli, A. (2013) 'Shopping and word-of-mouth intentions on social media', *Journal of Theoretical and Applied Electronic Commerce Research*, 8(1), pp. 17–34.
- Miles, M. B., Huberman, A. M. and Saldana, J. (2014) *Qualitative data analysis: A Methods Sourcebook, Qualitative Data Analysis.*
- Miller, D. and Friesen, P. H. (1984) 'A Longitudinal Study of the Corporate Life Cycle', *Management Science*, 30(10), pp. 1161–1183.
- Miller, K. and Tsang, E. (2010) 'Testing management theories: critical realist philosophy and research methods', *Strategic Management Journal*, 32(2), pp. 139-158.
- Miozzo, M. and DiVito, L. (2016) 'Growing fast or slow?: Understanding the variety of paths and the speed of early growth of entrepreneurial science-based firms', *Research Policy*. Elsevier B.V., 45(5), pp. 964–986.
- Möller, F. et al. (2020) 'Archetypes of digital business models in logistics startups', 27th European Conference on Information Systems - Information Systems for a Sharing Society, ECIS 2019, pp. 0–18.
- Monteiro, G. F. A. (2019) 'High-growth firms and scale-ups: a review and research agenda', *RAUSP Management Journal*, 54(1), pp. 96–111.
- Morecroft, J. D. W. (1982) 'A critical review of diagramming tools for conceptualizing feedback system models', *Dynamica*, pp. 20–29.
- Morecroft, J. D. W. (2015) *Strategic Modelling and Business Dynamics: A feedback systems approach.* 2nd edn. Chichester: John Wiley & Sons.
- Mudambi, R. and Treichel, M. Z. (2005) 'Cash crisis in newly public Internetbased firms: An empirical analysis', *Journal of Business Venturing*, 20(4), pp.

543-571.

- Muzellec, L., Ronteau, S. and Lambkin, M. (2015) 'Two-sided Internet platforms: A business model lifecycle perspective', *Industrial Marketing Management*. Elsevier Inc., 45(1), pp. 139–150.
- Nakano, A. and Kim, D. (2011) 'Dynamics of growth and profitability: The case of Japanese manufacturing firms', *Global Economic Review*, 40(1), pp. 67–81.
- Naldi, L. and Picard, R. G. (2012) "Let's start an online news site":Opportunities, resources, strategy, and formational myopia in startups', *Journal of Media Business Studies*, 9(4), pp. 69–97.
- Nambisan, S. (2017) 'Digital Entrepreneurship: Toward a Digital Technology Perspective of Entrepreneurship', *Entrepreneurship: Theory and Practice*, 41(6), pp. 1029–1055.
- Nambisan, S., Lyytinen, K. and Majchrzak, A. (2017) 'Digital Innovation Management: Reinventing Innovation Management Research in a Digital World', *MIS Quarterly*, 41(1), pp. 223–238.
- Naudé, W. and Liebregts, W. (2009) 'Digital Entrepreneurship Research: A Concise Introduction', *IZA Discussion Paper*, (13667).
- Nayak, A. and Chia, R. C. H. (2011) 'Thinking becoming and emergence:
 Process philosophy and organizational studies', in Tsoukas, H. and Chia, R. C.
 H. (eds) *Philosophy and organization theory*. Bingley: Emerald Group
 Publishing, pp. 281–309.
- Nelson, R. R. (1995) 'Recent Evolutionary Theorizing About Economic Change', *Journal of Economic Literature*, 33(1), pp. 48–90.
- Nelson, R. R. and Winter, S. G. (1978) Forces Generating and Limiting Concentration under Schumpeterian Competition, The Bell Journal of Economics.
- Nelson, R. R. and Winter, S. G. (1982) An evolutionary theory of economic change, An Evolutionary Theory of Economic Change. Cambridge: Harvard University Press.
- Nicholls-Nixon, C. L. (2005) 'Rapid Growth and High Performance: The Entrepreneur's "Impossible Dream?", *The Academy of Management Executive*, 19(1), pp. 77–89.
- Nikula, U. *et al.* (2010) 'Empirical validation of the Classic Change Curve on a software technology change project', *Information and Software Technology*.

Elsevier B.V., 52(6), pp. 680–696.

- Nkwor, N. N. and Ikpor, I. M. (2019) 'Firm size and growth: Testing gibrat's law in the Nigerian life insurance industry', *Pertanika Journal of Social Sciences and Humanities*, 27(3), pp. 2121–2142.
- Nzembayie, K. F., Buckley, A. P. and Cooney, T. (2019) 'Researching Pure Digital Entrepreneurship – A Multimethod Insider Action Research approach', *Journal of Business Venturing Insights*. Elsevier Inc., 11(November), p. e00103.
- Oestreicher-Singer, G. and Zalmanson, L. (2013) 'Content or Community? A Digital Business Strategy for Content Providers in the Social Age', *MIS Quartely*, 37(2), pp. 591–616.
- Ojala, A. (2016) 'Business models and opportunity creation: How IT entrepreneurs create and develop business models under uncertainty', *Information Systems Journal*, 26(5), pp. 451–476.
- Oliva, R. (2003) 'Model calibration as a testing strategy for system dynamics models', *European Journal of Operational Research*, 151(3), pp. 552–568.
- Oliva, R. (2004) 'Model structure analysis through graph theory: Partition heuristics and feedback structure decomposition', *System Dynamics Review*, 20(4), pp. 313–336.
- Oliva, R., Sterman, J. D. and Giese, M. (2003) 'Limits to growth in the new economy: Exploring the "get big fast" strategy in e-commerce', *System Dynamics Review*, 19(2), pp. 83–117.
- van Oorschot, K. E. *et al.* (2013) 'Anatomy of a decision trap in complex new product development projects', *Academy of Management Journal*, 56(1), pp. 285–307.
- Pagani, M. (2013) 'Digital business strategy and value creation: framing the dynamic cycle of control points', *MIS Quarterly*, 37(2), pp. 617–632.
- Paulk, M. C. et al. (1993) 'Capability Maturity Model for Software'.
- Peng, M. (2004) 'Outside directors and firm performance during institutional transitions', *Strategic Management Journal*, 25(5), pp. 453–471.
- Penrose, E. (1960) 'The Growth of the Firm: A Case Study: The Hercules Powder Company', *Business History Review*, 34(1), pp. 1–23.
- Penrose, E. (2009) *The Theory of the Growth of the Firm*. 4th edn. New York: Oxford University Press.

- Pentland, B. T. (1999) 'Building Process Theory with Narrative: From Description to Explanation', *The Academy of Management Review*, 24(4), pp. 711–724.
- Peteraf, M. A. (1993) 'The cornershones of competitive advantage: A resourcebased view', *Strategic Management Journal*, 14(3), pp. 179–191.
- Peteraf, M. A. and Barney, J. B. (2003) 'Unraveling the resource-based tangle', *Managerial and Decision Economics*, 24(4), pp. 309–323.
- Phelps, R., Adams, R. J. and Bessant, J. (2007) 'Life cycles of growing organizations: A review with implications for knowledge and learning', *International Journal of Management Reviews*, 9(1), pp. 1–30.
- Pitelis, C. N. (2009) 'Introduction by Christos N. Pitelis', in Penrose, E. (ed.) The Theory of the Growth of the Firm. 4th edn. Oxford University Press, pp. ix– xlvii.
- Porter, M. E. (1985) *Competitive Advantage: Creating and Sustaining Superior Performance*. New York: The Free Press.
- Porter, M. E. (1991) 'Towards a dynamic theory of strategy', *Strategic Management Journal*, 12(2), pp. 95–117.
- Porter, M. E. (1996) 'What is strategy?', *Harvard Business Review*, 74(11–12), pp. 44–55.
- Porter, M. E. (2008) 'The Five Competitive Forces That Shape Strategy', *Harvard Business Review*, 86(1), pp. 25–40.
- Power, R. (2018) *This Is How to Improve Retention at Your Startup*. Available at: https://www.forbes.com/sites/rhettpower/2018/11/25/this-is-how-to-improve-retention-at-your-startup/?sh=692be38d687d (Accessed: 12 August 2021).
- Prahalad, C. K. and Bettis, R. A. (1986) 'The dominant logic: A new linkage between diversity and performance', *Strategic Management Journal*, 7(6), pp. 485–501.
- Quinones, G. (2017) The Embeddedness of e-Entrepreneurship Institutional Constraints and Strategic Choice in Latin American Digital Start-ups. University of Manchester.
- Rahmandad, H. and Sterman, J. D. (2012) 'Reporting guidelines for simulationbased research in social sciences', *System Dynamics Review*, 28(4), pp. 396– 411.
- Rahmandad, H. and Weiss, D. M. (2009) 'Dynamics of concurrent software

development', System Dynamics Review, 25(3), pp. 224-249.

Ramadan, A. et al. (2016) Time To Market Cap: The New Metric That Matters.

- Rangaswamy, A. *et al.* (2020) 'The Role of Marketing in Digital Business Platforms', *Journal of Interactive Marketing*. Marketing EDGE.org., 51, pp. 72–90.
- Reid, G. C. (1995) 'Early Life-Cycle Behaviour of Micro-Firms in Scotland', Small Business Economics, 7(2), pp. 89–95.
- Remane, G. et al. (2016) 'A taxonomy of carsharing business models', 2016 International Conference on Information Systems, ICIS 2016, (December).
- Reuber, A. R. and Fischer, E. (2011) 'International entrepreneurship in internetenabled markets', *Journal of Business Venturing*. Elsevier Inc., 26(6), pp. 660–679.
- Ries, E. (2011) *The Lean Startup: How Constant Innovation Creates Radically Successful Businesses*. London: Portfolio Penguin.
- Roma, P. and Dominici, G. (2016) 'Understanding the price drivers of successful apps in the mobile app market', *International Journal of Electronic Marketing and Retailing*, 7(2), pp. 159–185.
- Romanelli, E. (1989) Environments and Strategies of Organization Start-Up: Effects on Early Survival, Source: Administrative Science Quarterly.
- Roper, S. (1999) 'Modelling Small Business Growth and Profitability', Small Business Economics, 13(3), pp. 235–252.
- Rosemann, M., Andersson, M. and Lind, M. (2011) 'Digital Complementary Assets', in *Proceedings of the International Conference on Information Systems 2011 (ICIS2011)*. Shanghai.
- Rotaru, K., Churilov, L. and Flitman, A. (2014) 'Philosophical positioning of discrete-event simulation and system dynamics as management science tools for process systems: A critical realist perspective', in Brailsford, S., Churivol, L., and Dangersfield, B. (eds) *Discrete-Event Simulation and System Dynamics for Management Decision Making*. 1st edn. Wiley, pp. 76–104.
- Rowbottom, N. and Lymer, A. (2010) 'Exploring the use and users of narrative reporting in the online annual report', *Journal of Applied Accounting Research*, 11(2), pp. 90–108.
- Runge, J. *et al.* (2014) 'Churn prediction for high-value players in casual social games', *IEEE Conference on Computatonal Intelligence and Games, CIG.*

IEEE.

- Russo, M. V. and Fouts, P. A. (1997) 'A Resource-Based Perspective on Corporate Environmental Performance and Profitability', *The Academy of Management Journal*, 40(3), pp. 534–559.
- Ruutu, S., Casey, T. and Kotovirta, V. (2017) 'Development and competition of digital service platforms: A system dynamics approach', *Technological Forecasting and Social Change*. The Authors, 117, pp. 119–130.
- Saaksjarvi, M. and Lassila, A. (2005) 'Role of the customer value in the software as a service concept', in Funabashi, M. and Grzech, A. (eds) *Challenges of Expanding Internet: E-Commerce, E-Business, and E-Government.* 189th edn. Boston, MA.: Springer, p. IFIP International Federation for Information Proc.
- Sahut, J. M., Iandoli, L. and Teulon, F. (2019) 'The age of digital entrepreneurship', *Small Business Economics*, (September).
- Santos, F. M. and Eisenhardt, K. M. (2009) 'Constructing Markets and Shaping Boundaries: Entrepreneurial Power in Nascent Fields', *Academy of Management Journal*, 52(4), pp. 643–671.
- Sapienza, H. J. *et al.* (2006) 'A capabilities perspective on the effects of early internationalization on firm survival and growth', *Academy of Management Review*, 31(4), pp. 914–933.
- Sargent, R. G. (2013) 'Verification and validation of simulation models', *Journal of Simulation*, 7(1), pp. 12–24.
- Saunders, M., Lewis, P. and Thornhill, A. (2012) *Research methods for business students*. 6th edn. Harlow: Pearson Education.
- Schlichter, J., Klyver, K. and Haug, A. (2021) 'The moderating effect of ERP system complexity on the growth–profitability relationship in young SMEs', *Journal of Small Business Management*. Routledge, 59(4), pp. 601–626.
- Schumpeter, J. (1928) The Instability of Capitalism, The Economic Journal.
- Schwaninger, M. and Grösser, S. (2008) 'System Dynamics as Model-Based Theory Building', Systems Research and Behavioral Science, 25(4), pp. 447– 465.
- Senderovitz, M., Klyver, K. and Steffens, P. (2016) 'Four years on: Are the gazelles still running? A longitudinal study of firm performance after a period of rapid growth', *International Small Business Journal: Researching Entrepreneurship*, 34(4), pp. 391–411.

- Simbaña-Taipe, L. E. et al. (2019) 'Key determinants for growth in high-growth Ecuadorian manufacturing firms', *International Journal of Management and Enterprise Development*, 18(4), pp. 293–315.
- Sirmon, D. G. and Hitt, M. A. (2003) 'Managing Resources: Linking Unique Resources, Management, and Wealth Creation in Family Firms', *Entrepreneurship: Theory & Practice*, 27(4), pp. 339–358.
- Slater, M. (1980) 'The Managerial Limitation to the Growth of Firms', *The Economic Journal*, 90(359), pp. 520–528.
- Smith, D. (2015) *Exploring Innovation*. 3rd edn. New York: McGraw-Hill Higher Education.
- Song, M., Parry, M. E. and Kawakami, T. (2009) 'Incorporating network externalities into the technology acceptance model', *Journal of Product Innovation Management*, 26(3), pp. 291–307.
- Sosa, M. E., Browning, T. and Mihm, J. (2007) 'Studying the dynamics of the architecture of software products', in *Proceedings of the ASME 2007 International Design Engineering Technical Conferences & Computers and Information in Engineering Conference.*
- Stam, E. and Garnsey, E. (2006) 'New Firms Evolving in the Knowledge Economy; problems and solutions around turning points', in Dolfsma, W. and Soete, L. (eds) Understanding the Dynamics of a Knowledge Economy. Edward Elgar Publishing.
- Stam, E. and Garnsey, E. (2007) 'Entrepreneurship in the knowledge economy', in Bessant, J and Venables, T. (eds) *Creating Wealth from Knowledge: Meeting the Innovation Challenge*. Edward Elgar Publishing, pp. 145–176.
- Stam, E., Garnsey, E. and Heffernan, P. (2006) 'A Penrosean Theory of the Firm: Implications and Applications for the Study of the Growth of Young Firms', in Dietrich, M. (ed.) *Economics of the Firm: Analysis, Evolution, History*. Milton Park: Taylor & Francis, pp. 224–262.
- Stampfl, G., Prügl, R. and Osterloh, V. (2013) 'An explorative model of business model scalability', *International Journal of Product Development*, 18(3/4), pp. 226–248.
- Standing, C. and Mattsson, J. (2016) "Fake it until you make it": business model conceptualization in digital entrepreneurship', *Journal of Strategic Marketing*. Routledge, 26(5), pp. 385–399.

- Stangler, D. (2010) High-Growth Firms and the Future of the American Economy, Kauffman Foundation Research Series: Firm Formation and Economic Growth.
- Staykova, K. S. and Damsgaard, J. (2015) 'A Typology of Multi-sided Platforms: The Core and the Periphery', in *Twenty-Third European Conference on Information Systems (ECIS)*.
- Steffens, P., Davidsson, P. and Fitzsimmons, J. (2009) 'Performance configurations over time: implications for growth- and profit-oriented strategies', *Entrepreneurship: Theory & Practice*, 33(1), pp. 125–148.
- Steininger, D. M. (2019) 'Linking information systems and entrepreneurship: A review and agenda for IT-associated and digital entrepreneurship research', *Information Systems Journal*, 29(2), pp. 363–407.
- Steininger, D. M., Wunderlich, P. and Pohl, F. (2013) 'Exploring competitive advantage of social networking sites: A business model perspective', in ECIS 2013 Proceedings of the 21st European Conference on Information Systems.
- Sterman, J. D. (1991) A Skeptic's Guide to Computer Models, Managing a Nation: The Microcomputer Software Catalog.
- Sterman, J. D. (2000) Business Dynamics Systems Thinking and Modeling for a Complex World. New York: McGraw-Hill Education.
- Sterman, J. D. (2001) 'System Dynamics Modeling: Tools for Learning in a Complex World', *California Management Review*, 43(4), pp. 8–25.
- Sterman, J. D. (2002) 'All models are wrong: Reflections on becoming a systems scientist', System Dynamics Review, 18(4), pp. 501–531.
- Sterman, J. D. (2005) *dt/time step in the real world*, *Ventana Systems*. Available at: http://www.ventanasystems.co.uk/forum/viewtopic.php?p=6534 (Accessed: 31 October 2020).
- Sterman, J. D. *et al.* (2007) 'Getting Big Too Fast: Strategic Dynamics with Increasing Returns and Bounded Rationality', *Management Science*, 53(4), pp. 683–696.
- Sterman, J. D. (2015) 'Booms, Busts, and Beer: Understanding the Dynamics of Supply Chains', in Bendoly, E., van Wezel, W., and Bachrach, D. (eds) *The Handbook of Behavioral Operations Management: Social and Psychological Dynamics in Production and Service Settings*. New York: Oxford University Press, pp. 203–237.

- Sterman, J. D. (2018) 'System dynamics at sixty: the path forward', System Dynamics Review, 34(1-2), pp. 5-47.
- Stuart, R. W. and Abetti, P. A. (1990) 'Impact of entrepreneurial and management experience on early performance', *Journal of Business Venturing*, 5(3), pp. 151–162.
- Stubbart, C. I. and Smalley, R. D. (1999) 'The Deceptive Allure of Stage Models of Strategic Processes', *Journal of Management Inquiry*, 8(3), pp. 272–286.
- Tan, D. and Mahoney, J. T. (2005) 'Examining the Penrose effect in an international business context: The dynamics of Japanese firm growth in US industries', *Managerial and Decision Economics*, 26(2), pp. 113–127.
- Tang, H. K. (2006) 'Adoption of navigation technologies: Five historical and contemporary cases', *Technological Forecasting and Social Change*, 73(7), pp. 845–859.
- Täuscher, K. (2018) 'Using qualitative comparative analysis and system dynamics for theory-driven business model research', *Strategic Organization*, 16(4), pp. 470–481.
- Täuscher, K. and Abdelkafi, N. (2018) 'Scalability and robustness of business models for sustainability: A simulation experiment', *Journal of Cleaner Production*, 170, pp. 654–664.
- Taylor, A. J. (1980) 'Loop Analysis and Model Validation', *Dynamica*, 6(1), pp. 2–8.
- Teece, D. J. (1986) 'Profiting from technological innovation: Implications for integration, collaboration, licensing and public policy', *Research Policy*, 15(2), pp. 285–305.
- Teece, D. J. (1998) 'Capturing Value from Knowledge Assets', *California Management Review*, 40(3), pp. 55–79.
- Teece, D. J. (2007) 'Explicating dynamic capabilities: The nature and microfoundations of (sustainable) enterprise performance', *Strategic Management Journal*, 28, pp. 1319–1350.
- Teece, D. J. (2010) 'Business models, business strategy and innovation', *Long Range Planning*. Elsevier Ltd, 43(2–3), pp. 172–194.
- Teece, D. J. (2018) 'Profiting from innovation in the digital economy: Enabling technologies, standards, and licensing models in the wireless world', *Research Policy*. Elsevier, 47(8), pp. 1367–1387.

- Teece, D. J. and Linden, G. (2017) 'Business models, value capture, and the digital enterprise', *Journal of Organization Design*. Journal of Organization Design, 6(1), pp. 1–14.
- Teece, D. J., Pisano, G. and Shuen, A. (1997) 'Dynamic capabilities and strategic management', *Strategic Management Journal*, 18(7), pp. 509–533.
- Teece, D. and Pisano, G. (1994) 'The dynamic capabilities of firms: An introduction', *Industrial and Corporate Change*, 3(3), pp. 537–556.
- Thiel, P. and Masters, B. (2014) Zero to One. London: Penguin Random House UK.
- Tidd, J., Bessant, J. and Pavitt, K. (1994) Managing Innovation: Integrating Technological, Market and Organizational Change. 3rd edn, Credit Control in Boom and Recession. 3rd edn. Chichester,: John Wiley & Sons.
- Tien, J. J. and Yang, S. S. (2014) 'The determinants of life insurer's growth for a developing insurance market: Domestic vs foreign insurance firms', *Geneva Papers on Risk and Insurance: Issues and Practice*, 39(1), pp. 1–24.
- Tong, Y. and Serrasqueiro, Z. (2020) 'A study on the influence of financial factors on the growth of small and medium-sized enterprises in portuguese high technology and medium-high technology sectors', *WSEAS Transactions on Business and Economics*, 17, pp. 703–716.
- Tranfield, D., Denyer, D. and Smart, P. (2003) 'Towards a Methodology for Developing Evidence-Informed Management Knowledge by Means of Systematic Review', *British Journal of Management*, 14(3), pp. 207–222.
- Tsoukas, H. (2017) 'Don't Simplify, Complexify: From Disjunctive to Conjunctive Theorizing in Organization and Management Studies', *Journal of Management Studies*, 54(2), pp. 132–153.
- Tsoukas, H. and Chia, R. (2002) 'On Organizational Becoming: Rethinking Organizational Change', *Organization Science*, 13(5), pp. 567–582.
- Tsoukas, H. and Hatch, M. J. (2001) 'Complex thinking, complex practice: The case of a narrative approach to a organizational complexity', *Human Relations*, 54(8), pp. 979–1013.
- Tucker, C. (2018) 'Network Effects and Market Power: What Have We Learned in the Last Decade?', *Antitrust*, Spring, pp. 72–29.
- Tyrväinen, P. and Selin, J. (2011) 'How to sell SaaS: A model for main factors of marketing and selling software-as-a-service', *Lecture Notes in Business*

Information Processing, 80 LNBIP, pp. 2–16.

- Van de Ven, A. H. (1992) 'Suggestions for Studying Strategy Process: A Research Note', *Strategic Management Journal*, 13(1), pp. 169–191.
- Van De Ven, A. H. and Huber, G. P. (1990) 'Longitudinal Field Research Methods for Studying Processes of Organizational Change', *Organization Science*, 1(3), pp. 213–219.
- Van De Ven, A. H. and Poole, M. S. (2005) 'Alternative Approaches for Studying Organizational Change', *Organization Studies*, 26(9), pp. 1377–1404.
- Ventana Systems (no date) 'Macros'. Available at: https://www.vensim.com/documentation/macros.html.
- Vhzquez, M., Liz, M. and Aracil, J. (1996) 'Knowledge and reality: some conceptual issues in system dynamics modeling', *System Dynamics Review*, 12(1), pp. 21-37.
- Voigt, S. and Hinz, O. (2016) 'Making Digital Freemium Business Models a Success: Predicting Customers' Lifetime Value via Initial Purchase Information', *Business and Information Systems Engineering*. Springer Fachmedien Wiesbaden, 58(2), pp. 107–118.
- Wagner, T. M., Benlian, A. and Hess, T. (2014) 'Converting freemium customers from free to premium—the role of the perceived premium fit in the case of music as a service', *Electronic Markets*, 24(4), pp. 259–268.
- Walker, R. and Wakeland, W. (2011) Calibration of Complex System Dynamics Models: A Practitioner's Report, Systems Science Faculty Publications and Presentations. 71. Available at:

http://www.systemdynamics.org/conferences/2011/proceed/papers/P1118.pdf.

- Warren, K. (1999) 'The dynamics of rivalry', *Business Strategy Review*, 10(4), pp. 41–54.
- Warren, K. (2000) 'The Softer Side of Strategy Dynamics', *Business Strategy Review*, 11(1), pp. 45–58.
- Warren, K. (2002) *Competitive Strategy Dynamics*. Baffins Lane: John Wiley & Sons.
- Wayne, M. L. (2018) 'Netflix, Amazon, and branded television content in subscription video on-demand portals', *Media, Culture and Society*, 40(5), pp. 725–741.

Wernerfelt, B. (1984) 'A resource-based view of the firm', Strategic Management

Journal, 5(2), pp. 171-180.

- Winch, G. and Arthur, D. (2002) 'User-parameterised generic models: A solution to the conundrum of modelling access for SMEs?', *System Dynamics Review*, 18(3), pp. 339-357.
- Wohlin, C. (2014) 'Guidelines for Snowballing in Systematic Literature Studies and a Replication in Software Engineering', in *Proceedings of the 18th international conference on evaluation and assessment in software engineering*.
- Wolstenholme, E. F. (1999) 'Qualitative vs quantitative modelling: The evolving balance', *Journal of the Operational Research Society*, 50(4), pp. 422–428.
- Woodard, C. J. et al. (2013) 'Design Capital and Design Moves: The Logic of Digital Business Strategy', MIS Quarterly, 37(2), pp. 537–564.
- Yadav, I. S., Pahi, D. and Gangakhedkar, R. (2021) 'The nexus between firm size, growth and profitability: new panel data evidence from Asia–Pacific markets', *European Journal of Management and Business Economics*.
- Yazdanfar, D. and Öhman, P. (2015) 'The growth-profitability nexus among Swedish SMEs', *International Journal of Managerial Finance*, 11(4), pp. 531–547.
- Yearworth, M. (2010) 'Inductive Modelling of an Entrepreneurial System', in 28th International Conference of the System Dynamics Society.
- Yin, R. K. (1981) 'The Case Study as a Serious Research Strategy', Science Communication, 3(1), pp. 97–114.
- Yoo, S. and Kim, J. (2015) 'The dynamic relationship between growth and profitability under long-term recession: The case of Korean construction companies', *Sustainability (Switzerland)*, 7(12), pp. 15982–15998.
- Zaheer, H. *et al.* (2019) 'Straight from the horse's mouth: Founders' perspectives on achieving "traction" in digital start-ups', *Computers in Human Behavior*. Elsevier Ltd, 95, pp. 262–274.
- Zaheer, H. (2020) *Digital Entrepreneurship: Review, Trajectory and Practice.*Macquarie University.
- Zaheer, H., Breyer, Y. and Dumay, J. (2019) 'Digital entrepreneurship: An interdisciplinary structured literature review and research agenda', *Technological Forecasting and Social Change*. Elsevier, 148(June), p. 119735.

Zahra, S. A., Sapienza, H. J. and Davidsson, P. (2006) 'Entrepreneurship and

dynamic capabilities: A review, model and research agenda', *Journal of Management Studies*, 43(4), pp. 917–955.

- Zhang, J. J., Lichtenstein, Y. and Gander, J. (2015) 'Designing scalable digital business models', in Baden-Fuller, C. and Mangematin, V. (eds) Advances in Strategic Management. Emerald Press, pp. 241–277.
- Zhang, J. and Niu, B. (2013) 'Optimal Strategy When Selling Software-as-a-Service'. IEEE.
- Zhao, Y. L., Song, M. and Storm, G. L. (2013) 'Founding Team Capabilities and New Venture Performance: The Mediating Role of Strategic Positional Advantages', *Entrepreneurship: Theory and Practice*, 37(4), pp. 789–814.
- Zhou, N., Park, S. H. and Ungson, G. R. (2013) 'Profitable growth: Avoiding the "growth fetish" in emerging markets', *Business Horizons*. 'Kelley School of Business, Indiana University', 56(4), pp. 473–481.
- Zhu, F. and Iansiti, M. (2007) 'Dynamics of platform competition: Exploring the role of installed base, platform quality and consumer expectations', in *ICIS* 2007 Proceedings - Twenty Eighth International Conference on Information Systems.
- Zhu, Y. and Bao, Z. (2018) 'The role of negative network externalities in SNS fatigue: An empirical study based on impression management concern, privacy concern, and social overload', *Data Technologies and Applications*, 52(3), pp. 313–328.
- Zott, C. (2003) 'Dynamic capabilities and the emergence of intraindustry differential firm performance: Insights from a simulation study', *Strategic Management Journal*, 24(2), pp. 97–125.
- Zott, C. and Amit, R. (2007) 'Business Model Design and the Performance of Entrepreneurial Firm', *Organization Science*, 18(2), pp. 181–199.
- Zott, C. and Amit, R. (2010) 'Business model design: An activity system perspective', *Long Range Planning*. Elsevier Ltd, 43(2–3), pp. 216–226.
- Zott, C., Amit, R. and Massa, L. (2011) 'The business model: Recent developments and future research', *Journal of Management*, 37(4), pp. 1019– 1042.

The development paths and relationships of value creation and capture: an investigation of digital ventures using System Dynamics

Appendices

Martin Alexander Lycko
Table of con	tent	xix
List of figur	es	xxii
List of table	s	xxiii
APPENDIX	X A: ADDITIONAL MODEL DETAILS	
A.1. Valu	e creation subsystem	
A.1.1. Va	lue creation for users and customers	
A.1.2. Va	lue drivers determining the product strength for users	
A.1.2.1.	Network effects for users	
A.1.2.2.	Content quality	289
A.1.3. Us	er usage intensity	
A.2. Value	e capture subsystem	
A.2.1. Sha	are of value captured from customers	
A.2.1.1.	Bargaining power to premium users	293
A.2.1.2.	User switching costs	295
A.3. Cust	omer and user subsystem	
A.3.1. Us	er and premium user lifetime	
A.3.2. Ca	pability development	
A.3.2.1.	Marketing and selling capabilities	299
A.3.2.2.	Value delivering capabilities	
A.4. Firm	managing subsystem	
A.4.1. Ma	nagement of human resources	
A.4.1.1.	Additional employee stocks	305
A.4.1.2.	Target value delivering employees	308
A.4.1.3.	Target technology developing employees	
A.4.1.4.	Target firm managing employees	
A.4.1.5.	Further details on target marketing and selling employees	
A.4.1.6.	Focus on servicing customers and users and creating content	
A.4.2. Ma	nagement of complementary assets	
A.4.2.1.	Delay to acquire complementary assets	320
A.4.3. Ma	nagement of financial resources	
A.4.3.1.	Additional liquidity requirements	321
A.4.3.2.	Delay to raise capital	
A.4.4. Fir	m management capabilities	
A.5. Cont	ent creation subsystem	

Table of content

A.5.1.	Content library	
A.5.1.1. Internal content creation		
А		
A.5.2.	Content quality	
A.5.3.	Content assets	
A.5.4.	Overview of subsystem inputs	
APPI	ENDIX B: CUSTOM VENSIM MACROS	
B.1.	HasValue	
B.2.	RollingAnnualFinancial	
APP	ENDIX C: OVERVIEW OF MODEL INPUTS	
C.1.	Subsystem outcomes	
C.2.	Manually set variables	
C.2.1.	Business model design	
C.2.2.	Environment	
C.2.3.	Dominant logic	
C.2.4.	Capability development	
C.2.5.	Accounting measures	
C.3 .	Automatically calibrated variables	
C.4 .	Fixed variables	
APP	ENDIX D: ADDITIONAL COMPANY INPUTS	
D.1 .	Additional inputs for all companies	
D.2 .	Additional inputs for Alpha	
D.3 .	Additional inputs for Beta	
D.4 .	Additional inputs for Gamma	
D.5 .	Additional inputs for Delta	
A DDI	NDIV E. OVEDVIEW OF SCENADIO INDUTS	259
	Daga aaga madal inguta	
E.I. F 2	Dase case model inputs	
Е.2.	Dominant logics with a cap on employee numbers	
E.2.1.	Different targets expressed in the dominant logic	
E.2.2.	Internal improvement for capability development	

E.3 .	Domi	nant logics without a cap on employee numbers	
E.3.1.	Em	ployee-uncapped base case	
E.3.2.	Dif	ferent targets expressed in the dominant logic	
E.3.3.	Inte	ernal improvement for capability development	
E.4 .	Trans	sitioning between dominant logics	
E.4.1.	Tra	nsitioning base case	
E.4.2.	Dif	ferent targets expressed in the dominant logic	
E	.4.2.1.	Different targets throughout the simulation	
E	.4.2.2.	Different restructuring magnitudes	
E	.4.2.3.	Different number of growth cycles	
E.4.3.	Inte	ernal improvement for capability development	
E.5.	Conte	extual influences and robustness	
E.5.1.	Dif	ferent environmental conditions	
E	.5.1.1.	Stable environments	
E	.5.1.2.	Dynamics environments	
E.5.2.	Dif	ferent business model designs	
E	.5.2.1.	Business models with network effects	
E	.5.2.2.	Business models with different efficiency levels	

List of figures

Figure A-1: Rates of value creation for users and premium users	286
Figure A-2: Strength of product for users	287
Figure A-3: Network effects for users	289
Figure A-4: Content quality	290
Figure A-5: User usage intensity	291
Figure A-6: Bargaining power to premium users	293
Figure A-7: User switching costs	296
Figure A-8: User and premium user lifetimes	297
Figure A-9: Marketing and selling capabilities	300
Figure A-10: Value delivering capabilities	302
Figure A-11: Human resource stocks	305
Figure A-12: Target value delivering employees	309
Figure A-13: Target technology developing employees	312
Figure A-14: Target firm managing employees	314
Figure A-15: Target marketing and selling employees	316
Figure A-16: Focus of value delivering employees	319
Figure A-17: Delay to acquire complementary assets	320
Figure A-18: Additional liquidity requirements	321
Figure A-19: Delay to raise capital	322
Figure A-20: Firm managing capabilities	323
Figure A-21: Content library	326
Figure A-22: Internal content creation	327
Figure A-23: External content creation	329
Figure A-24: Quality of new content items	330
Figure A-25: Content library	332

List of tables

Table A-1: Outputs and inputs of the content creation subsystem
Table C-1: Measurement of subsystem outcomes 338
Table C-2: Measurement of business model design inputs
Table C-3: Measurement of customer and user inputs 342
Table C-4: Measurement of competitor inputs
Table C-5: Measurement of input provider inputs 343
Table C-6: Measurement of complementary product inputs
Table C-7: Measurement of target and limit inputs
Table C-8: Measurement of human resource inputs
Table C-9: Measurement of asset requirement inputs
Table C-10: Measurement of capability development inputs
Table C-11: Measurement of accounting inputs
Table C-12: Automatically calibrated standard values
Table C-13: Automatically calibrated effect sizes
Table C-14: Automatically calibrated monetary scale adjustment
Table C-15: Overview of fixed variables
Table D-1: Additional dominant logic inputs for all companies
Table D-2: Content creation subsystem inputs for all companies
Table D-3: Initial values and accounting measures for Alpha
Table D-4: Initial values and accounting measures for Beta
Table D-5: Initial values and accounting measures for Gamma
Table D-6: Initial values and accounting measures for Delta
Table E-1: Firm management inputs for the base case
Table E-2: Technology developing inputs for the base case
Table E-3: Customer and user subsystem inputs for the base case
Table E-4: Value creation inputs for the base case
Table E-5: Value capture inputs for the base case
Table E-6: Content creation subsystem inputs for the base case
Table E-7: Initial values and accounting measures for the base case
Table E-8: Changes to employee-capped base case's dominant logic
Table E-9: Changes to employee-capped base case's capability development 364
Table E-10: Changes to develop BaseU 365
Table E-11: Changes to employee-uncapped base case's dominant logic

Table E-12: Changes to employee-uncapped base case's capability development
Table E-13: Changes to develop BaseT
Table E-14: Changes to transitioning base case's dominant logic's targets (1). 367
Table E-15: Changes to transitioning base case's dominant logic's targets (2). 367
Table E-16: Changes to transitioning base case's dominant logic's restructuring
Table E-17: Changes to transitioning base case's dominant logic's growth cycles
Table E-18: Changes to transitioning base case's capability development (1)369
Table E-19: Changes to transitioning base case's capability development (1)369
Table E-20: Changes to reflect differently attractive stable environments 370
Table E-21: Changes to reflect different environmental dynamics 371
Table E-22: Changes to reflect business model with different network effects. 371
Table E-23: Changes to reflect business model with different capability effect
sizes

Appendix A: Additional model details

This appendix outlines additional details of the formal model. The additional details described in this appendix reproduce different structures already explained in depth in Chapter 4 of this thesis. For example, the main body of the thesis described technology developing capabilities. This appendix uses the described structure and equations to formalise capabilities related to other activities. Together, the details in this appendix and Chapter 4 of the thesis illustrate all structures and equations required to reproduce the formal model. The appendix follows the same structure of subsystems utilised in Chapter 4. It first outlines additional model elements for the value creation and value capture subsystems. This appendix then presents additional details for the user and customer, firm managing, and content creation subsystems²⁹.

A.1. Value creation subsystem

The main body of the thesis has split digital ventures' value creation into the value created for customers and the value created for users. It has then further modelled the value created for customers. It depends on the customer base, product strength for customers, and customer usage intensity. The section noted that the mechanisms to create value for users and premium users follow the same structures. This section adapts and describes these structures and equations for users and premium users.

A.1.1. Value creation for users and customers

The main body of the thesis modelled the rate of value creation for customers (Subsection 4.1.2). However, digital ventures may distinguish between customers, users, and premium users (Afuah and Tucci, 2000; Kollmann, 2006; Teece, 2010; Zott, Amit and Massa, 2011; Roma and Dominici, 2016; Voigt and Hinz, 2016; Möller *et al.*, 2020). Thus, this subsection adapts the main body's structure to

²⁹ Chapter 4 of the thesis described the complete technology development subsystem.

model the rate at which digital ventures create value for users and premium users (Figure A-1). The value created for customers depends on the customer base and the rate of use value creation for customers. Therefore, the rates of value creation for users and premium users depend on their respective stocks and rates of use value creation. The use value depends on the product strength, the usage intensity, and the advantage of the premium product for its users.



Figure A-1: Rates of value creation for users and premium users

The model adapts the equations for customers in the main body. Thus, it calculates the *rate of value creation for users* and the *rate of value creation for premium users* by multiplying together the respective rates of use value creation and user bases (Equation A-1, A-2).

rate of value creation for users		Eq A-1
	= User base * rate of use value creation for users	
rate of value creation for premium users		Eq A-2
	= Premium user base	
	* rate of use value creation for premium users	

The customer and user subsystem in the main body has already modelled the user bases. The model calculates the *rate of use value creation for users* in the same manner as the use value for customers. The model thus derives it by multiplying together the *user usage intensity* and the *strength of product for users* (Equation A-3). In exchange for paying for the product, premium users get access to additional or improved features. Therefore, they generate value from a product at a higher rate (Teece, 2010; Roma and Dominici, 2016; Voigt and Hinz, 2016; Möller *et al.*, 2020). This higher rate is reflected in the *premium product* *advantage*. It captures the superiority of the premium product over the base version of the product or service. This advantage is multiplied with the use value for users to derive the *rate of user value creation for premium users* (Equation A-4).

rate of use value creation for users	
= User usage intensity * strength of product for users	
rate of value creation for premium users	Eq A-4
= rate of use value creation for users	-
* premium product advantage	

The product strength for users and the user usage intensity are further reviewed below.

A.1.2. Value drivers determining the product strength for users

The main body of the thesis has outlined different drivers of value creation in a digital context for customers (Subsection 4.1.3). These are the technological quality, content quality, direct and indirect network effects, and the quality of complementary products. These value drivers also apply to users and determine the product's strength for them (Figure A-2).



Figure A-2: Strength of product for users

Adapting the equation used to calculate the product strength for customers, the *product strength for users* depends on these value drivers. Each value driver has a value of one if the venture uses it and zero if the venture does not employ it. The model multiplies together these value drivers after confirming their value with the custom HasValue function (Equation A-5). The custom macro returns one if a value driver is not utilised to ensure that the unemployed drivers do not affect the multiplication.

Eq A-5

strength of product or service for customers

- = HasValue(Technological quality, 1)
- * HasValue(Content quality, 1)
- * HasValue(direct network effects for users, 1)
- * HasValue(indirect network effects for users, 1)
- * HasValue(Complementary quality, 1)

The main body of the thesis has modelled the technological quality (Subsubsection 4.1.3.1). The complementary quality is an exogenous input. The network effects for users and the content quality are modelled below. The model indexes these value drivers and thus the product strength for users in the same manner as the product strength for customers. All employed value drivers and product strengths have an initial value of one. This indexes the initial rate of value creation at one too.

A.1.2.1. Network effects for users

Users may be affected by positive and negative as well as direct and indirect network effects (Stampfl, Prügl and Osterloh, 2013; Staykova and Damsgaard, 2015; Gandia and Parmentier, 2017; Täuscher and Abdelkafi, 2018; Zhu and Bao, 2018). While direct network effects depend on the venture's user base, indirect network effects depend on its customer base. The initial bases normalise the network effects. It indexes them at an initial value of one. Moreover, the model uses policy switches to deactivate and activate positive and negative network effects (Figure A-3). Thus, the model uses the same structure and equations introduced in the main body of the thesis for customers (Subsubsection 4.1.3.2),



Figure A-3: Network effects for users

The model adapts the equations for customer network effects. It thus calculates the user network effects using an if-clause to check if a network effect is present. These are expressed in the respective *network effect settings*. If the respective setting is zero, there are no network effects, and they are thus zero. If network effect settings are positive (1), the functions increase network effect settings are negative (-1), the functions create a decline in network effects proportional to increases in the user or customer bases (Equation A-6, A-7).

direct network effects for usersEq A-6= IF THEN ELSE(user direct network effect setting== 0, 0, ((Premium user base + User base)/(initial user base++ initial premium user base))^user direct network effect setting)Eq A-7indirect network effects for usersEq A-7= IF THEN ELSE(user indirect network effect setting== 0, 0, (Customer base//initial customer base)^user indirect network effect setting)

The customer and user subsystem in Chapter 4 of this thesis has modelled the user and customer bases. The analyst sets the network effect settings as exogenous policy switches (Coyle, 1996).

A.1.2.2. Content quality

The content quality is a value driver for digital ventures that offer content-based products or services (Saaksjarvi and Lassila, 2005; Kim, Oh and Shin, 2010; Bradley *et al.*, 2012; Ojala, 2016; Kim and Kim, 2017; Cristofaro, 2020).

Examples of such business models are online newspapers and streaming services (Afuah and Tucci, 2000; Asvanund *et al.*, 2004; Saaksjarvi and Lassila, 2005; Mikalef, Giannakos and Pateli, 2013; Koch and Windsperger, 2017; von Briel, Recker and Davidsson, 2018). This value driver was included in the model because it has been identified in the literature on digital ventures (see Subsection 4.1.3). It is presented in this appendix because it is not employed by any of the case companies identified through the extensive and rigorous sampling process. Moreover, its structure is similar to the technological quality. The model uses the *content quality* to reflect the quality of the average content item offered by the venture. As all employed value drivers, the initial content quality has a value of one to index the rate of value creation. Like the technological quality, the content quality improves over time. While the technological quality improves along the technology s-curve towards its maximum, the content quality improves towards the *quality of new content items* depending on the *fraction of new content items* (Figure A-4).



Figure A-4: Content quality

The model uses the same structure employed repeatably in this thesis to update averages like the usage intensities or capabilities. The model thus relies on verified and validated structured and equations. It adjusts the *content quality* through the *rate of content quality improvements* (Equation A-8). This variable moves the content quality towards the *quality of new content items* depending on the *fraction of new content items* (Equation A-9, A-10).

Content quality = initial content quality
$$Eq A-8$$

+ (rate of content quality improvements)

rate of content quality improvements	
= (quality of new content items – Content quality)	
* fraction of new content items	
fraction of new content items	
= XIDZ(rate of content creation, Content library, 1)	

The content creation subsystem in this appendix presents the content library, rate of content creation, and quality of new content items.

A.1.3. User usage intensity

How intensively users make use of a digital product or service affects the value they derive from it. As for customers, this usage intensity changes over time due to learning, personalisation, and feature adoption (Amit and Zott, 2001; Kim, Oh and Shin, 2010; DaSilva *et al.*, 2013; Deng and Wang, 2016; Tucker, 2018). The model uses the customer usage intensity structure (Subsection 4.1.4) to model the user usage intensity. It increases for existing users and falls when new users are acquired (Figure A-5).



Figure A-5: User usage intensity

The model applies the customer's structure and equations to users. The *user usage intensity* accumulates based on *improvements in usage intensity for existing users* and *reductions in usage intensity* for new users (Equation A-11). The

improvements are modelled using Sterman's (2000) s-curve that adjusts a standard rate (Equation A-12). The adjustments for new users depend on the *new user usage intensity* and the *fraction of new users* (Equation A-13, A-14).

User usage intensity		Eq A-11
	= initial user usage intensity	
	+ \int (rate of improvements in user usage intensity	
	 rate of reduction in user usage intensity) 	
rate of improv	ements in user usage intensity	Eq A-12
	= standard rate of adoption by users	
	* User usage intensity	
	* ZIDZ(maximum user usage intensity	
	 User usage intensity, maximum user usage intensity) 	
rate of reduction	on in user usage intensity	Eq A-13
	= (User usage intensity – new user usage intensity)	
	* fraction of new users	
fraction of new users		Eq A-14
	= XIDZ(rate of user acquisition	
	+ rate of premium user acquisition, User base	
	+ Premium user base, 1)	

The *initial user usage intensity* has a value of one to index value creation. As for the customer usage intensity, analysts may use different data points to approximate the usage intensity. Please refer to customer usage intensity in the main body of the thesis (Subsection 4.1.4) or Appendix C for a list of examples. The analyst then uses these data points to calibrate the maximum usage intensity, new user usage intensity, and speed of adoption. They are constant during a growth state.

A.2. Value capture subsystem

The model conceptualisation in Chapter 3 has split the share of value captured into three fractions: the value captured from customers and users, the value lost to input providers, and the value lost to resource erosion. Chapter 4 has presented how value is captured from customers and lost to input providers and resource erosion on a more detailed level. It has presented all details regarding the shares of value lost. Additional details exist for the share of value captured from premium users. These are outlined below.

A.2.1. Share of value captured from customers

Chapter 4 of this thesis has modelled the share of value captured. It depends on the bargaining power to customers and the bargaining power to premium users. The descriptions in the main body proceeded to illustrate the development of the bargaining power to customers. It depends on the switching costs of customers and influences from other parts of the model. The two subsubsections below adapt the structures and equations for customers to model the development of the bargaining power to premium users and their switching costs.

A.2.1.1. Bargaining power to premium users

The bargaining power to premium users changes like the bargaining power to customers (Subsubsection 4.2.1.1). Over time, existing premium users develop switching costs that increase the venture's bargaining power. The venture's contract periods may delay these adjustments. However, the bargaining power is adjusted when new premium users are acquired that have not yet developed switching costs (Figure A-6).



Figure A-6: Bargaining power to premium users

The model reuses the structure and equations developed for customers. The *bargaining power to premium users* increased through *changes in the bargaining power to existing premium users and* reduces through *adjustments to bargaining power to new premium users* (Equation A-15). This adjustment reduces the bargaining power towards the *bargaining power to new premium users* based on the *fraction of new premium users* (Equation A-16, Equation A-17). The bargaining power to new premium users depends on the *strength of product of users* considering the *premium product advantage* and relative to the *value provided by competitors and substitutes* (Equation A-18). The changes for existing premium users over the *premium user contract period* (Equation A-19). In addition to the factors for new premium users, the bargaining power to exiting premium users also depends on the *switching costs* (Equation A-20). The model uses the same equation and factors to initialise the bargaining power to premium users (Equation A-21).

Bargaining power to premium users		
= initial bargaining power to premium users		
+ \int (change in bargaining power to exsting premium users		
 adjustment of bargaining power to new premium users) 		
adjustment to bargaining power to new premium users = (Bargaining power to premium users	Eq A-16	
– bargaining power to new premium users)		
* fraction of new premium users		
fraction of new premium users	Eq A-17	
- ADZ(rate of premium user acquisition, Fremium user base, 1)		
bargaining power to new premium users	Eq A-18	
= ZIDZ(strength of product for users		
* premium product advantage, strength of product for users		
* premium product advantage		
+ value provided by competitors and substitutes)		

change in barg	aining power to exsting premium users	Eq A-19
	= ZIDZ(bargaining power to existing premium users	
	 Bargaining power to premium users, 	
	premium user contract period)	
bargaining pov	ver to existing premium users	Eq A-20
	= ZIDZ(strength of product for users	
	* premium product advantage * (1	
	+ user switching costs), strength of product for users	
	* premium product advantage * (1	
	+ user switching costs)	
	+ value provided by competitors and substitutes)	
initial bargaini	ng power to premium users	Eq A-21
	= ZIDZ(strength of product for users	
	* premium product advantage * (1	
	+ user switching costs), strength of product for users	
	* premium product advantage * (1	
	+ user switching costs)	

+ value provided by competitors and substitutes)

The product strength for users has been reviewed in the value creation subsystem in this appendix. The switching costs for users are described below. The value provided by competitors and substitutes has been modelled in Chapter 4 of this thesis (Subsubsection 4.2.1.3). The contract period is set based on statements made in the company's annual reports and websites.

A.2.1.2. User switching costs

Chapter 4 of this thesis has modelled the development of switching costs for customers (Subsubsection 4.2.1.2). They affect the venture's bargaining power towards them. As shown above, the bargaining power to premium users also depends on their respective switching costs. Therefore, the model applies the structure and equations presented in Chapter 4 to model user switching costs. Thus, switching costs develop as users adopt features, learn how to use them, and customise a product (Amit and Zott, 2001; DaSilva *et al.*, 2013; Tucker, 2018). The model expresses switching costs as the relative increase in usage intensity (Figure A-7).



Figure A-7: User switching costs

The model reuses the equations for customer switching costs. It calculates user switching costs by calculating the difference between the current *user usage intensity* and the *average initial user usage intensity*. This difference is then expressed as a fraction of the average initial user usage intensity (Equation A-22). The average rate's initial value is the usage intensity of new users at the beginning of the simulation. It adjusts to reflect changes in the venture's target users (Equation A-23). The *change in average initial user usage intensity* moves the current average towards the *new customer usage intensity* depending on the *fraction of new users* (Equation A-24).

user switching costs	Eq A-22
= ZIDZ(User usage intensity	
– Average initial user usage intens	sity,
Average initial user usage intensit	y)
Average initial user usage intensity	Eq A-23
= new user usage intensity	
+ \int (change in average initial user	usage intensity)
change in average initial user usage intensity	Eq A-24
= (new user usage intensity	
– Average initial user usage intens	sity)
* fraction of new users	

The inputs to the user switching costs are derived from the customer and user subsystem in Chapter 4 of this thesis and the additional value creation subsystem details in this appendix.

A.3. Customer and user subsystem

The customer and user subsystem presented in Chapter 4 presented the customer base, user base, and premium user base (Section 4.3). The subsystem in the main body shows how the venture acquires new customer and user through marketing activities and word-of-mouth. The venture loses them depending on their lifetime. Chapter 4 presented this churn using the example of customers. The user and premium user lifetimes are formalised below using the same structure and equations. Moreover, ventures' marketing and value delivering activities rely on their respective capabilities. These have not been presented in Chapter 4 because they use the same structure as the venture's technology developing capabilities. They are also illustrated below.

A.3.1. User and premium user lifetime

The user and premium user lifetimes represent how many years the average user or premium user stays with the venture. The model applies the structure developed for customers (Subsection 4.4.4) to model these two lifetimes. It is based on Sterman's (2000) structure for multiplicative effects. It adjusts a standard lifetime for the relative product strength and service capacity adequacy while accounting for the contract terms of premium users (Figure A-8).



Figure A-8: User and premium user lifetimes

The model calculates the *user lifetime* and *premium user lifetime* by adjusting the *standard user lifetime* by the respective *relative strength to competitors* and the *user capacity adequacy* (Equation A-25). The *premium user contract period* serves as a minimum for the premium user lifetime (Equation A-26).

user lifetime = MAX(0, standard user lifetime * user capacity adequacy Eq A-25 * relative strength to competition for users)

premium user lifetime Eq A-26

= MAX(premium user contract period, standard user lifetime

* user capacity adequacy

* relative strength to competition for premium users)

The *user capacity adequacy* reflects the fraction of users and premium users that the venture can service adequately. The model calculates it by dividing the *capacity to service users* by the number of indexed users in the *user base* and *premium user base*. The model also caps the adequacy at one when capacity exceeds requirements (Equation A-27). The *relative strength to competition for users* expresses the *strength of product for users* considering *user switching costs* relative to the *value provided by competitors and substitutes* (Equation A-28). The *relative strength to competition for premium users* also considers the *premium product advantage* (Equation A-29).

user capacity adequacy	Eq A-27
= MIN (1, ZIDZ(capacity to service users, (User base	
+ Premium user base)))	
relative strength to competition for users	Eq A-28
= ZIDZ(strength of product for users * (1	

- + user switching costs), strength of product for users
- * (1 + user switching costs)
- + value provided by competitors and substitutes)

relative strength to competition for premium users

Eq A-29

- = ZIDZ(strength of product for users
- * premium product advantage * (1
- + user switching costs), strength of product for users
- * premium product advantage * (1
- + user switching costs)
- + value provided by competitors and substitutes)

The value creation and capture subsystems in this appendix have modelled the product strength for users, user switching costs, the premium user contract period, and premium product advantage. The user base and premium user base, the value provided by competitors and substitutes, and the capacity to service users have been modelled in Chapter 4 of this thesis.

A.3.2. Capability development

The rates at which the venture's employees can acquire new users and customers and their capacities to service them depend on their capabilities. All capabilities in the model use the same structure as the technology developing capabilities presented in Chapter 4 of the thesis (Subsection 4.3.3). Below, this structure is adapted to model the capabilities of marketing and selling employees and value delivering employees.

A.3.2.1. Marketing and selling capabilities

Marketing and selling employees may acquire the three different types of users and customers distinguished in this thesis. Therefore, their productivity levels are represented by three variables: the ability to acquire customers, acquire users, and acquire premium users. These abilities reflect the indexed number of respective customers each indexed employee acquires per period (Subsection 4.4.2). As with all other capabilities, these productivity levels depend on employee competence, the effect sizes that transform the competence levels to hard impacts, and the adequacy of complementary assets. The model also considers the product's strength relative to competitors for marketing capabilities (Figure A-9).



Figure A-9: Marketing and selling capabilities

Employee abilities depend on their *marketing and selling competence*, the respective *effect of that competence* on an ability, and the *adequacy of complementary assets*. The effect sizes transform competence from a soft to hard scale. In addition, marketing and selling employee productivity may be affected by the product's strength (Garnsey, 1998; Bhide, 2000; Garnsey, Lubik and Heffernan, 2015; Hesse and Sternberg, 2017). Therefore, the model adjusts abilities for the respective *strength of the product* relative to the *value provided by competitors and substitutes* (Equation A-30, Equation A-31, Equation A-32).

ability to acquire customers

Eq A-30

Eq A-31

- = Marketing and selling competence
- * effect of competence on ability to acquire customers
- * adequacy of complementary assets
- * ZIDZ(strength of product for customers, strength of product for customers
- + value provided by competitors and substitutes)

ability to acquire users

- = Marketing and selling competence
- * effect of competence on ability to acquire users
- * adequacy of complementary assets
- * ZIDZ(strength of product for users, strength of product for users
- + value provided by competitors and substitutes)

ability to acquire premium users

- = Marketing and selling competence
- * effect of competence on ability to acquire premium users
- * adequacy of complementary assets * ZIDZ(strength of product for users
- * premium product advantage, strength of product for users
- * premium product advantage
- + value provided by competitors and substitutes)

Like the technology developing competence, the *marketing and selling competence* is scaled between zero and one. It accumulates from the *initial marketing and selling competence* based on competence improvements and competence adjustments (Equation A-33). *Marketing and selling competence improvements* reflect capability improvements through learning and routinisation along Sterman's (2000) s-curve. The speed along the curve depends on the inverse of the *MS years of improvement* as the standard rate (Equation A-34). The *marketing and selling competence adjustment* reduced the competence level towards the *new MS employee competence*. The adjustment's magnitude depends on the *fraction of new marketing and selling employees* (Equation A-35, Equation A-36).

Marketing and	selling competence	Eq A-33
	= initial marketing and selling competence	
	$+\int$ (marketing and selling competence improvements	
	 marketing and selling competence adjustment) 	
marketing and	selling competence improvements	Eq A-34
0	= (1/MS years of improvement)	1
	* Marketing and selling competence * (1	
	 Marketing and selling competence) 	
marketing and	selling competence adjustment	Eq A-35
	= (Marketing and selling competence	
	 new MS employee competence) 	
	* fraction of new marketing and sales employees	

fraction of new marketing and sales employees

= XIDZ(rate of hiring marketing and selling employees,Marketing and selling employees, 1)

The marketing and selling employees, the rate of hiring, and the adequacy of complementary assets are modelled as part of the firm managing subsystem in Chapter 4. The analyst assesses the initial and new employee competence based on the capability maturity level (see Appendix C). The years of improvement are an input from the venture's annual reports. The effect sizes are derived through automatic model calibration. The strength of the product and value provided by competitors and substitutes are modelled as part of the value creation subsystem in Chapter 4 (for customers) and in this appendix (for users and premium users).

A.3.2.2. Value delivering capabilities

The venture must maintain the customers and users it has acquired through marketing and selling. The lifetime of these depends on the relative product strength and the adequacy of servicing capacity. This capacity has been modelled in Chapter 4 of this thesis. It depends on the venture's employees and the productivity of each employee. This productivity is reflected in their ability to service customers and their ability to service users (Subsection 4.4.5). As with other capabilities, these depend on the competence of employees, the effects of the competence, and the adequacy of complementary assets (Figure A-10).



Figure A-10: Value delivering capabilities

The *ability to service customers* and the *ability to service users* reflect the number of indexed customers and users each indexed employee can service. The model multiplies together the *value delivering competence*, the respective *effect of the competence*, and the *adequacy of complementary assets* to derive the abilities (Equation A-37, Equation A-38).

ability to service customers

Eq A-37

Eq A-38

- = Value delivering competence
- * effect of competence on ability to service customers
- * adequacy of complementary assets

ability to service users

- = Value delivering competence
- * effect of competence on ability to service users
- * adequacy of complementary assets

The development of the competence follows the same structure as the marketing and selling competence above. It accumulates from its initial value based on improvements and adjustments (Equation A-39). These *value delivering competence improvements* reflect learning and routinisation along the s-curve with the speed of improvement based on the *VD years of improvements* (Equation A-40). The *value delivering competence adjustments* lower the competence towards the *new VD employee competence* depending on the *fraction of new value delivering employees* (Equation A-41, Equation A-42).

Value delivering competence

Eq A-39

= initial value deliveirng competence

- + \int (value delivering competence improvements
- value delivering competence adjustment)

value delivering competence improvements

Eq A-40

- = (1/VD years of improvement)
- * Value delivering competence * (1
- Value delivering competence)

value delivering competence adjustment

- = (Value delivering competence
- new value delivering employee competence)
- * fraction of new value delivering employees

fraction of new value delivering employees

Eq A-41

= XIDZ(rate of hiring value delivering employees,Value delivering employees, 1)

The value delivering employees and their hiring rate are modelled as part of the additional details for the firm managing subsystem in this appendix. The analyst sets the initial value delivering competence and years of improvement based on information provided in annual reports (see Appendix C). The effect sizes are determined through automatic model calibration.

A.4. Firm managing subsystem

Chapter 4 of the thesis has split the tasks of the firm's management into three categories: managing human resources, managing complementary assets, and managing financial resources. The three subsections below provide further details on these three areas.

A.4.1. Management of human resources

Chapter 4 provided insight into the development of the venture's human resources, the formation of hiring targets, hiring delays, and priorities for employees. The chapter used the venture's marketing and selling employees as an example to illustrate the principles that determine these four elements. However, the model also includes value delivering, technology developing, and firm managing employees. The details for these different employees are outlined below. These details adapt the principles outlined in the main body of the thesis to the specific employee stock and their related activities.

A.4.1.1. Additional employee stocks

The other types of employees considered in the model develop through the same mechanisms as the marketing and selling employees. They increase through hiring and deteriorate through layoffs and employee turnover (Figure A-11). The hiring and layoffs are executed over a delay to adjust the employee numbers to the management's targets. The equations below model employee developments by adapting the structure and equation for marketing and selling employees presented in Chapter 4 (Subsubsection 1.5.1.1).



Figure A-11: Human resource stocks

Each employee stock has an initial value of one. They then accumulate based on their respective hiring, layoff, and leaving rates (Equation A-43, Equation A-44, Equation A-45).

Value delivering employees

Eq A-43

- = initial value delivering employees
- + (rate of hiring value delivering employees
- rate of value delivering employees lay offs
- rate of value delivering employees leaving)

Technology developing employees

= initial tech dev employees

- + (rate of hiring tech dev employees
- rate of tech dev employees lay offs
- rate of tech dev employees leaving)

Firm managing employees

Eq A-45

Eq A-44

- = initial firm managing employees + \int (rate of hiring firm managing employees - rate of firm managing employees lay offs
- rate of firm managing employees leaving)

The rate at which employees leave the venture depends on the employee stock and the *employee turnover rate* (Equation A-46, Equation A-47, Equation A-48). The turnover rate expresses the fractions of employees that leave the venture per period.

rate of value delivering employees leaving	Eq A-46
= Value delivering employees	
* employee turnover rate	
	F 4 45
rate of tech dev employees leaving	Eq A-47
= Technology developing employees	
* employee turnover rate	
rate of firm managing employees leaving	Eq A-48
= Firm managing employees	
* employee turnover rate	

The rate of hiring increases an employee stock when the management aims for a positive *target change* in the respective employee stock. This positive target is achieved over the *hiring delay* and executed depending on the *liquidity adequacy*. In addition, the venture needs to replace employees that it has lost due to churn (Equation A-49, Equation A-50, Equation A-51).

rate of hiring value delivering employees	Eq A-49
= MAX(0, target change in value delivering employees	
/hiring delay) * liquidity adequacy	
+ rate of value delivering employees leaving	
	F 4 50
rate of hiring tech dev employees	Eq A-50
= MAX(0, target change in technology developing employees	
/hiring delay) * liquidity adequacy + rate of tech dev employees leaving	
rate of hiring firm managing employees	Eq A-51
= MAX(0, target change in firm managing employees	
/hiring delay) * liquidity adequacy	
+ rate of firm managing employees leaving	

The rate of layoffs decreases an employee stock when the management aims for a negative *target change* in the respective employee stock. This planned reduction in employee numbers is executed over the *delay to lay off* (Equation A-52, Equation A-53, Equation A-54).

rate of value delivering employees lay offs = $MAX(0,$	Eq A-52
 target change in value delivering employees 	
/delay to lay off)	

rate of firm managing employees lay offs $=$ MAX(0,	Eq A-54
-target change in firm managing employees	
/delay to lay off)	

Each *target change* expresses the difference between the venture's current indexed employee numbers and the indexed number of employees the venture would like to employ in that activity. This latter target is reflected in the *target value delivering employees, target technology developing employees,* and *target firm managing employees* (Equation A-55, Equation A-56, Equation A-57).

target change in value delivering employees	Eq A-55
= target value delivering employees	
 Value delivering employees 	
target change in technology developing employees = target technology developig employees	Eq A-56
 Technology developing employees 	
target change in firm managing employees	Eq A-57
= target firm managing employees	
 Firm managing employees 	

The hiring delay, delay to lay off, and liquidity adequacy have been modelled in the firm managing subsystem in Chapter 4 of this thesis. The sections below describe how the management determines its employee targets across these three different activities.

A.4.1.2. Target value delivering employees

The target value delivering employees expresses how many employees the venture strives to employ for its customer service and content creation activities. Based on Sterman (2000) and Morecroft (2015), Chapter 4 has modelled targets for marketing and selling employees by determining the customer and user acquisition goals and expected acquisitions per employee (Subsubsection 4.5.1.2). These principles have also been applied to value delivering employees. They may service customers, service users, or create content. The number of required employees thus depends on the number of customers and users to service or the content it expects to create, and the expected productivity of each employee in those different tasks (Figure A-12).



Figure A-12: Target value delivering employees

The venture's *target value delivering employees* depend on the *employees required for value delivering activities* and the venture's *VD employee cap* (Equation A-58). If the venture has not set a cap (-1), the venture will target the number of required employees. Otherwise, the venture will limit the targeted employee number at the cap. The required employees are the sum of those required for each of the possible value delivering activities (Equation A-59).

target value delivering employees = IF THEN ELSE(VD employee capEq A-58= -1,
employees required for value delivering activities,
MIN (VD employee cap,
employees required for value delivering activities))Eq A-59

= employees required to service customers

- + employees required to service users
- + employees required to create content

The model adapts the goal and requirement formation from marketing and selling activities. For example, the employees required to acquire customers are calculated by determining the number of customers the management wants to acquire and the rate at which it expects each employee to acquire customers (Subsubsection 4.5.1.2). The model implements these principles again. It

determines how many customers and users the venture will need to service and how much new content it wants to create. The targets are also affected by the expected productivity of each employee in those tasks (Equation A-60, Equation A-61, Equation A-62).

employees required to service customers	Eq A-60
= ZIDZ(expected customer base, reported ability to service cusotmers)	
employees required to service users	Eq A-61
= ZIDZ(expected user base, reported ability to service users)	
employees required to create content	Eq A-62
= ZIDZ(target rate of content creation, reported ability to create content)	

The *expected customer base* and *expected user base* are determined using Vensim's forecast function. It provides an estimate of the respective base one year ahead (the reporting period). It also acknowledges that the management's information about the change in customer and user bases is delayed by the *reporting delay* (Equation A-63, Equation A-64). The model derives the *target rate of content creation* in the same manner as the target customer and user acquisition rates for marketing and selling employees. It assumes an annual *target content library growth* rate that is part of the management's dominant logic. This is the annual rate at which the management wants to grow its *content library*. However, the venture does not need to create all content items themselves to achieve this goal. Like customers acquired through word-of-mouth, the venture disregards content that it expects to be created externally but needs to replace expired content (Equation A-65).

expected custo	mer base	Eq A-63
= FORECAST(0	Customer base, reporting delay, reporting period)	
expected user = FORECAST(I	base User base, reporting delay, reporting period)	Eq A-64
target rate of c	ontent creation	Eq A-65
	= Content library * target content library growth	
	 reported rate of external content creation 	
	+ rate of content expiration	

Appendix 310

The model acknowledges that the management has no perfect information of current employee productivity levels and the rate at which connect is created externally. Therefore, the model uses Vensim's smooth function, which generates reported abilities and a reported rate of external content creation. These adjust over time to current abilities and rates (Equation A-66, Equation A-67, Equation A-68, Equation A-69).

reported ability to service cusotmers	Eq A-66
= SMOOTH(ability to service customers, reporting delay)	
reported ability to service users = SMOOTH(ability to service users, reporting delay)	Eq A-67
reported ability to create content = SMOOTH(ability to create content, reporting delay)	Eq A-68
reported rate of external content creation = SMOOTH(rate of external content creation, reporting delay)	Eq A-69

The value delivering employee cap is an exogenous influence from the dominant logic set by the analyst. All other inputs to this model part emerge from the customer and user subsystem in Chapter 4 or this appendix.

A.4.1.3. Target technology developing employees

As described in the technology developing subsystem (Section 4.3), the venture's technology developing employees maintain the existing technology and develop new technology. The model adapts the same principles as for all other managerial targets. It determines the employees required for these two goals and the expected employee productivity (Figure A-13). The management may cap the number of technology developing employees depending on their dominant logic.



Figure A-13: Target technology developing employees

If the venture has not set a limit for its technology developing employees, the *TD employee cap* variables has a value of -1. In such cases, the venture hires the employees required to maintain and develop its technology (Equation A-70). Otherwise, the venture hires the required employees up to the cap.

target technology developig employees

Eq A-70

= IF THEN ELSE (TD employee cap
= -1, required employees to develop technology
+ required employees to manage technology,
MIN(required employees to develop technology
+ required employees to manage technology,
TD employee cap))

The model calculates the employees required to maintain and develop the venture's technology through the goal these employees need to achieve and their expected employee productivity. As it does for other employee requirements, it divides the goal by the expected employee productivity (Equation A-71, Equation A-72).

```
required employees to maintain technology Eq A-71
= ZIDZ(expected technological quality,
reported ability to maintain technology)
```

The expected employee productivity is expressed by the reported abilities to maintain and develop the technology. As for other reported abilities, it is calculated using Vensim's smooth function (Equation A-73, Equation A-74), which can be used to model expectation formation over a reporting delay (Sterman, 2000).

reported ability to maintain technology	Eq A-73
= SMOOTH(ability to maintain technology, reporting delay)	
reported relative improvement in technology per employee	Eq A-74
= SMOOTH(ability to develop technology	

* effect of investments on technology improvements, reporting delay)

The management also needs to establish its targets in absolute amounts to determine the employees required to maintain and develop the technology. The model calculates the *expected technological quality* that needs maintaining using Vensim's forecast function (Equation A-75). The *target technology improvement* captured the management's improvement target along the technology s-curve. It has been calculated through the remaining improvements on the curve and the *target technological improvement rate*. This rate captures how much of the gap the venture aims to close per year (Equation A-76).

expected technological quality	Eq A-75
= FORECAST(Technological quality, reporting delay, reporting period)	

target technology improvement

- = (maximum technological quality
- Technological quality)
- * target technology improvement rate

The two abilities have been calculated as part of the technology developing subsystem in Chapter 4 of this thesis. Readers can find the technological quality and its maximum in the value creation subsystem in Chapter 4 too. The target

Eq A-76

technological improvement rate is an exogenous input set by the analyst. It reflects the management's dominant logic.

A.4.1.4. Target firm managing employees

Ventures also require firm managing employees that supervise the operating employees. The model treats marketing and selling, value delivering, and technology developing employees as those operating employees. Chapter 3 and 4 argue that firm managing employees are also responsible for developing growth plans and implementing them through hiring, acquiring complementary assets, and raising capital (Section 4.5). A larger number of employees requires additional managerial capacity to supervise and oversee. Therefore, digital ventures growth plans should also include increasing managerial employees (Penrose, 2009). The management must thus form a target for the required firm managing employees. As for other activities, this target depends on the goal these employees must achieve and the expected productivity of firm management employees (Figure A-14).



Figure A-14: Target firm managing employees

Like the other employee targets, the management may cap the firm managing employees depending on their dominant logic. If no limit is set, the firm targets the number of required managing employees (Equation A-77). These requirements depend on the number of *target operating employees* the management will need to supervise and the *reported ability to manage the firm*. In addition, the management needs to ensure sufficient managerial slack to hire, acquire complementary assets, and raise capital (Equation A-78). The fraction of
additional managerial resources above those required to manage the target operating employees is expressed in the *target managerial slack*. The total number of target operating employees is the sum of the target market and selling employees, target value delivering employees, and target technology developing employees (Equation A-79).

target firm managing employees = IF THEN ELSE(max FM employees	Eq A-77
= -1, required employees to manage the firm,	
MIN(required employees to manage the firm, max FM employees))	
required employees to manage the firm	Eq A-78
= ZIDZ(target operating employees, reported ability to manage the firm)	
* (1 + target managerial slack)	
target operating employees	Eq A-79

- = target marketing and selling employees
- + target technology developig employees
- + target value delivering employees

The reported ability to manage the firm expresses how many indexed operating employees each indexed firm managing employee can supervise. It is calculated using Vensim's smooth function that adjusts the reported ability towards the actual ability over the reporting delay (Equation A-80).

```
reported ability to manage the firm Eq A-80
= SMOOTH(ability to manage the firm, reporting delay)
```

The targets for the operating employee types are modelled in the firm managing subsystem in Chapter 4 of the thesis and this appendix. The ability to manage is modelled with the firm managing capabilities below. The firm managing employee cap is an exogenous influence reflecting the growth state's dominant logic and set by the analyst. The target managerial slack is set to 0.5 for all companies.

A.4.1.5. Further details on target marketing and selling employees The descriptions of the firm managing subsystem in Chapter 4 of this thesis have outlined how the venture forms targets for its marketing and selling employees. These included identifying the required employees to acquire the target number of customers, users, and premium users separately. The calculation of these three individual employee requirements was illustrated using the example of the employees required to acquire customers (Subsubsection 4.5.1.2). However, the venture also needs to determine the employees required to achieve its user and premium user growth targets. These two requirements are illustrated here utilising the same structure and equation as illustrated for the customer acquisition requirement. The two individual rates depend on the reported ability of employees to acquire the respective user type and the number of the respective type the venture wants to acquire over the next year (Figure A-15). This absolute number depends on the venture's current users base and the target growth rate. The venture also needs to make up for users lost to churn and those lost to conversion. However, the venture does not need to acquire users through its employees that are acquired through word-of-mouth marketing.



Figure A-15: Target marketing and selling employees

The employees required to acquire users and premium users are calculated by dividing the rate at which the venture strives to acquire users by the rate at which

the venture expects each employee to acquire users (Equation A-81, Equation A-82).

mployees required to acquire users	Eq A-81
 ZIDZ(target user acquisition, reported ability to acquire user) 	
mployees required to acquire premium users	Eq A-82
= ZIDZ(target premium user acquisition,	
reported ability to acquire premium users)	

The reported abilities capture the management's expectations of user acquisition per indexed employee. The model calculates them using Vensim's smooth function that acknowledges the management's delay in determining these abilities (Equation A-83, Equation A-84).

reported ability to acquire user	Eq A-83
= SMOOTH(ability to acquire users, reporting delay)	
reported ability to acquire premium users	Eg A-84

reported ability to acquire premium users Eq = SMOOTH(ability to acquire premium users, reporting delay)

The target user and premium user acquisition reflect the number of users and premium users the venture strives to acquire per period to achieve its target growth rates. They depend on the current user and premium user base and the fraction of new users and premium users the venture needs to acquire per period. These fractions are calculated by adjusting the target growth rates for users lost due to churn, users lost due to conversion to other types, users gained due to conversion from other types, and users gained through word-of-mouth marketing (Equation A-85, Equation A-86).

target user acquisition

Eq A-85

- = User base * (target user growth
- + ZIDZ(1, reported user lifetime)
- rate of user acquisition per customer
- rate of user acquisition per user
- + user to premium conversion rate
- premium to users conversion rate)

target premium user acquisition

Eq A-88

- = Premium user base * (target premium user growth
- + ZIDZ(1, reported premium user lifetime)
- rate of premium user acquisition per customer
- rate of premium user acquisition per user
- user to premium conversion rate
- + premium to user conversion rate)

While the model assumes constant conversion and acquisition rates through word-of-mouth, the customer lifetime adjusts dynamically. However, just as the management does not perfectly know employee abilities, it also does not perfectly know the user lifetimes. Therefore, the management expectation of user lifetimes is determined using the smooth function (Equation A-87, Equation A-88).

reported premium user lifetime

= SMOOTH(premium user lifetime, reporting delay)

The analyst sets the target user and premium user growth rates to reflect the venture's dominant logic in a growth state. All other inputs to these expectations emerge from the customer and user subsystem described in Chapter 4 of the thesis and this appendix.

A.4.1.6. Focus on servicing customers and users and creating content The venture's value delivering employees may have conflicting priorities and may compromise between servicing users, servicing customers, or creating content. The priorities represent the fraction of time spent on these different tasks (Subsection 4.4.5). Chapter 4 has modelled similar conflicting priorities for marketing and selling employees who may focus on acquiring customers, users, or premium users (Subsubsection 4.5.1.4). The model applies the same structure and equations to derive the foci of value delivering employees. These depend on the fraction of employees required to achieve each value delivering target (Figure A-16).



Figure A-16: Focus of value delivering employees

The model calculates each focus by expressing the indexed number of employees required to achieve a goal as a fraction of the indexed total number of required value delivering employees (Equation A-89, Equation A-90, Equation A-91). As outlined in the main body of the text for marketing and selling employees, this gives each focus a value between zero and one. While one indicates that the employees focus all their time on the activity, zero indicates that they do not engage in the activity at all.

focus on servicing customers		Eq A-89
	= employees required to service customers	
	/employees required for value delivering activities	
focus on servic	ing users	Eq A-90
	= employees required to service users	
	/employees required for value delivering activities	
focus on creati	ng content	Eq A-91
	= employees required to create content	
	/employees required for value delivering activities	

Each value delivering employee requirement has been introduced above.

A.4.2. Management of complementary assets

Chapter 4 of the thesis modelled the adequacy of complementary assets that influences capability levels. This adequacy depends on the complementary assets held by the venture, which depreciate and are acquired by the management. Therefore, the main body of the thesis has modelled acquisition targets and illustrated their implementation (Subsection 4.5.2). The *delay to acquire* *complementary assets* influences the acquisition of complementary assets. It is presented below.

A.4.2.1. Delay to acquire complementary assets

Just as hiring targets are implemented over a hiring delay, the acquisition of complementary assets is delayed by a delay to acquire complementary assets. The model uses the same structure and equation as for the hiring delay (Subsubsection 4.5.1.3). The delay is negatively affected by the number of firm managing employees, their managerial slack, and their ability to invest in complementary assets (Figure A-17). While the firm managing employees and their slack represent the resource inputs to acquiring complementary assets, their ability represents their productivity level. The delay thus implements Sterman's (2000) resource-productivity-flow relationship applied by the model to other activities.



Figure A-17: Delay to acquire complementary assets

The model calculates the delay like the hiring delay by dividing one by the product of the firm managing employees, their managerial slack, and their ability to acquire complementary assets (Equation A-92).

delay to acquire complementary assets

Eq A-92

- = XIDZ(1, Firm managing employees
- managerial slack
- * ability to acquire complementary assets, 100)

Please find the model elements for the firm managing employees above, the management's capabilities below, and the managerial slack in Chapter 4 of this thesis.

A.4.3. Management of financial resources

The financial resource management outlined as part of the firm managing subsystem in Chapter 4 includes determining liquidity requirements and raising capital to close liquidity shortfalls. This subsection presents additional details required for the liquidity requirements and the delay to raise capital. They make use of the structures and equations presented in Chapter 4 of this thesis.

A.4.3.1. Additional liquidity requirements

The venture may need to raise additional capital to finance its growth and expansion plans. Chapter 4 has modelled the financing requirements to acquire customers as an example (Subsubsection 4.5.3.3). However, the venture may also need to raise capital to finance its user acquisition, technology improvements, or content creation. As the example presented in the main body of the thesis, these depend on the magnitude of the target, the venture's planning horizon, and the financial resources it wishes to hold per unit of the target (Figure A-18).



Figure A-18: Additional liquidity requirements

The model adapts the implementation of liquidity requirements in Chapter 4. It calculates each liquidity requirement by multiplying the target rate with the firm's reporting period of one year and the financial resources it aims to hold per unit of the target (Equation A-93, Equation A-94, Equation A-95).

liquidity requirements for user acquisition

Eq A-93

- = (target user acquisition
- + target premium user acquisition)
- * financial resources required per user
- * reporting period

Appendix 321

liquidity requirements for technology improvements	Eq A-94
= target technology improvement rate	

Eq A-95

- * financial resources per technology improvement
- reporting period

liquidity requirements for content creation

- = target rate of content creation
- * financial resources required per content item
- * reporting period

The user acquisition, technology improvement, and content creation goals were modelled as part of the employee requirements in Chapter 4 of the thesis and this appendix. Automatic model calibration sets the financial resources required per user and technology improvement.

A.4.3.2. Delay to raise capital

Just like the management cannot instantaneously hire and acquire complementary assets, the management can also not raise capital instantaneously. The venture's capital raising is thus delayed (Subsection 4.5.3). Like the previous delays to hire and acquire complementary assets, the delay in raising capital depends on resources available for the activity and the resource's productivity (Sterman, 2000). While the firm managing employees and their managerial slack represent the available resources, their productivity is reflected by their ability to raise capital (Figure A-19).



Figure A-19: Delay to raise capital

The model calculates the delay to raise capital by dividing one by the product of the firm managing employees, their managerial slack, and their ability to raise capital (Equation A-96).

delay to raise capital

= XIDZ(1, Firm managing employees

* managerial slack * ability to raise capital, 100)

As described for the delay to acquire complementary assets, the firm managing employees and their capabilities are modelled as part of this subsystem. The managerial slack has been modelled in Chapter 4 of the thesis.

A.4.4. Firm management capabilities

The venture's firm management capabilities affect its capacity to manage employees, and thus the delays to hire, acquire complementary assets, and raise capital. The model uses the same structure and equations used for technology developing employees in Chapter 4 (Subsection 4.3.3) and for customer- and user-related capabilities above. Each ability depends on employee competence, the effect size that transforms the competence to a hard impact on the system, and the adequacy of complementary assets (Figure A-20).



Figure A-20: Firm managing capabilities

The model calculates the four firm managing abilities in the same manner as the other types of capabilities presented in this thesis. It multiplies employee competence with the effect of the competence on the respective ability and the adequacy of complementary assets (Equation A-97, Equation A-98, Equation A-99, Equation A-100).

ability to manage the firm Eq A-S		Eq A-97
	= Firm managing competence	
	effect of competence on ability to manage the firm	
	* adequacy of complementary assets	
	ability to hire = Firm managing competence * effect of competence on ability to hire	Eq A-98
	* adequacy of complementary assets	
	ability to invest in complementary assets	Eq A-99
	= Firm managing competence	
	* effect of competence on ability to acquire CA	
	 * adequacy of complementary assets 	
	ability to raise capital	Eq A-100
	= Firm managing competence	
	* effect of competence on ability to raise capital	

* adequacy of complementary assets

The competence development follows the same structure as the marketing and selling competence and value delivering competencies above and the technology developing competence in Chapter 4 of the thesis. The competence accumulates from its initial value based on improvements and adjustments (Equation A-101). These *firm managing competence improvements* reflect learning and routinisation along Sterman's (2000) s-curve. The speed along this s-curve depends on the *FM years of improvements* (Equation A-102). The *value delivering competence adjustments* lower the competence towards the *new FM employee competence* depending on the *fraction of new firm managing employees* (Equation A-103, Equation A-104).

Firm managing competence

Eq A-101

- = initial firm managing competence
- + (firm managing competence improvements
- firm managing competence adjustment)

firm managing competence improvements	Eq A-102
= (1/FM years of improvement)	
* Firm managing competence * (1	
 Firm managing competence) 	
firm managing competence adjustment	Eq A-103
= (Firm managing competence	
 new FM employee competence) 	
fraction of new firm managing employees	
fraction of new firm managing employees	Eq A-104
= XIDZ(rate of hiring firm managing employees,	

The firm managing employees and their hiring rate have been modelled above.

Firm managing employees, 1)

The analyst needs to set the initial competence of the firm and the competence of new employees (see Appendix C). The Vensim software calibrates automatically the effect of the competence on the ability to manage. The other effect sizes are fixed for all companies and set to a value of 12. Thus, managers can implement their hiring targets, acquire complementary asset targets, and fundraising targets at fully developed capabilities and managerial slack in one month.

A.5. Content creation subsystem

Some digital ventures employ business models that provide access to content items or artefacts (Afuah and Tucci, 2000; Mikalef, Giannakos and Pateli, 2013; Nambisan, 2017; Nzembayie, Buckley and Cooney, 2019). Examples include online newspapers and streaming services for different types of audio and video content. This thesis has implemented the content quality in the value creation subsystem to account for such business models (see Subsection 4.1.3). The supplementary material for the value creation subsystem in this appendix modelled this quality similar to the technological quality. It formalises the venture's content library, the quality at which it produces content, and the value of its content assets.

A.5.1. Content library

The *content library* is the collection of all content items that the venture offers to its customers and users (Afuah and Tucci, 2000; Wayne, 2018). It reflects, for example, how many records, videos, or newspaper articles a venture provides. The structure for technological resources is utilised for the content library. If ventures provide content, their initial content library is indexed at one. It then changes through new content creation and content expiration (Figure A-21). While content expires depending on its lifetime, different digital ventures may combine different mechanisms to create additional content. For example, music streaming services generally acquire the rights to external content (Rangaswamy *et al.*, 2020). Newspapers usually publish content that has been created internally by their employed journalists. Other content-based digital ventures may allow their users and customers to create content externally (Zhu and Iansiti, 2007; Naudé and Liebregts, 2009; Mikalef, Giannakos and Pateli, 2013; Gandia and Parmentier, 2017; Zhu and Bao, 2018; Rangaswamy *et al.*, 2020).



Figure A-21: Content library

The content library accumulates through the *rate of content creation* and *rate of the content expiration* (Equation A-105). Like technological resources' amortisation, the model calculates the expiration rate by dividing the content library by the *content lifetime* (Equation A-106). The rate of content creation is the sum of the *rate of internal content creation*, the *rate of external content creation*, and the *rate of content acquisition* (Equation A-107).

Content library	r = initial content library	Eq A-105
	+ \int (rate of content creation	
	 rate of content expiration) 	
rate of content	expiration = ZIDZ(Content library, content lifetime)	Eq A-106
rate of content	creation	Eq A-107
	= rate of internal content creation	
	+ rate of external content creation	

+ rate of content acquisition

The analyst sets the rate of content acquisition and the content lifetime based on the venture's annual reports. The rates of internal and external content creation are reviewed below.

A.5.1.1. Internal content creation

The rate of internal content creation captures how many content items the venture's value delivering employees create per period (Afuah and Tucci, 2000; Naldi and Picard, 2012). It depends on the venture's employees, their ability to create content, and their focus on creating content (Figure A-22). Like other capabilities, this ability depends on the venture's complementary assets and employee competence level, scaled using an effect variable.



Figure A-22: Internal content creation

The model uses Sterman's (2000) resource-productivity-flow relationship to model employees' *rate of internal content creation*. Like in previous implementations of this relationship, the resources are the number of *value delivering employees* and their *focus on creating content*. The *ability to create content* represents employee productivity i.e. the indexed items each indexed

employee creates per period (Equation A-108). The model calculates the *ability to create content* like the other abilities. It scales employees' value delivering *competence* using the *effect of the competence on the ability to create content* and multiplies it with the *adequacy of complementary assets* (Equation A-109).

rate of internal content creation

Eq A-108

= Value delivering employees * ability to create content* focus on creating content

ability to create content

Eq A-109

- = Value delivering competence
- * effect of competence on ability to create content
- * adequacy of complementary assets

Other subsystems have provided the details for most of the inputs for this part of the model. The value delivering competence is discussed as part of the customer and user subsystem. The value delivering employees, adequacy of complementary assets, and focus on creating content have been discussed as part of the firm managing subsystem in this appendix. The Vensim software sets the effect of the competence on the ability to create content through automatic model calibration.

A.5.1.2. External content creation

The *rate of external content creation* reflects the number of content items created by customers and users for the venture's library per period. This rate depends on the venture's customer and user bases and the rate at which each customer and user creates content (Figure A-23). Content creation may be limited to some customer types depending on the business model design. Moreover, these different groups may create content at different rates (Zhu and Iansiti, 2007; Naudé and Liebregts, 2009; Gandia and Parmentier, 2017; Rangaswamy *et al.*, 2020).



Figure A-23: External content creation

The *rate of external content creation* has been split into the *rate of content creation by customers* and rate *of content creation by users* (Equation A-110). The model calculates the rates at which customers and users create content by multiplying the respective customer and user bases and the rates at which each indexed customer and user creates content (Equation A-111, A-112).

rate of external	content creation	Eq A-110
	= rate of content creation by users	
	+ rate of content creation by customers	
rate of content	creation by customers	Eq A-111
	= Customer base	
	* rate of content creation per customer	
rate of content creation by users		Eq A-112
	= User base * rate of content creation per user	
	+ Premium user base	
	* rate of content creation per premium user	

The customer and user subsystem derives the number of customers, users, and premium users. The rates of content creation per user or customer are exogenous inputs set by the analyst.

A.5.2. Content quality

The *quality of new content items* reflects the quality of content items currently added to the venture's library. This quality affects the content quality in the value creation subsystem. As new items are added to the content library, the average content quality moves towards the quality of new items. The different mechanisms that add content to the content library may produce content of different qualities. The quality of internally developed content depends on employees' competence to create appealing content. The quality of externally created content depends on the users' experience (Kim, Oh and Shin, 2010; Oestreicher-Singer and Zalmanson, 2013; Kim and Kim, 2017; Täuscher and Abdelkafi, 2018). This experience has been modelled using their usage intensity as a proxy (Figure A-24).



Figure A-24: Quality of new content items

The model weights the different content qualities by their fraction of the total *rate* of content creation (Equation A-113). The model calculates the *quality of internally generated content* based on the value delivering competence. It scales the competence using the *effect of competence on content quality* (Equation A-114).

quality of new content items

- = ZIDZ((quality of internally created content
- * rate of internal content creation
- + quality of externally created content
- * rate of external content creation
- + quality of acquired content
- * rate of content acquisition), rate of content creation)

quality of internally created content

Eq A-114

- = Value delivering competence
- * effect of competence on content quality

The model derives the *quality of externally created content* like the overall quality. It calculates the weighted average of the *quality of customer content creation* and the *quality of user content creation* (Equation A-115). These respective qualities at which customers and users create new content depend on their abilities and experience. The model assumes that this ability develops as customers and users learn how to use the product and its features over time. The value creation subsystem has modelled this experience using the usage intensity. Intensities accumulate as customers become familiar with the product and adopt its features (Subsection 4.1.4). Therefore, the model uses it here to derive the quality at which customers and users create content. Like employee competence, the model scales the usage intensities using the *effect of experience on the content quality* (Equation A-116, Equation A-117).

quality of externally created content	Eq A-115
= ZIDZ((quality of customer content creation	
* rate of content creation by customers + quality of user content creation	
* rate of content creation by users), rate of external content creation)	
quality of customer content creation	Eq A-116
= Customer usage intensity	
* effect of experience on customer content quality	
quality of user content creation	Eq A-117
= User usage intensity	
* effect of experience on user content quality	

Eq A-113

The customer and user subsystem has illustrated the value delivering competence, while the value creation subsystem has illustrated the usage intensities. The effect sizes are set using automatic model calibration.

A.5.3. Content assets

The content library expresses the indexed number of content items that the venture offers to its customers and users. Some digital ventures may also account for this library in monetary terms on their balance sheet as *content assets*. For example, they may purchase or licence content for specified timeframes. They may also create content items themselves that they expect to have a useful life (Afuah and Tucci, 2000; Labes, Erek and Zarnekow, 2013; Rangaswamy *et al.*, 2020). As ventures can capitalise development costs, they may also capitalise investments into bought or internally created content items. Like technological resources, content assets amortise over their lifetime (Figure A-25).



Figure A-25: Content library

The *content assets* accumulate from their initial value through the *rate of investment in content assets* and the *rate of amortisation of content assets* (Equation A-118). The model calculates the amortisation rate similarly to the depreciation and amortisation of complementary assets and technological resources. It divides the stock of content assets by the *lifetime of content assets* (Equation A-119). The model calculates the investment rate by multiplying the indexed number of content items internally created and externally acquired with the respective indexed cost per item (Equation A-120).

Content assets = initial content assets	Eq A-118
+ \int (rate of investment in content assets - rate of amortisation of content assets)	
rate of amortisation of content assets	Eq A-119
= ZIDZ(Content assets, lifetime of content assets)	

rate of investment in content assets

Eq A-120

- = rate of internal content creation
- * cost per internally created content item
- + rate of content acquisition
- * cost per acquired content item

The analyst must set the initial content assets, their lifetime, and the costs per content item. They are based on information contained in the venture's financial statements (see Appendix C).

A.5.4. Overview of subsystem inputs

The content creation subsystem calculates the venture's content library and content assets (Table A-1). These outputs can be compared to annual reports regarding the number of content items offered by a digital venture and their value on its balance sheet. The subsystem relies on inputs from others to derive these stocks. The model covers the development of content through the venture's employees and its customers. Thus, inputs from the firm managing and customer subsystems are required.

Outputs	
Content library, content assets	
Inputs from other subsystems	Exogenous inputs
User and customer subsystem:	Business model design:
Value delivering competence	• Effect of competence on ability to create
• Customer base, user base,	content, effect of competence on content quality
premium user base	Initial content library
• Customer usage intensity, user	Environment – User and customer behaviour:
usage intensity	• Rate of content creation per customer, rate of
Firm management subsystem:	content creation per user, rate of content
Value delivering employees	creation per premium user
• Focus on content creation	• Effect of experience on customer content
• Adequacy of complementary	quality, effect of experience on user content
assets	quality
	Environment – Input providers:
	• Cost per internally created content item, cost per
	acquired content item
	Quality of acquired content
	Accounting measures:
	Initial content assets
	• Content lifetime, Lifetime of content asset
	Rate of content acquisition

Table A-1: Outputs and inputs of the content creation subsystem

The content creation subsystem also requires exogenous inputs. These relate to the business model design regarding content as a value driver and the effectiveness of a venture's content creation processes. Other inputs relate to the content creation by customers and the cost of creating content. Finally, accounting measures are required to set the initial stock of content assets, the lifetime of assets, and the rate of content acquisition.

Appendix B: Custom Vensim macros

Vensim macros allow repeating model structures without replicating the stockand-flow diagrams and retyping the equations. The Vensim software implements many of its functions via macros. Examples of such built-in macros are the smooth or forecast functions used in this thesis (Ventana Systems, no date). In addition to built-in macros, the detailed descriptions of the model in Chapter 4 of the thesis also include two custom macros: the HasValue and the RollingAnnualFinancial functions. The details of these two macros are provided below.

B.1. HasValue

The model uses the HasValue function to calculate the product strength for customers (Subsection 4.1.3) and users (Subsection A.1.2). It shortens the equations used to calculate these product strengths significantly. The product strengths depend on the five value drivers identified for digital ventures. The model multiplies together these value drivers to calculate the product strengths. If a value driver is employed, its initial value is one. If it is not employed, it is zero throughout the simulation. This setup allows identifying if a value driver is not employed and avoids mistaking an unemployed value driver with a constant one. However, using the value of zero in combination with multiplication causes problems when calculating the product strength if one or more value drivers are not employed. In such cases, the multiplication would create product strengths of zero. Therefore, an if-structure must be used to check if a value driver is used and set it from zero to one. This ensures that a value driver is not employed. However, this would create a long equation with five if-clauses.

The HasValue function executes these if-clauses externally to shorten the equations used to calculate product strengths. The function receives a value driver as the variable parameter and the value of one as the output parameter (Macro B-1).

<u>Macro B-1: HasValue</u>

:MACRO: HasValue(variable, output)				
HasValue=IF THEN ELSE(variable=0, output, variable)				
\sim variable				
~				
:END OF MACRO:				

If a value driver is employed and thus not zero, the function returns the driver's value. If the value driver is not employed and thus zero, the function returns one to avoid the multiplication of the product strength being erroneous.

B.2. RollingAnnualFinancial

The detailed descriptions of the model in Chapter 4 have outlined conceptual differences between accounting figures and variables representing instantaneous rates in System Dynamics models (Subsections 4.1.5, 4.2.3). Therefore, System Dynamics modellers must also model the reporting system (Sterman, 2000, 2005). The RollingAnnualFinancial function converts the rate variables of System Dynamics models to accounting figures over a reporting period. It has been based on the structure of equations used by Oliva, Sterman, and Giese (2003), who use their implementation to calculate quarterly accounting figures.

The RollingAnnualFinancial function requires two inputs (Macro B-2). The first parameter is a *current rate* in units of money per period reflecting, for example, the rate at which a company generates revenue or incurs costs in the model. The second parameter is the *reporting period* in units of time over which the function should generate the accounting figure. The function's core is a stock named *Rolling accumulation*, which accumulates from its initial value of zero through the *current rate* and a *delayed rate*. This delayed rate is the current rate delayed by the reporting period. Until a reporting period has passed, no outflows of the stocks occur. Thereby, the stock averages the current rate over the reporting period.

Using a stock with an initial value of zero has the disadvantage that estimates are too low until an entire reporting period has passed. Therefore, the RollingAnnualFinancial function also annualises the stock's value. It uses an ifclause checking the time passed in the model. If the model is being initialised (time=0), the function returns the fraction of the current rate appropriate for the reporting period. Otherwise, the function checks if a full reporting period has passed in the model. If it has not, the function annualises the stock. If a full reporting period or more has passed, the function returns the value of the stock.

Macro B-2: RollingAnnualFinancial

:MACRO: RollingAnnualFinancial(current rate, reporting period) RollingAnnualFinancial=IF THEN ELSE(Time\$=0, current rate*reporting period, IF THEN ELSE(Time\$<reporting period, ZIDZ(Rolling accumulation, Time\$)*reporting period, Rolling accumulation)) \$ ~ Rolling accumulation= INTEG (current rate-delayed rate, 0) \$ delayed rate= DELAY FIXED(current rate, reporting period, 0) \$/Year :END OF MACRO:

Overall, the function provides a flexible approach to convert instantaneous rate variables to accounting figures. The source code can be copied into the Vensim text editor (requires advanced Vensim versions) and used in future models for any accounting variable and reporting period.

Appendix C: Overview of model inputs

The model developed in this thesis and its subsystems rely on a series of inputs. The detailed model developed chapter (Chapter 4) has grouped these inputs into multiple categories. Moreover, each subsystem calculates one or more variables as its outputs. The tables at the end of each subsystem have summarised these exogenous inputs and subsystem outcomes. The methodology chapter has outlined the principles of setting System Dynamics models' hard, soft, and automatically calibrated inputs. The overviews below illustrate all variables and their estimation techniques using those principles. This appendix first outlines the outcomes of each subsystem. It then illustrates how an analyst can set hard and soft variables based on qualitative and quantitative company data. The third section outlines the tests performed on automatically calibrated variables. Some variables have been assigned fixed values across all companies. The final section of this appendix illustrates these variables.

C.1. Subsystem outcomes

Chapter 5 of this thesis has compared the model's subsystem outcomes to historical case data to validate the model. As illustrated in the methodology section, these outputs are hard variables. Thereby, they can be estimated with only minimal transformation and avoid researchers' bias and subjective assessment. Table C-1 illustrates the transformations required before importing the data to the model for comparison and validation.

Output	SubS ³⁰	Estimation
Rolling annual revenue	CR	 Estimated by indexing the annual revenue on the venture's income statement by the initial total assets and the value creation scale using the MonetaryScaler. Revenue in year / Initial total assets * MonetaryScaler
Rolling annual return on assets	CA	• Estimated by dividing the venture's annual net income on its income statement by, respectively, its total assets

Table C-1: Measurement of subsystem outcomes

³⁰ Indicates the subsystem that utilises the variables: value creation (CR), value capture (CA), technology development (TD), customer and user (CU), firm management (FM), and content creation (CC).

Rolling annual net profit margin	СА	 on its balance sheet or revenue on its income statement in the same year. Net income in year / Total assets in year Net income in year / Revenue in year
Technological resources	TD	 Estimates by indexing the venture's capitalised development expenses on its balance sheet by its initial total assets and the MonetaryScaler. Some companies may provide an aggregate figure for intangible assets on their balance sheet. In such cases, the notes on the financial reports provide a break-up of intangibles that should include a line item for capitalised development expenses. Capitalising development expenses is an optional accounting procedure. Companies that do not capitalise software development expenses have been excluded. Without capitalised expenses resources, no measurement of technological resources exists. Capitalised development costs / Initial total assets * MonetaryScaler
Customer base	CU	• The sum of all customer and user bases needs to be initialised with one to index value creation
User base	CU	• If only one type of customers/users is used calculate by customers in year / initial customer base
Premium user base	CU	• If customers and users are used, initialise customer and user at 0.5 respectively. If user and premium user are used ensure they are proportional to real data.
Marketing and selling employees	FM	• The initial number of employees in an activity indexes each employee stock. Thus, each stock has a value of one in the first year.
Value delivering employees	FM	• If a venture does not initially employ a type of employee, the stock is indexed by the first year in which it employs the type (only applies to Alpha's value
Technology developing employees	FM	 delivering employees). MS employees in year / Initial MS employees VD employees in year / Initial VD employees
Firm managing employees	FM	 VD employees in year / Initial VD employees TD employees in year / Initial TD employees FM employees in year / Initial FM employees
Complementary assets	FM	 Complementary assets are the venture's property, plant, and equipment in addition to its intangible assets, excluding capitalised development expenses and content assets. Initial total assets and the MonetaryScaler index them. (PPE + Intangibles – development expenses and content assets in year) / Initial total assets * MonetaryScaler
Financial resources	FM	 The venture's financial resources are the cash and equivalents on its balance sheet. Initial total assets and the MonetaryScaler index them Cash and equivalents in year / Initial total assets * MonetaryScaler
Equity employed	FM	 The venture's equity employed is the share capital and additional paid-in capital on the venture's balance sheet. Initial total assets and the MonetaryScaler index the equity. (Share capital + Additional paid in capital in year) / Initial total assets * MonetaryScaler
Debt employed	FM	The venture's debt employed is the long-term debt on its balance sheet.Initial total assets and the MonetaryScaler index the

		debt.		
		• Long-term debt in year / Initial total assets *		
		MonetaryScaler		
Content assets	CC	 Its content assets are intangible assets on the venture's balance sheet reflecting the capitalised costs of content production and acquisition. Initial total assets and the MonetaryScaler index these. Like technological resources, content assets may be part of an aggregated intangibles figure on the balance sheet with a line item provided in the corresponding note. Companies not employing content-based business models have no content assets. Its value is thus zero throughout the simulation. Value of content assets in year/ Initial total assets * MonetaryScaler 		
Content library	СС	 The size of the content library is estimated based on the number of content items offered by the venture. Analysts may use different data points depending on the type of content offered. Examples include the number of songs, videos, or articles; or the total duration of songs or videos. Companies not employing a content-based business model do not have a content library. Its value is thus zero throughout the simulation. Number of content items in year / Initial number of items 		

C.2. Manually set variables

Each subsystem's summary table has categorised the inputs analysts must set. These categories are outlined below with all variables included in them. For each variable, the estimation techniques based on qualitative and quantitative information in annual reports are illustrated. The list provided in this section is extensive. However, many variables reflect different input settings to reflect different business models that can quickly be set to, for example, zero or one. Moreover, many variables in the model exist multiple times. For example, the model uses the same structures for users and customers. It also uses the same structures for all four different employee types and capabilities. The inputs to these replicated structures are derived in the same manner. Therefore, the model is much less driven by exogenous influences as this detailed list would indicate.

C.2.1. Business model design

Chapter 4 shows digital ventures' business models. For example, some digital ventures may employ freemium business models, some rely on network effects, and some are content-based. These different business models have been implemented in the model using policy switches (Coyle, 1996). The analyst needs to set up the model to account for these different business model designs (Table C-2).

Input	SubS	Estimation
Premium product advantage	CR	 Assessed by the analyst by reviewing the premium product's features compared to the base product. Where available, quantitative information should be preferred to avoid bias. For example, the number of content or features free and paid to view and use. Set to zero if the venture does not differentiate between users and premium users
Maximum technological quality	CR	 Assess by reviewing the nature of changes in the venture's technology over the investigation period: 1.2 for ventures that do not make any significant changes to their technology and have a very mature technology. The initial technological quality is always set as one. Therefore, this allows a 20% increase in the venture's technology along the s-curve. 1.5 for ventures that have an established to mature technology but significantly improved their offer to customers. >2 for new companies with a minimum viable product and significant improvements.
Customer direct network effect settings	CR	• Each variable is set based on reviewing the venture's business model:
customer indirect network effect settings	CR	• 0 for business models that do not provide value via the respective network effects. For example, because the venture does not distinguish between
User direct network effect settings	CR	different user and customer groups.1 for business models that make use of the
User indirect network effect settings	CR	 respective network effect. For example, when offering a marketplace or social network. -1 for business models with negative network effects, for example, to reflect increasing competition on marketplaces.
Customer contract period	CA	• The contract periods in years are set based on
Premium user contract period	CA	information provided on the venture's website and annual reports.
Initial content quality	CR	• Both variables are indexed at an initial value of 1
Initial content library	CC	if employed; both variables are zero if they are not employed by a venture

Table C-2: Measurement of business model design inputs

C.2.2. Environment

The model also requires inputs regarding the venture's external environment. These relate to the different stakeholder groups that the model considers. The first set of inputs relate to customer and users. They include their initial values, the value they derive from using the product, and their behaviour regarding word-ofmouth and content creation (Table C-3).

Input	SubS	Estimation		
Initial customer base	CU	• The initial values of the inputs that are		
Initial user base	CU	determined in the performance outcomes section		
Initial premium user base	CU	(see Table C-1).		
Initial customer usage intensity	CR	• One if the venture services the type; zero if the		
Initial user usage intensity	CR	venture does not service it.		
New customer usage intensity	CR	• Set to reflect the development of usage-intensity variables (e.g. each user's features used, content		
New user usage intensity	CR	consumed, transactions facilitated,):Maximum set just above the maximum of the		
Maximum customer usage intensity	CR	development curve observed.New intensity set just below the minimum of the		
Maximum user usage intensity	CR	curve.Adoption rate set to fit the curve's development		
Standard rate of adoption by customers	CR	in the model to historical data.Values are zero if a venture does not service the		
Standard rate of adoption by users	CR	respective customer/user type.		
User to premium conversion rate	CU	 Zero if the company does not distinguish between users and premium users. Set based on the conversation rates provided by		
Premium to user conversion rate	CU	 companies. If not provided, estimate based on publicly available information for similar products. 		
Rate of user acquisition per user	CU	• Set based on the acquisition rates provided by		
Rate of user acquisition per customer	CU	companies.If not provided, estimate based on publicly		
Rate of premium user acquisition per user	CU	available information for similar products.Values are zero if the venture does not highlight		
Rate of premium user acquisition per customer	CU	word-of-mouth as a customer acquisition channel.		
Rate of customer acquisition per user	CU			
Rate of customer acquisition per customer	CU			
Rate of content creation per customer	CC	• Estimated by dividing the indexed number of content items produced by the average indexed number of customers/users.		
Rate of content creation per user	CC	 Indexed number of content produced in year / Average indexed number of users and customers Use the average value throughout a growth state 		
Rate of content creation per premium user	CC	 Use the average value inroughout a growth state Values are zero if customers and users do not create content for the venture. 		

Table C-3:	Measurement of	f customer	and	user	<i>inputs</i>

The second set of environmental inputs relates to the venture's competitors and substitutes (Table C-4).

Input	SubS	Estimation
Strength of competitors and substitutes	CA	 Expresses the strength of the average competitor's product or substitute relative to the venture's product: 0 if there are no competitors or substitutes Between 0 and 1 if competitors' products are inferior 1 if competitors offer products of about equal quality Greater than 1 if competitors' products are superior Each year's value is imported into the model
Number of competitors and substitutes	CA	 Based on the market characteristics outlined by (Lipczynski and Wilson, 2004; Grant, 2016), the following values have been set: 0 for ventures operating in a monopoly 0.25 for ventures operating in an oligopoly 1 for ventures operating under perfect competition Each year's value is imported into the model

Table C-4: Measurement of competitor inputs

The third set of environmental inputs relates to the venture's costs for its different activities, its cost of capital, and the costs per content item (Table C-5).

Input	SubS	Estimation
Cost of value delivering inputs	CA	• Express the indexed annual expenses (wages and other costs) per indexed employee.
Cost of marketing and selling inputs	CA	• (Respective cost in year on the income statement / Initial total assets * Monetary scaler) / (Respective
Cost of technology developing inputs	CA	average employees at beginning and end of year)Values are held constant at the average cost in a
Cost of firm managing inputs	CA	growth state.
Cost of equity	CA	Express the respective annual percentage cost of capital.Respective cost in year on income or cash flow
Cost of debt	CA	statement / Average of respective capital at beginning and end of yearValues are held constant at the average cost in a growth state.
Cost per internally created content item	CC	 Express the indexed expenses or investment per indexed content item. (Respective cost or investment in year on cash flow statement / Initial total assets * Monetary scaler) /
Cost per acquired content item	CC	 (Indexed produced or acquired content items in year) Values are held constant at the average cost in a growth state.
Quality of acquired content	CC	• Set by the analyst to reflect the relative quality of acquired content to the venture's initial content quality.

Table C-5: Measurement of input provider inputs

Lastly, the analyst needs to evaluate the strength of complementary products that integrate with the venture's technology and offer (Table C-6).

Input	SubS	Estimation		
Complementary quality	CR	 The initial value is always one if complementary products contribute to the value provided by the venture. The variable is zero throughout the simulation if complementary products do not add value to the venture's offer. The quality of complementary products is assessed annually relative to the initial value. For example, a value of 1.2 would indicate that the average complementary product is 20% better than at the simulation's start. 		

Table C-6: Measurement of complementary product inputs

C.2.3. Dominant logic

The dominant logic captures the management's goals and ambitions for the venture and what is required to succeed (Prahalad and Bettis, 1986; Von Krogh, Erat and Macus, 2000; Levie and Lichtenstein, 2010). The first element of the dominant logic relates to management targets and employee limits. The analyst needs to set these for each growth state (Table C-7).

Input Su		Estimation		
Target customer growth	FM	• Use the management growth's goals if provided		
Target user growth	FM	in advance.		
Target premium user growth	FM	 Assume achieved growth rates have been 		
Target content library growth	FM	targets if the management provides no goals in advance.Use the state's average growth rates (excluding negatives).		
Target technology improvement	FM	 Reflects the fraction of target improvements on the s-curve³¹: 0 if the company does not plan any improvements 0.1 if the company plans minor improvements 0.2 if the company plans major improvements 0.4 if technology improvements are the company's main priority 		
Max MS employees	FM	• Set to -1 if the company does not have any		
Max VD employees	FM	employee limits. • Set to the number of indexed employees at		
Max TD employees	FM	which the company stabilises. Use the highest		
Max FM employees	FM	indexed number in case of small fluctuations.		

Table C-7: Measurement of target and limit inputs

³¹ After reviewing the s-curves provided by Christensen (1992) and Nikula *et al.* (2010), these values for different improvement rates have been selected. In their empirical investigations of hardware and software innovations, these authors found improvements of up to 40-50% of the remaining s-curve per year. This maximum has been selected as the highest value for technology improvement targets.

The analyst must also set the initial employee levels (Table C-8). These are the initial values determined in the subsystem outcome section above. Besides, the analyst needs to set the employee turnover rate.

Input	SubS	Estimation
Initial marketing and selling employees	FM	• The initial values of the inputs that
Initial value delivering employees	FM	are determined in the subsystem
Initial technology development employees	FM	outcomes section above (see Table
Initial firm managing employees	FM	C-1).
Employee turnover rate	FM	 Use company provided figures if provided (average over the investigation period or growth states). If not provided³²: 20% for companies with normal turnover 30% for companies with high turnover and adverse employee relationships

Table C-8: Measurement of human resource inputs

The final selection of dominant logic variables relates to the firm's complementary assets and financial resource requirements (Table C-9). These can be approximated based on the ventures assets stocks and fundraising efforts.

Table C-9: Measurement of asset requirement inputs

Input	SubS	Estimation
Complementary assets required per employee	FM	 Estimated by calculating the complementary assets per total indexed employees. Initial total assets and MonetaryScaler index complementary assets. (Complementary assets / Initial total assets * MonetaryScaler) / Total indexed employees Use the average of the investigation period/growth state.
Financial resources required per customer	FM	• If the venture has only one goal, these can be estimated by dividing the indexed capital raised per magnitude of the goal. For example:
Financial resources required per user	FM	 (Customer base * Target growth rate) / (Capital raised / Initial total assets * MonetaryScaler)
Financial resources required per technology improvement	FM	• If the venture has multiple goals, it may be unclear which goal caused the fundraising. In such cases, automatic model calibration

³² Reported turnover rates in digital ventures vary widely. While the overall technology industry has turnover rates of around 15%, averages of 20-25% per year have been reported for new ventures (Founders Circle Capital, no date; Power, 2018; Bean, 2019). Therefore, the normal turnover rate has been set at 20% to reflect the average of these reported turnover rates.

Financial resources required per content item	FM	can be used. The values must be positive (feasible) and in the same order of magnitude but likely lower than each estimation (consistency).
Equity debt preference	FM	 Estimated by dividing each year's equity raised by the total of equity and debt raised. Equity raised in year / (Equity raised in year + debt raised in year) Import each year's figure into the model.
Fraction of equity repurchased	FM	 Estimated by dividing the equity or debt repaid in a year by average equity and debt at the beginning and end of the year. Equity repurchased in year / Average equity
Fraction of debt repaid	FM	 Debt repaid in year / Average debt Import each year's figure into the model.

C.2.4. Capability development

The four capabilities modelled in Chapter 4 of the thesis each develop from their initial levels based on their improvement times and new employee capabilities. The analyst needs to set these based on qualitative information in annual reports (Table C-10).

Input	SubS	Estimation
Initial marketing and selling competence	CU	• Capabilities have values between zero and one. The model divides this scale into the five levels of the capability maturity levels (Paulk <i>et al.</i> 1993):
Initial value delivering competence	CU	 0.1 when the activity is performed on an ad hoc basis 0.3 when the activity is executed on a
Initial technology developing competence	TD	 repeatable but reactive basis 0.5 when the processes and routines are defined and integrated 0.7 but distributions in the processes are defined and integrated
Initial firm management competence	FM	 0.7 when the activities are managed and controlled 0.9 when the routines have been continuously improved
New MS employee competence	CU	• Assessed on the same capability maturity
New VD employee competence	CU	 Only the 0.1 and 0.3 values should be
New TD employee competence	TD	used as new employees cannot be integrated yet by definition
New FM employee competence	FM	integrated yet by definition.
MS years of improvement	CU	• Use (maximum) training times if
VD years of improvement	CU	 Otherwise set to reflect the knowledge of
TD years of improvement	TD	capability improvements of the venture

Table C-10: Measurement of capability development inputs

FM years of improvement	FM	over time.If unknown, set training times based on similarly mature firms.
-------------------------	----	--

C.2.5. Accounting measures

The model also relies on some imported accounting measures (Table C-11). Some accounting measures initialise the model's stocks. Other inputs, like other assets and taxation, ensure consistency of variables when validating the model while not being theoretically relevant. This latter group relies on complex rules and discretionary decisions, for example, depreciation schedules. These are out of the model's scope.

T4	C L C	
Input	SUDS	Estimation
Other assets	CA	 Reflect all assets recognised on the firm's annual balance sheet but not included in the model. Calculated by indexing each year's total assets by initial total assets and the MonetaryScaler. Assets included in the model (see performance outcomes) are then subtracted. Total assets in year / Initial total assets * MonetaryScaler – Technological resources in year – Financial resources in year – Complementary assets in year – Content assets in year
Rolling annual taxation	СА	 Reflect the taxes paid (or tax credit received) by the venture in a given year. Taxes paid are a positive number. Taxes credits received are a negative number. Calculated by indexing each year's taxes on the venture's income statement by initial total assets and the MonetaryScaler. Taxes in year / Initial total assets * MonetaryScaler
Initial technological resources	TD	• The initial values of the inputs that are determined
Initial financial resources	FM	in the performance outcomes section (see Table C-
Initial complementary assets	FM	1).
Initial equity employed	FM	
Initial debt employed	FM	
Initial content assets	CC	
Lifetime of technological resources	TD	 Each year's lifetime is calculated by dividing the stock's average value (at the beginning and end of a year) by the loss during a year. If the venture does not employ a content-based
Lifetime of complementary assets	FM	 business model, the lifetime of content items and assets is zero throughout the simulation. Average technological resources at beginning and end of year / Amortisation of tech resources in year
Lifetime of content asset	CC	 Average complementary assets at beginning and end of year / Depreciation of comp assets in year Average content assets at beginning and end of

Table C-11: Measurement of accounting inputs

Content lifetime	CC	 year / Amortisation of content assets in year Average number of content items at beginning and end of year / Loss of content items in year
Rate of content acquisition	СС	 The value of content items purchased in a year, indexed by initial total assets and the MonetaryScaler. Like the investment in technological resources, the acquisition can be found on the venture's annual cash flow statement. It is zero throughout the simulation if the venture does not employ a content-based business model or purchases no content. Value of content acquisition per year / Initial total assets * MonetaryScaler

C.3. Automatically calibrated variables

The model also determines some input values using the Vensim software's calibration features. The software determines variable values by identifying inputs that maximise the model's fit to data for other variables. This process is commonly used in System Dynamics when no data is available to set input values manually, particularly for variables required for specific structures. In the model, these structures relate to three types of variables: standard rates, effect sizes, and the MontaryScaler. Calibration is an established process and critical for testing the model's ability to replicate the behaviour of a system (Lyneis and Pugh, 1996; Oliva, 2003, 2004; Homer, 2012). To ensure the reliability of the automated process, Oliva (2003) recommends confirming variables' feasibility and consistency (see subsection 2.4.1.2 of the methodology chapter for details). The automatically calibrated variables and their test are illustrated below. The first group of automatically calibrated variables relates to standard rates and effects (Table C-12). The model adjusts these variables as part of Sterman's (2000) multiplicative effects structure or s-curve.

Input	SubS	Validation	
Standard customer lifetime	CU	 <i>Feasibility</i>: Values need to be positive. No negative lifetimes are possible. <i>Consistency</i>: While customer and user lifetimes in a digital context differ widely, they can generally be assumed to be between two and eight years. Unless otherwise stated by the company, these values should be confirmed for the model's standard and adjusted lifetimes. 	
Standard user lifetime	CU		
Standard effect of investment in technology on technological improvement	CR	 <i>Feasibility</i>: Values need to be positive. No negative effect is possible. <i>Consistency</i>: The variable transforms investments into technology into improvements in technological quality. The development of the technological quality in the model should reflect qualitative statements about product changes in annual reports. 	

Table C-12: Automatically calibrated standard values

The second group of automatically calibrated variables relates to the effect sizes that transform employee competencies from a soft scale between zero and one to a hard impact on the system (Table C-13). These effect sizes are part of Warren's (2002) capability structure adopted in the model and reflect a business model's activity throughput at fully developed capabilities.

Table C-13: Automatically calib	brated ef	fect sizes
Innut	SubS	Validation

Input	SubS	Validation
Effect of competence on ability to acquire customers	CU	• <i>Feasibility</i> : Values need to be positive. No negative effects are possible.
Effect of competence on ability to acquire users	CU	• <i>Consistency</i> : The analyst should compare variable values to approximations of the effect size. For example:
Effect of competence on ability to acquire premium users	CU	• Indexed customers acquired in period / Average number of indexed MS employees at beginning and end of year / Estimated focus of employees /
Effect of competence on ability to service customers	CU	 Initial marketing and selling competence Indexed customers in year / Number of indexed VD employees in year / Estimated focus of
Effect of competence on ability to service users	CU	 Indexed customers in year / Number of indexed VD employees in year / Estimated focus of
Effect of competence on ability to develop technology	TD	 Indexed investment in technology in period / Average number of TD employees at beginning and end of year / Initial technology developing
Effect of competence on ability to maintain technology	TD	 Indexed number of operating employees in year * (1+Mangerial Slack) / Average number of FM
Effect of competence on ability to manage employees	FM	 employees / Initial firm managing competence Indexed content items created in period / Average number of indexed VD employees at beginning
Effect of competence on ability to create content	CC	and end of year / Estimated focus of employees / Initial value delivering competence

Effect of competence on content quality	CC	• <i>Feasibility</i> : Values need to be positive. No negative effects are possible.
Effect of experience on customer content quality	CC	• <i>Consistency</i> : These variables affect the development of content quality. The development
Effect of experience on user content quality	CC	of the content quality should reflect the statements made in annual reports.

The model indexes many hard variables to obscure the identity of case companies. Thereby, it transforms them from variable scales of absolute numbers to a reference point. This reference point is usually the initial value of a variable. For example, the model expresses the marketing and selling employees at any year relative to the venture's initial marketing and selling employees. Thereby, the stock in the model has an initial value of one. If the venture's employees increase from 50 to 55 employees over the first year, the model shows an increase from 1 to 1.1. This indexing also applies to all monetary amounts and rates. The reference point for revenues, costs, and assets owned by the venture is the initial rate of value creation of 1. In the above setting, monetary values are transformed to this scale by initial total assets and the MonetaryScaler. Indexing monetary values by initial assets already obscures companies' identities. The MonetaryScaler then transform the indexing from the initial total assets scale to the value creation scale (Table C-14). This is required to ensure that all monetary values in the model can be used in calculation while treating value creation as an indexed qualitative variable. This rescaling is executed through automatic model calibration after using the assets-indexed values as input (see Chapter 5 and Appendix D).

Input	Validation
MonetaryScaler	 <i>Feasibility</i>: Values need to be positive. No negative scaler is possible. <i>Consistency</i>: Analysts can approximate the scaler via the venture's share of value captured from users and customers. Because this share expresses the relationship between value creation and revenues, the share needs to be adjusted for the ratio of revenue and total assets. Initial share of value captured from customer and users / Estimated initial rate of revenue generation * Initial total assets The initial share of value captured can be calculated using the bargaining power equations (e.g. Equation 4-22) and the fraction of paying customer and premium users (Table C-3). The initial product strength is one, the initial strength and number of competitors are estimated by the analyst (Table C-4). The analyst can use different data points to estimate the initial rate of revenue generation. For example, some digital ventures report customer/user numbers and expected annual revenue per customer or user. These can be multiplied to determine the expected initial rate.

Table C-14: Automatically calibrated monetary scale adjustment
Some SaaS companies also publish annualised recurring revenue rates, which can be used for this rate. Suppose the venture provides no such data point. In that case, the analyst should use the reporting revenue for the financial year ending at the beginning of the simulation as the best
available estimate.

C.4. Fixed variables

The model assumes that some variables can be held constant across all companies. These fixed variables are outlined below with their values and explanation for the values (Table C-51).

Table C-15: Overview of fixed variables

Input	Value	Reasoning
Initial technological quality	1	 To index the initial product strength, use value creation, and rate of value creation. The technological quality is always set at 1. Other initial value drivers are zero or one, depending on the venture's business model. However, all digital ventures require a digital technology.
Reporting period	1	• The model is validated using company annual reports. Therefore, the model sets the reporting period to one year.
Reporting delay	0.25	 The reporting delay reflects company typical quarterly reporting intervals to, for example, shareholders. A value of one quarter also allows expectation formation to be reasonably quick without being too sensitive to short-term fluctuations.
Delay to lay off	0.1	• Reflects the time required to lay employees off. It is set to about 5.2 weeks (1/10 of a year) which is realistic considering managerial decision making and potential contract/legal delays to lay off.
Target managerial slack	0.5	• The target managerial slack has been set to 0.5. It means that the management has one-third of its time available to plan and execute growth plans (when operating at required staffing levels).
Effect of competence on ability to hire Effect of competence on ability to acquire CA Effect of competence on ability to raise capital	12	• With the effects set at 12, a fully capable manager with no other tasks requires one month to fulfil hiring, complementary asset acquisition, and fundraising targets.

Appendix D: Additional company inputs

Chapter 5 of the thesis has outlined the main inputs for each case company. However, besides those inputs noted in the Chapter, some minor additional inputs are required to run the model for each case company. They relate to goals regarding premium users and content creation. These inputs are zero for all companies because they do not employ freemium or content-based business models. While they are not required and relevant for any case company, they were implemented in the model to reflect the literature on digital ventures. Stocks' initial values and accounting measures also need to be set for each company separately.

D.1. Additional inputs for all companies

Chapter 5 of this thesis has outlined the managerial targets and limits for all companies (e.g. Table 5-1). In addition, analysts must set targets for the companies' premium user and content library growth (Table D-1). These are zero for the case companies because they do not distinguish between users and premium users. Nor do they employ a content-based business model. These need to be set to zero even though they are not relevant. Otherwise, the Vensim software creates an error as it expects some value for them.

Dominant logic – targets and HR	
Target premium user growth	0%
Target content library growth	0%
liquidity goal per content item	0%

Table D-1: Additional dominant logic inputs for all companies

In addition, each companies' content creation subsystem requires inputs (Table D-2). These inputs have been set to zero as the subsystem is not relevant for any identified company. The subsystem was built using structures verified and validated in other parts of the thesis (see Appendix A). This thesis has thus developed a model that reflects the theory on digital ventures wholistically. The model thus applies to any digital venture. However, the model has not been

validated for content-based business models. Therefore, users should use it with caution for any company employing such a business model design.

Content creation inputs			
Rate of content creation per customer	0	Cost per internally created content item	0
Rate of content creation per user	0	Cost per acquired content item	0
Rate of content creation per premium user	0	Quality of acquired content	0
Effect of experience on customer content quality	0	Effect of experience on user content quality	0
Effect of competence on ability to create	0	Effect of competence on content	0
Content lifetime	0	Acquisition of content	0

Table D-2: Content creation subsystem inputs for all companies

D.2. Additional inputs for Alpha

In addition to the inputs presented in Chapter 5, the model requires some additional inputs regarding initial values and accounting measures (Table D-3). All employee stocks are initialised with a value of one. Alpha does not employ any value delivering employees at the beginning of the simulation. Thus, this stock has an initial value of zero. Their value after one year of simulation is the reference point to index value delivering employees. Alpha does not have a content library because it does not employ a content-based business model. Its initial balance sheet resources are split into about 55% of financial resources, 5% of complementary assets, and 29% of technological resources. These are financed using no debt but ten times the value of its initial assets in equity. This number is high because the company went public via a reverse takeover where it bought a publicly traded, bankrupt company. This equity position includes the equity of the previous company.

Initial values for employee and balance sheet stocks								
Technology developing employees			Finan	cial resou		55.21%		
Marketing and selling employees		1	Comp	lementai	y assets			5.23%
Value delivering employees		0	Techr	nological	resource	s		28.73%
Firm managing employees		1	Conte	nt assets				0.00%
Content library		0	Equit	y employ	red		10	54.31%
·			Debt	employed		0.00%		
Accounting measures	t0		t1	t2	t3	t4	t5	t6
Lifetime of comp. assets			3.222	2.764	2.664	1.622	1.405	1.532
Lifetime of tech. resources			4.07	3.09	2.22	1.52	2.72	6.05
Lifetime of content assets			0	0	0	0	0	0
Equity debt preference			100%	100%	100%	0.8%	100%	100%
Equity repurchasing			0%	0%	0%	0%	0%	0%
Debt repayment			0%	0%	0%	0%	200%	0%
Other assets	119	V ₀	12%	22%	29%	38%	47%	58%
Rolling annual taxation	00	<i>/</i> ₀	-3%	-5%	-17%	-15%	-12%	-13%

Table D-3: Initial values and accounting measures for Alpha

The table also provides the lifetimes of Alpha's resources in each year. Its resources have lifetimes between 1.4 and 6.05 years. Alpha raises only equity in all but the fourth year of simulation. It repays the debt in year five. Other assets included on Alpha's balance sheet and the venture's taxation (both indexed by initial total assets) are also imported to the model on an annual basis. At the beginning of the simulation, assets unaccounted for in the model make up 11% of Alpha's initial total assets. They rise over six years to 58%. Alpha receives tax credits of between 0% and 17% of the initial total assets per year.

D.3. Additional inputs for Beta

Beta employs staff across the four activities considered in the model at the beginning of the simulation. Therefore, all employee stocks have an initial value of 1 (Table D-4). Beta's initial balance sheet assets are composed of about 35% financial resources, 8% complementary assets, and 6% technological resources. The venture has raised 118% of initial total assets in equity, and another 3% in debt before the simulation starts. Beta does not employ a content-based business model. Therefore, it does not have a content library or content assets. Their initial values thus are zero.

Initial values for employee and balance sheet stocks								
Technology developing employees			Financial resou	Financial resources				
Marketing and selling emplo	yees	1	Complementar	ry assets	8.49%	6		
Value delivering employees		1	Technological	resources	5.56%	6		
Firm managing employees		1	Content assets		0.00%	6		
Content library		0	Equity employ	red	118.45%	6		
			Debt employed	1	2.76%	6		
Accounting measures	t0		t1	t2	t3			
Lifetime of comp. assets			1.84	1.77	1.77	7		
Lifetime of tech. resources			4.13	2.35	1.51	1		
Lifetime of content assets			0	0	0	0		
Equity debt preference			100%	100%	100%	6		
Equity repurchasing			0%	0%	0%	6		
Debt repayment			38.42%	54.73%	91.46%	6		
Other assets	519	%	52%	53%	46%	6		
Rolling annual taxation	-0.379	%	0.41%	0.45%	0.44%	6		

Table D-4: Initial values and accounting measures for Beta

The lifetime of Beta's complementary assets stays nearly constant at a level of about 1.8 years throughout the simulation. The lifetime of its technological assets falls from 4.13 to 1.51 years over the three years of simulation. The venture raises all its capital in equity but repays parts of its initial debt over the simulation years. Beta holds additional balance sheet assets of around 50% of initial total assets, which are not considered in the model. Beta receives a small tax credit in the year ending at the beginning of the simulation and pays small amounts of tax in each of the following years.

D.4. Additional inputs for Gamma

Because Gamma has employees across four activities, their initial indexed values are 1 (Table D-5). Its initial balance sheet resources include about 29% of initial total assets as financial resources. Moreover, Gamma holds about 3% of complementary assets and technological resources each. The venture has raised 88% of its initial total assets in equity and 0.1% in debt. Gamma does not employ a content library or hold content assets.

Initial values for employee and balance sheet stocks								
Technology developing employees			Financial resou	Financial resources				
Marketing and selling emplo	yees	1	Complementar	ry assets	2.98%			
Value delivering employees		1	Technological	resources	2.94%			
Firm managing employees		1	Content assets		0.00%			
Content library		0	Equity employ	red	87.85%			
			Debt employed	1	0.10%			
Accounting measures	t0		t1	t1 t2				
Lifetime of comp. assets			3.12	3.13	2.79			
Lifetime of tech. resources			2.69	1.50	1.33			
Lifetime of content assets			0	0	0			
Equity debt preference			94.2%	100.0%	94.6%			
Equity repurchasing			0.00%	0.00%	4.52%			
Debt repayment			119.30%	16.14%	0.56%			
Other assets	659	%	72%	81%	63%			
Rolling annual taxation	0.469	%	-0.15%	0.21%	-0.01%			

Table D-5: Initial values and accounting measures for Gamma

Throughout the years, the lifetime of Gamma's complementary assets falls from 3.12 to 2.79 years. The lifetime of its technological resources falls from 2.69 to 1.33 years. Every year, the capital raised by Gamma is composed of at least 94% equity. The company repays significant fractions of debt in the first year of the simulation. Depending on the year, assets not considered in the model amount to between 63% and 81% of initial total assets. Gamma switches from receiving small amounts of tax credits to paying small tax charges every year.

D.5. Additional inputs for Delta

Delta's employee stocks have initial values of one to reflect that the venture employs staff across all four activities considered in the model (Table D-6). The assets considered in the model include about 23% of initial total assets in financial resources and less than 1% of initial total assets as complementary assets and technological resources. The venture has financed these assets using equity and debt of 83% and 4% respectively of initial total assets. Delta does not employ a content-based business model. It thus does not have a content library or content assets.

Initial values for employee and balance sheet stocks								
Technology developing employees			Financial resou	Financial resources				
Marketing and selling emplo	yees	1	Complementar	y assets	0.74%			
Value delivering employees		1	Technological	resources	0.35%			
Firm managing employees		1	Content assets		0.00%			
Content library		0	Equity employ	red	82.81%			
			Debt employed	1	3.83%			
Accounting measures	t0		t1	t2	t3			
Lifetime of comp. assets			1.40	1.56	2.72			
Lifetime of tech. resources			1.26	2.02	1.21			
Lifetime of content assets			0	0	0			
Equity debt preference			100%	100%	100%			
Equity repurchasing			0%	0%	0%			
Debt repayment			150%	0%	45%			
Other assets	769	%	87%	91%	87%			
Rolling annual taxation	0.66	%	-0.47%	0.11%	2.08%			

Table D-6: Initial values and accounting measures for Delta

The lifetime of Delta's complementary assets rises from 1.4 to 2.7 years over the three simulation years. The lifetime of its technological assets fluctuated between 1.21 and 2.02 years. During its simulation, it raises its new capital entirely as equity. The company also repays significant amounts of debt in the first and third simulation years. Besides the assets recognised in the model, the company holds assets worth between 76% and 91% of its initial total assets. The company pays small amounts of taxes in all but the first year of simulation.

Appendix E: Overview of scenario inputs

Chapter 6 of this thesis has simulated the model for a range of hypothetical scenarios. These scenarios investigate and illustrate the performance development and trade-offs between value creation and capture for digital ventures. The chapter has outlined the principles guiding the design of the base case, its most essential inputs, and changes to the base case for additional scenarios. This appendix provides an overview of all model inputs required to run the base case. It presents inputs in the format used for case companies. It then illustrates changes to the base case for each scenario. Thereby, this appendix creates full transparency over the scenario simulations.

E.1. Base case model inputs

The base case with caps on employee numbers (BaseC) has the following inputs for its firm managing subsystem (Table E-1):

- The venture targets a customer growth goal and technology improvement goal of 25% and 10% per year.
- Its employee caps have been set at their initial indexed level of one. Thus, BaseC does not hire additional employees. The company operates at the maximum number of employees it is willing to hire.
- BaseC employs a SaaS business model. Thus, it does not distinguish between customers, users, and premium users. It also does not employ a content library. Therefore, all targets regarding users, premium users, and content are irrelevant for BaseC and set to zero.
- BaseC's initial capability level has been set to 0.5. This midpoint allows the venture to improve its employee productivity over time. It takes the venture one year to train its employees. This value is realistic compared to the case companies in Chapter 5 and allows the venture to improve capabilities within the five-year simulation period. New employees have a capability of 0.3 because they require integration into the venture's processes.
- The higher employee turnover rate of 30% has been set for BaseC.

- The venture does not raise any capital to acquire customers or users, improve its technology, or create content. Thus, the financial resources accumulate based on the venture's value creation and capture. Therefore, they can be used as a cumulative performance measure.
- The venture requires 7.5% of initial total assets per employee. The value has been set to ensure that the venture has initial complementary assets that are sufficient to equip its initial employees. Therefore, no acquisition of complementary assets and cash outflow takes place at the beginning of the simulation to correct imbalances.
- Similarly, BaseC's effect of competence on its ability to manage the firm has been set to 9. Thereby, the venture's initial firm managing employees can manage its operating employees with 50% managerial slack. Thus, no initial hiring or layoffs eliminate imbalances between required and employed firm managing employees.

Dominant logic – targets and HR	time: 0 - 5			
Target customer growth	25%			
Target user growth		0%		
Target premium user growth		0%		
Target technology improvement		10%		
Target content library growth		0%		
Marketing and selling (MS) employee ca	р	1		
Value delivering (VD) employee cap		1		
Tech. developing (TD) employee cap		1		
Firm managing (FM) employee cap		1		
Capabilities - firm managing		Capital asset requirements		
initial firm managing competence	0.5	complementary assets per emp.	7.5%	
firm managing years of improvement	1	liquidity goal per customer	0	
new FM employee competence	0.3	liquidity goal per user	0	
Employee turnover rate 30%		liquidity goal per tech improvm.	0	
effect of competence on ability to manage the firm	9	liquidity goal per content item	0	

Table E-1: Firm management inputs for the base case

BaseC's technology developing capability settings are the same values as its firm managing capabilities (Table E-2). These allow the venture to improve its productivity over time. Each indexed employee at fully developed capabilities can maintain four times the venture's initial technological quality and develop 40% of initial total assets in new technological resources. These values ensure that the venture's initial technology developing employees can achieve its technology improvement goal of 10% (see Table E-1).

Capability development					
initial technology developing competence	0.5	effect of competence on ability to develop technology	40%		
TD years of improvement	1	effect of competence on ability to	4000/		
new TD employee competence	0.3	maintain technology	400%		

Table E-2: Technology developing inputs for the base case

BaseC does not distinguish between customers, users, and premium users.

Therefore, it employs only the model's customer stock, and its initial value is one (Table E-3). Variables related to users and premium users are zero. Moreover, the venture's word-of-mouth variables have been set to zero. They are altered as part of the environmental scenarios. The standard customer lifetime is five years. It reflects a mid-point of the values observed for case companies. Marketing and value delivering capabilities have the same values as the firm managing and technology developing capabilities (see Table E-1 and E-2). These allow for productivity improvements over time. Each indexed marketing and selling employee can acquire 1.5 indexed customers per year. Each indexed value delivering employee of BaseC can service two indexed customers. These effect sizes ensure that the venture is operating at full capacity at the beginning of the simulation. There is no idle capacity causing layoffs of employees or a capacity shortage requiring additional employees. Moreover, the venture's marketing and selling employees can initially achieve the venture's customer growth goal. This further prevents initial hiring and layoffs and ensures that all changes in performance outcomes are due to theoretically relevant internal changes.

User and customer behaviour					
initial customer base	1	rate of customer acquisition per customer	0		
initial user base	0	rate of customer acquisition per user	0		
initial premium user base	0	rate of user acquisition per customer	0		
user to premium conversion rate	0	rate of user acquisition per user	0		
premium to user conversion rate	0	rate of premium user acquisition per cust.	0		
standard customer lifetime	5	rate of premium user acquisition per user	0		
standard user lifetime	0				
Capability development					
initial marketing selling competence	0.5	initial value delivering competence	0.5		
MD years of improvement	1	VD years of improvement	1		
new MS employee competence	0.3	new VD employee competence	0.3		
effect of competence on ability to acq	uire c	ustomers	1.5		
effect of competence on ability to acq	uire u	sers	0		
effect of competence on ability to acquire premium users					
effect of competence on ability to serv	vice ci	ustomers	2		
effect of competence on ability to serv	vice u	sers	0		

Table E-3: Customer and user subsystem inputs for the base case

As a SaaS company, BaseC does not utilise network effects (Table E-4). Because the venture does not distinguish between users and premium users, the premium product advantage and initial user usage intensity are zero. The same applies to other variables regarding user usage intensity. The maximum technological quality of BetaC and standard effect of investment has been set to 1.5. They align with the venture's technology developing capabilities in preventing immediate hiring and layoff. Moreover, the maximum quality allows the venture to improve the value it creates over time. Customer usage intensity variables also incorporate this potential for improvements in value creation. The maximum usage intensity of 1.5 allows the venture to increase its value creation.

Business model designs	Business model designs					
Customer direct network effect settings	0	Premium product advantage	0			
Customer indirect network effect settings	0	Maximum technological quality	1.5			
User direct network effect settings	0	Standard effect of investment in	15			
Customer indirect network effect settings	0	technology on tech. improvement	1.3			
Initial customer usage intensity	1	Initial user usage intensity	0			
User and customer behaviour		time: 0 - 5				
Maximum customer usage intensity		1				
New customer usage intensity		1.5				
Standard rate of adoption by customers	0.5					
Maximum user usage intensity		0				
New usage intensity		0				
Standard rate of adoption by users		0				
Complementary product		time: 0 - 5				
Quality of complementary products		1				

Table E-4: Value creation inputs for the base case

A contract period of one year has been set (Table E-5). This length is similar to the value observed in case companies. BaseC loses value initially. The input values achieve this through costs of 25% of initial value creation. The value created is the maximum value that can be captured. At best, the venture thus breaks even with these costs. However, the venture operates in an oligopolist (0.5) with equally strong competitors (1). This combination reduced the share of value capture from customers. These values ensure that BaseC does not have an initial ability to capture value. The venture does not compensate providers of equity and debt.

Table E-5: Value capture inputs for the base case

Business model designs			
Customer contract period	1	Premium user contract period	0
Input providers		time: 0 - 5	
Cost of value delivering inputs		25%	
Cost of marketing and selling inputs		25%	
Cost of technology developing inputs		25%	
Cost of firm managing inputs		25%	
Cost of equity		0%	
Cost of debt		0%	
Competition		time: 0 - 5	
Number of competitors		0.5	
Strength of competitors		1	

The venture does not employ a content-based business model. Therefore, all inputs regarding the content creation subsystem are zero (Table E-6).

Content creation inputs			
Rate of content creation per customer	0	Cost per internally created content item	0
Rate of content creation per user	0	Cost per acquired content item	0
Rate of content creation per premium user	0	Quality of acquired content	0
Effect of experience on customer content quality	0	Effect of experience on user content quality	0
Effect of competence on ability to create content	0	Effect of competence on content quality	0
Content lifetime	0	Acquisition of content	0

Table E-6: Content creation subsystem inputs for the base case

BaseC is assumed to employ staff across all activities (Table E-7). Therefore, the venture's initial employee stocks have a value of one. Its initial financial resources are 100% of initial total assets. Its initial complementary assets and technological resources are 30% and 20% of initial total assets. These values are unrealistic because initial total assets amount to more than 100%. However, the initial value of financial resources at one allows an easy performance comparison. For example, a scenario that achieves financial resources of two has doubled its initial resources. The venture has financed these assets entirely by equity. Complementary assets and technological resources have a lifetime of three years. The value is with the range of lifetimes observed among case companies. BaseC does not repay any debt or equity. If it were to raise capital, it would do so entirely in equity. However, the model requires an input for the variable between zero and one. No other assets or taxation are imported to the model for BaseC. These are not theoretically relevant. The model used them to ensure consistency between accounting measures and model variables for testing purposes.

Initial values for employee and balance sheet stocks							
Technology developing emplo	yees	1	Financ	cial resou	urces		100%
Marketing and selling employe	ees	1	Comp	lementai		30%	
Value delivering employees		1	Techn	ological		20%	
Firm managing employees		1	Conter	nt assets			0%
Content library		0 Equity employed				150%	
-		Debt employed				0%	
Accounting measures	t0	t1		t2	t3	t4	t5
Lifetime of comp. assets					3		
Lifetime of tech. resources					3		
Lifetime of content assets					0		
Equity debt preference					1		
Equity repurchasing					0%		
Debt repayment		0%					
Other assets				0	%		
Rolling annual taxation				0	%		

Table E-7: Initial values and accounting measures for the base case

The sections below outline the changes made to the above base case inputs for individual scenarios. The tables outline only the changes to the base case to avoid repetition. The sections mirror Chapter 6's structure to allow the reader to easily compare this appendix's model inputs and the chapter's scenario outcomes.

E.2. Dominant logics with a cap on employee numbers

This section outlines the changes made to BaseC to simulate additional scenarios for companies with a cap on employee numbers. These scenarios change the targets expressed in the management's dominant logic and variables related to capability development to reflect internal improvements.

E.2.1. Different targets expressed in the dominant logic

Section 6.1.2 simulates different targets expressed in the management's dominant logic for companies with a cap on employee numbers. Three scenarios alter BaseC. These scenarios increase the venture's technology improvement target (TechC), its customer growth target (ScaleC), or both targets (AmbitiousC). The variable values are doubled to reflect higher targets (Table E-8).

Table E-8: Changes to employee-capped base case's dominant logic

Variables	BaseC	TechC	ScaleC	AmbitiousC
Target customer growth	25%	25%	50%	50%
Target technology improvement	10%	20%	10%	20%

E.2.2. Internal improvement for capability development

Section 6.1.3 alters variables affecting capability development to represent internal improvement programs (Table E-9). These maintain all but one input of BaseC. One scenario reduces the capability improvement times to half a year (TrainingC). Another scenario decreased new employee capabilities from 0.3 in BaseC to 0.1 (NewEmpsC). The last scenario in this section reduces the employee turnover rate from 30% to 10% (TurnoverC).

BaseC Variables TrainingC **NewEmpsC** TurnoverC Capability improvement times 0.5 1 1 New employee capability level 0.3 0.1 0.3 30% 10% Employee turnover rate

Table E-9: Changes to employee-capped base case's capability development

E.3. Dominant logics without a cap on employee numbers

The second section of Chapter 6 simulates the development of companies without a cap on employee numbers. It outlines the development of a new base case without a cap on employee numbers (BaseU). It then investigates the impact of different managerial targets and internal improvements on BaseU.

E.3.1. Employee-uncapped base case

BaseU maintains all inputs from BaseC except its employee caps (Table E-10). BaseC is unwilling to hire more than its initial employee of one indexed employee per activity. BaseU does not have employee caps. These are reflected by a -1 in the model. The development of BaseU compared to BaseC is outlined in section 6.2.1.

Table E-10: Changes to develop BaseU

Dominant logic – targets and HR	BaseC	BaseU
Marketing and selling (MS) employee cap	1	-1
Value delivering (VD) employee cap	1	-1
Tech. developing (TD) employee cap	1	-1
Firm managing (FM) employee cap	1	-1

E.3.2. Different targets expressed in the dominant logic

Section 6.2.2 simulates the impact of different managerial targets on companies without a cap on employee numbers. These scenarios build on BaseU and maintain no cap on employee numbers. The scenarios implement the same changes used for BaseC's different targets (Table E-11).

Table E-11: Changes to employee-uncapped base case's dominant logic

Variables	BaseU	TechU	ScaleU	AmbitiousU
Target customer growth	25%	25%	50%	50%
Target technology improvement	10%	20%	10%	20%

E.3.3. Internal improvement for capability development

Section 6.2.3 changes variables related to capability development to simulate the impact of internal improvements on companies without a cap on employee numbers. These scenarios build on BaseU and do not have a cap on employee numbers. They apply the same changes used to simulate internal improvement for BaseC (Table E-12).

Table E-12: Changes to employee-uncapped base case's capability development

Variables	BaseU	TrainingU	TrainingU NewEmpsU	
Capability improvement times	1	0.5		1
New employee capability level	(0.3	0.1	0.3
Employee turnover rate		30%		10%

E.4. Transitioning between dominant logics

The third section of Chapter 6 investigates the transition from dominant logics without a cap on employee numbers to logics with caps on employee numbers.

The section first describes a first transitioning base case. The section then explores different targets, layoffs, growth cycles, and internal improvements.

E.4.1. Transitioning base case

The transitioning base case (BaseT) in subsection 6.3.1 continues the development of BaseU for another five years. During the first five years, the settings are the same as for BaseU. Another five years of simulation are then added to BaseU to investigate the introduction of employee caps halfway through the simulation. Therefore, the venture has two growth states, similar to the case companies illustrated during model testing. During the first five years, the venture grows without a cap on employee numbers (Table E-13). During the second five years, it introduces caps on all employee numbers. These caps have been set by running the model without caps at first. Each activity's cap is the respective employee number after five years of simulation.

Table E-13: Changes to develop BaseT

Dominant logic – targets and HR	time: 0 – 5 (=BaseU)	time: 5 – 10
Marketing and selling (MS) employee cap	-1	2.15
Value delivering (VD) employee cap	-1	2.9
Tech. developing (TD) employee cap	-1	0.7
Firm managing (FM) employee cap	-1	1.5

E.4.2. Different targets expressed in the dominant logic

The scenarios for employee-capped and -uncapped companies have investigated the impact of different targets. Transitions increase the possibilities and complexities of the dominant logic. Besides different targets, the simulations below also outline the development of companies with different restructuring sizes and growth cycles.

E.4.2.1. Different targets throughout the simulation

The first set of dominant logic-related scenarios investigates the development of transitioning companies with different growth and improvement targets. The scenarios use the same targets as for BaseC and BaseU (Table E-14). Subsubsection 6.3.2.1 shows the outcomes of these scenarios.

Variables	BaseT	TechT	ScaleT	AmbitiousT
Target customer growth	25%	25%	50%	50%
Target technology improvement	10%	20%	10%	20%

Table E-14: Changes to transitioning base case's dominant logic's targets (1)

The principles of the design of BaseT have been applied to these scenarios. The companies grow without caps on employee numbers during the first five years. Thus, their inputs for the first growth state are the same as for BaseU. During the second growth state, employee caps have been introduced. The employee cap for each activity reflects the number of employees employed by the venture at the end of the first growth state. Each scenario employs different numbers of employees to achieve its targets. Therefore, their employee caps differ (Table E-15).

TechT	time: 0 – 5 (=BaseU)	time: 5 – 10
Marketing and selling (MS) employee cap	-1	2.15
Value delivering (VD) employee cap	-1	2.9
Tech. developing (TD) employee cap	-1	0.96
Firm managing (FM) employee cap	-1	1.5
ScaleT	time: 0 – 5 (=BaseU)	time: 5 – 10
Marketing and selling (MS) employee cap	-1	9.94
Value delivering (VD) employee cap	-1	10.6
Tech. developing (TD) employee cap	-1	0.65
Firm managing (FM) employee cap	-1	5.9
AmbitiousT	time: 0 – 5 (=BaseU)	time: 5 – 10
Marketing and selling (MS) employee cap	-1	9.9
Value delivering (VD) employee cap	-1	10.75
Tech. developing (TD) employee cap	-1	0.9
Firm managing (FM) employee cap	-1	6

Table E-15: Changes to transitioning base case's dominant logic's targets (2)

E.4.2.2. Different restructuring magnitudes

BaseT sets the caps of employee numbers during the second growth state at the number of employees employed at the end of the first growth state. However, model validation has revealed that all companies reduce their employee numbers as part of restructuring programs. Therefore, the impact of layoffs should also be evaluated. Subsubsection 6.3.2.2. illustrates two scenarios. They are based on and adapt BaseT. During the first five years, they grow without employee limits. During the second half of the simulation, they introduce caps on employee numbers. However, the caps of employee numbers are 20% and 50% below those

in BaseT (Table E-16). Therefore, the companies lay off 20% and 50% of employees when transitioning between growth states.

Reductions20	time: 0 – 5 (=BaseU)	time: 5 – 10
Marketing and selling (MS) employee cap	-1	1.72
Value delivering (VD) employee cap	-1	2.32
Tech. developing (TD) employee cap	-1	0.56
Firm managing (FM) employee cap	-1	1.2
Reductions50	time: 0 – 5 (=BaseU)	time: 5 – 10
Markating and calling (MS) amplayee can	1	4 0
Marketing and sening (MS) employee cap	-1	1.075
Value delivering (VD) employee cap	-1 -1	<u>1.075</u> 1.45
Value delivering (VD) employee cap Tech. developing (TD) employee cap	-1 -1 -1	1.075 1.45 0.35

Table E-16: Changes to transitioning base case's dominant logic's restructuring

E.4.2.3. Different number of growth cycles

BaseT involves one growth cycle with an initial period of growth followed by an employee-capped period. However, companies may switch more regularly between dominant logics. Therefore, subsection 6.3.2.3 created scenarios with two and five growth cycles. These build on and follow the same principles as BaseT. After an initial period of growth, the scenarios introduce caps on employee numbers. These caps are set to the employee numbers at the end of the previous growth period (Table E-17).

2Cycles	time	: 0 - 2.	5 t	ime: 2.	5 – 5	time	: 5 – 7.	5 t	time: 7.5 – 10				
MS employee cap		-1		1.3		-1			3.1				
VD employee cap	-1		-1 1.8		1.8		1.8			-1		4.4	
TD employee cap		-1		0.75		0.75 -1		-1 0		0.7			
FM employee cap		-1		1		-1			2.2				
5Cycles	0-1	1-2	2-3	3-4	4-5	5-6	6-7	7-8	8-9	9-10			
MS employee cap	-1	1.1	-1	1.4	-1	1.97	-1	2.9	-1	4.4			
VD employee cap	-1	1.4	-1	1.8	-1	2.55	-1	3.9	-1	6			
TD employee cap	-1	0.93	-1	0.75	-1	0.7	-1	0.7	-1	0.7			
FM employee cap	-1	1.1	-1	1	-1	1.3	-1	1.9	-1	2.9			

Table E-17: Changes to transitioning base case's dominant logic's growth cycles

E.4.3. Internal improvement for capability development

Subsection 6.3.3 simulates the impact of internal improvements on transitioning companies. These adapt BaseT by changing the capability development variables in the same manner as for BaseC and BaseU (Table E-18).

Variables	BaseT	TrainingT	NewEmpsT	TurnoverT
Capability improvement times	1	0.5		1
New employee capability level	0.3		0.1	0.3
Employee turnover rate		30%		10%

Table E-18: Changes to transitioning base case's capability development (1)

During the first growth state, the scenarios grow without caps on employee numbers. They follow the same development as BaseU. The ventures then introduce employee caps for the second half of the simulation. These caps are the number of employees at the end of the first growth state. Like the simulations for different targets, each internal improvement scenario may set a different employee cap. Therefore, these values must be set individually for each scenario (Table E-19).

Table E-19: Changes to transitioning base case's capability development (1)

TrainingT	time: 0 – 5 (=BaseU)	time: 5 – 10	
Marketing and selling (MS) employee cap	-1	2.5	
Value delivering (VD) employee cap	-1	3.17	
Tech. developing (TD) employee cap	-1	0.76	
Firm managing (FM) employee cap	-1	1.74	
NewEmpsT	time: 0 – 5 (=BaseU)	time: 5 – 10	
Marketing and selling (MS) employee cap	-1	1.84	
Value delivering (VD) employee cap	-1	2.49	
Tech. developing (TD) employee cap	-1	0.63	
Firm managing (FM) employee cap	-1	1.08	
TurnoverT	time: 0 – 5 (=BaseU)	time: 5 – 10	
Marketing and selling (MS) employee cap	-1	1.93	
Value delivering (VD) employee cap	-1	2.52	
Tech. developing (TD) employee cap	-1	0.62	
Firm managing (FM) employee cap	-1	1.15	

E.5. Contextual influences and robustness

The chapter also confirms the findings of the above scenario outcomes across different environments and business models. Section 4 of the scenario simulation chapter illustrates this contextual robustness.

E.5.1. Different environmental conditions

Chapter 6 confirms the findings of employee-capped and -uncapped companies across different stable and dynamically changing environments.

E.5.1.1. Stable environments

Subsubsection 6.4.1.1 confirms the development patterns observed for employeecapped and -uncapped simulations. It simulates a favourable environment with improved customer usage intensities, lifetimes, word-of-mouth marketing, and complementary quality (Table E-20). It also simulates a hostile environment with higher input costs, more competition, and stronger competitors. These simulations are run for BaseC and BaseU to evaluate the robustness of findings for both types of dominant logic.

Variables	Hostile	Base	Favourable
Maximum customer usage intensity	1.5		2
Standard rate of adoption	1		2
Standard customer lifetime	5		8
Rate of customer acquisition per customer	0%		10%
Quality of complementary products	1		2
Costs of each type of input	30%	25%	
Strength of competitors and substitutes	1.5	1	
Number of competitors and substitutes	1	0.5	

Table E-20: Changes to reflect differently attractive stable environments

E.5.1.2. Dynamics environments

Subsubsection 6.4.1.2 evaluates the robustness of findings in dynamically changing environments. It simulates an improving and a worsening environment (Table E-21). These environments change from their base case values to favourable and hostile values over five years. These simulations are run for BaseC and BaseU to evaluate the robustness of findings for both types of dominant logic.

Variables	Hostile	Base	Favourable
Maximum customer usage intensity	1.5		1.5 -> 2
Standard rate of adoption	1		1 -> 2
Standard customer lifetime	5		5 -> 8
Rate of customer acquisition per customer	0%		0 -> 10%
Quality of complementary products	1		1 -> 2
Costs of each type of input	25% -> 30%	25%	
Strength of competitors and substitutes	1 -> 1.5	1	
Number of competitors and substitutes	0.5 -> 1	0.5	

Table E-21: Changes to reflect different environmental dynamics³³

E.5.2. Different business model designs

The fourth section of Chapter 6 also evaluates the robustness of findings regarding different business model designs and activity effect sizes.

E.5.2.1. Business models with network effects

Subsubsection 6.4.2.1 outlines the business models of a social network and marketplace. These distinguish between customers and users, have different sets of network effects, target user growth, and have effect sizes for user acquisition and servicing (Table E-22). These settings have been applied to BaseC and BaseU to confirm their development patterns for different business model designs.

Variables	Base Network		Marketplace	
Initial customer base	1	0.5	0.5	
Initial user base	0	0.5	0.5	
Target user growth rate	0	25%	25%	
Customer direct network effect settings	0	-1	-1	
Customer indirect network effect settings	0	1	1	
User direct network effect settings	0	1	-1	
User indirect network effect settings	0	0	1	
effect of competence on ability to acquire	0	1.5	1.5	
users	0	1.5		
effect of competence on ability to service	0	2	2	
users	0	2	2	

Table E-22: Changes to reflect business model with different network effects

³³ For improving and worsening conditions, the left and right values reflect, respectively, the value at the beginning and end of the simulation. For example, the term "5 -> 8" for customer lifetime represents that the customer lifetime is five years at the beginning of the simulation. At the end of the simulation, it is eight years. During the five years of simulation, the model interpolates between these two values linearly. For example, after one year of simulation, the customer lifetime was 5.6 years.

E.5.2.2. Business models with different efficiency levels

Companies may also develop different activity systems. These are reflected in the effect sizes that transform capabilities to hard impacts on resource flows. Subsubsection 6.4.2.2 designs three scenarios with higher and lower effect sizes.

These different effect sizes are evaluated for BaseC and BaseU.

Variables	Lower50	Lower20	Base	Higher20
effect of competence on ability to develop technology	0.2	0.32	0.4	0.48
effect of competence on ability to maintain technology	2	3.2	4	4.80
effect of competence on ability to acquire customers	0.75	1.2	1.5	1.8
effect of competence on ability to service customers	1	1.6	2	2.4
effect of competence on ability to manage the firm	4.5	7.2	9	10.8

Table E-23: Changes to reflect business model with different capability effect sizes